09-50026-mg Doc 14413 Filed 02/04/19 Entered 02/04/19 15:31:37 Main Document

Pg 1 of 27 HEARING DATE AND TIME: February 27, 2019 at 10:00 a.m. (Eastern Time) OBJECTION DEADLINE: February 20, 2019 at 4:00 p.m. (Eastern Time)

BINDER & SCHWARTZ LLP

Eric B. Fisher
Neil S. Binder
Lindsay A. Bush
Lauren K. Handelsman
366 Madison Avenue, 6th Floor
New York, New York 10017
Telephone: (212) 510-7008

Attorneys for the Motors Liquidation Company Avoidance Action Trust

UNITED STATES BANKRUPTCY COURT SOUTHERN DISTRICT OF NEW YORK

X	
In re:	
	Chapter 11
MOTORS LIQUIDATION COMPANY, f/k/a	
GENERAL MOTORS CORPORATION, et al.,	Case No. 09-50026 (MG)
	(Jointly Administered)
Debtors.	
X	

MOTION OF MOTORS LIQUIDATION COMPANY AVOIDANCE ACTION TRUST FOR ENTRY OF ORDER PURSUANT TO SECTIONS 105 AND 1142 OF THE BANKRUPTCY CODE AND BANKRUPTCY RULE 3020(d) (I) APPROVING AMENDMENTS TO THE THIRD AMENDED AND RESTATED AVOIDANCE ACTION TRUST AGREEMENT AND AUTHORIZING THE AVOIDANCE ACTION TRUST TO ENTER INTO THE FOURTH AMENDED AND RESTATED AVOIDANCE ACTION TRUST AGREEMENT (II) AUTHORIZING THE AVOIDANCE ACTION TRUST TO ENTER INTO THE LW CAPITAL PROVISION AGREEMENT AND TO GRANT A LIEN TO THE LW CAPITAL PROVIDER

TABLE OF CONTENTS

		Page
TABLE OF AU	THORITIES	ii
JURISDICTIO	N AND VENUE	2
PRELIMINAR'	Y STATEMENT	2
BACKGROUN	D	4
A.	Old GM Files for Bankruptcy, the Term Loan Is Paid Off, and the Committee Commences the Term Loan Avoidance Action	4
В.	The Plan Is Confirmed, and the Committee Transfers Prosecution of the Term Loan Avoidance Action to the Avoidance Action Trust	6
C.	The Avoidance Action Trust Would Require Additional Funding to Prosecute th Litigation to Completion	
D.	The Avoidance Action Trust Sought Additional Funding	12
E.	The LW Capital Provision Agreement	14
BASIS FOR RI	EQUESTED RELIEF	17
A.	The Avoidance Action Trust Amendment Should Be Approved	18
В.	The LW Capital Provision Agreement and Other LW Transaction Documents Provide Necessary Funding to the Avoidance Action Trust and Should Be Approved	20
NOTICE		23
CONCLUSION	Į	24

TABLE OF AUTHORITIES

Page(s)
Cases Hosp. & Univ. Prop. Damage Claimants v. Johns-Manville Corp. (In re Johns-Manville Corp.), 7 F.3d 32 (2d Cir. 1993)
In re Petition of Bd. of Dirs. of Hopewell Int'l Ins., Ltd., 272 B.R 396 (Bankr. S.D.N.Y. 2002)
LTV Corp. v. Back (In re Chateaugay Corp.), 201 B.R. 48 (Bankr. S.D.N.Y. 1996)
Official Comm. of Unsecured Creditors of Motors Liquidation Co. v. JPMorgan Chase Bank, N.A. (In re Motors Liquidation Co.), 755 F.3d 78 (2d Cir. 2014)
Official Comm. of Unsecured Creditors of Motors Liquidation Co. v. JPMorgan Chase Bank, N.A. (In re Motors Liquidation Co.), 777 F.3d 100 (2d Cir. 2015)
Official Comm. of Unsecured Creditors of Motors Liquidation Co. v. JPMorgan Chase Bank, N.A., 103 A.3d 1010 (Del. 2014)
Penthouse Media Grp. v. Guccione (In re Gen. Media, Inc.), 335 B.R. 66 (Bankr. S.D.N.Y. 2005)
Statutes 11 U.S.C. § 1142(b)
Rules Fed. R. Bankr. P. 3020(d)

TO: THE HONORABLE MARTIN GLENN UNITED STATES BANKRUPTCY JUDGE

Wilmington Trust Company, solely in its capacity as trust administrator and trustee (the "Avoidance Action Trust Administrator") of the Motors Liquidation Company Avoidance Action Trust (the "Avoidance Action Trust" or the "AAT"), as established under the Debtors' Second Amended Joint Chapter 11 Plan dated as of March 18, 2011 [Bankr. Dkt. No. 9836] (as confirmed, the "Plan") of the above-captioned post-effective date debtors (the "Debtors"), submits this motion (the "Motion"), pursuant to sections 105 and 1142 of title 11 of the United States Code (the "Bankruptcy Code") and Rule 3020(d) of the Federal Rules of Bankruptcy Procedure (the "Bankruptcy Rules"), seeking entry of an order, substantially in the form attached hereto as Exhibit A (the "Approval Order"), (i) approving the capital provision agreement attached hereto as Exhibit B (the "LW Capital Provision Agreement") between the Avoidance Action Trust and the private litigation funder LW Holdco VI LLC (the "LW Capital Provider"); (ii) authorizing the Avoidance Action Trust and the Avoidance Action Trust Administrator to enter into the Fourth Amended and Restated Motors Liquidation Company Avoidance Action Trust Agreement (the "Fourth Amended Avoidance Action Trust Agreement") substantially in the form attached hereto as Exhibit C, which amends the Third Amended and Restated Motors Liquidation Company Avoidance Action Trust Agreement, dated as of June 29, 2017 and attached hereto as Exhibit D (the "Third Amended Avoidance Action Trust Agreement"); 1 (iii) authorizing the Avoidance Action Trust to take all actions necessary or appropriate to effectuate the LW Capital Provision Agreement, including the granting of a third-priority lien in favor of the LW Capital Provider on specified property of the Avoidance Action Trust; and

¹ A redline comparing the proposed Fourth Amended Avoidance Action Trust Agreement to the Third Amended Avoidance Action Trust Agreement is attached hereto as <u>Exhibit E</u>.

(iv) granting such other and further relief as may be necessary. In support of the foregoing, the Avoidance Action Trust Administrator respectfully states as follows:

JURISDICTION AND VENUE

1. The Bankruptcy Court has jurisdiction to consider this matter under 28 U.S.C. §§ 157 and 1334, paragraph II of the order of the Bankruptcy Court dated as of March 29, 2011, confirming the Plan [Bankr. Dkt. No. 9941], Article XI of the Plan, and Sections 6.1(b), 8.1 and 13.13 of the Third Amended Avoidance Action Trust Agreement. This is a core proceeding pursuant to 28 U.S.C. § 157(b). Venue is proper before this Court pursuant to 28 U.S.C. § 1409. The statutory predicates for the relief requested are sections 105 and 1142 of the Bankruptcy Code and Bankruptcy Rule 3020.

PRELIMINARY STATEMENT

Agreement that would provide it with up to \$10 million in additional funding to cover litigation costs and other trust administrative expenses. The AAT chose the terms of the LW Capital Provision Agreement after soliciting and receiving indicative terms from other potential third-party capital providers. As the Court was advised on February 1, 2019, the parties have reached an agreement in principle to fully resolve the Avoidance Action Trust's action (the "Term Loan Avoidance Action") against JPMorgan Bank, N.A. ("JPMorgan") and hundreds of other defendants (collectively, the "Term Loan Defendants" or "Term Loan Lenders"). The AAT entered into the LW Capital Provision Agreement before the settlement in principle was reached to ensure that it would have sufficient funds to prosecute to resolution the Term Loan Avoidance Action. Although the AAT and the Term Loan Defendants were continuing to mediate this dispute, the likelihood of reaching a settlement was uncertain at the time the funding was sought.

09-50026-mg Doc 14413 Filed 02/04/19 Entered 02/04/19 15:31:37 Main Document Pg 6 of 27

The funding commitment eliminated the risk to the AAT that it would run out of funds to prosecute this action before an acceptable settlement was reached.

- 3. Moreover, even though a settlement in principle has been reached, the need for the LW Capital Provision Agreement remains. Although the Avoidance Action Trust expects that the parties will succeed in negotiating a definitive settlement agreement and securing the Court's approval of that settlement agreement, significant work is still required before a settlement can be finalized and presented for approval. In addition, there always remains the possibility, however unlikely and undesirable, that a settlement will not be effectuated, in which case the Avoidance Action Trust would be faced with the possibility of not having sufficient funds to cover the fees and costs associated with continued prosecution of the Term Loan Avoidance Action.
- 4. Further, the terms of the funding contained in the LW Capital Provision Agreement take into account the fact that the parties were in active mediation and, though uncertain, there existed a reasonable likelihood of settlement. In a scenario where the settlement is finalized and then approved by the Court, the Avoidance Action Trust will not be obligated to draw more than \$1.5 million of the \$10 million facility and will owe a return only on the drawn amount. Should the settlement not be effectuated, there is no guarantee that the AAT could obtain funding on equally favorable terms. The LW Capital Provision Agreement ensures that in the event that the settlement is not effectuated, there will be no delay in the Avoidance Action Trust's ability to proceed with the litigation.
- 5. After careful consideration, the Trustee, in consultation with the Trust Monitor, has determined that it is in the AAT's interest to have the LW Capital Provision Agreement approved and respectfully requests that the LW Capital Provision Agreement be approved no later than March 7, 2019, to ensure that the Avoidance Action Trust will have the

financial means to fully and finally reach a resolution of the Term Loan Avoidance Action and maximize the value of the Avoidance Action Trust's assets for its beneficiaries.

BACKGROUND

A. Old GM Files for Bankruptcy, the Term Loan Is Paid Off, and the Committee Commences the Term Loan Avoidance Action

- 6. The history of this litigation is well known to this Court and will be summarized briefly. General Motors Corporation ("Old GM") obtained a syndicated secured term loan (the "Term Loan") of approximately \$1.5 billion pursuant to a term loan agreement, dated as of November 29, 2006, as amended on March 4, 2009 (the "Term Loan Agreement"). Official Comm. of Unsecured Creditors of Motors Liquidation Co. v. JPMorgan Chase Bank, N.A. (In re Motors Liquidation Co.), 777 F.3d 100, 101 (2d Cir. 2015) ("Second Circuit 2015 Decision"). To secure repayment of the Term Loan, the Term Loan Lenders were granted a security interest in a large number of Old GM's assets located at forty-two Old GM facilities (the "Collateral"). Id. Among other filings to perfect the Term Loan Lenders' security interest in the Collateral, JPMorgan caused the filing of a UCC-1 financing statement with the Delaware Secretary of State that covered all of the equipment and fixtures constituting Collateral at the forty-two Old GM facilities (the "Main Lien"). Id.
- 7. In connection with its bankruptcy filing, Old GM sought authority from the Court to use a portion of the \$33 billion in post-petition financing (the "**DIP Financing**") from the DIP Lenders to repay the Term Loan in full. Bankr. Dkt. No. 64 ¶¶ 75-78. Old GM repaid the

² All references to the Bankruptcy Docket are to *In re Motors Liquidation Company f/k/a General Motors Corporation*, Case No. 09-50026. All references to the Adversary Docket are to *Motors Liquidation Company Avoidance Action Trust v. JPMorgan Chase Bank, N.A.*, Adv. Pro. No. 09-00504.

09-50026-mg Doc 14413 Filed 02/04/19 Entered 02/04/19 15:31:37 Main Document Pg 8 of 27

Term Loan Lenders in full, ahead of other creditors of Old GM, on the assumption that their claims arising under the Term Loan Agreement were fully secured. Bankr. Dkt. No. 2529 ¶ 19.

- 8. However, shortly before entry of the Final Order Pursuant to Bankruptcy Code Sections 105(a), 361, 362, 363, 364 and 507 and Bankruptcy Rules 2002, 4001 and 6004 (A) Approving a DIP Credit Facility and Authorizing the Debtors to Obtain Post-Petition Financing Pursuant Thereto, (B) Granting Related Liens and Super-Priority Status, (C) Authorizing the Use of Cash Collateral and (D) Granting Adequate Protection to Certain Pre-Petition Secured Parties (the "DIP Order") [Bankr. Dkt. No. 2529], which provided for the final approval of the DIP Financing, the Official Committee of Unsecured Creditors of Motors Liquidation Company f/k/a General Motors Corporation (the "Committee") learned that the Term Loan Lenders' security interests, in fact, may not all have been perfected as of the date that Old GM filed for bankruptcy on June 1, 2009 (the "Petition Date") due to the filing of a termination statement relating to the Main Lien (the "2008 Termination Statement") months before the Petition Date. Second Circuit 2015 Decision, 777 F.3d at 102. Therefore, the DIP Order, while conditionally approving Old GM's repayment of the Term Loan, expressly preserved the right of the Committee to investigate and bring actions based upon the purported perfection of the security interests related to the Term Loan. Bankr. Dkt. No. 2529 ¶ 19(d).
- 9. In order to recover amounts alleged to have been improperly paid by Old GM to the Term Loan Lenders (the "**Transfers**"), based on the erroneous assumption that the Term Loan Lenders' security interests were perfected and the loan fully secured, the Committee commenced the Term Loan Avoidance Action on July 31, 2009. Adv. Pro. Dkt. No. 1.

B. The Plan Is Confirmed, and the Committee Transfers Prosecution of the Term Loan Avoidance Action to the Avoidance Action Trust

- 10. JPMorgan and the Committee agreed to litigate the issue of whether the 2008 Termination Statement terminated the security interest in the Collateral covered by the Main Lien ("**Phase I**"), before litigating any other issues in the case, and the Bankruptcy Court approved this bifurcation of the case. *See, e.g.*, Adv. Pro. Dkt. Nos. 10, 17, and 82.
- 11. By August 2010, approximately one year after commencement of the Term Loan Avoidance Action, the Avoidance Action Trust and JPMorgan had completed discovery, and fully briefed cross-motions for summary judgment, in connection with Phase I of the Avoidance Action. *See* Adv. Pro. Dkt. Nos. 17, 20, and 23.
- 12. On March 29, 2011, the Bankruptcy Court entered an order (the "Confirmation Order") confirming the Plan. Bankr. Dkt. No. 9941. The Plan provided for, among other things, the creation of the Avoidance Action Trust, which was established to liquidate and distribute its non-administrative assets, which consist entirely of the proceeds, if any, of the Term Loan Avoidance Action. Bankr. Dkt. No. 9836 § 6.5 (Plan). Thereafter the Committee was dissolved, and on December 15, 2011, while the cross-motions for summary judgment relating to Phase I were pending, prosecution of the Term Loan Avoidance Action was transferred to the Avoidance Action Trust. *See id.* § 6.5 (Plan); Ex. D (Third Amended Avoidance Action Trust Agreement).

C. The Avoidance Action Trust Would Require Additional Funding to Prosecute the <u>Litigation to Completion</u>

13. The initial administrative assets of the Avoidance Action Trust consisted of approximately \$1.6 million in cash to be held and maintained by the Avoidance Action Trust Administrator for fees and expenses in connection with trust administration and prosecution of the

Term Loan Avoidance Action (the "Avoidance Action Trust Administrative Cash"). Bankr. Dkt No. 11330 ¶ 18.³

- 14. The \$1.6 million of Avoidance Action Trust Administrative Cash set aside under the Plan quickly proved to be insufficient, and on January 20, 2012, the GUC Trust filed a motion seeking, *inter alia*, to liquidate securities to fund additional Avoidance Action Trust fees, costs, and expenses primarily related to the prosecution of the Term Loan Avoidance Action. *See generally id* ¶¶ 18-21. The Court granted the GUC Trust's motion and entered an Order, which, among other things, allocated an additional \$13,714,000 to the Avoidance Action Trust to satisfy the Avoidance Action Trust's estimated fees, costs and expenses for 2012, 2013, and 2014 (the "GUC Trust Supplemental Cash"). Bankr. Dkt. No. 11507.
- 15. On March 1, 2013, the Bankruptcy Court denied the Committee's motion for partial summary judgment and granted summary judgment in favor of JPMorgan, ruling that the filing of the 2008 Termination Statement was not effective and that the security interest covered by the Main Lien was therefore perfected as of the Petition Date. Adv. Pro. Dkt. No. 71. The Committee appealed to the Second Circuit. Adv. Pro. Dkt. No. 74.
- 16. A little less than two years later, on January 21, 2015, after receiving an answer to a Delaware UCC question that the Second Circuit had certified to the Delaware Supreme Court, see Official Comm. of Unsecured Creditors of Motors Liquidation Co. v. JPMorgan Chase Bank, N.A. (In re Motors Liquidation Co.), 755 F.3d 78, 86 (2d Cir. 2014) and Official Comm. of Unsecured Creditors of Motors Liquidation Co. v. JPMorgan Chase Bank, N.A., 103 A.3d 1010

³ The Avoidance Action Trust also received \$500,000 of Avoidance Action Trust SEC Reporting Cash to be used solely for SEC Reporting Costs to the extent there is no other available source of funds to pay such costs. Bankr. Dkt. No. 11330 ¶ 18. Any unused portion of the Avoidance Action Trust SEC Reporting Cash will be returned to the GUC Trust. Ex. C § 2.3(e) (Third Amended Avoidance Action Trust Agreement).

(Del. 2014), the Second Circuit reversed the Bankruptcy Court's grant of summary judgment and remanded the matter to the Bankruptcy Court with instructions to enter partial summary judgment in favor of plaintiff, the Avoidance Action Trust. *Second Circuit 2015 Decision*, 777 F.3d at 101.

- 17. On June 12, 2015, following issuance of the Second Circuit's mandate, the Bankruptcy Court entered partial summary judgment in favor of the Avoidance Action Trust as to the termination of the Main Lien. Adv. Pro. Dkt. No. 96. Thereafter, the Avoidance Action Trust filed an amended complaint and proceeded to serve it upon all the Term Loan Lenders. Adv. Pro. Dkt. Nos. 91, 94, 95, 163, 164. The Term Loan Lenders moved to dismiss the complaint on various grounds but the Court ruled on June 30, 2016, that the Term Loan Avoidance Action could proceed. Adv. Pro. Dkt. No. 643.
- Action was the Avoidance Action Trust's contention that because the perfected security interest with respect to the Main Lien was terminated, the Term Loan Lenders should not have been paid as fully secured creditors, and the value of the Transfers in excess of the value of any surviving perfected collateral should be avoided and recovered for the benefit of the estate under Bankruptcy Code section 549 and 550. *See, e.g.*, Adv. Pro. Dkt. No. 903. The Court decided in an attempt to streamline the Phase II proceedings and promote a global settlement between the parties, to hold an initial trial to determine the collateral classification and valuation issues of forty "Representative Assets" to be selected by the parties (the "Representative Assets Trial"). Adv. Pro. Dkt. No. 547.
- 19. In the spring of 2016, it became clear that the GUC Trust Supplemental Cash would be insufficient to fund the Avoidance Action Trust's fees, costs, and expenses related to the prosecution of Phase II of the litigation. The Avoidance Action Trust entered into a

financing agreement with the United States Department of the Treasury ("**Treasury**") and Export Development Canada ("**EDC**" and together with Treasury, the "**DIP Lenders**") pursuant to which the DIP Lenders provided \$15 million in financing (the "**DIP Lenders Litigation Financing**"), which was approved by the Court on August 30, 2016. Bankr. Dkt. No. 13748.

- 20. At the same time the Avoidance Action Trust was seeking funding, the parties began extensive discovery and preparations in the lead up to the Representative Assets Trial. In the fall of 2016, depositions were taken of 37 fact witnesses and more than 200,000 documents were produced for review. Both sides retained large expert teams to review the voluminous and technical discovery and assist in litigation strategy. The parties deposed 18 experts, and 22 expert reports, including rebuttal reports, were submitted. The parties went on four site visits to GM facilities, including the judicial site visit with the Court.
- 21. After discovery ended, there was extensive pre-trial briefing in the months immediately before the Representative Assets Trial, including nine motions in limine and briefs to the Court laying out the issues to be addressed at trial. Additionally, the opening statements of the Representative Assets Trial and the direct testimony of the experts were submitted in writing before trial. The live testimony portion of the Representative Assets Trial lasted two weeks, from April 24, 2017 to May 5, 2017, and three fact witnesses and 15 expert witnesses testified. Hundreds of pages of post-trial briefing were submitted on May 25, 2017, and closing arguments were heard by the Court on June 5, 2017.
- 22. By June 2017, the Avoidance Action Trust had exhausted the DIP Lenders Litigation Financing. The Avoidance Action Trust sought a second round of litigation funding and on June 23, 2017, entered into a second financing agreement (the "CE Capital Providers Agreement") with the private litigation funders Cynthiana LLC and Earlham LLC (collectively,

the "CE Capital Providers") pursuant to which the CE Capital Providers provided up to \$15 million in financing (the "CE Capital Providers Litigation Financing"). Bankr. Dkt. No. 13957. The Court approved the CE Capital Providers Agreement on June 29, 2017, Bankr. Dkt. No. 13980.

- 23. On September 26, 2017, the Court issued its decision on the Representative Assets Trial with the expectation that the parties would use the decision to reach a global settlement of the entire case. Adv. Pro. Dkt. No. 1015 at 2-3.
- 24. From the date of the decision to the summer of 2018, the parties engaged in extensive mediation, conducting five days of mediation aimed at global resolution as well as five days of mediation aimed at reaching consensus on individual asset line items. As part of the mediation process, the Avoidance Action Trust, among other preparations, shared with the Term Loan Defendants a mediation statement as well as additional memoranda and presentations on discrete topics; engaged in high-level strategizing and legal analysis; and worked with their expert team to analyze the impact of the Court's decision on the remaining thousands of assets in dispute and model different outcome scenarios.
- 25. On July 3, 2018, the parties informed the Bankruptcy Court that a global settlement was not at that time possible. Adv. Pro. Dkt. No. 1055 at 2. In their communication with the Court, the parties identified issues that were ripe for summary judgment briefing and additional issues that, if resolved at trial, would materially facilitate reaching a global settlement (the "Initial Discovery and Trial Issues"). Adv. Pro. Dkt. No. 1075 at 3-6. On September 14, 2018, the Court entered the *Order Amending and Superseding Certain Prior Orders Regarding Discovery and Scheduling*, establishing a briefing schedule for a number of summary judgment issues, and a discovery and trial schedule for the Initial Discovery and Trial Issues (the "Phase III

Trial"). Adv. Pro. Dkt. No. 1080 at 6-8. The Court-ordered schedule contemplated an additional trial on the Initial Discovery and Trial Issues and prescribed that (i) after the Court's decision on the Initial Discovery and Trial Issues, the parties would have 30 days to reach a global settlement; and (ii) if no resolution after this second trial were reached, there would be an expedited schedule for a third trial by the Court for all remaining assets in dispute. *Id.* at 8-9.

- 26. A settlement was not reached at that time, and the parties began discovery on issues for the Phase III Trial. In accordance with the scheduling order, the parties fully briefed four summary judgment motions and an estoppel motion. At the same time, the parties engaged in discovery, serving more than 20 document subpoenas which resulted in the production of approximately 36,000 documents. In preparation for the Phase III Trial, the parties have also conducted eight additional fact depositions and conducted an additional five site visits to current and former GM facilities.
- 27. In preparing for the Phase III Trial, the Avoidance Action Trust retained additional experts and also continued to work with its experts from Phase II of the case. The Avoidance Action Trust has been working with its team of experts to analyze the additional discovery and prepare initial expert reports, which, prior to the settlement in principle, were to be served by February 12, 2019.
- 28. In addition, the Trust continues to incur expenses in connection with trust administration.
- 29. The Avoidance Action Trust estimates that as of January 31, 2019, the Trust had \$2 million in undrawn funds remaining on the current litigation finance facility plus approximately \$2.5 million of cash in its accounts. Thus after accounting for billed and estimated unbilled fees and expenses in excess of approximately \$1.2 million, the Trust has less than \$3.3

million available to fund the fees and costs associated with prosecution of the Term Loan Avoidance Action, the administrative expenses of the Avoidance Action Trust, and the fees of the Administrator of the Avoidance Action Trust (the "Trust Administrator") and trust monitor (the "Avoidance Action Trust Monitor"). Declaration of Arthur J. Gonzalez, dated February 4, 2019 (the "Gonzalez Declaration") ¶ 4.

30. As the Court is aware, the parties have now reached an agreement in principle to fully resolve the Term Loan Avoidance Action. Adv. Pro. Dkt. No. 1168 (E. Fisher letter to Court). However, the parties have yet to enter into a definitive settlement agreement for which the parties will seek Court approval. In the unexpected event that a final settlement is not entered into and approved, the Trust's remaining cash holdings are insufficient to fund the Term Loan Avoidance Action to its conclusion. Gonzalez Decl. ¶ 4.

D. The Avoidance Action Trust Sought Additional Funding

- Trust Administrator "shall at all times, to the extent practicable, retain ... sufficient Avoidance Action Trust Administrative Cash and Supplemental Avoidance Action Trust Cash as the Trust Administrator shall determine, with the approval of the Trust Monitor and subject to the Budget, is necessary (x) to pay the reasonable incurred or anticipated fees and expenses of the Trust . . . and (y) to satisfy other liabilities incurred or anticipated by the Trust in accordance with the Plan, the Confirmation Order and this Trust Agreement." Ex. D § 5.5(b) (Third Amended Avoidance Action Trust Agreement).
- 32. The Avoidance Action Trust Agreement sets forth a mechanism for the Avoidance Action Trust to seek additional funding in the event that the Avoidance Action Trust Administrative Cash and the GUC Trust Supplemental Cash "is not reasonably likely to be

09-50026-mg Doc 14413 Filed 02/04/19 Entered 02/04/19 15:31:37 Main Document Pg 16 of 27

adequate to satisfy the current and projected future fees, costs and expenses" of the Avoidance Action Trust. *Id.* § 6.1(d).

- 33. Specifically, the Avoidance Action Trust Agreement permits, upon approval of the Bankruptcy Court, the granting of a lien on the Term Loan Avoidance Action or other property of the Avoidance Action Trust in exchange for proceeds that may be used to satisfy the fees, costs and expenses of the Avoidance Action Trust (the "Other Supplemental Cash"). *Id.*
- Agreement, the Avoidance Action Trust Administrator, in consultation with the Avoidance Action Trust Monitor, made a determination that the cash available to the Avoidance Action Trust was not sufficient to meet its projected fees and expenses. *See* Gonzalez Decl. ¶ 4. The Avoidance Action Trust Administrator, through counsel, solicited and received indicative terms from potential third-party capital providers in order to obtain Other Supplemental Cash for the Avoidance Action Trust. *See id.* ¶ 8. The LW Capital Provider's proposed terms were competitive with those offered by other potential funders. *See id.* ¶ 9.
- 35. The Avoidance Action Trust engaged in negotiations with the LW Capital Provider in order to obtain an agreement on terms that would ensure adequate litigation funding for the Avoidance Action Trust on the best economic terms. *Id.* ¶¶ 7-9. Ultimately, after these negotiations, the parties arrived at the LW Capital Provision Agreement that is the subject of this Motion.

E. The LW Capital Provision Agreement⁴

- 36. The LW Capital Provision Agreement provides that the LW Capital Provider will provide up to \$10 million to the Avoidance Action Trust to cover all costs, which include the costs of litigation as well as the costs and expenses of the Avoidance Action Trust. The AAT is required to draw \$1,500,000 upon Court approval of the LW Capital Provider Agreement. In the event that a settlement is not effectuated within a year or this case proceeds to a Phase III Trial, the AAT would be required to draw the remaining \$8,500,000. The full terms of the litigation funding are set out in the LW Capital Provision Agreement attached hereto as Exhibit B.
- 37. Critically, the LW Capital Provision Agreement provides that the LW Capital Provider has no rights with respect to oversight of the Avoidance Action Trust's prosecution of the Term Loan Avoidance or any other action, nor does the LW Capital Provider have any rights with respect to any decision whether or not to settle any action. Ex. B § 5.2(b) (LW Capital Provision Agreement).
- 38. The key terms of the LW Capital Provision Agreement, all of which are consistent with Section 6.1(d) of the Avoidance Action Trust Agreement, are as follows:
 - Following, *inter alia*, this Court's approval of the LW Capital Provision Agreement (the "**Initial Funding Date**"), the LW Capital Provider shall provide the first \$1,500,000 (the "**Initial Funding**") to the Avoidance Action Trust (*id.* § 2.1(a)(ii));
 - The Avoidance Action Trust agrees to accept the Initial Funding (id. § 2.1(a)(i));
 - Thereafter, the Avoidance Action Trust may make further draws in tranches of not less than \$1 million in its sole discretion (all funds actually paid by the LW Capital Provider to the Avoidance Action Trust are collectively referred to as the "Invested Amount") (id. § 2.1(a)(ii), (iii));

⁴ To the extent there is any conflict between this summary and the LW Capital Provision Agreement, the LW Capital Provision Agreement will govern in all respects.

- The LW Capital Provider shall provide the Avoidance Action Trust with up to \$10,000,000 in financing until the earlier of (a) the one year anniversary of the Initial Funding Date, and (b) the commencement of the Phase III Trial (id. § 2.1(a)(i); and
- The LW Capital Provider shall be entitled (the "LW Capital Provider's Entitlement") to a return calculated as (a) the product of the Invested Amount multiplied by two plus (b) after twelve months, an annual interest rate of 15%, compounded annually, accruing on the amounts outstanding under clause (a) above until the LW Capital Provider's Entitlement is paid in full to the LW Capital Provider; such fee shall be payable from the Proceeds of the Avoidance Action Trust ahead of all other obligations and beneficiaries of, or investors in, the Avoidance Action Trust other than the DIP Lenders and the CE Capital Providers (id. § 2.2(a)).
- 39. Effectively, under the LW Capital Provision Agreement, if the settlement that has in principle been reached by the parties is effectuated prior to a year from when the LW Capital Provider provided the AAT the Initial Funding, the cost to the AAT will be \$1,500,000. If the settlement does not become effective and it is necessary to continue litigation, the AAT will draw the full \$10,000,000, and the LW Capital Provider will be entitled to a \$20,000,000 payment (plus applicable interest) from litigation proceeds of the Avoidance Action Trust (as defined in the LW Capital Provision Agreement) after payment to the DIP Lenders and the CE Capital Providers.
- 40. In addition to the key terms summarized above, the LW Capital Provision Agreement contains other provisions, including but not limited to, those concerning conditions to

consummating the LW Capital Provision Agreement, and termination events, rights, and procedures.⁵

- Provision Agreement, the Avoidance Action Trust and the LW Capital Provider also executed a Security Agreement attached hereto as Exhibit F (the "LW Security Agreement") pursuant to which the Avoidance Action Trust granted a third-priority security interest and lien to the LW Capital Provider on (i) the Avoidance Action Proceeds and (ii) the LW Capital Provider Funding Account and all Supplemental Capital therein (all terms in this paragraph shall retain the meanings ascribed thereto in the LW Security Agreement) (the "LW Capital Provision Agreement Collateral"). Ex. F §§ 2.1 and 6.3 (LW Security Agreement).
- 42. The Avoidance Action Trust has advised and consulted with the DIP Lenders throughout the process of obtaining additional funding and shared with the DIP Lenders the initial term sheet outlining the terms of the LW Capital Provision Agreement and drafts of the LW Capital Provision Agreement. The DIP Lenders have consented to the AAT entering into this agreement for additional funding.
- 43. In addition, the DIP Lenders and the CE Capital Providers have both approved, and are parties to, a subordination agreement (the "LW Subordination Agreement,"

⁵ For example, in exchange for the LW Capital Provider reserving the funding for the Avoidance Action Trust notwithstanding the ongoing mediation, the Avoidance Action Trust has entered into a Termination Fee Agreement with the LW Capital Provider (the "**Termination Fee Agreement**"). Ex. B § 27 (LW Capital Provider Agreement). Under the Termination Fee Agreement, the Avoidance Action Trust agreed that if for any reason the LW Capital Provider Agreement was not approved within 45 days after the agreement was executed—for example, because the AAT terminated the agreement prior to Court approval or because such approval was denied—the AAT would, subject to Court approval, pay the LW Capital Provider a fee of \$200,000 from Distributable Trust Assets (as that term is defined in the Third Amended Avoidance Action Trust Agreement). *Id*.

and together with the LW Capital Provision Agreement and the LW Security Agreement, the "LW Transaction Documents"), attached hereto as Exhibit G.

44. As required by Section 6.1(d)(ii) of the Third Amended Avoidance Action Trust Agreement, the Avoidance Action Trust Monitor has submitted a Declaration herewith setting forth his approval of the LW Capital Provision Agreement and the other LW Transaction Documents. *See* Gonzalez Decl. ¶¶ 6-10.

BASIS FOR REQUESTED RELIEF

- 45. This Court has the authority to enter the proposed order (a) authorizing the amendments to the Third Amended Avoidance Action Trust Agreement (the "Avoidance Action Trust Amendment") and execution of the Fourth Amended Avoidance Action Trust Agreement; and (b) approving the LW Capital Provision Agreement and the other LW Transaction Documents under the terms of the Plan and under section 1142(b) of the Bankruptcy Code.
- 46. The Plan specifies that the Bankruptcy Court retains exclusive jurisdiction "of all matters arising under, arising out of, or related to the Chapter 11 Cases and the Plan pursuant to, and for the purposes of, sections 105(a) and 1142 of the Bankruptcy Code" and for, *inter alia*, the following purposes: (1) to "ensure that distributions to holders of Allowed Claims are accomplished as provided herein"; (2) to "hear and determine disputes arising in connection with or related to the interpretation, implementation, or enforcement of the Plan, . . . the Avoidance Action Trust, . . . and the Avoidance Action Trust Agreement"; and (3) to "take any action and issue such orders as may be necessary to construe, enforce, implement, execute, and consummate the Plan to maintain the integrity of the Plan following consummation." Bankr. Dkt. No. 9836 §§ 11.1(c), (i), (j) (Plan).

Further, section 1142(b) of the Bankruptcy Code authorizes the Court to 47. "direct the debtor and any other necessary party to execute or deliver or to join in the execution or delivery of any instrument required to effect a transfer of property dealt with by a confirmed plan. and to perform any other act . . . that is necessary for the consummation of the plan." 11 U.S.C. § 1142(b); see also Hosp. & Univ. Prop. Damage Claimants v. Johns-Manville Corp. (In re Johns-Manville Corp.), 7 F.3d 32, 34 (2d Cir. 1993) (finding that bankruptcy courts retain postconfirmation jurisdiction in chapter 11 proceedings to the extent provided by the plan); Penthouse Media Grp. v. Guccione (In re Gen. Media, Inc.), 335 B.R. 66, 73 (Bankr. S.D.N.Y. 2005) (finding that bankruptcy courts retain post-confirmation jurisdiction to matters related to the implementation of a plan); In re Petition of Bd. of Dirs. of Hopewell Int'l Ins., Ltd., 272 B.R 396, 407 n.11 (Bankr. S.D.N.Y. 2002) ("[T]he Court may direct parties to perform any act necessary to consummate the plan.") (citing 11 U.S.C. § 1142(b)); LTV Corp. v. Back (In re Chateaugay Corp.), 201 B.R. 48, 66 (Bankr. S.D.N.Y. 1996), aff'd in part, 213 B.R. 633 (S.D.N.Y. 1997) ("The clear intent of Section 1142(b) of the Bankruptcy Code is to assure that the terms and provisions of a confirmed chapter 11 plan are carried out until the plan is completed and the final decree is entered closing the case."). In addition, Bankruptcy Rule 3020(d) provides that "[n]otwithstanding the entry of the order of confirmation, the court may issue any other order necessary to administer the estate." Fed. R. Bankr. P. 3020(d).

A. The Avoidance Action Trust Amendment Should Be Approved

48. The Avoidance Action Trust Amendment is intended to implement the terms of the LW Capital Provision Agreement. While the Third Amended Avoidance Action Trust Agreement provided a mechanism for the Avoidance Action Trust to obtain additional funding from private litigation funders, it now needs to be revised to implement the terms of the LW Capital

Provision Agreement. In addition, the Third Amended Avoidance Action Trust Agreement needs to be amended to reflect the priority of payments to the LW Capital Provider and the other creditors and beneficiaries of the Avoidance Action Trust.

49. The Third Amended Avoidance Action Trust Agreement permits that:

The Trust Administrator may amend or supplement this Trust Agreement for any other purpose, but only on petition to, and with the approval of, the Bankruptcy Court; <u>provided</u> that (x) no amendment or supplement to this Trust Agreement shall be inconsistent with the purpose and intent of the Trust to dispose of in an expeditious but orderly manner the Avoidance Action Trust Assets in accordance with the terms of the Plan, the Confirmation Order and this Trust Agreement, and (y) this Trust Agreement shall not be amended in a manner that is inconsistent with the Plan in the form confirmed by the Bankruptcy Court, subject to any post confirmation modifications to the Plan pursuant to Section 1127 of the Bankruptcy Code.

Ex. D § 13.13(b) (Third Amended Avoidance Action Trust Agreement).

liquidate and distribute its assets, which consist entirely of the proceeds, if any, of the Avoidance Action. Bankr. Dkt. No. 9836 § 6.5 (Plan). The additional funding provided to the Avoidance Action Trust through the LW Capital Provision Agreement is necessary to ensure the AAT's ability to continue to meet its responsibility to seek the recovery and liquidation of the "Avoidance Action Trust Assets" – the Term Loan Avoidance Action. As discussed in greater detail below, although the parties have recently reached an agreement in principle to fully resolve the Term Loan Avoidance Action, the LW Capital Provision Agreement provides necessary funding in the event the settlement in principle reached by the parties is not effectuated and the Avoidance Action Trust must restart the prosecution of the Term Loan Avoidance Action. Accordingly, the Avoidance Action Trust, as well as the Plan.

51. The Avoidance Action Trust Amendment has been narrowly tailored to implement only the terms of the LW Capital Provision Agreement and the other LW Transaction Documents and, for all the reasons stated above, should be approved.

B. The LW Capital Provision Agreement and Other LW Transaction Documents Provide Necessary Funding to the Avoidance Action Trust and Should Be Approved

- 52. Here, the LW Capital Provision Agreement allows the Avoidance Action Trust to fulfill its basic purpose under the Plan. The Plan specifies that the "sole purpose" of the Avoidance Action Trust is to liquidate and distribute its assets, which consist of the Term Loan Avoidance Action. Bankr. Dkt. No. 9836 §§ 1.23, 6.5 (Plan). The LW Capital Provision Agreement promotes this goal.
- Term Loan Defendants, requiring the Trust's counsel, in conjunction with a team of experts, to mount a tremendous effort as part of the Trust's continuing duty to attempt to realize value from this litigation for the benefit of the Trust's beneficiaries, which include the unsecured creditors of this bankruptcy estate, United States Treasury, and Export Development Canada. As this litigation progressed during 2018, building towards a potential second trial before the Court later this year, the Trust Administrator, in consultation with the Trust Monitor and the Trust's counsel, determined that the Trust's available cash would be insufficient to meet the needs of the Trust if this case were to continue to a litigated resolution. Specifically, the Trust Administrator determined that the Trust would run out of funds by the spring of this year.
- 54. The Avoidance Action Trust entered into the LW Capital Provision Agreement because, at the time the agreement was executed by the Trust, it was uncertain that the parties would be able to find a way to resolve this case, and the Trust had to ensure that it would have the financial wherewithal to see the case through to completion. Further, the Avoidance

Action Trust's ability to demonstrate to the Term Loan Defendants the Trust's financial capability to meet the demands of the case was necessary to negotiate for the best possible settlement for the Trust.

55. Although the parties have now succeeded in reaching an agreement in principle to settle this case, a settlement is still subject to the negotiation of definitive settlement documents and the Court's approval. The LW Capital Provision Agreement remains critical because it provides a source of funding to the Trust in the event that the settlement in principle that has been reached does not ultimately come to fruition. Should no settlement be reached, the Avoidance Action Trust would be required to restart the prosecution of the Term Loan Avoidance Action, which would not be possible with the cash currently available to the Avoidance Action Trust.

56. Further, the terms of the LW Capital Provision Agreement take into account the fact that the parties were in active mediation, and though uncertain, there existed a reasonable likelihood of settlement at the time the agreement was reached. In a scenario where the settlement is finalized and then approved by the Court, the Avoidance Action Trust will not be obligated to draw more than \$1.5 million of the \$10 million facility and will owe a return only on the drawn amount. Should the settlement not be effectuated, there is no guarantee that the AAT could obtain funding on equally favorable terms. Further, the process of obtaining and negotiating additional funding would take time and result in additional costs to the AAT.⁶ Though the AAT has every hope that a final settlement will be effectuated, the Trustee, in consultation with the Trust Monitor,

⁶ In addition to costs associated with soliciting and finalizing other funding, the Avoidance Action Trust is already obligated to reimburse the LW Capital Provider up to \$65,000 for its costs incurred in connection with the Transaction Documents. Ex. B § 22 (LW Capital Provision Agreement).

has determined that it is in the best interest of the Trust to enter into the LW Capital Provision Agreement.

- 57. In sum, the LW Capital Provision Agreement, which was the result of the solicitation of bids from private funders and good faith, arm's-length negotiations, resolved and continues to resolve significant funding issues of the Avoidance Action Trust, fulfills the goals of the Plan, and benefits all affected parties.
- Agreement is necessary in order for the Avoidance Action Trust to meet the requirement that any additional funding be junior and subordinate to the funding provided by the DIP Lenders and the CE Capital Providers. The LW Subordination Agreement should have no material impact on the Avoidance Action Trust or its assets, as the LW Subordination Agreement merely documents and gives effect to the subordination of the claims and interests of the LW Capital Provider to the claims and interests of the DIP Lenders and the CE Capital Providers as provided in the Settlement Agreement and Third Amended Avoidance Action Trust Agreement. Because the Avoidance Action Trust's entry into the LW Subordination Agreement is necessary in order for the Avoidance Action Trust to obtain funding under the LW Capital Provision Agreement, the LW Subordination Agreement should be approved.
- 59. Finally, the LW Capital Provision Agreement provides that the LW Capital Provider may terminate the agreement if the Court's approval is not obtained by March 7, 2019, which is 45 days after the execution of the agreement. This requirement was an essential part of the agreement. The AAT wanted (and received) the benefit of committed funding in advance of what it expected would be the last chance to reach a settlement before the Phase III Trial, and, in exchange, the LW Capital Provider did not want to provide the AAT with an open-ended option

to fund up to \$10 million in funding. Although the AAT does have the right to terminate the LW Capital Provision Agreement at this point, in light of the risk associated with a failure to effectuate a settlement, the Trustee in consultation with the Trust Monitor has determined that such funding remains in the best interest of the AAT. Accordingly, the Avoidance Action Trust respectfully requests that by no later than March 7, 2019, the Court enter the proposed order (a) authorizing the amendments to the Third Amended Avoidance Action Trust Agreement and execution of the Fourth Amended Avoidance Action Trust Agreement; and (b) approving the LW Capital Provision Agreement and the other LW Transaction Documents under the terms of the Plan and under section 1142(b) of the Bankruptcy Code.

NOTICE

Office of the United States Trustee for the Southern District of New York, 33 Whitehall Street, 21st Floor, New York, New York 10004; (b) the DIP Lenders; (c) the other parties in interest in accordance with the Sixth Amended Order Pursuant to 11 U.S.C. § 105(a) and Fed. R. Bankr. P. 1015(c) and 9007 Establishing Notice and Case Management Procedures, dated May 5, 2011 [Docket No. 10183]; (d) JPMorgan and each of the Term Loan Defendants, and (e) any other required notice parties under Section 6.1(b)(ii) of the Avoidance Action Trust Agreement. ⁷ The

⁷ The Avoidance Action Trust Agreement requires notice to "the Trust Monitor, the holders of Units and the holders of Disputed General Unsecured Claims." Ex. C § 6.1(d)(ii) (Third Amended Avoidance Action Trust Agreement). However, because of the now-resolved dispute over who is entitled to the proceeds of the Term Loan Avoidance Action (*see supra* p. 6, n.2), no Units have been issued, and there are no holders of Units. Accordingly, notice has been provided to all potential Unit holders and/or beneficiaries of the Avoidance Action Trust and other interested parties, including the DIP Lenders; the Committee; the holders of Motors Liquidation Company (f/k/a General Motors Company) debentures and notes with the following CUSIP Nos.: 370ESCAN5; 370ESCAJ4; 370ESCAR6; 370ESCAG3; 370ESCAS7; 370ESCAT2; 370ESCAU9; 370ESCAV7; 370ESCAZ8; 370ESCBB0; 370ESCBQ7; 370ESCBT1; 370ESCBW4; 370ESCBS3; 370ESC816; 370ESC774; 370ESC766; 370ESC758; 370ESC741; 370ESC733; 370ESC725; 370ESC717; 370ESC121; 370ESC691; 616ESC AA2; 616ESC AB0; 349ESC AT1; 677ESC AU2; 677ESC BC2; 455ESC AB8; 594ESC AQ6; XS0171942757; XS0171943649; CH0008769264 (served

09-50026-mg Doc 14413 Filed 02/04/19 Entered 02/04/19 15:31:37 Main Document Pg 27 of 27

Avoidance Action Trust submits that such notice is sufficient and no other or further notice need

be provided.

CONCLUSION

WHEREFORE, the Avoidance Action Trust respectfully requests that the Court enter an

Order: (i) approving the LW Transaction Documents; (ii) authorizing the Avoidance Action Trust

and the Avoidance Action Trust Administrator to enter into the Fourth Amended and Restated

Motors Liquidation Company Avoidance Action Trust Agreement; (iii) authorizing the Avoidance

Action Trust to take all actions necessary or appropriate to effectuate the LW Transaction

Documents, including the granting of a third-priority lien (subject to the DIP Lenders' and the CE

Capital Providers' liens, interests and rights) on the LW Capital Provision Agreement Collateral;

and (iv) granting such other and further relief as may be necessary.

Dated: February 4, 2019

New York, New York

Respectfully submitted,

BINDER & SCHWARTZ LLP

/s/ Eric B. Fisher

Eric B. Fisher

Neil S. Binder

Lindsay A. Bush

Lauren K. Handelsman

366 Madison Avenue, 6th Floor

New York, New York 10017

Tel: (212) 510-7008

Attorneys for the Motors Liquidation

Company Avoidance Action Trust

through the Depository Trust Company (DTC)); the non-bondholder holders of Allowed General Unsecured Claims (as defined in the Third Amended Avoidance Action Trust Agreement); and any holders of disputed General Unsecured Claims.

24

Exhibit A

SOUTHERN DISTRICT OF NEW YORK		
	X	
In re:		Chapter 11
MOTORS LIQUIDATION COMPANY, f/k/a GENERAL MOTORS CORPORATION, et al.,		Case No. 09-50026 (MG) (Jointly Administered)
Debtors.		
	X	

HNITED STATES BANKBUPTCV COURT

ORDER (I) APPROVING AMENDMENTS TO THE THIRD AMENDED AND RESTATED AVOIDANCE ACTION TRUST AGREEMENT AND AUTHORIZING ENTRY INTO THE FOURTH AMENDED AND RESTATED AVOIDANCE ACTION TRUST AGREEMENT AND (II) AUTHORIZING THE AVOIDANCE ACTION TRUST TO ENTER INTO THE LW CAPITAL PROVISION AGREEMENT AND TO GRANT A LIEN TO THE LW CAPITAL PROVIDER

Upon the motion (the "Motion")¹ of Motors Liquidation Company Avoidance Action Trust for entry of an order, pursuant to sections 105 and 1142 of title 11 of the United States Code: (1) approving the amendment and restatement of the Third Amended Avoidance Action Trust Agreement and (2) authorizing the Avoidance Action Trust to enter into the LW Transaction Documents and to grant a lien to the LW Capital Provider; and upon the Declaration of Arthur J. Gonzalez in support of the Motion; and it appearing that the Court has jurisdiction to consider the Motion and relief requested therein pursuant to 28 U.S.C. §§ 157 and 1334 and the Amended Standing Order of Reference M-431, dated January 31, 2012 (Preska, C.J.), consideration of the Motion and the relief requested therein being a core proceeding pursuant to 28 U.S.C. § 157(b) and venue being proper before this Court pursuant to 28 U.S.C. § 1409; and any objections to the

¹ Capitalized terms not otherwise defined herein shall have the meaning ascribed to them in the Motion. Capitalized terms not otherwise defined herein or in the Motion shall have the meaning ascribed to them in the applicable LW Transaction Document.

09-50026-mg Doc 14413-1 Filed 02/04/19 Entered 02/04/19 15:31:37 Exhibit A Pg 3 of 5

Motion having been settled, resolved, withdrawn, or overruled; and it appearing that the relief requested in the Motion is in the best interest of the above-captioned post-effective date debtors' creditors and estates; and it appearing that due and appropriate notice of the Motion has been given under the circumstances in these Chapter 11 cases; and it appearing that no other or further notice need be provided; and upon the record in these Chapter 11 cases; and after due deliberation thereon; and good and sufficient cause appearing therefor; it is hereby

ORDERED, that the Motion is granted as provided herein and the Avoidance Action Trust Administrator and Trust Monitor are authorized to enter into the Fourth Amended Avoidance Action Trust Agreement, in substantially the form attached as Exhibit C to the Motion; and it is further

ORDERED, that the Avoidance Action Trust is authorized to enter into the LW Transaction Documents, and grant a priority lien to the LW Capital Provider (junior in priority to the liens of the DIP Lenders and the CE Capital Providers under the Fourth Amended Avoidance Action Trust Agreement) in the Collateral as set forth in the LW Transaction Documents; and it is further

ORDERED, this Order shall be sufficient and conclusive evidence of the validity, enforceability, perfection, and priority of the liens granted to the LW Capital Provider without the necessity of filing or recording any financing statement or other instrument or document which may otherwise be required under the law of any jurisdiction, or taking any other action to validate or perfect the liens or to entitle the liens to the priorities granted herein; and it is further

ORDERED, that the Avoidance Action Trust and the Avoidance Action Trust Administrator are authorized to take all necessary actions to carry out this Order and execute and perform all of the obligations of the Avoidance Action Trust under the LW Transaction

Documents, including the (i) satisfaction of all conditions precedent to the transactions contemplated thereby and (ii) payment to the LW Capital Provider of any fees or expense reimbursements due under the LW Transaction Documents, but solely to the extent permitted under the LW Subordination Agreement; and it is further

ORDERED, the Avoidance Action Trust shall at all times maintain the security interest of the LW Capital Provider as set forth in the LW Capital Provision Agreement and the LW Security Agreement as a valid, perfected lien, junior in priority to the liens of the DIP Lenders and the CE Capital Providers in the Collateral and senior in priority to any lien or security interest of any other party, and it is further;

ORDERED, the Avoidance Action Trust and the Avoidance Action Trust Administrator irrevocably waive any right to seek any amendment, modification, or extension of this Order without the prior written consent of the LW Capital Provider, the DIP Lenders, and the CE Capital Providers and no such consent shall be implied by any other action, inaction, or acquiescence of the LW Capital Provider, the DIP Lenders, or the CE Capital Providers; and it is further;

ORDERED, the LW Capital Provider shall not be authorized, nor shall they be deemed, to control or direct in any manner whatsoever the method, manner and/or timing of the Avoidance Action Trust's actions in prosecuting, pursuing, negotiating and/or settling the Term Loan Avoidance Action or any contested matter, civil action, adversary proceeding or other matter of litigation to which the Avoidance Action Trust is a party; and it is further

ORDERED, that in the event of any conflicting terms between the Fourth Amended Avoidance Action Trust Agreement and the LW Transaction Documents, the Fourth Amended Avoidance Action Trust Agreement shall govern and control; and it is further

ORDERED, that the Avoidance Action Trust shall not distribute Proceeds in

09-50026-mg Doc 14413-1 Filed 02/04/19 Entered 02/04/19 15:31:37 Exhibit A Pg 5 of 5

contravention of the Fourth Amended Avoidance Action Trust Agreement; and it is further

ORDERED, that this Order shall be effective immediately upon entry; and it is further

ORDERED, that the Court shall retain jurisdiction to hear and determine any and all matters concerning this Order.

IT IS SO ORDERED.

Dated:	, 2019	
	New York, New York	
		MARTIN GLENN
		United States Bankruptcy Judge

Exhibit B

Dated as of January 21, 2019

Capital Provision Agreement

between

The Counterparty named in Annex I hereto

and

The LW Capital Provider, as defined in the introductory paragraph hereof

CONTENTS

SEC	CTION		PAGE
1.	Defin	nitions	2
2.	Capit	tal Provision Obligations and LW Capital Provider's Entitlement	3
	2.1	Capital Amounts	3
	2.2	LW Capital Provider's Entitlement	5
3.	Paym	nent Obligations of the Counterparty	6
4.	Paym	nents Generally	7
	4.1	Place and Account for Payment	7
	4.2	Interest on Overdue Amounts	7
	4.3	Waiver of Right of Set-Off	7
5.	Cove	enants	7
	5.1	Covenants of Each Party	7
	5.2	Covenants of the LW Capital Provider	8
	5.3	Covenants of the Counterparty	8
6.	Repre	esentations and Warranties	11
	6.1	Representations and Warranties of Each Party	11
	6.2	Representations and Warranties of the Counterparty	12
	6.3	Representation and Warranty of the LW Capital Provider	14
7.	Conf	identiality	14
	7.1	Exclusive Ownership of Information by Disclosing Party	14
	7.2	Non-Disclosure of Information	15
	7.3	Confidentiality Procedures	15
	7.4	Judicial and Official Disclosure Requests	15
8.	Lega	l Privilege	16
	8.1	Common Interest Privilege Applies	16
	8.2	Information Subject to Privilege Protection	16
	8.3	Information Subject to Work Product Protection	16
	8.4	No Obligation to Provide Privileged Information	17
9.	Inder	nnification of the LW Capital Provider	17

10.	Exculpation; Reinstatement	18
11.	Remedy Events	19
	11.1 Failure to Pay or Deliver	19
	11.2 Failure to Discharge Claim Costs	19
	11.3 Breach or Repudiation	19
	11.4 Misrepresentation	19
12.	Remedies	19
	12.1 Available Remedies Generally	19
	12.2 Remedies Cumulative	20
13.	Limitations on Transfer, Successors and Assigns; Third Party	• •
	Beneficiaries	
14.	Tax Withholding	20
15.	Notices	21
	15.1 Effectiveness of Notices	21
	15.2 Change of Details	21
16.	Amendments	22
17.	Entire Agreement	22
18.	Counterparts	22
19.	Survival of Obligations	22
20.	No Waiver	22
21.	Severability	22
22.	Expenses	23
23.	Relationship of Parties	23
24.	Governing Law	23
25.	Dispute Resolution	24
26.	Rules of Construction	26
27.	Termination of Agreement	26
28	Subordination Agreement	27

This CAPITAL PROVISION AGREEMENT, dated as of January 21, 2019 (this "Agreement"), is by and between Motors Liquidation Company Avoidance Action Trust, an entity organized or formed under the laws of the jurisdiction identified in Annex I (the "Counterparty"), and LW Holdco VI LLC, a limited liability company formed under the laws of the State of Delaware ("LW Capital Provider").

WHEREAS, prior to August 2016, there was a dispute between the Official Committee of Unsecured Creditors of Motors Liquidation Company (the "Committee") and the DIP Lenders as to entitlements to Avoidance Action Proceeds, which dispute was resolved by mutual agreement (the "Litigation Settlement") pursuant to which, among other things, (i) the DIP Lenders provided the Litigation Cost Advance and (ii) the DIP Lenders are entitled to be repaid the Litigation Cost Advance out of Distributable Trust Assets; and

WHEREAS, the Trust Administrator and the Trust Monitor determined that the Avoidance Action Trust Administrative Cash and Supplemental Avoidance Action Trust Cash held by the Counterparty at the time of the Litigation Settlement were insufficient to satisfy projected fees, costs and expenses of the Counterparty, and, as an integral part of the Litigation Settlement, the DIP Lenders agreed to provide the Litigation Cost Advance to the Counterparty to fund the prosecution of the Term Loan Avoidance Action, on the terms set forth in the Litigation Cost Advance Agreement;

WHEREAS, on August 24, 2016, the Bankruptcy Court entered an order approving the Settlement Agreement and the Litigation Cost Advance Agreement; and

WHEREAS, subsequent to the entering into of the Litigation Cost Advance Agreement, the Trust Administrator and the Trust Monitor determined that the Avoidance Action Trust Administrative Cash and Supplemental Avoidance Action Trust Cash held by the Counterparty was insufficient to satisfy projected fees, costs and expenses of the Counterparty, for which reason the Counterparty sought additional funding by entering into the CE Capital Provision Agreement (as defined below) with the CE Capital Providers (as defined below); and

WHEREAS, one of the preconditions to the DIP Lenders' provision of the Litigation Cost Advance was the Counterparty's agreement, as reflected in Section 6.1(d)(i) of the Trust Agreement, that any additional litigation funding provided to the Counterparty would be (i) junior and subordinate to the Litigation Cost Advance and any other amounts owed to the DIP Lenders on account of prior funding of the Counterparty and (ii) subject to a form of subordination acceptable to the DIP Lenders in all respects; and

WHEREAS, the Trust Administrator and the Trust Monitor have determined that the Avoidance Action Trust Administrative Cash and Supplemental Avoidance Action Trust Cash currently held by the Counterparty remain insufficient to satisfy projected fees, costs and expenses of the Counterparty, for which reason the Counterparty has determined to seek additional funding from the LW Capital Provider; and

WHEREAS, the Counterparty, before interacting with the LW Capital Provider in regard to the matters described herein and without any influence from the LW Capital Provider, decided to pursue one or more legal claims identified on Annex II to this Agreement; and

WHEREAS, the Counterparty has provided confidential materials protected by the attorney-client privilege and/or the attorney work product doctrine to the LW Capital Provider and its Affiliates, subject to a non-disclosure agreement and in reliance on the common interest privilege, the work product doctrine and other existing law protecting such materials from disclosure; and

WHEREAS, the LW Capital Provider and its Affiliates have conducted research and analysis and communicated views and opinions both internally and to the Counterparty, all in reliance on that same common interest privilege, work product doctrine and other existing law protecting such research, analysis and communications from disclosure; and

WHEREAS, the LW Capital Provider is a passive provider of external capital and has not become owner of, partner in or party to the Claim or any part thereof or acquired any rights as to their control or resolution, and the Counterparty remains in full control of the assertion and resolution of the Claim; and

WHEREAS, the Counterparty is sophisticated and is entering into this Agreement freely and entirely of its own volition following independent legal advice from experienced counsel, and does not believe that this Agreement or the transactions it contemplates are inconsistent with any relevant law or public policy; and

WHEREAS, pursuant to Section 6.1(d)(i) of the Trust Agreement, the entry into the Subordination Agreement (as defined herein), by and among the DIP Lenders, the LW Capital Provider, and the Counterparty, is a precondition of the DIP Lenders' consent to the LW Capital Provider's provision of the Supplemental Capital to the Counterparty pursuant to this Agreement.

NOW, THEREFORE, in consideration of the mutual agreements contained herein, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, hereby agree as follows:

1. **DEFINITIONS**

Certain capitalized terms used herein have the meanings assigned thereto in <u>Exhibit A</u>. Capitalized terms used but not otherwise defined in <u>Exhibit A</u> have the respective meanings assigned to such terms elsewhere in this Agreement or in the Trust Agreement.

2. CAPITAL PROVISION OBLIGATIONS AND LW CAPITAL PROVIDER'S ENTITLEMENT

2.1 Capital Amounts

The LW Capital Provider shall provide capital to the Counterparty as set forth below.

- (a) Capital Amounts:
- (i) Subject to satisfaction of conditions set forth in clause (e) of this Section 2.1 and as otherwise set forth in this Agreement, the LW Capital Provider shall provide to the Counterparty up to Ten Million United States Dollars (US\$10,000,000), until the earlier of (A) the one year anniversary of the Initial Funding Date (as defined below) and (B) the commencement of the Phase III Trial. The Counterparty agrees to accept, at a minimum, Capital Amounts in the aggregate amount of One Million Five Hundred Thousand United States Dollars (US\$1,500,000).
- (ii) The LW Capital Provider shall provide the first \$1,500,000 to the Counterparty upon receipt of the Approvals (as defined below) (the date of such initial funding, the "*Initial Funding Date*"). The remaining committed Capital Amounts shall be paid to the Counterparty, in tranches of no less than One Million Dollars (\$1,000,000) each, at such times as the Counterparty may determine in its sole discretion; *provided*, *that*, if the amount available for the final tranche is less than \$1,000,000, such tranche shall be for such lesser amount.
- (iii) The sum of all amounts actually paid by the LW Capital Provider to the Counterparty hereunder constitutes the "Invested Amount".
- (b) The LW Capital Provider shall make each payment of a Capital Amount owing to the Counterparty hereunder by the tenth (10th) Business Day following the relevant due date.
- (c) The LW Capital Provider shall make each such payment of a Capital Amount to a segregated capital provider funding account established at Wilmington Trust Company (the "LW Capital Provider Funding Account").
- (d) Subject to the requirements of Section 5.3(c)(iii) hereof and the Trust Agreement, the Trust shall be entitled to make withdrawals from the LW Capital Provider Funding Account at any time and in any amount determined by the

Trust Administrator to be reasonable and appropriate to meet the needs of the Trust.

- (e) The LW Capital Provider's obligation to provide the Capital Amounts shall be subject to satisfaction of the following conditions:
 - (i) the Trust Administrator shall have made an application to the United States Bankruptcy Court for the Southern District of New York (the "Bankruptcy Court") seeking approval of the transactions contemplated hereby, in accordance with Section 6.1(d) of the Trust Agreement, including a statement of the Trust Monitor's position in respect thereof and proposed amendments to the Trust Agreement, in form and substance approved in writing by the LW Capital Provider, to implement the terms of this Agreement (the "Bankruptcy Court Application");
 - (ii) the Trust Administrator shall have given proper notices to the parties entitled to receive the same, pursuant to Section 6.1(d)(ii) of the Trust Agreement, of any hearing on the matter of selling or granting liens;
 - (iii) the Bankruptcy Court shall have issued an order: (A) (1) approving the transactions contemplated hereby, in such form as is necessary to ensure the LW Capital Provider of the validity, perfection and second priority status of its lien, granted pursuant to the Security Agreement described on Annex II to this Agreement, (2) authorizing the payment to the LW Capital Provider of the LW Capital Provider's Entitlement as and when due pursuant to the terms of this Agreement and the Trust Agreement, and (3) providing that the Counterparty shall not otherwise distribute Proceeds in contravention of the Trust Agreement, and (B) approving the amendments to the Trust Agreement described in the preceding clause 2.1(e)(i) hereof;
 - (iv) the Trust Administrator shall have delivered to the LW Capital Provider a certificate signed by an authorized officer thereof stating without qualification that the Trust has fully complied with the DIP Lenders' right of first refusal pursuant to Section 6.1(d)(i) of the Trust Agreement and Section 12 of the Litigation Cost Advance Agreement; and
 - (v) the DIP Lenders shall have provided their consent to the transactions contemplated hereby, pursuant to Section 6.1(d)(i) of the Trust Agreement and Section 12 of the Litigation Cost Advance Agreement, and the DIP Lenders and the CE Capital Providers shall have provided their consent that the Subordination Agreement is acceptable, or, with respect to the CE Capital Providers, the Trust Administrator and the Trust Monitor have determined that the CE Capital Providers have unreasonably withheld consent to the Subordination Agreement, pursuant to Section 6.1(d)(i) of the Trust Agreement, is acceptable (all

consents, approvals and other documents required pursuant to this Section 2.1(e), collectively, the "*Approvals*").

2.2 LW Capital Provider's Entitlement

In consideration of their agreement to provide the Capital Amounts, the LW Capital Provider shall be entitled to receive the amounts set forth below.

- (a) LW Capital (i) Amount. The LW Capital Provider's Entitlement shall Provider's be:
 Entitlement:
 - (A) the product of the Invested Amount <u>multiplied by</u> 2 (two) plus
 - (B) beginning on the 12-month anniversary of the Initial Funding Date, an annual interest rate of fifteen percent (15.0%), compounded annually, accruing on the amounts outstanding under clause (A) above until the LW Capital Provider's Entitlement is paid in full to the LW Capital Provider.
 - (ii) Timing and Priority. Subject to and in accordance with the terms of the Trust Agreement, the Counterparty shall pay the LW Capital Provider's Entitlement as a distribution of Distributable Trust Assets, on applicable Distribution Dates, in the order of priority set forth in Section 5.1(d) of the Trust Agreement, which priority for the distribution of Distributable Trust Assets shall be: (A) first, to the DIP Lenders, each in their relative proportion pursuant to the terms of the DIP Credit Agreement, in the amount of the DIP Lender Advances, until such obligation is satisfied; (B) second, to the CE Capital Providers, to pay the CE Capital Providers' Entitlement, until such obligation is satisfied; (C) third, to the LW Capital Provider, to pay the LW Capital Provider's Entitlement, until such obligation is satisfied; (D) fourth, to the Segregated Account (as defined in the Trust Agreement) in the amount of the GUC Trust Advances to be distributed to the holders of Allowed General Unsecured Claims, until such obligation is satisfied; and (E) fifth, to the DIP Lenders, each in their relative proportion pursuant to the terms of the DIP Credit Agreement, and to the holders of Allowed General Unsecured Claims.

(iii) Obligation Absolute. The payment obligation to the LW Capital Provider, to the extent not subject to a reasonable dispute among the parties, shall be absolute and unconditional and shall not under any circumstances, except pursuant to Section 25(c) of this Agreement, be delayed, suspended or avoided.

3. PAYMENT OBLIGATIONS OF THE COUNTERPARTY

- (a) Non-Recourse Agreement; Circumstances of All Payment Obligations. The Capital Amounts are provided to the Counterparty on a non-recourse basis, and the LW Capital Provider's Entitlement is derived from, computed on the basis of and paid solely from Proceeds recovered on or after the first Capital Amounts are provided. If the amount of Proceeds is not sufficient to pay the DIP Lenders pursuant to Section 5.1(d) of the Trust Agreement and Section 2.2(a)(ii)(A) of this Agreement, or the CE Capital Providers pursuant to Section 5.1(d) of the Trust Agreement and Section 2.2(a)(ii)(B) of this Agreement, then the Counterparty shall not have any obligation to pay the LW Capital Provider's Entitlement (including any repayment of the Invested Amount). Other than any payment obligation that arises under Section 2.2 of this Agreement, the only other circumstances under which the Counterparty may be obligated to make a payment to the LW Capital Provider hereunder are set forth in Sections 4.2, 9, 10(d), 12, 22 and 27 of this Agreement.
- (b) Notice. If at any time Proceeds arise, the Counterparty shall as soon as practicable notify the LW Capital Provider of the amount of such Proceeds. Upon the request of the LW Capital Provider, the Counterparty shall promptly provide to the LW Capital Provider in writing a calculation of the portion of the LW Capital Provider's Entitlement payable to the LW Capital Provider from the Proceeds.
- (c) Application of Payments. The Counterparty's payment of any and all portions of the LW Capital Provider's Entitlement shall be applied as follows: (i) first, towards payment of the returns on the applicable Capital Amounts, as specified in Section 2 of this Agreement, and (ii) second, towards the repayment of Capital Amounts paid by the LW Capital Provider to the Counterparty.
- (d) Termination Fee. In exchange for the LW Capital Provider reserving the Capital Amounts as necessary to satisfy its obligations hereunder, in the event that (i) the Counterparty terminates the LW Capital Provision Agreement pursuant to Section 27(b) of this Agreement; or (ii) the Counterparty fails to secure the Approvals within forty-five (45) days of this Agreement for any reason and the LW Capital Provider terminates this Agreement pursuant to Section 27(c) of this Agreement, the Counterparty shall be obligated to pay to the LW Capital Provider the Termination Fee pursuant to the terms of the Termination Fee Agreement included as Exhibit A to this Agreement (the "Termination Fee Agreement").

4. PAYMENTS GENERALLY

4.1 Place and Account for Payment

Each payment to a party required under this Agreement shall be made (a) to the payment account for such party specified on Annex II to this Agreement; (b) in the currency specified in Section 2 of this Agreement; (c) in the place where such account is located; (d) unless otherwise agreed to in writing by the LW Capital Provider, by wire transfer of freely transferable and immediately available funds; and (e) otherwise in the manner customary for payments in the specified currency.

4.2 Interest on Overdue Amounts

If the Counterparty defaults in the performance of any payment obligation under this Agreement, the Counterparty shall on demand pay interest (before as well as after judgment, if applicable) at the Default Rate on the overdue amount to the LW Capital Provider for the period from (and including) the original due date for payment to (but excluding) the date of actual payment.

4.3 Waiver of Right of Set-Off

Each amount that the Counterparty is obligated to pay under this Agreement shall be paid without Set-off or other deduction.

5. COVENANTS

5.1 Covenants of Each Party

The Counterparty, on the one hand, and the LW Capital Provider, on the other hand, agrees that, so long as such party has or may have any obligation under this Agreement or any other Transaction Document to the other:

- (a) It shall preserve and maintain its corporate existence, except to the extent that the failure to do so would not have a Material Adverse Effect.
- (b) It shall use all reasonable efforts to maintain in full force and effect all consents, approvals, actions, authorizations, exceptions, notices, filings and registrations of or with any Governmental Authority that are required to be obtained by it with respect to this Agreement or any other Transaction Document and shall use all reasonable efforts to obtain any that may become necessary in the future.
- (c) It shall comply with all applicable laws and orders of any Governmental Authority to which it may be subject if failure so to comply could reasonably be expected, either individually or in the aggregate, to have a Material Adverse Effect.

5.2 Covenants of the LW Capital Provider

With respect to each Claim, the LW Capital Provider agrees with the Counterparty that:

- (a) the LW Capital Provider is not, and shall not by virtue of entering into this Agreement or any other Transaction Document become, a party to such Claim; and
- (b) the LW Capital Provider shall have no authority to control any litigation affecting the Counterparty, including but not limited to the Claim, and the decisions of the Counterparty, the Trust Administrator and the Trust Monitor regarding the conduct of any such litigation, including but not limited to a settlement of all or a portion of the Claim, shall be conclusive and binding upon the LW Capital Provider.

5.3 Covenants of the Counterparty

The Counterparty agrees with the LW Capital Provider that:

- (a) Subject to Section 8 of the Trust Agreement, the Counterparty shall use its best efforts and exercise good faith and reasonable commercial judgment, but at all times within the bounds of any applicable law and rules of professional responsibility, to (i) pursue the Claim; (ii) bring about the reasonable monetization thereof through a Claim Resolution; (iii) collect and enforce any settlement, final judgment or award (all costs of doing the foregoing, plus Trust Administration Expenses, collectively, "Claim Costs"); (iv) manage Claim Costs incurred in doing the foregoing and promptly discharge the same.
- (b) The Counterparty shall use its best efforts to obtain the Approvals as soon as practicable, and shall file the Bankruptcy Court Application as soon as reasonably practicable, but within ten (10) Business Days in any event, after the full execution of this Agreement.
- (c) So long as the Counterparty has or may have any obligation under this Agreement or any other Transaction Document, the Counterparty shall:
 - (i) comply in all material respects with the terms and conditions of the Trust Agreement, the Litigation Cost Advance Agreement and the Settlement Agreement;
 - (ii) provide annual Budgets and quarterly updates in accordance with Section 6.3 of the Trust Agreement. The quarterly updates shall reconcile actual expenditures with projections that had been included in the relevant Budget, and, if and when the Trust identifies a need for or the occurrence of actual expenditures that materially differ from the relevant Budget, they will notify the LW Capital Provider as promptly as practicable, and no later than 15 days after identifying the issue;
 - (iii) provide in the Trust Agreement that Professionals and any other persons

entitled to payment from the Trust (other than the Trust Administrator and the Trust Monitor) shall be required to submit reasonably detailed invoices on a monthly basis to the Trust Administrator, the Trust Monitor and the LW Capital Provider, including in such invoices a description of the work performed, the individuals who performed such work, and the hourly rate of such Person, plus an itemized statement of expenses for which payment is sought. Invoices submitted to the LW Capital Provider may be in a form reasonably necessary to protect the attorney-client privilege. For the avoidance of doubt, any and all payments by the Trust from funds that originated from the Capital Amounts shall be subject to this invoice disclosure requirement. No payment on account of such invoices shall be made until 15 days after presentation of the invoices to the LW Capital Provider. The Trust Administrator shall timely pay all such invoices that are not disputed by the Trust Administrator or the Trust Monitor. If the LW Capital Provider has questions or concerns about any invoice submitted to the Trust, or about any other issue relating to the Trust's finances or administration, it shall be entitled to discuss the matter with the Trust Monitor, as well as with the Trust Administrator and/or counsel for the Trust, as appropriate, and the Trust Monitor shall consult with the LW Capital Provider and give good faith consideration to any objection of the LW Capital Provider before approval of payment by the Trust Administrator or non-objection by the Trust Monitor;

- (iv) open the LW Capital Provider Funding Account and provide the LW Capital Provider with monthly account statements for the LW Capital Provider Funding Account, and provide immediate access to any or all LW Capital Provider Funding Account records upon request by the LW Capital Provider;
- (v) subject to Section 8.4 of this Agreement, (A) shall itself, and cause the applicable Nominated Lawyers to, keep the LW Capital Provider fully and promptly apprised of each material development in relation to the Claim and the Trust and respond fully and promptly to any request by the LW Capital Provider and its Affiliates or Representatives for any information regarding such Claim and (B) provide to the LW Capital Provider copies of all reports provided by it to Trust Beneficiaries pursuant to the Trust Agreement contemporaneously with delivery to such Trust Beneficiaries;
- (vi) provide notice by email as soon as practicable to the LW Capital Provider of a Claim Resolution or any receipt of any Proceeds, or of any event that is expected to generate a Claim Resolution or the receipt of any Proceeds;
- (vii) other than pursuant to the Litigation Cost Advance Agreement, the Settlement Agreement, Section 6.1(d) of the Trust Agreement, any Security Agreement or any Engagement Agreement(s), and with the prior written consent of the LW Capital Provider, (A) not dispose of, transfer, encumber

or assign, nor otherwise create, incur, assume or permit to exist any Adverse Claim with respect to, all or any portion of such Claim (or any interest therein) or any Proceeds thereof (or any right to such Proceeds); or (B) not increase the amount of existing indebtedness that is secured by the collateral under any Security Agreement, if, in the case of either (A) or (B), doing so would either (x) create a lien senior to or *pari passu* with the lien securing the LW Capital Provider's Entitlement or, (y) except as set forth in Section 2.7 of the Subordination Agreement, result in the subordination of any portion of the payment of the LW Capital Provider's Entitlement pursuant to Section 5.1(d) of the Trust Agreement;

- (viii) not permit or apply any Set-off or agree to permit or apply any Set-off against such Claim;
 - (ix) not use Capital Amounts to fund distributions to other creditors (including the DIP Lenders and the CE Capital Providers) or, after a Claim Resolution, to pay Trust Professionals (as defined in the Trust Agreement) without the prior written consent of the LW Capital Provider; and
 - (x) not amend or seek to amend the Trust Agreement without the prior written consent of the LW Capital Provider if such amendment(s) would negatively impact the amount or payment priority of the LW Capital Provider's Entitlement or modify the treatment of Capital Amounts as Other Supplemental Cash, as such term is defined in the Trust Agreement.
- (d) Within five (5) Business Days after the Counterparty knows or has reason to believe that any of the following has occurred or is likely to occur, the Counterparty shall deliver to the LW Capital Provider a notice describing the same in reasonable detail and, together with such notice or as soon thereafter as possible, a description of any action that the Counterparty has taken or proposes to take with respect thereto:
 - (i) any Potential Remedy Event or Remedy Event;
 - (ii) any action, suit or proceeding of the kind described in Section 6.2(b) of this Agreement has been or shall be brought; and
 - (iii) a material failure by the Counterparty, or the Nominated Lawyers on the Counterparty's behalf, to pay when due any undisputed amounts owing to any litigation services providers (e.g., experts) retained in connection with a Claim (after giving effect to any applicable notice requirement or grace period).
- (e) If the Counterparty replaces the Nominated Lawyers for any reason or retains additional legal counsel for the Claim, the Counterparty shall promptly deliver to the LW Capital Provider copies of any Engagement Agreements with any successor or new attorneys, and all references to the "Nominated Lawyers" herein shall be

deemed to refer to and include such successor or new counsel.

(f) The Counterparty shall not, without the LW Capital Provider's prior written consent, (x) amend the economic terms of any Engagement Agreement if such amendment(s) would, directly or indirectly, negatively impact the amount or payment priority of the LW Capital Provider's Entitlement or (y) enter into any new Engagement Agreements that would have such an effect.

6. REPRESENTATIONS AND WARRANTIES

6.1 Representations and Warranties of Each Party

On each Representation Date, the Counterparty, on the one hand, and the LW Capital Provider, on the other hand, represents and warrants to the other as follows:

- (a) It (i) is duly organized or formed and validly existing under the laws of the jurisdiction of its organization or formation and, if relevant under such laws, in good standing, (ii) is qualified to do business in each jurisdiction in which the nature of its business so requires, and (iii) has not filed any certificates of dissolution or liquidation, or any certificates of domestication, transfer or continuance in any jurisdiction.
- (b) It has the power to execute this Agreement and the other Transaction Documents to which it is a party, to deliver this Agreement and the other Transaction Documents it is required by this Agreement to deliver, and to perform its obligations under this Agreement and the other Transaction Documents; and it has taken all necessary action to authorize such execution, delivery and performance.
- (c) Such execution, delivery and performance do not and shall not violate or conflict with any law applicable to it, any provision of its constitutional documents, any order or judgment of any court or other Governmental Authority applicable to it or any of its assets or any contractual restriction binding on or affecting it or any of its assets (including, with respect to the Counterparty, the Engagement Agreement(s) and any provisions therein).
- (d) All consents, approvals, actions, authorizations, exceptions, notices, filings and registrations that are required to have been obtained by it with respect to this Agreement or any other Transaction Document, other than the Approvals, have been obtained and are in full force and effect, and all conditions of any such consents, approvals, actions, authorizations, exceptions, notices, filings and registrations have been complied with.
- (e) This Agreement and the other Transaction Documents constitute its legal, valid and binding obligations, enforceable in accordance with their respective terms (subject to applicable bankruptcy, reorganization or similar laws and to general equitable principles).

- (f) It is acting for its own account, and it has made its own independent decisions to enter into this Agreement and the other Transaction Documents, and the transactions contemplated hereby and thereby, and as to whether this Agreement and the other Transaction Documents and such transactions are appropriate or proper for it based upon its own judgment and upon advice from such advisers as it has deemed necessary.
- (g) No communication (written or oral) received from the other party shall be deemed to be an assurance or guarantee as to the expected results of this Agreement or any other Transaction Document, or any of the transactions contemplated hereby or thereby.
- (h) It is capable of assessing the merits of and understanding (on its own behalf or through independent professional legal advice), and understands and accepts, the terms, conditions and risks of this Agreement and the other Transaction Documents, and the transactions contemplated hereby and thereby.

6.2 Representations and Warranties of the Counterparty

On each Representation Date, the Counterparty represents and warrants to the LW Capital Provider as follows:

- (a) No Remedy Event or Potential Remedy Event has occurred and is continuing and no such event or circumstance would occur as a result of its entering into or performing its obligations under this Agreement or any other Transaction Document.
- (b) No litigation or other proceedings before any court or other Governmental Authority, official, tribunal or arbitrator that could reasonably be expected, either individually or in the aggregate, to have a Material Adverse Effect, have been commenced by or against the Counterparty or, to the best of the Counterparty's knowledge, are threatened against, the Counterparty, any other Person or any collateral under any Security Agreement.
- (c) It has provided the most recent financial reporting for the Trust, which financial reporting accurately reflects the present financial condition and budget of the Trust. As of the date of this Agreement, such financial reporting includes the Q3 2018 balance sheet and income statement and the 2019 budget.
- (d) It is not materially overdue in the filing of any Tax return nor overdue in the payment of any amount in respect of Tax.
- (e) It has, and reasonably expects to continue to have, sufficient assets (taking into account the Capital Amounts) available to it with which to discharge such Claim Costs as may be necessary to comply with its covenants under this Agreement.

- (f) As of the date of this Agreement, it has entered into the Engagement Agreements in respect to the Claim as described on <u>Annex II</u> to this Agreement.
- (g) With respect to the Claim:
 - (i) it is the sole legal and beneficial owner of, has good title to, and possesses sole control of, the Claim, free and clear of any Adverse Claim, other than pursuant to Article V of the Trust Agreement, the Litigation Cost Advance Agreement, the Settlement Agreement, any CE Security Agreement or any Security Agreement;
 - (ii) it has full power and authority to bring such Claim and has obtained all necessary corporate, contractual and other authorizations to do so;
 - (iii) other than pursuant to this Agreement, the CE Capital Provision Agreement, the Litigation Cost Advance Agreement, the Settlement Agreement, any CE Security Agreement, any Security Agreement or any Engagement Agreement or as described in Section 6.1(d) of the Trust Agreement, it has not disposed of, transferred, encumbered or assigned all or any portion of such Claim (or any interest therein) or any Proceeds thereof, whether by way of security, subrogation, assignment to an insurer or otherwise;
 - (iv) to the extent that the Counterparty purports to grant to the LW Capital Provider any security interest in such Claim, any proceeds thereof and/or related collateral pursuant to any Security Agreement (without limiting any representation or warranty made by the Counterparty in such Security Agreement), such security interest is a legal, validly created, perfected security interest, with priority over all other secured parties other than the DIP Lenders and the CE Capital Providers;
 - (v) it has not permitted or applied any Set-off nor agreed to permit or apply any Set-off against such Claim, and there exist no rights of Set-off or similar rights against the Counterparty that could permit any Set-off against such Claim;
 - (vi) it has not taken any steps or executed any documents, nor is it aware of any asserted or unasserted claim, lien or judgment against it, which could reasonably be expected, either individually or in the aggregate, to have a Material Adverse Effect, and it is not aware of anyone else doing or purporting to do so;
 - (vii) the Counterparty believes (and does not have, and has not been informed by any of its Representatives of, any belief to the contrary) that such Claim is meritorious and likely to succeed; and

- (viii) any proceeds arising directly or indirectly from the Reconstitution Claims shall constitute Avoidance Action Proceeds, as defined in the Trust Agreement.
- (h) It is not relying on any communication (written or oral) of the LW Capital Provider or any of the LW Capital Provider's Affiliates as legal advice or as a recommendation to enter into this Agreement or any other Transaction Document, or any of the transactions contemplated hereby or thereby, it being understood that information and explanations related to the terms and conditions of this Agreement and the other Transaction Documents, and the transactions contemplated hereby and thereby, shall not be considered legal advice or as such a recommendation.
- (i) Schedule 6.2(i) sets forth, as of January 15, 2019, the amounts which, if recovered, would be due pursuant to the Trust Agreement from Distributable Trust Assets: (i) to the DIP Lenders pursuant to the Litigation Cost Advance Agreement; (ii) to the CE Capital Providers pursuant to the CE Capital Provision Agreement; and (iii) to the GUC Trust for DIP Lender Advances. The amounts set forth in Schedule 6.2(i) are a representation by the Counterparty and are not binding on the DIP Lenders or CE Capital Providers.

6.3 Representation and Warranty of the LW Capital Provider

On each Representation Date, the LW Capital Provider represents and warrants to the Counterparty that it has sufficient access to capital to enable it to fulfill its funding commitment to the Counterparty under this Agreement.

7. CONFIDENTIALITY

7.1 Exclusive Ownership of Information by Disclosing Party

Each recipient of Confidential Information hereunder (the "Recipient") agrees that all Confidential Information provided to it is and shall remain at all times the exclusive property of and owned by the disclosing party (the "Disclosing Party"), or any of its Affiliates or contract counterparties, as the case may be, and that the Recipient's use or awareness of such Confidential Information shall create no rights, at law or in equity, in the Recipient in or to such information, or any aspect or embodiment thereof. Neither the execution of this Agreement or any other Transaction Document, nor the furnishing of any Confidential Information hereunder, shall be construed as granting, whether expressly or by implication, estoppel or otherwise, any license to distribute or title to any patent, trademark, copyright, service mark, business and trade secret or other proprietary right, title or interest in or to such Confidential Information, or to use such Confidential Information for any purpose other than as specified in this Agreement or any other Transaction Document, or to constitute a waiver of any attorney-client privilege or work product protection.

7.2 Non-Disclosure of Information

The Recipient shall not for any reason, (a) until the later of (i) the seventh (7th) anniversary of the date of this Agreement, or (ii) the second (2nd) anniversary of a Claim Resolution, disclose, reveal, report, publish, transfer or make available, directly or indirectly, to any Person other than its Representatives authorized pursuant to Section 7.3 of this Agreement, nor use for any purpose other than the fulfillment of its obligations hereunder (collectively, "Disclose"), any Confidential Information provided to it; nor (b) in perpetuity, Disclose any Common Interest Material provided to it. Moreover, notwithstanding any other provision of this Agreement, at no time shall the Recipient knowingly disclose any Confidential Information or Common Interest Material to the Opponent or any other adversary, potential adversary or conduit to an adversary of the Counterparty, and the Recipient shall treat such materials in a manner that does not substantially increase the likelihood that any of the foregoing shall come into possession of it. Notwithstanding the foregoing, the LW Capital Provider shall be free (A) following a Claim Resolution, to disclose publicly its involvement in the Claim and (B) at any time, to make disclosures as part of any public reporting obligations.

7.3 Confidentiality Procedures

The Recipient shall ensure that the Confidential Information it receives is not divulged or disclosed to any Person except its Representatives who have a "need to know" such information in order to perform their job responsibilities. The Recipient shall procure its Representatives' compliance with Section 7.2 of this Agreement and all other provisions of this Agreement and shall be responsible for its or its Representatives' failure to so comply.

7.4 Judicial and Official Disclosure Requests

If a Recipient receives a request, including in any judicial, arbitral or administrative proceeding or by any Governmental Authority, to disclose any Confidential Information, then such Recipient, before complying with such request, shall, unless prohibited from doing so by applicable law or court order, promptly provide written notice of such request, including a copy of such request, to the Disclosing Party, sufficiently in advance, so that the Disclosing Party may contest the requested disclosure. If, upon such notice, the Disclosing Party elects to contest the requested disclosure (all such contests being at the Disclosing Party's expense and under its control; provided that where the Counterparty is the Disclosing Party, the Counterparty shall consult with the LW Capital Provider about such contests), the Recipient shall not disclose any Confidential Information until such time as an order has been entered compelling such disclosure, and the Recipient shall cooperate with the Disclosing Party in its contest. Moreover, if any such request for the disclosure of Confidential Information seeks disclosure of this Agreement or any other Transaction Document, any terms hereof or thereof, the identities of the LW Capital Provider or any communications between the Counterparty and the LW Capital Provider, then the Recipient shall promptly notify the LW Capital Provider and shall object to such disclosure on all applicable bases, including work product doctrine, common interest privilege and relevance, as applicable, and shall abide by the instructions of the LW Capital Provider with respect to the management of such objections, including using counsel nominated by the LW Capital Provider with experience in addressing such objections for that narrow purpose. Should the Disclosing Party not contest the requested disclosure, the Recipient shall not have any obligation to do so; the Recipient may, however, contest the requested disclosure even if the Disclosing Party elects not to do so. Notwithstanding anything herein to the contrary, (i) any disclosure in connection with the Claim ordered by the United States Bankruptcy Court for the Southern District of New York or any other court of competent jurisdiction is permitted upon notice to the other party; and (ii) the obligations of the Recipient in connection with requests for the disclosure of Confidential Information described in this Section 7.4 shall not apply with respect to requests made by a regulatory or self-regulatory body having jurisdiction over the Recipient when such disclosures are required by law as a matter of general or routine examinations in which no specific request is made for Confidential Information relating to the Disclosing Party or the Claim(s).

8. LEGAL PRIVILEGE

8.1 Common Interest Privilege Applies

The parties agree that the Counterparty, the LW Capital Provider and the LW Capital Provider's Affiliates have a "common legal interest" in each Claim and its successful prosecution, this Agreement and the other Transaction Documents, and any discussion, evaluation and negotiation and other communications and exchanges of information relating thereto.

8.2 Information Subject to Privilege Protection

Notwithstanding any contrary provision of this Agreement, the parties agree that any Common Interest Material shall at all times remain subject to all applicable Privileges, it being the express intent of the parties and their Affiliates to preserve intact to the fullest extent applicable, and not to waive, by virtue of this Agreement or otherwise, in whole or in part, any and all Privileges to which Common Interest Material, or any part of it, is, may be or may in the future become subject. It is the good faith belief of the Disclosing Party that Privileges, including the common interest privilege, attach to the Common Interest Material and disclosure of Common Interest Material is made in reliance on that good faith belief.

8.3 Information Subject to Work Product Protection

The parties agree that without limiting the foregoing, any materials prepared in anticipation of litigation by or for the Counterparty, the LW Capital Provider, any of their respective Affiliates or any of their respective Representatives shall remain subject to work product protection. The LW Capital Provider and its Affiliates shall be considered "representatives" of the Counterparty given that the Counterparty requested assistance and advice from them regarding the plausibility of the Counterparty's obtaining external capital

to finance its legal claims. Moreover, the parties agree that information shared between the Counterparty and the LW Capital Provider is shared pursuant to a common interest and non-disclosure agreement, and the exchange of such information does not increase the risk of disclosure, inadvertent or otherwise, to the Opponent or any other adversary and does not lessen or waive the protection secured under work product doctrine. Disclosure of work product is made in reliance on that good faith belief.

8.4 No Obligation to Provide Privileged Information

Notwithstanding other provisions of this Agreement, the Counterparty is not obliged to provide to the LW Capital Provider any information that is subject to attorney-client privilege; <u>provided</u> that if the Counterparty withholds from the LW Capital Provider information otherwise required by this Agreement on the basis of attorney-client privilege, any other Privilege or any similar doctrine of confidentiality, it shall provide prompt notice thereof to the LW Capital Provider.

9. INDEMNIFICATION OF THE LW CAPITAL PROVIDER

The Counterparty agrees to indemnify, defend and hold the LW Capital Provider and its Representatives ("*Indemnitees*") free and harmless from and against any and all losses, costs, charges, damages and expenses (including reasonable attorneys' fees and costs of experts and advisors) which any Indemnitee sustains at any time by reason of one or more of:

- (a) any costs, sanctions, awards or penalties assessed or awarded against the LW Capital Provider in connection with any Claim (other than in connection with any dispute with the Counterparty), in each case which results from any act committed by the Counterparty; and
- (b) any claim by any agent or broker claiming to represent the Counterparty for compensation on account of the transactions contemplated by this Agreement.

Any amounts owed as a result of the indemnification obligations set forth in this Section 9 are independent of the Counterparty's obligation to pay the LW Capital Provider's Entitlement. Any party who receives notice of a claim for which it shall seek indemnification hereunder (the "Indemnified Party") shall promptly notify the party from which the Indemnified Party shall seek indemnification (the "Indemnifying Party") of such claim in writing. The Indemnifying Party shall have the right to assume the defense of such action at its cost with counsel approved by the Indemnified Party but shall not have the right to settle or compromise any claim or action without the written consent of the Indemnified Party. The Indemnified Party shall have the right to participate in such defense with its own counsel at its cost. Any failure by the Indemnified Party to give prompt notice of a claim hereunder shall not relieve the Indemnifying Party of its indemnification obligation except to the extent the Indemnifying Party is actually prejudiced by such failure.

10. EXCULPATION; REINSTATEMENT

- (a) Other than its obligations to provide the Capital Amounts to the Counterparty, the LW Capital Provider shall not have any obligation to fund any fees, expenses or other sums in relation to any Claim, and all such fees, expenses or other sums shall be the sole responsibility of the Counterparty. Without limiting the generality of the foregoing, the LW Capital Provider shall not have any obligation to pay any sums awarded against, or penalties incurred by, the Counterparty, including any costs orders, awards, interest, damages, expenses or penalties against the Counterparty, nor to fund any legal fees or any other costs whatsoever incurred as a result of defending any counterclaim brought against the Counterparty in relation to any Claim or defending any enforcement or other proceedings against the Counterparty.
- (b) Subject to Section 10(c) of this Agreement, the liability of the LW Capital Provider to the Counterparty under this Agreement and the other Transaction Documents, and the transactions contemplated hereby and thereby, in relation to any Claim is equal to the aggregate of the Capital Amounts committed by the LW Capital Provider in relation to such Claim.
- (c) There shall be no other liability of either LW Capital Provider to the Counterparty hereunder or otherwise in connection with the transactions contemplated hereby or any of the activities of the parties related thereto, except in the event of fraud of the LW Capital Provider or reckless activity of the LW Capital Provider amounting to fraud. This limitation of liability is absolute and applies to liability for (without limitation) breach of contract, negligence and gross negligence, and for any damages that may constitute compensatory damages, lost profit, expenses, costs, losses or charges, or consequential, exemplary or punitive damages or otherwise. This limitation of liability extends to the LW Capital Provider and its Representatives.
- To the extent any payment received by the LW Capital Provider, or obligation (d) incurred by the Counterparty pursuant to any of the Transaction Documents, is subsequently invalidated, declared to be fraudulent or preferential, set aside or required to be repaid in whole or in part by the LW Capital Provider or paid over to a trustee, receiver or any other entity, whether under any bankruptcy law, insolvency, fraudulent transfer law or otherwise (any such payment or obligation being hereinafter referred to as a "Challenged Item"), then the obligations of the Counterparty pursuant to this Agreement with respect to such Challenged Item shall (i) be fully reinstated and revived, as the case may be, notwithstanding such payment or incurrence, and (ii) to the extent of each such Challenged Item, remain effective and continue in full force and effect as if said Challenged Item had not occurred or been made. The Counterparty shall indemnify the LW Capital Provider on demand for all reasonable third party costs and expenses (including the reasonable fees and disbursements of outside counsel) incurred by the LW Capital Provider in connection with such rescission or revival, if caused by a wrongful act

committed by the Counterparty, including any such costs and expenses incurred in defending against any claim alleging that such payment constituted a preference, fraudulent transfer or similar payment.

11. REMEDY EVENTS

The occurrence at any time with respect to the Counterparty of any of the following events constitutes a "*Remedy Event*":

11.1 Failure to Pay or Deliver

The Counterparty fails to make, when and where due, any payment under Section 3 of this Agreement.

11.2 Failure to Discharge Claim Costs

The Counterparty fails to comply with, or perform any of its obligations to discharge Claim Costs under, Section 5.3(a) of this Agreement.

11.3 Breach or Repudiation

The Counterparty fails to comply with or perform any of its obligations in accordance with this Agreement, any Security Agreement or any other Transaction Document if (a) such failure, if remediable, is not remedied within five (5) Business Days after written notice thereof is given to the Counterparty and (b) (i) such failure could reasonably be expected to have a Material Adverse Effect; (ii) such failure could reasonably be expected to cause the expiration or termination of any Security Agreement or the failing or ceasing of any Security Agreement to be in full force and effect, in each case (A) other than in accordance with its terms and (B) without the written consent of the LW Capital Provider; or (iii) the Counterparty or any of its Representatives disaffirms, disclaims, repudiates or rejects in writing, in whole or in part, or challenges the validity of, this Agreement, any Security Agreement or any other Transaction Document.

11.4 Misrepresentation

A representation by the Counterparty in this Agreement or any other Transaction Document proves to have been incorrect or misleading in any material respect when made or deemed to have been made.

12. REMEDIES

12.1 Available Remedies Generally

If at any time a Remedy Event has occurred with respect to one or more Claims, the LW Capital Provider shall be entitled to exercise one or more of the following remedies (in each case without thereby relieving the Counterparty of any of its obligations under Section 3 of this Agreement):

- (a) the LW Capital Provider may by notice to the Counterparty suspend or terminate all or any portion of the obligations of the LW Capital Provider hereunder to pay Capital Amounts due and payable on or after the date such Remedy Event occurred, regardless of which Claim(s) such Capital Amounts may relate to or be used for;
- (b) with respect to those Claim(s) to which the Remedy Event relates, the LW Capital Provider may exercise any rights and remedies available to the LW Capital Provider under any Security Agreement; and
- (c) the LW Capital Provider may pursue all other legal and equitable remedies available to the LW Capital Provider under applicable law in connection with the enforcement of their rights under this Agreement and the other Transaction Documents.

12.2 Remedies Cumulative

Except as provided otherwise in this Agreement, the rights, powers, remedies and privileges provided in this Agreement and the other Transaction Documents are cumulative; may be exercised singularly, concurrently or successively at the LW Capital Provider's option; and may be exercised or enforced without constituting a bar to the exercise or enforcement of any other such rights, powers, remedies and privileges.

13. LIMITATIONS ON TRANSFER, SUCCESSORS AND ASSIGNS; THIRD PARTY BENEFICIARIES

- (a) Neither this Agreement nor any other Transaction Document, nor any right or obligation in or under this Agreement or any other Transaction Document, may be transferred (whether by way of security or otherwise) or delegated by the Counterparty without the prior written consent of the LW Capital Provider, which consent shall not be unreasonably withheld. The LW Capital Provider may assign its rights and obligations under this Agreement and the other Transaction Documents, in whole or in part, to another Person without the consent of the Counterparty. Any purported transfer that is not in compliance with this provision shall be void.
- (b) This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns.
- (c) No Person other than the parties hereto (and the Indemnitees and any permitted transferee hereunder) shall have any rights under this Agreement.

14. TAX WITHHOLDING

The Counterparty shall make all payments under or in connection with this Agreement without any deduction or withholding for or on account of any Tax, save only as may be required by applicable law. If any such deduction or withholding is required by law to be

made, the Counterparty shall (a) promptly notify the LW Capital Provider upon becoming aware that it must make such a deduction or withholding; (b) pay to the relevant authorities (within the time allowed) the full amount required to be deducted or withheld (including the full amount required to be deducted or withheld from any additional amount paid by the Counterparty to the LW Capital Provider under clause (d) of this Section 15 of this Agreement); (c) promptly provide an official receipt (or a certified copy or such other evidence reasonably acceptable to the LW Capital Provider) evidencing the relevant withholding and payment to such authorities; and (d) pay to the LW Capital Provider such additional amounts as are necessary to ensure that (after making any such withholdings or deductions) the net amount actually received by the LW Capital Provider in respect of the payment due from the Counterparty equals the amount which the LW Capital Provider would have received if no such withholdings or deductions had been required.

15. NOTICES

15.1 Effectiveness of Notices

Any notice or other communication in respect of this Agreement shall be in writing and may be given in any manner described below to the address or number provided for the recipient in <u>Annex I</u> to this Agreement and shall be deemed effective as indicated:

- (a) if delivered in person or by courier, on the date it is received;
- (b) if sent by certified or registered mail or the equivalent (return receipt requested), on the date it is received; or
- (c) if sent by email, on the date sent in the absence of any immediate automated response indicating the message was not received or would not be timely read, and if such an immediate automated response is received by the sender, on the date the sender receives an acknowledgement from the recipient,

unless the date of receipt is not a Business Day or the communication is received after the close of business on a Business Day, in which case such communication shall be deemed given and effective on the first following day that is a Business Day. Notices to the Counterparty shall be copied to the Nominated Lawyers, and the time of effectiveness of each such notice shall be the effective time of receipt by either the Counterparty or the Nominated Lawyers, whichever occurs earlier.

15.2 Change of Details

Either party may by notice to the other in accordance with Section 15.1 of this Agreement change the address or email details at which notices or other communications are to be given to it.

16. AMENDMENTS

No amendment, modification or waiver in respect of this Agreement shall be effective unless in writing and executed by the Counterparty and the LW Capital Provider.

17. Entire Agreement

This Agreement and the other Transaction Documents collectively constitute the entire agreement between the parties relating to the subject matter hereof and are the final and complete expression of their intent. The parties represent and warrant that no prior or contemporaneous negotiations, promises, agreements, covenants, or representations of any kind or nature, whether made orally or in writing, have been made or relied upon by them, whether in negotiations leading to this Agreement or relating to the subject matter hereof, which are not expressly contained in the Transaction Documents, or which have not become merged and finally integrated therein; it being the intention of the parties that in the event of any subsequent litigation, controversy, or dispute concerning the terms and provisions of the Transaction Documents, no party shall be permitted to offer or introduce oral or extrinsic evidence concerning the terms and conditions thereof that are not included or referred to therein and not reflected in writing.

18. COUNTERPARTS

This Agreement and the other Transaction Documents (and each amendment, modification and waiver in respect thereof) may be executed and delivered in counterparts (including by facsimile or digital transmission), each of which shall be deemed an original.

19. SURVIVAL OF OBLIGATIONS

With respect to each Claim, the rights and obligations of the parties under Sections 3, 4, 6, 7, 8, 9, 10 and 12 through 27 of this Agreement in relation to such Claim shall survive any Claim Resolution or the exercise by the LW Capital Provider of any remedies with respect to such Claim pursuant to Section 12 of this Agreement. In the event of a termination of this Agreement pursuant to Section 27, the following provisions shall survive to the extent applicable: Sections 2, 3, 4, 6, 7, 8, 9, 10, 13, 17 and 19 - 26.

20. No Waiver

A failure or delay in exercising any right, power or privilege in respect of this Agreement or any other Transaction Document shall not be presumed to operate as a waiver, and a single or partial exercise of any right, power or privilege shall not be presumed to preclude any subsequent or further exercise, of that right, power or privilege or the exercise of any other right, power or privilege.

21. SEVERABILITY

If any term of this Agreement or any other Transaction Document, or the application thereof to either party or any circumstance, is held to be unenforceable, invalid or illegal

(in whole or in part) for any reason (in any relevant jurisdiction), the remaining terms, modified by the deletion of the unenforceable, invalid or illegal portion, shall continue in full force and effect, and such unenforceability, invalidity, or illegality shall not otherwise affect that of the remaining terms, so long as this Agreement and the other Transaction Documents as so modified continues to express, without material change, the original intentions of the parties as to the subject matter hereof and the deletion of such portion of this Agreement or any other Transaction Document shall not substantially impair the respective expectations or reciprocal obligations of the parties or the practical realization of the benefits that would otherwise be conferred upon the parties. The parties shall endeavour in good faith negotiations to replace any prohibited or unenforceable provision with a valid provision the economic effect of which comes as close as possible to that of the prohibited or unenforceable provision.

22. EXPENSES

The Counterparty shall reimburse the LW Capital Provider for its expenses as set forth in such invoice and for all other expenses incurred in connection with this Agreement and the other Transaction Documents, up to an aggregate reimbursement under this Section 22 of US\$65,000. Such reimbursement shall be payable regardless of whether the Approvals are obtained and regardless of whether any Capital Amounts are provided by the LW Capital Provider pursuant to this Agreement. The parties acknowledge and agree that such reimbursement obligation may, at the LW Capital Provider's option, be satisfied by netting such amount against the amount to be funded by the LW Capital Provider pursuant to Section 2.1(a)(ii).

23. RELATIONSHIP OF PARTIES

- (a) The LW Capital Provider and its Affiliates are not law firms and are not engaged in the practice of law with respect to any Claim or the Counterparty. The Counterparty may not, and shall not, rely on any of the LW Capital Provider or its Affiliates for legal advice.
- (b) Nothing in this Agreement or any other Transaction Document shall give rise to or be construed to create a fiduciary, lawyer-client, lender-borrower, agency or other non-contractual relationship between the parties.
- (c) Neither this Agreement nor any other Transaction Document creates any joint venture, partnership or any other type of affiliation between the parties, nor does this Agreement or any other Transaction Document create a joint interest in any Claim for any purpose, including for U.S. federal, state and local income tax purposes.

24. GOVERNING LAW

This Agreement shall be construed in accordance with, and this Agreement and all matters arising out of or relating in any way whatsoever to this Agreement (whether in contract,

tort or otherwise) shall be governed by, the law of the State of New York (without reference to any conflict of law principles or choice of law doctrine that would have the effect of causing this Agreement to be construed in accordance with or governed by the law of any other jurisdiction).

25. DISPUTE RESOLUTION

- Any dispute, controversy or claim arising out of, under or in connection with this (a) Agreement or any other Transaction Document, or any amendment or supplementation thereof, including any question regarding formation, existence, validity, interpretation, amendment, supplementation, performance, breach or termination and any application for interim, preliminary, equitable or injunctive relief, shall (to the exclusion of any other forum) be resolved by an action or judicial proceeding pursuant to the remainder of this Section 25. If and to the extent an action or judicial proceeding is commenced, then, subject to clause (v) of the following sentence, it shall be brought exclusively in the Bankruptcy Court (as defined herein). If the Bankruptcy Court declines jurisdiction, the action or judicial proceeding may be brought in another court of competent jurisdiction in the State of New York, City of New York, Borough of Manhattan. By executing and delivering this Agreement, each party irrevocably (i) accepts generally and unconditionally the exclusive jurisdiction and venue of such courts; (ii) waives any defense of forum non conveniens or improper venue; (iii) agrees that service of all process in any such proceeding in any such court may be made by registered or certified mail, return receipt requested, to such party at its address provided in accordance with this Agreement; (iv) agrees that service as provided in clause (iii) above is sufficient to confer personal jurisdiction over the applicable party in any such proceeding in any such court, and otherwise constitutes effective and binding service in every respect; and (v) agrees that the other party retains the right to serve process in any other manner permitted by law.
- (b) EACH OF THE PARTIES HERETO HEREBY IRREVOCABLY AGREES TO WAIVE ITS RESPECTIVE RIGHTS TO A JURY TRIAL OF ANY CLAIM OR CAUSE OF ACTION BASED UPON OR ARISING HEREUNDER OR ANY DEALINGS BETWEEN THEM RELATING TO THE SUBJECT MATTER HEREOF. THE SCOPE OF THIS WAIVER IS INTENDED TO BE ALL-ENCOMPASSING OF ANY AND ALL DISPUTES THAT MAY BE FILED IN ANY COURT AND THAT RELATE TO THE SUBJECT MATTER HEREOF, INCLUDING CONTRACT CLAIMS, TORT CLAIMS, BREACH OF DUTY CLAIMS AND ALL OTHER COMMON LAW AND STATUTORY CLAIMS. EACH PARTY HERETO ACKNOWLEDGES **THAT THIS** WAIVER IS INDUCEMENT TO ENTER INTO A BUSINESS RELATIONSHIP, THAT EACH HAS ALREADY RELIED ON THIS WAIVER IN ENTERING INTO THIS AGREEMENT. AND THAT EACH WILL CONTINUE TO RELY ON THIS WAIVER IN ITS RELATED FUTURE DEALINGS. EACH PARTY HERETO FURTHER WARRANTS AND REPRESENTS THAT IT HAS

REVIEWED THIS WAIVER WITH COUNSEL AND THAT IT KNOWINGLY AND VOLUNTARILY WAIVES ITS JURY TRIAL RIGHTS FOLLOWING CONSULTATION WITH COMPETENT COUNSEL. THIS WAIVER IS IRREVOCABLE, MEANING THAT IT MAY NOT BE MODIFIED EITHER ORALLY OR IN WRITING OTHER THAN BY A MUTUAL WRITTEN WAIVER SPECIFICALLY REFERRING TO THIS SECTION AND EXECUTED BY EACH OF THE PARTIES HERETO, AND THIS WAIVER SHALL APPLY TO ANY SUBSEQUENT AMENDMENTS, RENEWALS, SUPPLEMENTS OR MODIFICATIONS HERETO. IN THE EVENT OF LITIGATION, THIS PROVISION MAY BE FILED AS A WRITTEN CONSENT TO A TRIAL BY THE COURT.

- (c) The commencement of an action or proceeding in connection herewith shall not suspend or interfere with the Counterparty's obligation to make payment to the LW Capital Provider pursuant to this Agreement; provided, however, that if the Counterparty commences an action or proceeding contemplated by this Section 25 the Nominated Lawyers and/or the Counterparty, as applicable, shall immediately deposit any Proceeds in a dedicated account with the Bankruptcy Court, which Proceeds shall be released, including any interest thereon, as directed by any applicable order of the Bankruptcy Court, unless the parties expressly agree otherwise in writing.
- (d) Any attempt by the Counterparty or the LW Capital Provider to seek relief or remedies in any forum that contravenes this Section 25 shall constitute a breach of this Agreement and entitle the other party to damages, equitable relief and full indemnification against all costs and expenses incurred in connection therewith. No claim by any third party, including the Nominated Lawyers, shall interfere with the operation of this Section 25.
- (e) The Counterparty, being a sophisticated commercial entity with access to counsel, irrevocably waives and forever and unconditionally releases, discharges and quitclaims any claims, counterclaims, defenses, causes of action, remedies and/or rights that it has or may have in the future arising from any doctrine, rule or principle of law or equity that this Agreement or any other Transaction Document, or any of the relationships and transactions contemplated hereby or thereby, (i) are against the public policy of any relevant jurisdiction; (ii) are unconscionable or contravene any laws relating to consumer protection; (iii) are usurious or call for payment of interest at a usurious rate; (iv) were entered into under duress; (v) were entered into as a result of actions by the LW Capital Provider that violated its obligations of good faith and/or fair dealing; (vi) constitute illegal gambling or the sale of unregistered securities; (vii) constitute malicious prosecution, abuse of process or wrongful initiation of litigation; or (viii) constitute champerty, maintenance, barratry or any impermissible transfer or assignment of property or choses in action. The parties specifically agree that any issues concerning the scope or validity of the foregoing waiver shall be within the exclusive jurisdiction of the Tribunal.

26. Rules of Construction

Unless the context otherwise clearly requires: (a) the definitions of terms herein shall apply equally to the singular and plural forms of the terms defined; (b) whenever the context may require, any pronoun shall include the corresponding masculine, feminine and neuter forms; (c) the words "include", "includes" and "including" shall be deemed to be followed by the phrase "without limitation"; (d) the word "shall" shall be construed to have the same meaning and effect as the word "will"; (e) any definition of or reference to any agreement, instrument or other document herein shall be construed as referring to such agreement, instrument or other document as from time to time amended, supplemented, restated or otherwise modified (subject to any restrictions on such amendments, supplements or modifications set forth herein); (f) any reference herein to any Person shall be construed to include such Person's successors and assigns; (g) the phrase "to its knowledge" and phrases of similar import shall be construed to mean the best knowledge, after due inquiry, of any director, officer, manager, member, partner, principal or employee of the entity to which the phrase refers; (h) the words "herein", "hereof" and "hereunder", and words of similar import, shall be construed to refer to this Agreement in its entirety and not to any particular provision hereof; (i) all references herein to Sections, Annexes, Exhibits and Schedules, as applicable, shall be construed to refer to Sections of, and Annexes, Exhibits and Schedules to, this Agreement, and the same are incorporated herein as part of this Agreement; and (i) the headings used in this Agreement are for convenience of reference only and are not to affect the construction of or to be taken into consideration in interpreting this Agreement.

27. TERMINATION OF AGREEMENT

- (a) The Counterparty, at its option, by notice pursuant to Section 15 of this Agreement stating that it is terminating this Agreement, shall have the right to terminate this Agreement upon its receipt of written confirmation from the LW Capital Provider that payment of the LW Capital Provider's Entitlement has been received by it in full, such confirmation, if requested by the Counterparty, not to be unreasonably withheld. Upon the LW Capital Provider's receipt of a termination notice from the Counterparty in accordance herewith, this Agreement shall be terminated, subject to Section 19 of this Agreement.
- (b) At any time prior to the Bankruptcy Court's issuance of an order approving the LW Capital Provision Agreement, the Counterparty may terminate this Agreement by sending a notice, pursuant to Section 15 of this Agreement, representing to the LW Capital Provider in good faith that a settlement of the Claim has been reached in principle and informing the LW Capital Provider that the LW Capital Provision Agreement has been terminated. Upon such termination of the LW Capital Provision Agreement pursuant to this subsection, the LW Capital Provider is entitled to the Termination Fee under the terms of the Termination Fee Agreement.
- (c) The LW Capital Provider, at its option, by notice pursuant to Section 15 of this Agreement stating that it is terminating this Agreement, shall have the right to terminate this Agreement if (i) LW Capital Provider is in compliance with this

Agreement and (ii) the Counterparty fails to secure the Approvals for any reason within forty-five (45) days of this Agreement. Upon the Counterparty's receipt of a termination notice from the LW Capital Provider in accordance herewith, this Agreement shall be terminated, subject to Section 19 of this Agreement. Upon termination of the LW Capital Provision Agreement pursuant to this subsection, the LW Capital Provider is entitled to the Termination Fee under the terms of the Termination Fee Agreement, and to any unreimbursed expenses payable pursuant to Section 22.

28. SUBORDINATION AGREEMENT

The terms and conditions of this Agreement are expressly subject to the terms and conditions of the Subordination Agreement.

[Remainder of this page intentionally left blank.]

Signature page to Capital Provision Agreement

IN WITNESS WHEREOF, the parties have executed and delivered this Capital Provision Agreement as of the date first written above.

Counterparty:

MOTORS LIC	DUIDATION	COMPANY	AVOIDANCE	ACTION TRUST
	2 CIDILILA CIT	COLTAX X XX 1 X		

By: Wilmington Trust Company, as Trust Administrator and Trustee

Name: Son Andrews
Title: Via Hasidest

LW Capital Provider:

LW HOLDCO VI LLC

By:_____

Name: Boaz Weinstein Title: Authorized Signatory

Acknowledged and approved:

ARTHUR J. GONZALEZ, as Trust Montor

Signature page to Capital Provision Agreement

IN WITNESS WHEREOF, the parties have executed and delivered this Capital Provision Agreement as of the date first written above.

Counterparty:

MOTORS LIQUIDATION COMPANY AVOIDANCE ACTION TRUST

By: Wilmington Trust Company, as Trust Administrator and Trustee

By:
Name:
Title:
LW Capital Provider:
LW HOLDCO VI LLC
By: Sy WA
Name: Boaz Weinstein
Title: Authorized Signatory
Acknowledged and approved:
ARTHUR J. GONZALEZ, as Trust Monitor

EXHIBIT A

Defined Terms

- "Adverse Claim" means, with respect to any Claim or any interest therein or any Proceeds thereof, (i) any mortgage, deed of trust, lien, pledge, hypothecation, encumbrance, charge or security interest in, on or affecting such Claim or any interest therein or any Proceeds thereof, (ii) the interest of a vendor or a lessor under any conditional sale agreement, capital lease or title retention agreement (or any financing lease having substantially the same economic effect as any of the foregoing) relating to such Claim or any interest therein or any Proceeds thereof, (iii) any purchase option, call or similar right of a third party with respect to such Claim or any interest therein or any Proceeds thereof and (iv) any other claim that a claimant has an interest in such Claim or any interest therein or any Proceeds thereof or that it is a violation of the rights of the claimant for another person to hold, transfer or deal with such Claim or any interest therein or any Proceeds thereof.
- "Affiliate" means, with respect to a specified Person, another Person that directly, or indirectly through one or more intermediaries, Controls or is Controlled by or is under common Control with the Person specified. For this purpose, "Control" means the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a Person, whether through the ability to exercise voting power, by contract or otherwise, and "Controlling" and "Controlled" have meanings correlative thereto.
- "Allowed General Unsecured Claims" has the meaning given thereto in the Trust Agreement.
- "Approvals" has the meaning set forth in Section 2.1 of this Agreement.
- "Bankruptcy Court" has the meaning set forth in Section 2.1 of this Agreement.
- "Bankruptcy Court Application" has the meaning set forth in Section 2.1 of this Agreement.
- "Budget" has the meaning given thereto in the Trust Agreement.
- "Business Day" means a day on which commercial banks and foreign exchange markets settle payments and are open for general business (including dealings in foreign exchange and foreign currency deposits) in New York.
- "Capital Amounts" means the amounts set forth as such in Section 2.1 of this Agreement.
- "CE Capital Providers" means Earlham LLC, a limited liability company formed under the laws of the State of Delaware, and Cynthiana LLC, a limited liability company formed under the laws of the State of Delaware.

- "CE Capital Providers' Entitlement' means the amounts set forth in Section 2.2(a)(i) of the CE Capital Provision Agreement.
- "CE Capital Provision Agreement" means that certain Capital Provision Agreement, dated as of June 2, 2017 between the Counterparty and the CE Capital Providers.
- "CE Security Agreement" means any security agreement, pledge agreement or similar agreement pursuant to which the Counterparty provides to the CE Capital Providers a security interest in certain collateral to secure the obligations of the Counterparty under the CE Capital Provision Agreement.
- "Claim" means the claim(s) described on Annex II to this Agreement. If two or more claims are set forth on Annex II, all such claims together shall be referred to collectively as "the Claim".
- "Claim Costs" has the meaning set forth in Section 5.3(a) of this Agreement.
- "Claim Resolution" means either full and final settlement of the Claim or the entry of a final, non-appealable and enforceable judgment, in either case resolving with prejudice all aspects and elements of the Claim. In circumstances where a final, non-appealable and enforceable judgment does not automatically come into being upon the rendering of a dispositive decision, a Claim Resolution shall be deemed to have occurred on the date that is sixty (60) days following such dispositive decision in the absence of any subsequent challenge thereto.
- "Committee" has the meaning set forth in the Recitals of this Agreement.
- "Common Interest Material" means any Confidential Information that is the work product of qualified legal advisers and/or attorney work product, protected by the attorney-client privilege or any similar privilege in any jurisdiction including, for the avoidance of doubt, legal professional privilege and/or litigation or arbitral privilege, or that is protected by any rules of professional secrecy in any jurisdiction (collectively, "Privileges"), including: (i) information prepared by a party to a Claim or any of its Affiliates or their counsel, advisors, consultants or agents; or (ii) information prepared by the LW Capital Provider or any of its Representatives in connection with the review of a Claim, including legal and factual memoranda, case analyses and evaluations.
- "Confidential Information" means any non-public, confidential or proprietary information relating to: (i) the LW Capital Provider and its Affiliates and Representatives, information provided by them about their business and operations or the structures and economic arrangements they use in their business, including the nature, terms and existence of this Agreement and the other Transaction Documents, the existence of any relationship between the Counterparty and the LW Capital Provider or any of its Affiliates or Representatives, and the discussions and negotiations related thereto; (ii) any Claim, including the names of the parties and potential other parties to such Claim; the factual, legal, technical, economic and financial background of such Claim; and the procedural status, theories, strategies and

tactics for the prosecution or defense of such Claim; (iii) billing arrangements, rates, financial or fee arrangements; (iv) any financial statements, accounts or other similar information or materials; (v) business or financial information, business plans and relationships, marketing or product data; (vi) algorithms, computer databases, computer programs, computer software and systems, intellectual property, trade secrets and trademarks; (vii) research, scientific data, specifications, technical data, techniques and technology; and (viii) other proprietary or non-public information, data or material; in all cases regardless of whether such information is (A) written or oral, irrespective of the form or storage medium, or (B) specifically identified as "Confidential". Notwithstanding the foregoing, Confidential Information does not include information that (x) was or becomes generally available to the public other than as a result of a disclosure by the Recipient; (y) was available to the Recipient on a non-confidential basis prior to its disclosure; or (z) was developed independent of the information derived from the Confidential Information.

- "Counterparty" has the meaning set forth in the introductory paragraph of this Agreement.
- "Default Rate" means a rate per year of 17%.
- "DIP Lender Advances" has the meaning given thereto in the Trust Agreement.
- "DIP Lenders" has the meaning given thereto in the Trust Agreement.
- "Disclosing Party" has the meaning set forth in Section 7.1 of this Agreement.
- "Distributable Trust Assets" has the meaning given thereto in the Trust Agreement.
- "Distribution Date" has the meaning given thereto in the Trust Agreement.
- "Engagement Agreement" means any engagement, retainer or similar agreement between the Counterparty and the Nominated Lawyers (including any successor or new Nominated Lawyers designated in accordance with Section 5.3(e) of this Agreement) governing or purporting to govern the terms of the representation of the Counterparty by such legal counsel with respect to the Claim, in each case as set forth on Annex II to this Agreement and as provided to the LW Capital Provider. In the case of any Engagement Agreements entered into after the date hereof, Annex II shall be deemed amended to include such agreements.
- "Governmental Authority" means the government of the United States of America, any other nation or any political subdivision thereof, whether state or local, and any agency, authority, instrumentality, regulatory body, court, central bank or other entity exercising executive, legislative, judicial, taxing, regulatory or administrative powers or functions of or pertaining to government.
- "GUC Trust Advances" has the meaning given thereto in the Trust Agreement.
- "Invested Amount" has the meaning set forth in Section 2.1 of this Agreement.

- "Litigation Settlement" has the meaning set forth in the Recitals of this Agreement.
- "Litigation Cost Advance" has the meaning given thereto in the Trust Agreement.
- "Litigation Cost Advance Agreement" has the meaning given thereto in the Trust Agreement.
- "LW Capital Provider" has the meaning set forth in the introductory paragraph of this Agreement.
- "LW Capital Provider Funding Account" has the meaning set forth in Section 2.1(c) of this Agreement.
- "LW Capital Provider's Entitlement" means the amounts set forth as such in Section 2.2(a) of this Agreement.
- "Material Adverse Effect" means, with respect to any event or circumstance and either party, one or more of (i) the impairment of its ability to perform any of its obligations under this Agreement or any other Transaction Document, (ii) the impairment of the rights or remedies available under this Agreement or any other Transaction Document to the other party and (iii) solely in the case of the Counterparty, an adverse effect on any Claim or the value or collectability thereof.
- "Nominated Lawyers" means, with respect to any Claim, the Person or Persons identified as such on Annex II to this Agreement.
- "Opponent" means, with respect to any Claim, individually and collectively, the Person(s) named as defendants or counterclaim defendants in such Claim, as set forth on Annex II to this Agreement, and any other Person added or joined to the Claim from time to time as a defendant or indemnitor or against whom proceedings are asserted or threatened even if such Person is not named or served, and in each case their respective Affiliates and successors.
- "Person" means any natural person, corporation, limited liability company, trust, joint venture, association, company, partnership or other entity or Governmental Authority.
- "Phase III Trial" means the trial on the Initial Discovery and Trial Issues as defined in the Bankruptcy Court's order of September 14, 2018 in case number 09-00504-mg (docket number 1080).
- "Potential Remedy Event" means any event which, with the giving of notice or the lapse of time or both, would constitute a "Remedy Event".
- "Privileges" has the meaning set forth in the definition of "Common Interest Material".

- "Proceeds" means Distributable Trust Assets as defined in the Trust Agreement, to wit: Avoidance Action Proceeds, as defined in the Trust Agreement, together with earnings (including interest) thereon.
- "Recipient" has the meaning set forth in Section 7.1 of this Agreement.
- "Reconstitution Claims" has the meaning set forth in Annex II to this Agreement.
- "Remedy Event" has the meaning set forth in Section 11 of this Agreement.
- "Representation Date" means (i) the date hereof, (ii) each date on which the parties agree to cause any Claim to become subject to this Agreement, (iii) each date on which the LW Capital Provider is obligated to or do make any payment of Capital Amounts and (iv) in the case of the Counterparty, each date on which a Monthly Report is delivered (provided that, in the event of a failure to deliver any Monthly Report by its due date, the Representation Date shall instead be such due date); provided that, for the avoidance of doubt, even though an agreement or payment referred to in clause (ii) or (iii) above relates to a specific Claim, a "Representation Date" shall be deemed to occur with respect to each Claim that is then subject to this Agreement.
- "Representatives" means, with respect to any person or entity, as applicable, its directors, officers, managers, members, partners, principals, employees, shareholders and participants (or potential shareholders and participants), Affiliates, related entities, agents, assignees (or potential assignees), reinsurers, lawyers, accountants, consultants, advisors and independent contractors.
- "Security Agreement" means, if required by the LW Capital Provider in connection with the transactions contemplated hereby, any security agreement, pledge agreement or similar agreement pursuant to which the Counterparty provides to the LW Capital Provider a security interest in certain collateral to secure the obligations of the Counterparty hereunder, as set forth on Annex I and in the form required by the LW Capital Provider.
- "Set-off" means a direct or indirect right of one party (Party A) to deduct from a debt owed by it to another party (Party B) the amount of a different debt that Party B owes to Party A, including but not limited to a counterclaim in a legal proceeding, but excluding a substantive defense successfully asserted in a legal proceeding.
- "Settlement Agreement" has the meaning given thereto in the Trust Agreement.
- "Subordination Agreement" means the subordination agreement to be entered into by and among the LW Capital Provider, the Counterparty and the DIP Lenders promptly after the date hereof, in a form acceptable to the DIP Lenders and the CE Capital Providers pursuant to Section 6.1(d)(i) of the Trust Agreement, which acceptance shall not be unreasonably withheld.

- "Tax" means any tax, duty, contribution, impost, withholding, levy or other charge or withholding of a similar nature (including use, sales and value added taxes), whether domestic or foreign, and any fine, penalty, surcharge or interest in connection therewith.
- "*Termination Fee*" means US\$200,000 payable by the Counterparty to the LW Capital Provider pursuant to the Termination Fee Agreement.
- "Term Loan Avoidance Action" has the meaning set forth in Annex II.
- "Transaction Documents" means, collectively, this Agreement, any Security Agreement, the Subordination Agreement, and any other agreements, documents, instruments or certificates entered into or delivered in connection with this Agreement.
- "Tribunal" has the meaning set forth in Section 26 of this Agreement.
- "Trust Administration Expenses" means reasonable costs and expenses of the Trust incurred in connection with the administration thereof, as described in Section 6.1(a)(i) of the Trust Agreement, in accordance with the Budget.
- "Trust Administrator" has the meaning given thereto in the Trust Agreement.
- "Trust Agreement" means the Third Amended and Restated Motors Liquidation Company Avoidance Action Trust Agreement, as the same shall be amended pursuant to this Agreement.
- "Trust Beneficiaries" has the meaning given thereto in the Trust Agreement.
- "Trust Monitor" has the meaning given thereto in the Trust Agreement.

EXHIBIT B

Termination Fee Agreement

[Attached]

Dated as of January 21, 2019

Termination Fee Agreement

between

Motors Liquidation Company Avoidance Action Trust

and

The LW Capital Provider, as defined in the introductory paragraph hereof

This TERMINATION FEE AGREEMENT, dated as of January 21, 2019 (this "Agreement"), is by and between Motors Liquidation Company Avoidance Action Trust, an entity organized or formed under the laws of the State of Delaware (the "Counterparty"), and LW Holdco VI LLC, a limited liability company formed under the laws of the State of Delaware ("LW Capital Provider").

WHEREAS, prior to August 2016, there was a dispute between the Official Committee of Unsecured Creditors of Motors Liquidation Company (the "Committee") and the DIP Lenders as to entitlements to Avoidance Action Proceeds, which dispute was resolved by mutual agreement (the "Litigation Settlement") pursuant to which, among other things, (i) the DIP Lenders provided the Litigation Cost Advance and (ii) the DIP Lenders are entitled to be repaid the Litigation Cost Advance out of Distributable Trust Assets; and

WHEREAS, the Trust Administrator and the Trust Monitor determined that the Avoidance Action Trust Administrative Cash and Supplemental Avoidance Action Trust Cash held by the Counterparty at the time of the Litigation Settlement were insufficient to satisfy projected fees, costs and expenses of the Counterparty, and, as an integral part of the Litigation Settlement, the DIP Lenders agreed to provide the Litigation Cost Advance to the Counterparty to fund the prosecution of the Term Loan Avoidance Action, on the terms set forth in the Litigation Cost Advance Agreement;

WHEREAS, on August 24, 2016, the Bankruptcy Court entered an order approving the Settlement Agreement and the Litigation Cost Advance Agreement; and

WHEREAS, subsequent to the entering into of the Litigation Cost Advance Agreement, the Trust Administrator and the Trust Monitor determined that the Avoidance Action Trust Administrative Cash and Supplemental Avoidance Action Trust Cash held by the Counterparty was insufficient to satisfy projected fees, costs and expenses of the Counterparty, for which reason the Counterparty sought additional funding by entering into a Capital Provision Agreement with Cynthiana LLC and Earlham LLC (the "CE Capital Provision Agreement"); and

WHEREAS, on June 20, 2017, the Bankruptcy Court entered an order approving the CE Capital Provision Agreement;

WHEREAS, subsequently, the Trust Administrator and the Trust Monitor determined that the Avoidance Action Trust Administrative Cash and Supplemental Avoidance Action Trust Cash currently held by the Counterparty remained insufficient to satisfy projected fees, costs and expenses of the Counterparty, for which reason the Counterparty determined to seek additional funding from the LW Capital Provider; and

WHEREAS, the LW Capital Provider and the Counterparty entered into the LW Capital Provision Agreement, dated as of January 21, 2019 (the "LW Capital Provision Agreement"), for the purpose of the LW Capital Provider providing additional funding to the Counterparty; and

WHEREAS, the LW Capital Provision Agreement provides that the LW Capital Provider's obligation to provide additional funding is subject to, among other conditions, the obtaining of certain Approvals, as defined at Section 2.1(e) of the LW Capital Provision Agreement; and

WHEREAS, the LW Capital Provision Agreement provides, at Section 3(d), that:

Termination Fee. In exchange for the LW Capital Provider reserving the Capital Amounts as necessary to satisfy its obligations hereunder, in the event that (i) the Counterparty terminates the LW Capital Provision Agreement pursuant to Section 27(b) of this Agreement; or (ii) the Counterparty fails to secure the Approvals within forty-five (45) days of this Agreement for any reason and the LW Capital Provider terminates this Agreement pursuant to Section 27(c) of this Agreement, the Counterparty shall be obligated to pay to the LW Capital Provider the Termination Fee pursuant to the terms of the Termination Fee Agreement included as Exhibit A to this Agreement (the "Termination Fee Agreement").

WHEREAS, the present Termination Fee Agreement is Exhibit A to the LW Capital Provision Agreement, referred to in Section 3(d) of the LW Capital Provision Agreement; and

WHEREAS, the LW Capital Provision Agreement defined the term Termination Fee as "US\$200,000 payable by the Counterparty to the LW Capital Provider";

NOW, THEREFORE, in consideration of the mutual agreements contained herein, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, hereby agree as follows:

ENTITLEMENT TO TERMINATION FEE

- (a) In the event that (i) the Counterparty terminates the LW Capital Provision Agreement pursuant to Section 27(b) of the LW Capital Provision Agreement, or (ii) the LW Capital Provider terminates the LW Capital Provision Agreement pursuant to Section 27(c) of the LW Capital Provision Agreement, in the next motion filed by the Counterparty with the Bankruptcy Court seeking authorization to distribute Distributable Trust Assets, whether based upon the settlement of the Claims or otherwise, or any other appropriate motion or process, the Counterparty shall seek approval to pay, and shall use its best efforts to take all other actions necessary for such approval and payment of, the Termination Fee to the LW Capital Provider.
- (b) The Counterparty's obligation to pay the Termination Fee is contingent upon the Bankruptcy Court's approval of the payment of the Termination Fee.

(c) Payment of the Termination Fee is contingent upon the availability of Distributable Trust Assets after full payment to the DIP Lenders of the DIP Lender Advances and full payment to the CE Capital Providers' Entitlement.

[Signature Page Follows.]

Signature page to Termination Fee Agreement

IN WITNESS WHEREOF, the parties have executed and delivered this Termination Fee Agreement as of the date first written above.

Counterparty:

MOTORS LIQUIDATION COMPANY AVOIDANCE ACTION TRUST

By: Wilmington Trust Company, as Trust Administrator and Trustee

By: St. Musicest

Name: Sem Andrews

Title: Via Assidest

LW Capital Provider:

LW HOLDCO VI LLC

By:_____

Name: Boaz Weinstein Title: Authorized Signatory

Acknowledged and approved:

ARTHUR J. GONZALEZ, as Trust Monitor

Signature page to Termination Fee Agreement

IN WITNESS WHEREOF, the parties have executed and delivered this Termination Fee Agreement as of the date first written above.

Counterparty:

MOTORS LIQUIDATION COMPANY AVOIDANCE ACTION TRUST

By: Wilmington Trust Company, as Trust Administrator and Trustee

By:	
Name:	
Title:	
LW Capital Provider:	
LW HOLDCO VI LLC	
By: /Syn TWA	
Name: Boaz Weinstein	
Title: Authorized Signatory	

Acknowledged and approved:

ARTHUR J. GONZALEZ, as Trust Monitor

ANNEX I Counterparty: Motors Liquidation Company Avoidance Action Trust

1.	Type of Entity:	Statutory trust
2.	Jurisdiction of Organization or Formation:	State of Delaware
3.	Security Agreement(s):	Security Agreement between the Counterparty, as grantor, and the LW Capital Provider, as secured parties, dated the date hereof.
4.	Notice Information:	Address: Wilmington Trust Company Rodney Square North 1100 North Market Street Wilmington, Delaware, 19890-1615 Fax Number: (302) 636-4145 Email Address: bandrews@WilmingtonTrust.com Attn: Corporate Trust Administration
5.	Notice Information for Nominated Lawyers:	Address: Binder & Schwartz LLP 366 Madison Avenue 6th Floor New York, NY 10017 Fax Number: (212) 510-7299 Email Address: nbinder@binderschwartz.com Attn: Neil Binder

LW Capital Provider: LW Holdco VI LLC

6.	Type of Entity:	limited liability company
7.	Jurisdiction of Organization or Formation:	State of Delaware
8.	Notice Information:	Address: 1350 Avenue of the Americas, 2 nd Floor New York, New York 10019 Fax Number: (646) 389-1032 Email Address: weinstein@lakewhillans.com Attn: Boaz Weinstein

ANNEX II

1. Description of the Claim:

Any litigation commenced by the Counterparty, including In re: Motors Liquidation Company, et al., Chapter 11 Case No. 09-50026 (REG) (Jointly Administered), including but not limited to (i) Motors Liquidation Co. Avoidance Action Trust, by and through the Wilmington Trust Company, solely in its capacity as Trust Administrator and Trustee v. J.P. Morgan Chase Bank, N.A., individually and as administrative Agent for various lenders party to the Term Loan Agreement, et al., Adversary Proceeding Case No. 09-00504 (REG), U.S.B.C. S.D.N.Y. (the "Term Loan Avoidance Action"); (ii) the action commenced in the Court of Chancery of the State of Delaware against Oaktree Loan Fund, LP, including any related actions; (iii) the action commenced in the Court of Chancery of the State of Delaware against SSS Funding II, LLC (the two actions described in the preceding clauses (ii) and (iii), the "Reconstitution Claims"); and (iv) any other actions or proceedings brought by the Counterparty.

The Claim also includes any variations or expansions of the above claims by the addition of any claims and/or parties from time to time, as well as the following:

- (i) any and all related pre- and post-trial proceedings or processes (or pre- and post-hearing proceedings or processes, where applicable) in or in connection with such claim(s), including the pursuit of costs or post-judgment or post-arbitral award remedies;
- (ii) all proceedings seeking to appeal, challenge, confirm, enforce, modify, correct, vacate or annul a judgment or award, as well as proceedings on remand or retrial or rehearing;
- (iii) all ancillary, parallel or alternative dispute resolution proceedings and processes arising out of or related to the acts or occurrences alleged in such claim(s) (including conciliation or mediation

			or court filings seeking discovery for or filed in aid of a contemplated or pending arbitration);
		(iv)	re-filings or parallel filings of such claim(s) and any other legal, diplomatic or administrative proceedings or processes founded on the underlying facts giving rise to or forming a basis for such claim(s);
		(v)	ancillary or enforcement proceedings related to the facts or claims alleged from time to time or that could have been alleged in such claim(s) at any time;
		(vi)	all arrangements, settlements, negotiations, or compromises made between the Counterparty and any adverse party having the effect of resolving any of the Counterparty's claims against any adverse party that are or could be or could have been brought in such claim(s); and
		(vii)	all rights of the Counterparty to collect any damages or awards or otherwise exercise remedies in connection with any of the foregoing.
2.	Payment account of the Counterparty:	Bank: Wilmington Trust Company ACCT Name: LW Capital Provider Funding Account Address: 1100 N. Market Street Wilmington, DE 19890-0100 Wire Routing: ABA# 031100092 Account Number: To be provided prior to Initial Funding Date	
3.	Payment account of LW Capital Provider:	To be provided.	
4.	Identity of the Opponent:	Defendants in the Term Loan Avoidance Action Oaktree Loan Fund, LP SSS Funding II, LLC	
5.	Identity of the Nominated Lawyers:	1	r & Schwartz LLP
6.	Engagement Agreement(s):	Agreement between Binder & Schwartz LLP and the Counterparty, dated January 5, 2016, as amended by letter dated August 8, 2016	
		1	

Schedule 6.2(i)

As of January 15, 2019, the amounts which, if recovered, would be due pursuant to the Trust Agreement from Distributable Trust Assets:

		Amount in US Dollars
(i.)	to the DIP Lenders pursuant to the	\$16,600,000.00
	Litigation Cost Advance	
	Agreement	
(ii.)	to the CE Capital Providers	\$24,870,531.00
	pursuant to the CE Capital	
	Provision Agreement	
(iii.)	to the GUC Trust for DIP Lender	\$31,575,574.00
	Advances	

Exhibit C

FOURTH AMENDED AND RESTATED MOTORS LIQUIDATION COMPANY AVOIDANCE ACTION TRUST AGREEMENT

This FOURTH AMENDED AND RESTATED MOTORS LIQUIDATION COMPANY AVOIDANCE ACTION TRUST AGREEMENT, dated as of [], 2019 (as it may be amended from time to time, this "Trust Agreement"), by and among Wilmington Trust Company, as trust administrator and trustee (together with any successor appointed under the terms hereof, the "Trust Administrator") of the Motors Liquidation Company Avoidance Action Trust (the "Trust") for the benefit of the Trust Beneficiaries (as defined below), and Arthur J. Gonzalez, as trust monitor (together with any predecessor or successor appointed under the terms hereof, the "Trust Monitor") of the Trust, amends and restates in its entirety the Third Amended and Restated Trust Agreement (as defined below). Capitalized terms used herein and not otherwise defined herein shall have the meanings ascribed to such terms in the Debtors' Second Amended Joint Chapter 11 Plan of liquidation pursuant to chapter 11 of Title 11 of the United States Code, 11 U.S.C. §§ 101 et seq., as amended (the "Bankruptcy Code"), dated March 18, 2011, as confirmed (including all exhibits thereto, as the same may be further amended, modified, or supplemented from time to time, the "Plan").

WITNESSETH:

WHEREAS, the Trust Administrator and the Trust Monitor are party to the Motors Liquidation Company Avoidance Action Trust Agreement, dated as of March 30, 2011, by and among Motors Liquidation Company ("MLC"), MLC of Harlem, Inc., MLCS, LLC, MLCS Distribution Corporation, Remediation and Liability Management Company, Inc., and Environmental Corporate Remediation Company, Inc. (collectively, the "Debtors"), as debtors and debtors-in-possession, Wilmington Trust Company, as Trust Administrator, and FTI Consulting, Inc., as Trust Monitor (the "Original Trust Agreement"); and

WHEREAS, each of the Debtors has, prior to the date hereof, ceased to operate and dissolved; and

WHEREAS, the Original Trust Agreement was amended and restated in its entirety, with the approval of the Bankruptcy Court (as defined below), pursuant to that certain Amended and Restated Motors Liquidation Company Avoidance Action Trust Agreement dated as of May 11, 2012 by and between the Trust Administrator and the Trust Monitor (the "First Amended and Restated Trust Agreement"); and

WHEREAS, there was a dispute between the Official Committee of Unsecured Creditors of Motors Liquidation Company (the "Committee") and the DIP Lenders (defined below) as to entitlements to Avoidance Action Proceeds, which dispute was resolved by mutual agreement (the "Litigation Settlement") pursuant to which (a) the DIP Lenders provided the Litigation Cost Advance (as defined below), (b) the DIP Lenders shall be entitled to be repaid the Litigation Cost Advance out of Distributable Trust Assets (as defined below) and (c) the DIP Lenders shall be entitled to receive thirty percent (30%) of all remaining Distributable Trust Assets and, subject to the terms of this Trust Agreement, the holders of Allowed General Unsecured Claims (or Units) shall be entitled to receive seventy percent (70%) of all remaining Distributable Trust Assets, with each such distribution to be made at or about the same time and on a pari passu basis, as set forth

more fully in that certain Stipulation and Agreed Order (I) Settling Disputed Entitlements of Debtor-in-Possession Lenders and Official Committee of Unsecured Creditors to Potential Term Loan Avoidance Action Proceeds, and (II) Modifying Avoidance Action Trust Agreement to Implement Settlement, executed by the Committee, the DIP Lenders and the Trust and dated July 14, 2016 (the "Settlement Agreement"); and

WHEREAS, the Trust Administrator and the Trust Monitor determined that the Avoidance Action Trust Administrative Cash and Supplemental Avoidance Action Trust Cash (each defined below) held at that time by the Trust was insufficient to satisfy projected fees, costs and expenses of the Trust, and, as an integral part of the Litigation Settlement, the DIP Lenders agreed to provide the Litigation Cost Advance to the Trust on the terms set forth in that certain agreement, dated July 14, 2016, executed by the Trust and the DIP Lenders (the "<u>Litigation Cost Advance Agreement</u>"), which included, among other things, that the DIP Lenders would provide the Litigation Cost Advance to fund the prosecution of the Avoidance Action; and

WHEREAS, on July 15, 2016, the Trust, jointly with the Committee, filed a motion with the Bankruptcy Court (the "9019 Motion") seeking approval of the Settlement Agreement and the Litigation Cost Advance Agreement; and

WHEREAS, on August 24, 2016, the Bankruptcy Court entered an order approving the 9019 Motion (the "9019 Order"); and

WHEREAS, pursuant Section 13.13(b) of the First Amended and Restated Trust Agreement, the 9019 Motion and the 9019 Order, the Trust Administrator petitioned and received from the Bankruptcy Court approval to amend and restate in its entirety the First Amended and Restated Trust Agreement; and

WHEREAS, the First Amended and Restated Trust Agreement was amended and restated in its entirety pursuant to that certain Second Amended and Restated Motors Liquidation Company Avoidance Action Trust Agreement dated as of August 25, 2016, by and between the Trust Administrator and the Trust Monitor (the "Second Amended and Restated Trust Agreement"); and

WHEREAS, one of the preconditions to the DIP Lenders' provision of the Litigation Cost Advance was the Trust's agreement, as reflected in Section 6.1(d)(1) of the Second Amended and Restated Trust Agreement, that any additional litigation funding provided to the Trust would be (i) junior and subordinate to the Litigation Cost Advance and any other amounts owed to the DIP Lenders on account of prior funding of the Trust and (ii) subject to a form of subordination acceptable to the DIP Lenders in all respects; and

WHEREAS, on or about June 29, 2017, the Trust Administrator and the Trust Monitor determined that the Avoidance Action Trust Administrative Cash and Supplemental Avoidance Action Trust Cash held by the Trust remained insufficient to satisfy projected fees, costs and expenses of the Trust; and

WHEREAS, with the consent of the DIP Lenders, the Trust Administrator and Trust Monitor entered into a Capital Provision Agreement, dated June 2, 2017 (the "CE Capital Provision Agreement"), with Delaware limited liability companies Cynthiana LLC and Earlham LLC (together, the "CE Capital Providers"), pursuant to which the CE Capital Providers agreed to

provide up to \$15,000,000 in additional funding to the Trust (the "<u>CE Supplemental Capital</u>") as Other Supplemental Cash; and

WHEREAS, pursuant to Section 6.1(d)(i) of the Second Amended and Restated Trust Agreement, entry into that certain Subordination Agreement (the "CE Subordination Agreement"), dated as of June 2, 2017, by and among the DIP Lenders, the CE Capital Providers, and the Trust, was a precondition of the DIP Lenders' consent to the CE Capital Providers' provision of the CE Supplemental Capital to the Trust; and

WHEREAS, on June 7, 2017, the Trust moved for entry of an order, pursuant to sections 105 and 1142 of title 11 of the United States Code: (1) approving the amendment and restatement of the Second Amended Avoidance Action Trust Agreement, (2) authorizing the Avoidance Action Trust to enter into the CE Capital Provision Agreement and to grant a lien to the CE Capital Providers, and (3) authorizing the Avoidance Action Trust and Avoidance Action Trust Administrator to use \$1,750,759.93 of the Holdback to fund necessary fees, costs and expenses of the Avoidance Action Trust (the "CE Capital Provision Motion"); and

WHEREAS, on June 29, 2017, the Bankruptcy Court entered an order granting the CE Capital Provision Motion (the "<u>CE Capital Provision Order</u>"); and

WHEREAS, pursuant Section 13.13(b) of the Second Amended and Restated Trust Agreement, the CE Capital Provision Motion and the CE Capital Provision Order, the Trust Administrator petitioned and received from the Bankruptcy Court approval to amend and restate in its entirety the Second Amended and Restated Trust Agreement; and

WHEREAS, the Second Amended and Restated Trust Agreement was amended and restated in its entirety pursuant to that certain Third Amended and Restated Motors Liquidation Company Avoidance Action Trust Agreement dated as of June 29, 2017, by and between the Trust Administrator and the Trust Monitor (the "Third Amended and Restated Trust Agreement"); and

WHEREAS, on or about the date hereof, for all of the reasons set forth in the LW Capital Provision Motion (as defined below), the Trust Administrator and the Trust Monitor have determined that the Avoidance Action Trust Administrative Cash and Supplemental Avoidance Action Trust Cash currently held by the Trust are likely to be insufficient to satisfy projected fees, costs and expenses of the Trust; and

WHEREAS, with the consent of the DIP Lenders and the CE Capital Providers, the Trust Administrator and Trust Monitor entered into a Capital Provision Agreement, dated January 21, 2019 (the "LW Capital Provision Agreement"), with LW Holdco VI LLC, a Delaware limited liability company (the "LW Capital Provider," and together with the DIP Lenders and the CE Capital Providers, the "Capital Providers"), pursuant to which the LW Capital Provider agreed to provide up to \$10,000,000 in additional funding to the Trust (the "LW Supplemental Capital," and together with the CE Supplemental Capital, the "Supplemental Capital") as Other Supplemental Cash; and

WHEREAS, pursuant to Section 6.1(d)(i) of the Third Amended and Restated Trust Agreement, entry into that certain Subordination Agreement, dated as of January 24, 2019 (the

"<u>LW Subordination Agreement</u>"), by and among the DIP Lenders, the LW Capital Provider and the Trust, the form of which was approved by the CE Capital Providers, is a precondition of the DIP Lenders' consent to the LW Capital Provider's provision of the LW Supplemental Capital to the Trust; and

WHEREAS, on February 4, 2019, the Trust moved for entry of an order, pursuant to sections 105 and 1142 of title 11 of the United States Code: (1) approving the amendment and restatement of the Third Amended Avoidance Action Trust Agreement, and (2) authorizing the Avoidance Action Trust to enter into the LW Capital Provision Agreement and to grant a lien to the LW Capital Provider (the "LW Capital Provision Motion"); and

WHEREAS, on [], 2019, the Bankruptcy Court entered an order granting the LW Capital Provision Motion (the "LW Capital Provision Order"); and

WHEREAS, pursuant Section 13.13(b) of the Third Amended and Restated Trust Agreement, the LW Capital Provision Motion and the LW Capital Provision Order, the Trust Administrator petitioned and received from the Bankruptcy Court approval to amend and restate in its entirety the Third Amended and Restated Trust Agreement.

NOW, THEREFORE, in accordance with Section 13.13(b) of the Third Amended and Restated Trust Agreement and the LW Capital Provision Agreement, the Third Amended and Restated Trust Agreement is hereby amended and restated as follows:

Background

- A. Beginning on June 1, 2009, the Debtors filed in the United States Bankruptcy Court for the Southern District of New York (the "Bankruptcy Court") voluntary petitions for relief under chapter 11 of the Bankruptcy Code (the "Chapter 11 Cases").
- B. On or about August 31, 2010, the Debtors filed their Plan and Disclosure Statement in the Bankruptcy Court. The Debtors filed an amended Plan and Disclosure Statement on December 7, 2010. The Debtors filed a second amended Plan on March 18, 2011.
- C. The Disclosure Statement was approved by the Bankruptcy Court on December 8, 2010.
- D. On or about March 29, 2011, the Bankruptcy Court entered an order (the "Confirmation Order") confirming the Plan.
- E. The Plan provides for the creation of the Trust as a post-confirmation successor to MLC within the meaning of Section 1145(a) of the Bankruptcy Code, to hold and administer the Avoidance Action Trust Assets for the benefit of the Trust Beneficiaries and, to the extent received by the Trust, to distribute the Distributable Trust Assets to the Trust Beneficiaries in accordance with the terms of the Plan, the Confirmation Order and this Trust Agreement.
- F. The Plan also provides that the Trust Administrator may determine to hold and administer Other Debtor Residual Trust Assets, if any, for the benefit of the DIP Lenders.

- G. The Trust was created, with respect to the Avoidance Action Trust Assets, on behalf of, and for the benefit of, the Trust Beneficiaries, and, with respect to the Other Debtor Residual Trust Assets, if any, on behalf of, and for the benefit of, the DIP Lenders.
- The Trust Administrator shall have all powers necessary to implement the H. provisions of this Trust Agreement and administer the Trust in respect of the Avoidance Action Trust Assets, including the power to: (i) prosecute for the benefit of the Trust Beneficiaries, through counsel and other professionals selected by the Trust Administrator, the Term Loan Avoidance Action and any other causes of action that may from time to time be held by the Trust in respect thereof; (ii) preserve and maintain the Avoidance Action Trust Assets; (iii) distribute the Distributable Trust Assets, if any, to the Trust Beneficiaries in accordance with the Plan, the Confirmation Order and this Trust Agreement; (iv) expend the Avoidance Action Trust Administrative Cash, the Avoidance Action Trust SEC Reporting Cash (if applicable), and the Supplemental Avoidance Action Trust Cash to cover fees and expenses of the Trust, including payment of taxes and the fees and expenses of the Trust Professionals (other than in respect thereof of the Other Debtor Residual Trust Assets); (v) enter into any contracts or agreements necessary or desirable to facilitate the implementation of the provisions of this Trust Agreement, including but not limited to loan agreements or sale agreements for the purposes of funding the fees and expenses of the Trust; and (vi) sell or grant liens on the Term Loan Avoidance Action or any other property of the Trust (other than the Other Debtor Residual Trust Assets) pursuant to the terms of this Trust Agreement for the purposes of implementing the provisions of this Trust Agreement and the 9019 Order, including but not limited to funding the fees and expenses of the Trust.
- I. The Trust Administrator shall also have all powers necessary to implement the provisions of this Trust Agreement and administer the Trust in respect of the Other Debtor Residual Trust Assets, if any, including the power to: (i) prosecute for the benefit of the DIP Lenders, through counsel and other professionals selected by the Trust Administrator, any causes of action that may from time to time be held by the Trust in respect thereof; (ii) preserve and maintain the Other Debtor Residual Trust Assets; (iii) distribute the Distributable Other Debtor Residual Trust Assets, if any, to the DIP Lenders in accordance with the Plan, the Confirmation Order and this Trust Agreement; and (iv) expend the Other Debtor Residual Trust Administrative Cash to cover fees and expenses of the Trust, including payment of the fees and expenses of the Trust Professionals, in respect thereof.
- K. The Trust is subject to the continuing jurisdiction of the Bankruptcy Court, whose approval is required to pay or distribute money or property to, or on behalf of, a Trust Beneficiary, except as expressly provided in this Trust Agreement.
- L. The Trust (other than the Avoidance Action Trust Claims Reserve) is intended to qualify as a liquidating trust under Treasury Regulation section 301.7701-4(d) that is treated as a "grantor trust" for federal and applicable state and local income tax purposes.

Agreement

NOW, **THEREFORE**, in consideration of the promises and the mutual covenants contained herein, the Trust Administrator and the Trust Monitor agree as follows:

ARTICLE I DEFINED TERMS

- 1.1. <u>Definitions</u>. Whenever used in this Trust Agreement, unless the context otherwise requires, the following words and phrases shall have the respective meanings ascribed to them as follows:
- (a) "9019 Motion" has the meaning set forth in the preamble to this Trust Agreement.
- (b) "9019 Order" has the meaning set forth in the preamble to this Trust Agreement.
- (c) "<u>Affiliates</u>" means, with respect to any Person, any other Person which directly or indirectly controls, is controlled by or is under common control with such Person. For purposes of this definition "control" means, with respect to any Person, the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of such Person, whether through ownership of voting securities, by contract or otherwise.
- (d) "<u>Aggregate Maximum Amount</u>" means the sum of the Maximum Amounts of all Disputed General Unsecured Claims, Unresolved Term Loan Avoidance Action Claims and Unresolved Other Avoidance Action Claims, as set forth in the applicable GUC Trust Report as the Aggregate Maximum Amount as of a given date.
- (e) "<u>Agreed Allocation</u>" means the DIP Lenders shall be entitled to receive thirty percent (30%), and holders of Allowed General Unsecured Claims (or Units) shall be entitled to receive seventy percent (70%), with each such distribution to be made at or about the same time and on a pari passu basis.
- (f) "Allowed General Unsecured Claims" means, collectively, (i) the Initial Allowed General Unsecured Claims and (ii) the Resolved Allowed General Unsecured Claims.
- (g) "Avoidance Action Proceeds" means the proceeds of the Term Loan Avoidance Action.
- (h) "Avoidance Action Trust Administrative Cash" means the Trust Cash (other than the Supplemental Avoidance Action Trust Cash and the Avoidance Action Trust SEC Reporting Cash), including the fifteen million (\$15,000,000) Litigation Cost Advance and the \$1.6 million initially received from the DIP Lenders by the Debtors on the Avoidance Action Trust Transfer Date, held and maintained by the Trust Administrator for the purpose of paying the fees and expenses incurred by the Trust Administrator (including fees and expenses for Trust Professionals) in connection with the Trust and any obligations imposed on the Trust Administrator or the Trust, including fees and expenses relating to the performance of the Trust

Administrator's obligations under this Trust Agreement and the Plan, other than in respect of the Other Debtor Residual Trust Assets.

- (i) "Avoidance Action Trust Assets" means, collectively, (i) the Term Loan Avoidance Action transferred to the Trust, (ii) the Avoidance Action Proceeds, (iii) the Avoidance Action Trust Administrative Cash, (iv) the Avoidance Action Trust SEC Reporting Cash; (v) the Supplemental Avoidance Action Trust Cash; (vi) the Litigation Cost Advance; (vii) the CE Supplemental Capital and (viii) the LW Supplemental Capital.
- (j) "Avoidance Action Trust SEC Reporting Costs" means any costs, fees or expenses incurred by the Trust that are directly or indirectly related to reports that may be required to be filed by the Trust with the SEC pursuant to applicable rules, regulations and interpretations of the SEC (including, without limitation, any legal, accounting or registration fees, costs and expenses incurred by the Trust with respect thereto).
- (k) "Avoidance Action Trust SEC Reporting Cash" has the meaning set forth in Section 2.3(e) of this Trust Agreement.
- (l) "Avoidance Action Trust Transfer Date" means the date selected by the Debtors on which the Avoidance Action Trust Assets (other than the Supplemental Avoidance Action Trust Cash and the Avoidance Action Proceeds) are transferred to the Trust, which transfer shall occur on or before December 15, 2011.
- (m) "<u>Bankruptcy Code</u>" has the meaning set forth in the preamble to this Trust Agreement.
 - (n) "<u>Bankruptcy Court</u>" has the meaning set forth in Background paragraph A.
 - (o) "<u>Budget</u>" has the meaning set forth in <u>Section 6.3</u>.
- (p) "<u>calendar quarter</u>" means the relevant three-month period ending on the last day of March, June, September or December, as applicable, of each calendar year; <u>provided</u>, <u>however</u>, that the calendar quarter that contains the Avoidance Action Trust Transfer Date shall be the period commencing on the Avoidance Action Trust Transfer Date and concluding on the date on which the relevant calendar quarter would naturally end in accordance with the foregoing.
- (q) "<u>CE Capital Providers</u>" has the meaning set forth in the preamble to this Trust Agreement.
- (r) "<u>CE Capital Providers' Entitlement</u>" means the funds to which the CE Capital Providers are entitled pursuant to the Capital Provision Agreement.
- (s) "<u>CE Capital Providers Funding Account</u>" means that segregated account, in the name of the Trust, established at Wilmington Trust Company, in which the Trust shall hold the CE Supplemental Capital. For the avoidance of doubt, the CE Capital Providers Funding Account shall contain Other Supplemental Cash as defined in this Trust Agreement.

- (t) "<u>CE Capital Provision Agreement</u>" has the meaning set forth in the preamble to this Trust Agreement, as the same may be amended, restated, supplemented or modified from time to time after the date hereof.
- (u) "<u>CE Capital Provision Motion</u>" has the meaning set forth in the preamble to this Trust Agreement.
- (v) "<u>CE Capital Provision Order</u>" has the meaning set forth in the preamble to this Trust Agreement.
- (w) "<u>CE Supplemental Capital</u>" has the meaning set forth in the preamble to this Trust Agreement.
- (x) "<u>Capital Provider</u>" and "<u>Capital Providers</u>" have the meanings set forth in the preamble to this Trust Agreement.
- (y) "Certificate of Trust" means the certificate of trust of the Trust as required by Section 3810 of the Delaware Act.
 - (z) "Chapter 11 Cases" has the meaning set forth in Background paragraph A.
 - (aa) "Claim Conflict Resolution" has the meaning set forth in Section 3.6.
- (bb) "<u>Confidential Party</u>" and "<u>Confidential Parties</u>" have the meanings set forth in Section 13.12.
- (cc) "<u>Confirmation Order</u>" has the meaning set forth in Background paragraph D.
- (dd) "<u>Current Total Amount</u>" means as of a given date, the sum of (A) the Total Allowed Amount as of such date and (B) the Aggregate Maximum Amount as of such date, as set forth in the applicable GUC Trust Report as the Current Total Amount as of a given date.
- (ee) "<u>Debtors</u>" has the meaning set forth in the preamble to this Trust Agreement.
- (ff) "<u>Delaware Act</u>" means the Delaware Statutory Trust Act, 12 <u>Del</u>. <u>C</u>. §3801 <u>et seq</u>.
- (gg) "<u>DIP Credit Agreement Claims</u>" means all Claims arising under the DIP Credit Agreement and Orders approving the DIP Credit Agreement dated June 25, 2009 and July 5, 2009.
- (hh) "<u>DIP Lender Advances</u>" means an amount equal to (i) the amounts of Cash advanced directly or indirectly by MLC and the DIP Lenders to fund the costs and expenses associated with realizing the proceeds of the Term Loan Avoidance Action, including, without limitation, any such amounts expended to fund the costs and expenses of professionals retained by the defendants in the Term Loan Avoidance Action, (ii) without duplication, the amount of the

Avoidance Action Trust Administrative Cash, and (iii) without duplication, the Litigation Cost Advance.

- (ii) "<u>DIP Lender Distributable Trust Assets</u>" means the assets distributable to the DIP Lenders, in accordance with the Agreed Allocation and the Settlement Agreement.
- (jj) "<u>DIP Lenders</u>" means the U.S. Treasury and Export Development Canada, as lenders under the DIP Credit Agreement.
- (kk) "<u>Disputed General Unsecured Claims</u>" means the General Unsecured Claims against the Debtors that are Disputed (as defined in the Plan) as of the Initial GUC Record Date, until a time when such claims become Resolved General Unsecured Claims or are otherwise resolved pursuant to the claims resolution procedures contained in the Plan.
- (ll) "<u>Distributable Other Debtor Residual Trust Assets</u>" means the Other Debtor Residual Assets Proceeds, if any, together with any earnings (including interest) thereon.
- (mm) "<u>Distributable Other Debtor Residual Trust Cash</u>" means any Cash or cash equivalents included in the Distributable Other Debtor Residual Trust Assets.
- (nn) "<u>Distributable Trust Assets</u>" means the Avoidance Action Proceeds together with any earnings (including interest) thereon.
- (oo) "<u>Distributable Trust Cash</u>" means any Cash or cash equivalents included in the Distributable Trust Assets.
- (pp) "<u>Distribution Date</u>" means the date of any distribution made by the Trust Administrator to the Trust Beneficiaries pursuant to this Trust Agreement, whether to the DIP Lenders pursuant to <u>Section 5.1(d)</u> or on account of Allowed General Unsecured Claims and/or Units.
 - (qq) "<u>Distribution Threshold</u>" means \$10,000,000.
- (rr) "<u>Excess Distributable Trust Assets</u>" means Distributable Trust Assets that the Trust Administrator, with the approval of the Trust Monitor, determines shall be distributed following the Initial GUC Distribution Date.
- (ss) "Excess GUC Distributable Trust Assets Determination Date" has the meaning set forth in Section 5.4(a).
- (tt) "<u>Excess GUC Distributable Trust Assets</u>" has the meaning set forth in <u>Section 5.4(c)</u>.
 - (uu) "Final Recovery Date" has the meaning set forth in Section 5.1(a).
- (vv) "<u>First Amended and Restated Trust Agreement</u>" has the meaning set forth in the preamble to this Trust Agreement.

- (ww) "Funding Account" means that segregated account, in the name of the Trust, established at a bank reasonably acceptable to the Trust and the DIP Lenders, in which the Trust shall hold the Litigation Cost Advance.
- (xx) "GUC Beneficiaries" means the holders of Allowed General Unsecured Claims or Units received in respect of such claims."
- (yy) "<u>GUC Distributable Trust Assets</u>" means: (i) the Segregated Account Assets; and (ii) the assets distributable to the holders of Allowed General Unsecured Claims in accordance with the Agreed Allocation and the Settlement Agreement.
- (zz) "GUC Trust Advances" means the Distributable Trust Assets distributable to the holders of Allowed General Unsecured Claims, in an amount equal to the aggregate amount of GUC Trust Supplemental Cash received from the GUC Trust pursuant to Section 2.3(f)(i) hereof, together with any earnings (including interest) thereon, minus the aggregate amount of Cash held in the Segregated Account on the date of measurement.
- (aaa) "GUC Trust Reports" means the reports prepared by the GUC Trust Administrator each quarter as provided in the GUC Trust Agreement, which shall be delivered to the Trust Administrator pursuant to the terms of the GUC Trust Agreement.
- (bbb) "GUC Trust Supplemental Cash" has the meaning set forth in Section 2.3(f)(i).
 - (ccc) "Holdback" has the meaning set forth in Section 6.1(b).
- (ddd) "<u>Incompetency</u>" means, with respect to any Person, the incompetency of such Person if such Person is a natural person.
- (eee) "<u>Initial Allowed General Unsecured Claims</u>" has the meaning set forth in <u>Section 5.2(a)</u>.
 - (fff) "Initial GUC Distribution Date" has the meaning set forth in Section 5.2(a).
 - (ggg) "Initial GUC Record Date" has the meaning set forth in Section 5.2(a).
 - (hhh) "IRS" means the Internal Revenue Service.
- (iii) "<u>LW Capital Provider</u>" has the meaning set forth in the preamble to this Trust Agreement.
- (jjj) "<u>LW Capital Provider's Entitlement</u>" means the funds to which the LW Capital Provider is entitled pursuant to the LW Provider Capital Provision Agreement.
- (kkk) "<u>LW Capital Provider Funding Account</u>" means that segregated account, in the name of the Trust, established at Wilmington Trust Company, in which the Trust shall hold the LW Supplemental Capital.

- (III) "<u>LW Capital Provision Agreement</u>" has the meaning set forth in the preamble to this Trust Agreement, as the same may be amended, restated, supplemented or modified from time to time after the date hereof.
- (mmm)"<u>LW Capital Provision Motion</u>" has the meaning set forth in the preamble to this Trust Agreement.
- (nnn) "<u>LW Capital Provision Order</u>" has the meaning set forth in the preamble to this Trust Agreement.
- (000) "<u>LW Supplemental Capital</u>" has the meaning set forth in the preamble to this Trust Agreement.
- (ppp) "<u>Litigation Cost Advance</u>" means the fifteen million dollars (\$15,000,000) to be provided by the DIP Lenders pursuant to the Litigation Cost Advance Agreement and the Settlement Agreement and subject to a deposit account control agreement acceptable to the DIP Lenders and the Trust.
- (qqq) "<u>Litigation Cost Advance Agreement</u>" has the meaning set forth in the preamble to this Trust Agreement.
- (rrr) "<u>Litigation Settlement</u>" has the meaning set forth in the preamble to this Trust Agreement.
- (sss) "Maximum Amount" means the maximum amount of any Disputed General Unsecured Claim, Unresolved Term Loan Avoidance Action Claim or Unresolved Other Avoidance Action Claim, calculated in accordance with the GUC Trust Agreement and as set forth in the applicable GUC Trust Report as the Maximum Amount of any Claim or group of Claims as of a given date.
 - (ttt) "MLC" has the meaning set forth in the preamble to this Trust Agreement.
- (uuu) "Original Trust Agreement" has the meaning set forth in the preamble to this Trust Agreement.
- (vvv) "Other Avoidance Action Claims" means the additional General Unsecured Claims that have arisen as a result of recovery of proceeds of the Avoidance Actions other than the Term Loan Avoidance Action (and any related unsecured claims).
- (www) "Other Debtor Residual Assets" means any assets of MLC remaining at such time as the Debtors shall be liquidated, other than the Term Loan Avoidance Action and the Term Loan Avoidance Action Administrative Cash and any other assets of MLC whose disposition is specifically provided for under the Plan or the Confirmation Order.
- (xxx) "Other Debtor Residual Accepted Assets" means Other Debtor Residual Assets accepted by the Trust Administrator for transfer to the Trust pursuant to Section 2.3(b).

- (yyy) "<u>Other Debtor Residual Assets Proceeds</u>" means any proceeds realized in respect of the Other Debtor Residual Accepted Assets.
- (zzz) "Other Debtor Residual Trust Administrative Cash" means the Cash, if any, held and maintained by the Trust Administrator for the purpose of paying the fees and expenses incurred by the Trust Administrator (including fees and expenses for Trust Professionals) in connection with the Trust and any obligations imposed on the Trust Administrator or the Trust, including fees and expenses relating to the performance of the Trust Administrator's obligations under this Trust Agreement and the Plan, but only in respect of the Other Debtor Residual Trust Assets, which Cash may be obtained by transfer to the Trust by the Debtors, from the DIP Lenders (in their sole discretion) or from the proceeds of the Other Debtor Residual Accepted Assets.
- (aaaa) "Other Debtor Residual Trust Assets" means, if any, collectively, (i) the Other Debtor Residual Accepted Assets transferred to the Trust, (ii) the Other Debtor Residual Assets Proceeds and (iii) the Other Debtor Residual Trust Administrative Cash.
- (bbbb) "Other Debtor Residual Assets Transfer Date" means the date selected by the Debtors on which the Other Debtor Residual Assets Transfer, if any, are transferred to the Trust, which transfer shall occur on or before December 15, 2011.
- (cccc) "Other Supplemental Cash" means any Cash received by the Trust pursuant to Section 6.1(d) hereof from the sale of, or granting of liens on, all or a portion of the Term Loan Avoidance Action or any other property held by the Trust (other than the Other Debtor Residual Trust Assets), or from any source other than the GUC Trust, including the CE Supplemental Capital and the LW Supplemental Capital.
 - (dddd) "Permissible Investments" means investments in any of the following:
- (i) Marketable securities issued by the U.S. Government and supported by the full faith and credit of the U.S. Treasury, either by statute or an opinion of the Attorney General of the United States;
- (ii) Marketable debt securities, rated Aaa by Moody's and/ or AAA by S&P, issued by U.S. Government-sponsored enterprises, U.S. Federal agencies, U.S. Federal financing banks, and international institutions whose capital stock has been subscribed for by the United States;
- (iii) Certificates of deposit, time deposits, and bankers acceptances of any bank or trust company incorporated under the laws of the United States or any state, provided that, at the date of acquisition, such investment, and/or the commercial paper or other short term debt obligation of such bank or trust company has a short-term credit rating or ratings from Moody's and/or S&P, each at least P-1 or A-1;
- (iv) Commercial paper of any corporation incorporated under the laws of the United States or any state thereof which on the date of acquisition is rated by Moody's and/or S&P, provided each such credit rating is least P-1 and/or A-1;

- (v) Money market mutual funds that are registered with the Securities and Exchange Commission under the Investment Company Act of 1940, as amended, and operated in accordance with Rule 2a-7 and that at the time of such investment are rated Aaa by Moody's and/or AAAm by S&P, including such funds for which the Trust Administrator or an Affiliate provides investment advice or other services;
- (vi) Tax-exempt variable rate commercial paper, tax-exempt adjustable rate option tender bonds, and other tax-exempt bonds or notes issued by municipalities in the United States, having a short-term rating of "MIG-1" or "VMIG-1" or a long term rating of "AA" (Moody's), or a short-term rating of "A-1" or a long term rating of "AA" (S&P); and
- (vii) Repurchase obligations with a term of not more than thirty days, 102 percent collateralized, for underlying securities of the types described in clauses (i) and (ii) above, entered into with any bank or trust company or its respective affiliate meeting the requirements specified in clause (iii) above.
 - (eeee) "Plan" has the meaning set forth in the preamble to this Trust Agreement.
- (ffff) "Resolved Allowed General Unsecured Claims" means, collectively, (I) the Disputed General Unsecured Claims that are allowed after the Initial GUC Record Date in accordance with the claims resolution procedures administered under the Plan (to the extent so resolved); (II) the Term Loan Avoidance Action Claims, to the extent and in the amount collected by the Trust against the respective defendants in the underlying litigation (including by way of settlement); and (III) the Other Avoidance Action Claims, to the extent and in the amount collected against the respective defendants in the underlying litigations (including by way of settlement). For the avoidance of doubt, unless and until a Disputed General Unsecured Claim, Unresolved Term Loan Avoidance Action Claim or Unresolved Other Avoidance Action Claim becomes a Resolved Allowed General Unsecured Claim, the holders of such claim shall not receive any distribution from the Trust.
- (gggg) "Resolved Allowed General Unsecured Claims Determination Date" has the meaning set forth in Section 5.3(a).

(hhhh) RESERVED

- (iiii) RESERVED
- (jjjj) "SEC" means the Securities and Exchange Commission.
- (kkkk) "<u>Second Amended and Restated Trust Agreement</u>" has the meaning set forth in the preamble to this Trust Agreement.
- (IIII) "<u>Secretary of State</u>" means the Office of the Secretary of State of the State of Delaware.
 - (mmmm) "Segregated Account" has the meaning set forth in Section 2.3(f)(i).

- (nnnn) "Segregated Account Assets" means the Cash held in the Segregated Account.
- (0000) "Settlement Agreement" has the meaning set forth in the preamble to this Trust Agreement.
- (pppp) "<u>Supplemental Avoidance Action Trust Cash</u>" means the GUC Trust Supplemental Cash and the Other Supplemental Cash.
- (qqqq) "<u>Supplemental Capital</u>" has the meaning set forth in the preamble to this Trust Agreement.
- (rrrr) "<u>Tax Returns</u>" means all tax returns, reports, certificates, forms or similar statements or documents.
- (ssss) "Term Loan Avoidance Action" means the Avoidance Action commenced by the Creditors' Committee against JPMorgan Chase Bank, N.A., individually and as Administrative Agent, and various lenders party to a term loan agreement, dated as of November 29, 2006, between General Motors Corporation, as borrower, JPMorgan Chase Bank, N.A., as agent, and various institutions as lenders and agents, styled *Official Committee of Unsecured Creditors of Motors Liquidation Co. v. JPMorgan Chase Bank, N.A. et al.*, Adv. Pro. No. 09-00504 (Bankr. S.D.N.Y. July 31, 2009).
- (tttt) "<u>Term Loan Avoidance Action Claims</u>" means the additional General Unsecured Claims that have arisen as a result of recovery of proceeds of the Term Loan Avoidance Action (or any related unsecured claims).
- (uuuu) "<u>Third Amended and Restated Trust Agreement</u>" has the meaning set forth in the preamble to this Trust Agreement.
- (vvvv) "<u>Total Allowed Amount</u>" means the sum of the amount of all Initial Allowed General Unsecured Claims plus the amount of all Resolved Allowed General Unsecured Claims, as set forth in the applicable GUC Trust Report.
- (wwww) "<u>Treasury Regulations</u>" means the income tax regulations promulgated under the Tax Code, including any amended or successor income tax regulations thereto.
 - (xxxx) "Trust" has the meaning set forth in the preamble to this Trust Agreement.
- (yyyy) "<u>Trust Administrator</u>" has the meaning set forth in the preamble to this Trust Agreement.
- (zzzz) "<u>Trust Administrator Parties</u>" means the Trust Administrator and its principals, directors, officers, employees, agents, representatives, attorneys, accountants, advisors and other professionals (including the Trust Professionals).

- (aaaaa) "<u>Trust Agreement</u>" has the meaning set forth in the preamble to this Trust Agreement.
- (bbbbb) "<u>Trust Beneficiaries</u>" means the holders of the DIP Credit Agreement Claims and the holders of Allowed General Unsecured Claims (or Units received in respect of such claims).
- (cccc) "<u>Trust Cash</u>" means the Cash or cash equivalents included in the Avoidance Action Trust Assets or the Other Debtor Residual Trust Assets, if any.
- (ddddd) "<u>Trust Monitor</u>" has the meaning set forth in the preamble to this Trust Agreement.
- (eeeee) "<u>Trust Monitor Parties</u>" means the Trust Monitor and its principals, directors, officers, employees, agents, representatives, attorneys, accountants, advisors and other professionals.
- (fffff) "<u>Trust Professionals</u>" means, collectively, independent contractors, including attorneys, accountants, appraisers, disbursing agents or other parties deemed by the Trust Administrator to have the qualifications necessary or desirable to assist in the proper administration of the Trust and that are employed or retained by the Trust in such capacities.
- (ggggg) "<u>Unit Issuance Ratio</u>" means the ratio of one Unit for each \$1,000 in amount of Allowed General Unsecured Claims.
- (hhhhh) "<u>Units</u>" means the units of beneficial interest issued by the Trust to holders of Allowed General Unsecured Claims.
- (iiii) "<u>Unresolved Other Avoidance Action Claim</u>" means an Other Avoidance Action Claim that has not or has not yet arisen because no determination (including by way of settlement) has been made in the respective Avoidance Action against the respective defendant who would be entitled to such claim in the event of such determination (or if a determination has been made against the defendant, the proceeds related to such resolution have not been recovered in full).
- (jjjjj) "Unresolved Term Loan Avoidance Action Claim" means a Term Loan Avoidance Action Claim that has not or has not yet arisen because no determination (including by way of settlement) has been made in the Term Loan Avoidance Action against the respective defendant who would be entitled to such claim in the event of such determination (or if a determination has been made against the defendant, the proceeds related to such resolution have not been recovered in full).
- (kkkk) "<u>Wind-Down Facility</u>" means the \$1.175 billion wind-down facility provided to the Debtors pursuant to the DIP Credit Agreement.

ARTICLE II DECLARATION OF TRUST

- 2.1. <u>Creation of Trust</u>. The Debtors and the Trust Administrator, pursuant to the Plan and the Confirmation Order and in accordance with the applicable provisions of chapter 11 of the Bankruptcy Code, hereby constitute and create the Trust, in the form of a statutory trust under the Delaware Act, which shall bear the name "Motors Liquidation Company Avoidance Action Trust." In connection with the exercise of the Trust Administrator's power hereunder, the Trust Administrator may use this name or such variation thereof as the Trust Administrator sees fit. The Trust Administrator, as trustee of the Trust, is hereby authorized and directed to execute and file a Certificate of Trust for the Trust in the form attached hereto as <u>Exhibit B</u>.
- 2.2. <u>Purpose of Trust</u>. The sole purpose of the Trust is to liquidate and distribute its assets pursuant to the Plan in accordance with Treasury Regulation section 301.7701-4(d), with no objective to continue or engage in the conduct of a trade or business.

2.3. Transfer of Avoidance Action Trust Assets to the Trust.

- Effective upon the Avoidance Action Trust Transfer Date, the Debtors (a) hereby transfer to the Trust, pursuant to Bankruptcy Code Sections 1123(a)(5)(B) and 1123(b)(3)(B), and in accordance with the Plan and the Confirmation Order, the Avoidance Action Trust Assets (other than the Avoidance Action Trust SEC Reporting Cash, and the Supplemental Avoidance Action Trust Cash), as they exist on the Avoidance Action Trust Transfer Date, free and clear of any and all liens, claims, encumbrances and interests (legal, beneficial or otherwise) of all other persons to the maximum extent contemplated by and permissible under Bankruptcy Code Section 1141(c); provided, however that notwithstanding anything to the contrary in the Plan, Disclosure Statement, Confirmation Order, this Trust Agreement or any other agreement, the DIP Lenders shall maintain their liens on the Avoidance Action Trust Administrative Cash, provided that for the avoidance of doubt, the DIP Lenders shall not demand acceleration of their liens on the Avoidance Action Trust Administrative Cash except in accordance with the provisions of section 7.2 of the DIP Credit Agreement. Such transfers shall be exempt from any stamp, real estate transfer, mortgage reporting, sales, use or other similar tax pursuant to Bankruptcy Code Section 1146. The Debtors and their successors and assigns shall be released from any and all liability with respect to the transfer of such Avoidance Action Trust Assets to the Trust as aforesaid. Nothing in this Trust Agreement is intended to, or shall be construed to, effect a release, extinguishment or compromise of any claim or cause of action transferred to the Trust pursuant to this Trust Agreement. The Avoidance Action Trust Assets and all other property held from time to time by the Trust under this Trust Agreement (other than the Other Debtor Residual Trust Assets) and any earnings (including interest) thereon are to be managed, applied and disposed of by the Trust Administrator in accordance with the terms hereof, the Plan and the Confirmation Order for the benefit of the Trust Beneficiaries and the Capital Providers, and for no other party, subject to the further covenants, conditions and terms hereinafter set forth, including the provisions of Section 2.6.
- (b) To the extent any Avoidance Action Trust Assets (other than the Supplemental Avoidance Action Trust Cash) cannot be transferred to the Trust, because of a restriction on transferability under applicable non-bankruptcy law that is not superseded by

Bankruptcy Code Section 1123 or any other provision of the Bankruptcy Code, such assets shall be retained by the Debtors or any successor thereto including, without limitation, the GUC Trust. The proceeds of the sale of any such assets retained by the Debtors (or any successor thereto) shall be allocated to the Trust pursuant to the Plan as if such transfer had not been restricted under applicable non-bankruptcy law. The Trust Administrator may commence an action in the Bankruptcy Court to resolve any dispute regarding the allocation of the proceeds of any assets retained by the Debtors (or any successor thereto) pursuant to the Plan and Confirmation Order.

- (c) On the Avoidance Action Trust Transfer Date, the Debtors shall also deliver, or cause to be delivered, to the Trust a complete list of all General Unsecured Claims, both Allowed and Disputed, reflected on the claims registry as of the Avoidance Action Trust Transfer Date, including the names and addresses of all holders of such General Unsecured Claims, whether such claims have been Allowed or are Disputed, and the details of all objections in respect of Disputed General Unsecured Claims.
- Effective upon the Other Debtor Residual Assets Transfer Date, the Debtors (d) hereby transfer to the Trust, pursuant to Bankruptcy Code Sections 1123(a)(5)(B) and 1123(b)(3)(B), and in accordance with the Plan and the Confirmation Order, such of the Other Debtor Residual Assets, as they exist on the Other Debtor Residual Assets Transfer Date, as the Trust Administrator, in its sole discretion but with the approval of the Trust Monitor, shall determine to accept, free and clear of any and all liens, claims, encumbrances and interests of all other persons to the maximum extent contemplated by and permissible under Bankruptcy Code Section 1141(c); provided that, for the avoidance of doubt, the Trust Administrator may determine not to accept the transfer to the Trust of any or all of the Other Debtor Residual Assets for any reason or for no reason. Any such transfer shall be exempt from any stamp, real estate transfer, mortgage reporting, sales, use or other similar tax pursuant to Bankruptcy Code Section 1146. The Debtors and their successors and assigns shall be released from any and all liability with respect to the transfer of the Other Debtor Residual Accepted Assets to the Trust as aforesaid. Nothing in this Trust Agreement is intended to, or shall be construed to, effect a release, extinguishment or compromise of any claim or cause of action transferred to the Trust pursuant to this Trust Agreement, and notwithstanding anything to the contrary in the Plan, Disclosure Statement, Confirmation Order, this Trust Agreement or any other agreement, the DIP Lenders shall maintain their liens on the Other Debtor Residual Accepted Assets. The Other Debtor Residual Trust Assets and all other property held from time to time by the Trust under this Trust Agreement in respect thereof, and any earnings (including interest) thereon, are to be managed, applied and disposed of by the Trust Administrator in accordance with the terms hereof, the Plan and the Confirmation Order for the benefit of the DIP Lenders, and for no other party, subject to the further covenants, conditions and terms hereinafter set forth, including the provisions of Section 2.6.
- (e) (i) On the Avoidance Action Trust Transfer Date, the Debtors shall, pursuant to Section 2.3(e) of the GUC Trust Agreement, transfer Cash to the Trust in an amount of \$500,000 (the "Avoidance Action Trust SEC Reporting Cash"). The Avoidance Action Trust SEC Reporting Cash shall be used solely for the satisfaction of Avoidance Action Trust SEC Reporting Costs. Any taxes imposed on the Trust in respect of the Avoidance Action Trust SEC Reporting Cash shall be satisfied from the income realized thereon.

- (ii) The Trust Administrator shall only use Avoidance Action Trust SEC Reporting Cash to satisfy Avoidance Action Trust SEC Reporting Costs to extent there is no other available source of funds to satisfy such expenses (other than the Supplemental Avoidance Action Trust Cash), including, without limitation, any funds obtained through the reservation and application of all or a portion of the Holdback pursuant to Section 6.1(b) hereof.
- (iii) If the Trust Administrator determines that (x) reports are not, and at no time will be, required to be filed by the Trust with the SEC pursuant to applicable rules, regulations and interpretations of the SEC or (y) the Trust has other available funds (as set forth in Section 2.3(e)(ii) hereof) which are sufficient to satisfy any current or future projected, fees, costs or expenses that are directly or indirectly related to reports that may be required to be filed by the Trust with the SEC pursuant to applicable rules, regulations and interpretations of the SEC (including, without limitation, any legal, accounting or registration fees, costs and expenses incurred by the Trust with respect thereto), then the Trust shall transfer to the GUC Trust any Avoidance Action Trust SEC Reporting Cash that has not been applied as provided in this Section 2.3(e).
- (iv) Any income earned on the Avoidance Action Trust SEC Reporting Cash, net of taxes paid thereon, shall be Avoidance Action Trust Administrative Cash.
- (f) (i) From time to time, in accordance with a Final Order of the Bankruptcy Court if so required, the GUC Trust may deliver Cash to the Trust to be used solely for the satisfaction of fees and expenses of the Trust (including for the payment of the fees and expenses of Trust Professionals) or the satisfaction of any federal, state or local taxes incurred by the Trust. Any Cash so delivered to the Trust shall be designated as "GUC Trust Supplemental Cash." GUC Trust Supplemental Cash shall be held by the Trust in a segregated account (the "Segregated Account"), and shall be used solely for the satisfaction of fees and expenses of the Trust (including for the payment of the fees and expenses of Trust Professionals), for the satisfaction of taxes incurred by the Trust or for distribution to holders of Allowed General Unsecured Claims pursuant to Section 5.1(d)(iv) hereof.
- (ii) The Trust Administrator shall only use the GUC Trust Supplemental Cash to satisfy fees and expenses of the Trust and tax liabilities of the Trust to the extent that there is no other source of funds to satisfy such expenses (other than the Avoidance Action Trust SEC Reporting Cash, and other than through the sale or encumbrance of the Term Loan Avoidance Action or any other property of the Trust pursuant to Section 6.1(d) hereof), including, without limitation, Avoidance Action Trust Administrative Cash, and any funds obtained through the reservation and application of all or a portion of the Holdback pursuant to Section 6.1(b) hereof. Any income earned on the GUC Trust Supplemental Cash, net of taxes paid thereon, shall be GUC Trust Supplemental Cash.
- 2.4. <u>Appointment and Acceptance of Trust Administrator</u>. The Trust Administrator shall be deemed to be appointed pursuant to Bankruptcy Code Section 1123(b)(3)(B) and is hereby appointed trustee of the Trust under the Delaware Act. The Trust Administrator hereby accepts such appointments, including trusteeship of the Trust created by this Trust Agreement and the grant, assignment, transfer, conveyance and delivery by the Debtors to the Trust Administrator, (i) on behalf of the Trust, and for the benefit of the Trust Beneficiaries, of all of their respective right,

title and interest in the Distributable Trust Assets, and (ii) on behalf of the Trust, and for the benefit of the DIP Lenders, of all of their respective right, title and interest in the Other Debtor Residual Trust Assets, upon and subject to the terms and conditions set forth in the Plan, the Confirmation Order and this Trust Agreement. The Trust Administrator's powers are exercisable solely in a fiduciary capacity consistent with, and in furtherance of, the purpose of the Trust and not otherwise, and in accordance with applicable law, including the Delaware Act. The Trust Administrator shall have the authority to bind the Trust within the limitations set forth herein, but shall for all purposes hereunder be acting in the capacity as Trust Administrator, and not individually.

2.5. <u>Distribution of Distributable Trust Assets</u>. The Trust Administrator shall, in an expeditious but orderly manner and subject to the provisions of the Plan, the Confirmation Order and this Trust Agreement, make timely distributions of the Distributable Trust Assets and the Distributable Other Debtor Residual Assets in accordance with the terms hereof and not unduly prolong the existence of the Trust. The Trust Administrator may incur and pay any reasonable and necessary expenses in connection with the administration of the Trust, including the fees and expenses of the Trust Professionals <u>provided</u>, <u>however</u>, that all such expenditures (other than in respect of the Other Debtor Residual Trust Assets) shall be made in accordance with the Budget.

2.6. No Reversion to Debtors.

- (a) In no event shall any part of the Avoidance Action Trust Assets or the Other Debtor Residual Trust Assets revert to or be distributed to or for the benefit of any Debtor. All Distributable Trust Assets shall be applied as provided in Section 5.1(d), including to the satisfaction of Allowed General Unsecured Claims, including through distributions made in respect of the Units. All Distributable Other Debtor Residual Trust Assets shall be applied as provided in Article 5A.
- To the extent that after satisfaction in full of all of the costs and expenses of (b) the administration of the Trust, after all Allowed General Unsecured Claims have been paid pursuant to the Plan, after satisfaction of all other obligations or liabilities of the Trust incurred or assumed in accordance with the Plan, Confirmation Order or this Trust Agreement (or to which the Avoidance Action Trust Assets are otherwise subject), and after the affairs of the Trust have been finally wound up and concluded in accordance with the provisions of Section 4.3 hereof and Section 3808 of the Delaware Act, there shall remain any Avoidance Action Trust Administrative Cash, the Trust Administrator is authorized to and shall distribute any such remaining Avoidance Action Trust Administrative Cash to the DIP Lenders in accordance with the terms of the DIP Credit Agreement and the Plan. To the extent any portion of such residue is not accepted by the respective DIP Lenders, the Trust Administrator shall (i) be authorized to distribute up to \$100,000 of such remaining Avoidance Action Trust Administrative Cash to an organization described in Section 501(c)(3) of the Tax Code and exempt from U.S. federal income tax under section 501(a) of the Tax Code that is unrelated to the Debtors, the Trust, any Trust Administrator Parties or any Trust Monitor Parties, or (ii) request an order from the Bankruptcy Court authorizing the Trust Administrator to distribute any such remaining Avoidance Action Trust Administrative Cash to such an organization, or authorizing such other disposition as recommended by the Trust Administrator and approved by the Bankruptcy Court.

To the extent that after satisfaction in full of all of the costs and expenses of (c) the administration of the Trust, after all Allowed General Unsecured Claims have been paid pursuant to the Plan, after satisfaction of all other obligations or liabilities of the Trust incurred or assumed in accordance with the Plan, Confirmation Order or this Trust Agreement (or to which the Avoidance Action Trust Assets are otherwise subject), and after the affairs of the Trust have been finally wound up and concluded in accordance with the provisions of Section 4.3 hereof and Section 3808 of the Delaware Act, there shall remain any GUC Trust Supplemental Cash, the Trust Administrator is authorized to and shall distribute any such remaining GUC Trust Supplemental Cash to the holders of Allowed General Unsecured Claims (as defined in the GUC Trust Agreement), regardless of whether such amount of GUC Trust Supplemental Cash is less than the Distribution Threshold, provided that, if the remaining GUC Trust Supplemental Cash is less than \$100,000, the Trust Administrator shall (i) be authorized to distribute such remaining GUC Trust Supplemental Cash to an organization described in Section 501(c)(3) of the Tax Code and exempt from U.S. federal income tax under section 501(a) of the Tax Code that is unrelated to the Debtors, the Trust, any Trust Administrator Parties or any Trust Monitor Parties, or (ii) request an order from the Bankruptcy Court authorizing the Trust Administrator to distribute any such remaining GUC Trust Supplemental Cash to such an organization, or authorizing such other disposition as recommended by the Trust Administrator and approved by the Bankruptcy Court.

ARTICLE III TRUST BENEFICIARIES; UNITS

3.1. Rights of Beneficiaries.

- (a) Except as provided in <u>Section 2.6</u> hereof, the Trust Beneficiaries shall be the sole beneficiaries of the Trust (to the extent of the Avoidance Action Trust Assets) and the Distributable Trust Assets, and the Trust Administrator shall retain only such incidents of ownership as are necessary to undertake the actions and transactions authorized in the Plan, the Confirmation Order and this Trust Agreement, including those powers set forth in <u>Articles VI</u> and <u>VIII</u> hereof.
- (b) The beneficial interest of a Trust Beneficiary in the Trust is hereby declared and shall be in all respects and for all purposes intangible personal property.
- c) Except as expressly provided herein, a Trust Beneficiary shall have no title or right to, or possession, management or control of, the Trust, or the Avoidance Action Trust Assets, or to any right to demand a partition or division of such assets or to require an accounting of the Trust Administrator or the Trust Monitor. The whole legal title to the Avoidance Action Trust Assets shall be vested in the Trust as a separate legal entity under the Delaware Act or, if necessary, in the Trust Administrator on behalf of the Trust and the sole beneficial interest of the Trust Beneficiaries shall be as set forth in this Trust Agreement.
- 3.2. <u>Limited Liability</u>. No provision of the Plan, the Confirmation Order or this Trust Agreement, and no mere enumeration herein of the rights or privileges of any Trust Beneficiary, shall give rise to any liability of such Trust Beneficiary solely in its capacity as such, whether such liability is asserted by any Debtor, by creditors or employees of any Debtor, or by any other Person. GUC Beneficiaries are deemed to receive the GUC Distributable Trust Assets in accordance with

the provisions of the Plan, the Confirmation Order and this Trust Agreement in exchange for their Allowed General Unsecured Claims or on account of their Units, as applicable, without further obligation or liability of any kind, but subject to the provisions of this Trust Agreement.

3.3. No Control by Trust Beneficiaries. A Trust Beneficiary shall have no title to, or any right to possess, manage or control, the Avoidance Action Trust Assets, or any portion thereof or interest therein, except as expressly provided herein. No surviving spouse, heir or devisee of any deceased Trust Beneficiary shall have any right of dower, homestead or inheritance, or of partition, or any other right, statutory or otherwise, in the Avoidance Action Trust Assets, but the whole title to all the Avoidance Action Trust Assets shall be vested in the Trust Administrator and the sole interest of the Trust Beneficiaries shall be the rights and benefits provided to such persons under this Trust Agreement.

3.4. Issuance of Units.

- (a) The Trust shall issue Units to holders of Allowed General Unsecured Claims as provided in this Trust Agreement. On the Initial GUC Distribution Date, holders of Initial Allowed General Unsecured Claims shall receive the number of Units equal to the amount of such Initial Allowed General Unsecured Claims multiplied by the Unit Issuance Ratio, rounded up or down to the nearest whole Unit. Following the Initial GUC Distribution Date, holders of Resolved Allowed General Unsecured Claims shall receive the number of Units equal to the amount of such Resolved Allowed General Unsecured Claims multiplied by the Unit Issuance Ratio, rounded up or down to the nearest whole Unit. Units will represent the contingent right to receive, on a pro rata basis as provided in the Plan, the Confirmation Order and this Trust Agreement, Excess GUC Distributable Trust Assets. The Units shall be issued subject to all the terms and conditions of the Plan, the Confirmation Order and this Trust Agreement. References in this Trust Agreement to holders of Units shall be to the record holders of such Units.
- (b) As provided in <u>Section 7.5</u> hereof, the Trust Administrator may retain Units otherwise issuable pursuant to this section with respect to Allowed General Unsecured Claims that are subject to withholding, and the Trust Administrator shall apply amounts distributed in respect of such retained Units to satisfy such withholding obligations.
- (c) Notwithstanding the foregoing, if (i) as of the Initial GUC Distribution Date, the total amount of the Disputed General Unsecured Claims in the aggregate is less than 0.5% of the Current Total Amount, or (ii) the Initial GUC Distribution Date is determined by the Trust Administrator to also be the final Distribution Date, no Units shall be distributed and any GUC Distributable Trust Assets remaining after satisfaction of all Initial Allowed General Unsecured Claims and any other obligations of the Trust shall be disposed of as set forth in the last sentence of Section 2.6(c).

3.5. Ownership of Units; Transfers of Units.

(a) The interest of a Trust Beneficiary is in all respects personal property, and upon the death, insolvency or incapacity of an individual GUC Beneficiary, such GUC Beneficiary's Units shall pass to the legal representative of such GUC Beneficiary.

- (b) The Units will be issued and evidenced by appropriate notation on the books and records of the Trust Administrator. The Units shall not be certificated and shall not be transferable, assignable, pledged or hypothecated in whole or in part, except by applicable laws of descent and distribution (in the case of a deceased individual GUC Beneficiary); by operation of law; in accordance with applicable Bankruptcy law; or as otherwise approved by the Bankruptcy Court. The Trust Administrator shall not be required to recognize any equitable or other claims to such interest by the transferee thereof, and the named GUC Beneficiary shall remain as such for all purposes hereunder.
- Conflicting Claims to Units. If the Trust Administrator has actual knowledge of 3.6. any conflicting claims or demands that have been made or asserted with respect to a Unit, or a beneficial interest therein, the Trust Administrator shall be entitled, at its sole election, to refuse to comply with any such conflicting claims or demands. In so refusing, the Trust Administrator may elect to make no payment or distribution with respect to the Unit subject to the claims or demands involved, or any part thereof, and the Trust Administrator shall be entitled to refer such conflicting claims or demands to the Bankruptcy Court, which shall have exclusive and continuing jurisdiction over resolution of such conflicting claims or demands. The Trust Administrator shall not be or become liable to any party for either (i) its election to continue making distributions pursuant to its books and records, without regard to the conflicting claims or demands; or (ii) its election to cease payments or distributions with respect to the subject Unit. In the event that the Trust Administrator elects to cease payments, it shall be entitled to refuse to act until either (x) the rights of the adverse claimants have been adjudicated by a Final Order of the Bankruptcy Court (or such other court of proper jurisdiction) or (y) all differences have been resolved by a written agreement among all of such parties and the Trust Administrator, which agreement shall include a complete release of the Trust, the Trust Administrator Parties and the Trust Monitor Parties in form and substance reasonably satisfactory to the Trust Administrator and Trust Monitor (the occurrence of either (x) or (y), a "Claim Conflict Resolution"). Until a Claim Conflict Resolution is reached with respect to such conflicting claims or demands, the Trust Administrator shall hold in a segregated account any payments or distributions from the Trust to be made with respect to the Unit(s) at issue. Promptly after a Claim Conflict Resolution is reached, the Trust Administrator shall transfer the payments and distributions, if any, held in the segregated account, together with interest and income earned thereon, if any, in accordance with the terms of such Claim Conflict Resolution.
- 3.7. <u>Distributions Relating to Note Claims and Eurobond Claims</u>. The Trust shall distribute GUC Distributable Trust Assets and Units to the Indenture Trustees and Fiscal and Paying Agents, to the extent necessary to provide each beneficial holder of debt securities arising out of or relating to the Note Claims and Eurobond Claims with an amount of GUC Distributable Trust Assets and Units equal to the amount of GUC Distributable Trust Assets and Units such holder would receive had its claim been treated as an Initial Allowed General Unsecured Claim hereunder. For the avoidance of doubt, Units will be issued and evidenced by appropriate notation on the books and records of the Trust Administrator in the names of the Indenture Trustees and the Fiscal and Paying Agents, as applicable, and not in the individual names of the beneficial holders of debt securities arising out of or relating to the Note Claims and Eurobond Claims.

ARTICLE IV DURATION AND TERMINATION OF THE TRUST

- Duration. The Trust shall become effective upon (x) the earlier to occur of the (i) Avoidance Action Trust Transfer Date and (ii) the Other Debtor Residual Assets Transfer Date, if any, and (y) the execution of this Trust Agreement and the filing of the Certificate of Trust with the Secretary of State and shall remain and continue in full force and effect until (a) all of the Distributable Trust Assets and all Distributable Other Debtor Residual Trust Assets, if any, have been distributed pursuant to the Plan and this Trust Agreement, (b) the Trust Administrator determines, in its discretion and with the approval of the Trust Monitor, that the administration of the Avoidance Action Trust Assets is not likely to yield sufficient additional Distributable Trust Assets or Distributable Other Debtor Residual Trust Assets to justify further pursuit, and (c) all other distributions required to be made by the Trust Administrator under the Plan and this Trust Agreement have been made, but in no event shall the Trust be dissolved later than three (3) years from the earlier of (i) the Avoidance Action Trust Transfer Date and (ii) the Other Debtor Residual Assets Transfer Date, unless the Bankruptcy Court, upon motion within the six (6) month period prior to the third (3rd) anniversary of the earlier of (i) the Avoidance Action Trust Transfer Date and (ii) the Other Debtor Residual Assets Transfer Date (or at least six (6) months prior to the end of an extension period), determines that a fixed period extension (such that, together with any prior extensions, the dissolution of the Trust shall occur no later than five (5) years from the date on which the Trust became effective, without a favorable private letter ruling from the IRS that any further extension would not adversely affect the status of the Trust as a liquidating trust for U.S. federal income tax purposes) is necessary to facilitate or complete the recovery and liquidation of the Avoidance Action Trust Assets or the Other Debtor Residual Trust Assets, as the case may be. If at any time the Trust Administrator determines, in reliance upon such professionals as the Trust Administrator may retain and with the approval of the Trust Monitor, that (x) the expense of administering the Trust so as to make a final distribution to the Trust Beneficiaries is likely to exceed the value of the Avoidance Action Trust Assets remaining in the Trust and (y) the expense of administering the Trust so as to make a final distribution to the DIP Lenders is likely to exceed the value of the Other Debtor Residual Trust Assets, if any, remaining in the Trust, the Trust Administrator may apply to the Bankruptcy Court for authority to (i) reserve any amounts necessary to dissolve the Trust, transfer the balance to the DIP Lenders and the holders of Allowed General Unsecured Claims (as defined in the GUC Trust Agreement), in accordance with the Agreed Allocation and the Settlement Agreement, or donate any balance to a charitable organization described in section 501(c)(3) of the Tax Code and exempt from U.S. federal income tax under section 501(a) of the Tax Code that is unrelated to the Debtors, the Trust, any Trust Administrator Parties or any Trust Monitor Parties, and (iii) dissolve the Trust.
- 4.2. <u>Dissolution of the Trust</u>. Notwithstanding anything to the contrary in this Trust Agreement, in no event shall the Trust Administrator unduly prolong the duration of the Trust, and the Trust Administrator shall, in the exercise of its reasonable business judgment and in the interests of all Trust Beneficiaries, at all times endeavor to terminate the Trust as soon as practicable in accordance with the purposes and provisions of this Trust Agreement and the Plan. Upon final dissolution and wind-up of the Trust, the Trust Administrator shall file a certificate of cancellation for the Trust with the Secretary of State.

4.3. Continuance of Trust for Purposes of Winding Up. After the dissolution of the Trust and solely for the purpose of liquidating and winding up its affairs, the Trust Administrator shall continue to act in such capacity until its duties hereunder have been fully performed. The Trust Administrator shall retain the books, records and files that shall have been delivered to or created by the Trust Administrator until distribution or resolution of all the Avoidance Action Trust Assets and Other Debtor Residual Trust Assets, if any. At the Trust Administrator's discretion, all of such books, records and files may be destroyed at any time following the later of (x) final distribution of the Avoidance Action Trust Assets and Other Debtor Residual Trust Assets, if any, unless such books, records and files are necessary to fulfill the Trust Administrator's obligations pursuant to Articles VI and VIII hereof and subject to any joint prosecution and common interests agreement(s) to which the Trust Administrator may be party, and (y) the date until which the Trust Administrator is required by applicable law to retain such books, records and files.

ARTICLE V DISTRIBUTIONS TO TRUST BENEFICIARIES

5.1. General.

- Until such date as the Term Loan Avoidance Action shall have been (a) completely and finally resolved in full against all defendants (including by way of settlement) and the Trust Administrator determines that all Avoidance Action Proceeds have been collected in respect thereof (such date, the "Final Recovery Date"), the Trust Administrator shall establish Distribution Dates no less frequently than once each calendar year and no more frequently than once a calendar quarter for the distribution of Distributable Trust Assets as provided in this Article V; provided that distributions need not be made in any calendar year to the extent (A) there are no Distributable Trust Assets held by the Trust or (B) the Trust Administrator, with the approval of the Trust Monitor, determines (i) that it is reasonably necessary to retain the Distributable Trust Assets to meet contingent liabilities and maintain the value of the Avoidance Action Trust Assets (such as, for example, in the event that the Trust Administrator determines that the Distributable Trust Assets are so small in amount as not to justify making a distribution, taking into account the costs that would be incurred in making the distribution, the anticipated total amount of Distributable Trust Assets expected to be available for distribution over time and the timing of the distribution or distributions thereof), or (ii) that it is necessary to retain the Distributable Trust Assets to pay reasonable incurred and anticipated expenses (including any taxes imposed on the Trust or in respect of the Avoidance Action Trust Assets) or to satisfy liabilities incurred and anticipated by the Trust in accordance with the Plan, the Confirmation Order and this Trust Agreement. Following the Final Recovery Date, distributions shall be made on a quarterly basis, as provided in this Article V.
- (b) Except as otherwise set forth herein, no distributions shall be made with respect to any portion of a Disputed General Unsecured Claim, Unresolved Term Loan Avoidance Action Claim or Unresolved Other Avoidance Action Claim unless and until such Disputed General Unsecured Claim, Unresolved Term Loan Avoidance Action Claim or Unresolved Other Avoidance Action Claim has become an Allowed General Unsecured Claim.
- (c) To the extent that a Disputed General Unsecured Claim, Unresolved Term Loan Avoidance Action Claim or Unresolved Other Avoidance Action Claim has become an

Allowed General Unsecured Claim, distributions (if any) shall be made to the holder of such Allowed General Unsecured Claim in accordance with the provisions of the Plan, the Confirmation Order and this Trust Agreement.

- (d) The Distributable Trust Assets shall be distributed
- (i) first, to the DIP Lenders, each in their relative proportion pursuant to the terms of the DIP Credit Agreement, in the amount of the DIP Lender Advances;
- (ii) second, to the CE Capital Providers, in the amount of the CE Capital Providers' Entitlement;
- (iii) third, to the LW Capital Provider, in the amount of the LW Capital Provider's Entitlement;
- (iv) fourth, to the Segregated Account in the amount of the GUC Trust Advances to be distributed to the holders of Allowed General Unsecured Claims; and
- (v) fifth, to the DIP Lenders, each in their relative proportion pursuant to the terms of the DIP Credit Agreement, and the holders of Allowed General Unsecured Claims, in accordance with the Agreed Allocation.
- (e) On any Distribution Date following the Final Recovery Date on which the Trust does not hold sufficient Distributable Trust Assets, after taking into account any amounts necessary to satisfy the current and projected future fees and expenses of the Trust (including those of any Trust Professionals) pursuant to Section 6.1(b), to satisfy all Disputed General Unsecured Claims or other Claims that became Resolved Allowed General Unsecured Claims since the next preceding Resolved Allowed General Unsecured Claims Determination Date following which there was a distribution pursuant to Section 5.3(c), the Trust Administrator shall distribute all Distributable Trust Assets that remain in the Trust to the DIP Lenders and the holders of such Resolved Allowed General Unsecured Claims pro rata by Claim amount, in accordance with the Agreed Allocation. Following such distribution, any remaining unsatisfied portion of such Resolved Allowed General Unsecured Claims, together with all remaining Disputed General Unsecured Claims and other Claims (including any Term Loan Avoidance Action Claims and any Other Avoidance Action Claims) shall be forever barred from assertion against the Trust.
- (f) Anything to the contrary herein notwithstanding but subject to <u>Section 5.1(a)</u>, the Trust Administrator shall not make a distribution of Distributable Trust Assets on any Distribution Date pursuant to <u>Sections 5.2 or 5.4</u>, if the amount to be distributed pursuant thereto does not exceed the Distribution Threshold. In such case, any Distributable Trust Assets then available for distribution shall be held by the Trust and distributed on a subsequent Distribution Date when the amount of the Distributable Trust Assets to be distributed shall exceed the Distribution Threshold; <u>provided</u> that if on the date determined by the Trust Administrator to be the final Distribution Date the Distributable Trust Assets do not exceed the Distribution Threshold, then such Distributable Trust Assets shall be disposed of as provided in the final sentence of <u>Section 2.6(b)</u> and in <u>Section 2.6(c)</u>.

- (g) For the avoidance of doubt, if the Trust fails to recover any Avoidance Action Proceeds or if the Avoidance Action Proceeds recovered by the Trust through the Final Recovery Date do not exceed the amount of DIP Lender Advances then no distributions shall be made hereunder in respect of any Allowed General Unsecured Claims.
- (h) For the avoidance of doubt, if the Avoidance Action Proceeds recovered by the Trust through the Final Recovery Date exceed the amount of DIP Lender Advances but do not exceed the amount of the Segregated Account Assets, the Segregated Account Assets shall be distributed as GUC Distributable Trust Assets pursuant to Section 5.2(a) hereof.

5.2. <u>Distribution to the DIP Lenders and to Holders of Initial Allowed General Unsecured Claims.</u>

- (a) Once (i) the DIP Lender Advances have been indefeasibly paid in full in cash, (ii) the CE Capital Providers' Entitlement has been indefeasibly paid in full in cash, (iii) the LW Capital Provider's Entitlement has been indefeasibly paid in full in cash, and (iv) the Segregated Account repaid up to the amount of the GUC Trust Advances, the Trust Administrator shall make distributions of the DIP Lender Distributable Trust Assets and the GUC Distributable Trust Assets, from time to time in accordance with Section 5.1(a). Prior to the first such distribution, the Trust Administrator shall establish a record date (the "Initial GUC Record Date") for the holders of the General Unsecured Claims that are allowed as of the Initial GUC Record Date (the "Initial Allowed General Unsecured Claims") who shall be entitled to participate in the first distribution of GUC Distributable Trust Assets, which date shall be the last day of the calendar quarter next preceding the date of such distribution (such Distribution Date, the "Initial GUC Distribution Date"). On the Initial GUC Distribution Date, the Trust Administrator shall distribute to each holder of an Initial Allowed General Unsecured Claim a distribution consisting of:
 - (1) an amount of GUC Distributable Trust Assets at the time available for distribution, in proportion to the amount of such Initial Allowed General Unsecured Claim as prescribed in subsection (d) below; and
 - (2) a number of Units as provided in Section 3.4.
- (b) During the period, if any, following the Initial GUC Distribution Date and prior to the Final Recovery Date, and as promptly as practicable following the Final Recovery Date, as and to the extent that additional GUC Distributable Trust Assets become available for distribution from time to time as a result of additional recoveries by the Trust in the Term Loan Avoidance Action, the Trust Administrator shall, from time to time in accordance with Section 5.1(a), make additional distributions to the holders of Initial Allowed General Unsecured Claims, on a Distribution Date or Dates designated by the Trust Administrator for such purpose, in the amount prescribed in subsection (d) below. Subject to the proviso in Section 5.1(a), the Trust Administrator shall make such distributions no less frequently than once each calendar year and no more frequently than once each calendar quarter.
- (c) The amount of GUC Distributable Trust Assets that the holder of an Initial Allowed General Unsecured Claim shall be entitled to receive pursuant to Section 5.2 (or, in the

case of a holder of a Resolved Allowed General Unsecured Claim being treated as if it were a holder of an Initial Allowed General Unsecured Claim in the amount of its Resolved Allowed General Unsecured Claim pursuant to Section 5.3(c) on any Distribution Date (including the Initial Distribution Date) shall be determined in accordance with the following formula:

$$D = (\underline{A}) \times G$$

Where—

- D is the distribution that the holder of an Initial Allowed General Unsecured Claim (or the holder of a Resolved Allowed General Unsecured Claim pursuant to Section 5.3(c)) will be entitled to receive;
- A is the amount of the Initial Allowed General Unsecured Claim (or the holder of a Resolved Allowed General Unsecured Claim pursuant to Section 5.3(c));
- C is the Current Total Amount as of the last day of the calendar quarter next preceding the respective Distribution Date; and
- G is the amount available for distribution determined as of the last day of the calendar quarter next preceding the respective Distribution Date, after taking account of any Holdback, or the application of such Holdback, pursuant to Section 6.1(b).

5.3. Distributions to Holders of Resolved Allowed General Unsecured Claims.

- (a) Following the Initial GUC Distribution Date, the Trust Administrator, with the approval of the Trust Monitor, shall periodically make a determination (the date of any such determination, which shall in all cases be made as of the last day of a calendar quarter, being a "Resolved Allowed General Unsecured Claims Determination Date") whether any holders of Disputed General Unsecured Claims or other Claims have become holders of Resolved Allowed General Unsecured Claims between Determination Date or, in the case of the first such determination, since the Initial GUC Record Date. During the period, if any, following the Initial GUC Distribution Date and prior to the Final Recovery Date, the Trust Administrator shall make such determination no less frequently than once each calendar year and no more frequently than once each calendar quarter, and following the Final Recovery Date such determination shall be made once each calendar quarter.
- (b) On a Distribution Date scheduled as soon as practicable following each Resolved Allowed General Unsecured Claims Determination Date, the Trust Administrator shall, subject to the proviso in Section 5.1(a), distribute to each holder of a Resolved Allowed General Unsecured Claim identified on such Resolved Allowed General Unsecured Claims Determination Date, if any:
- (i) the pro rata amount of GUC Distributable Assets that the holder of such Resolved Allowed General Unsecured Claim would have received had such Resolved Allowed General Unsecured Claim been an Initial Allowed General Unsecured Claim, including the aggregate amount of Excess GUC Distributable Trust Assets that the holder would have

received had it been the holder of Units referred to in clause (ii) below on each Excess GUC Distributable Trust Assets Determination Date occurring on or prior to the date of such distribution; provided that a holder of a Resolved Allowed General Unsecured Claim shall not receive pursuant to this clause (i) an amount of Excess GUC Distributable Assets distributed in respect of any prior Excess GUC Distributable Trust Assets Determination Date to the extent that it will be receiving such Excess GUC Distributable Assets as a distribution on the Units to be received by such holder pursuant to clause (ii) below; and

(ii) a number of Units as provided in Section 3.4.

- (c) Once a holder of a Resolved Allowed General Unsecured Claim has received the distribution prescribed in the preceding subsection (b), such holder shall be treated as if it were a holder of an Initial Allowed General Unsecured Claim in the amount of its Resolved Allowed General Unsecured Claim on all subsequent Distribution Dates established for purposes of Section 5.2(c).
- (d) For the avoidance of doubt, it is intended that the distributions to be made to holders of Resolved Allowed General Unsecured Claims in accordance with this <u>Section 5.3</u> shall provide such holders, as nearly as possible, with the exact same amount of distributions as if such holders had been holders of Initial Allowed General Unsecured Claims.

5.4. Distribution of Excess Distributable Trust Assets.

- (a) Following the Initial GUC Distribution Date, the Trust Administrator, with the approval of the Trust Monitor, shall periodically make a determination (the date of any such determination, which shall in all cases be as of the last day of a calendar quarter, being an "Excess Distributable Trust Assets Determination Date") of the Excess Distributable Trust Assets as of such date, taking into account the extent to which Disputed General Unsecured Claims are disallowed or the Term Loan Avoidance Action or other Avoidance Actions are resolved in favor of the defendants therein. During the period, if any, following the Initial GUC Distribution Date and prior to the Final Recovery Date, the Trust Administrator shall make such determination no less frequently than once each calendar year and no more frequently than once each calendar quarter, and following the Final Recovery Date such determination shall be made once each calendar quarter.
- (b) On a Distribution Date scheduled as soon as practicable following each Excess Distributable Trust Assets Determination Date, the Trust Administrator shall, subject to the proviso of Section 5.1(a), distribute the Excess Distributable Trust Assets, in each case determined as of the respective Excess Distributable Trust Assets Determination Date, to the DIP Lenders and the holders of Units outstanding on such Excess Distributable Trust Assets Determination Date (including Units distributed or to be distributed to holders of Resolved Allowed General Unsecured Claims pursuant to Section 5.3(b)(ii) on such Distribution Date) in accordance with the Agreed Allocation.
- (c) Any Excess Distributable Trust Assets allocable to the holders of Units outstanding on such Excess Distributable Trust Assets Determination Date (the "Excess GUC Distributable Trust Assets"), shall be distributed pro rata according to the following formula:

$$DU = \left(\begin{array}{c} \underline{U_H} \end{array} \right) x \left(\sum G - H \right) x \left[\begin{array}{c} \underline{T_U} - \underline{T_U} \\ C_U \end{array} \right]$$

Where—

- D_U is the distribution of Excess GUC Distributable Trust Assets that a holder of Units will be entitled to receive;
- U_H is the number of Units held by the holder;
- U_O is the total number of Units outstanding (including Units distributed to holders of Resolved Allowed General Unsecured Claims pursuant to Section 5.3(b)(ii) on such Distribution Date);
- ∑G is the sum of the amounts that are or were available for distribution to holders of Initial Allowed General Unsecured Claims (and holders of Resolved Allowed General Unsecured Claims pursuant to Section 5.3(c)) on the relevant Distribution Date and each prior Distribution Date;
- T_U is the Total Allowed Amount as of the Excess GUC Distributable Trust Assets Determination Date;
- C_U is the Current Total Amount as of the Excess GUC Distributable Trust Assets Determination Date;
- L is the aggregate amount of all (i) Disputed General Unsecured Claims disallowed since the next preceding Excess GUC Distributable Trust Assets Determination Date (or, in the case of the first Excess GUC Distributable Trust Assets Determination Date, since the Initial GUC Record Date), (ii) Unresolved Term Loan Avoidance Action Claims to the extent resolved (including by way of settlement) in favor of the respective defendants since the next preceding Excess GUC Distributable Trust Assets Determination Date (or, in the case of the first Excess GUC Distributable Trust Assets Determination Date, since the Initial GUC Record Date); and (iii) all Unresolved Other Avoidance Action Claims to the extent resolved (including by way of settlement) in favor of the respective defendants since the next preceding Excess GUC Distributable Trust Assets Determination Date (or, in the case of the first Excess GUC Distributable Trust Assets Determination Date, since the Initial GUC Record Date); and
- H is the amount, if any, of any holdback pursuant to <u>Section 6.1</u> that was not otherwise deducted from the amounts available for distribution on a Distribution Date.
- (d) Notwithstanding the foregoing, if the Trust Administrator becomes aware of previously unknown potential Allowed General Unsecured Claims, the Trust Administrator may, with the approval of the Trust Monitor, withhold distribution of Excess GUC Distributable Trust Assets to the holders of Units in an amount that the Trust Administrator, with the approval of the Trust Monitor, estimates to be the maximum amount reasonably allowable in respect of such previously unknown claims.

- 5.5. <u>Retention of Avoidance Action Trust Assets</u>. Notwithstanding anything in this Trust Agreement to the contrary, the Trust Administrator shall at all times, to the extent practicable, retain
- (a) sufficient GUC Distributable Trust Assets as the Trust Administrator shall determine, with the approval of the Trust Monitor, as would be distributable (I) to all holders of Disputed General Unsecured Claims at the time outstanding as if all Disputed General Unsecured Claims were allowed at the Maximum Amount, but only until such Disputed General Unsecured Claims are resolved, (II) to the holders of all Resolved Allowed General Unsecured Claims at the time outstanding, to the extent not previously distributed, (III) in respect of any Unresolved Term Loan Avoidance Action Claims at the Maximum Amount thereof but only until the Term Loan Avoidance Action is dismissed by Final Order or such claims become Resolved Allowed General Unsecured Claims, and (IV) in respect of any Unresolved Other Avoidance Action Claims at the Maximum Amount thereof but only until such claims become Resolved Allowed General Unsecured Claims or the related other Avoidance Actions are dismissed by Final Order; and
- (b) sufficient Avoidance Action Trust Administrative Cash and Supplemental Avoidance Action Trust Cash as the Trust Administrator shall determine, with the approval of the Trust Monitor and subject to the Budget, is necessary (x) to pay the reasonable incurred or anticipated fees and expenses of the Trust (including any taxes imposed on the Trust or in respect of the Avoidance Action Trust Assets) and (y) to satisfy other liabilities incurred or anticipated by the Trust in accordance with the Plan, the Confirmation Order and this Trust Agreement.
- 5.6. Minimum Distributions. Notwithstanding anything to the contrary contained herein, no Cash payment in an amount less than \$25 shall be made by the Trust Administrator to any holder of an Allowed General Unsecured Claim or Unit under any circumstance; *provided* that the Trust Administrator shall carry the entitlement of such holder of an Allowed General Unsecured Claim or Unit to such amount on its books and records, shall aggregate such amount with any subsequent amount to which such holder shall become entitled and shall make payment of such amount to such holder at such time as the amounts due such holder in the aggregate shall equal \$25 or more; *provided further* that if any such amount shall be owing to a holder of an Allowed General Unsecured Claim or Unit as of the date determined by the Trust Administrator to be the final Distribution Date, such amount shall be disposed of as provided in the final sentence of Section 2.6(c).
- 5.7. Distributions Not in Compliance with this Article. Subject to Section 5.3(d), in the event that the Trust Administrator determines in good faith that it is necessary or desirable in order to carry out the intent and purposes of the Plan, the Confirmation Order and this Trust Agreement to receive any assets or make any distribution in a manner that is not in technical compliance with this Trust Agreement, the Trust Administrator shall be permitted to receive assets or make, or cause to be made, distributions in such manner, but only with the approval of the Trust Monitor; provided, however, that no such distribution shall result in any holder of an Allowed General Unsecured Claim receiving a distribution in excess of the distribution that such holder would have received had such claim been an Initial Allowed General Unsecured Claim or shall discriminate among the holders of Units. Except as aforesaid, no payment or distribution of Avoidance Action Trust Assets shall be made to, or on behalf of, a Trust Beneficiary or any other person except in strict accordance with the terms of this Trust Agreement, the Plan, and the

Confirmation Order, unless such payment or distribution shall have been approved by the Bankruptcy Court.

5.8. <u>No Accounting</u>. Except as otherwise provided in the Plan, the Confirmation Order or this Trust Agreement, nothing shall require the Trust Administrator to file any accounting or seek approval of any court with respect to the administration of the Trust or as a condition for making any payment or distribution out of the Avoidance Action Trust Assets.

ARTICLE VA DISTRIBUTIONS TO DIP LENDERS

If any Other Debtor Residual Accepted Assets shall be transferred to the Trust, the Trust Administrator shall make distributions to the DIP Lenders of Distributable Other Debtor Residual Assets, if any, from time to time (but no less frequently than once each calendar year), pro rata as their interests appear, as the Trust Administrator shall determine with the approval of the Trust Monitor or as the Trust Administrator shall be directed by a majority in interest of the DIP Lenders; provided that distributions need not be made in any calendar year to the extent (A) there are no Distributable Other Debtor Residual Assets held by the Trust or (B) the Trust Administrator, with the approval of the Trust Monitor, determines that it is necessary to retain the Distributable Other Debtor Residual Assets to (i) meet any contingent liabilities of the Trust or maintain the value of the Other Debtor Residual Trust Assets (such as for example, in the event that the Trust Administrator determines that the Distributable Other Debtor Residual Assets are so small in amount as not to justify making a distribution, taking into account the costs that would be incurred in making the distribution, the anticipated total amount of Distributable Other Debtor Residual Assets expected to be available for distribution over time and the timing of the distribution or distributions thereof), or (ii) pay reasonable incurred and/or anticipated expenses of the Trust (including any taxes imposed on the Trust or in respect of the Other Debtor Residual Trust Assets) or to satisfy liabilities incurred and/or anticipated by the Trust in accordance with the Plan, the Confirmation Order and this Trust Agreement.

ARTICLE VI ADMINISTRATION OF THE TRUST

- 6.1. <u>Payment of Costs, Expenses and Liabilities (other than in respect of the Other</u> Debtor Residual Accepted Assets).
- (a) Subject to the Budget, the Trust Administrator shall use the Avoidance Action Trust Administrative Cash:
- (i) to pay reasonable costs and expenses of the Trust that are incurred in connection with the administration thereof (including any taxes imposed on the Trust, actual reasonable fees and out-of-pocket expenses incurred by Trust Professionals retained by the Trust Administrator in connection with the administration of the Avoidance Action Trust Assets and preservation of books and records);
- (ii) to satisfy other obligations or other liabilities incurred or assumed by the Trust (or to which the Avoidance Action Trust Assets are otherwise subject) in accordance with the Plan, the Confirmation Order, or this Trust Agreement, including fees and expenses

incurred and in connection with, the prosecution and resolution of the Term Loan Avoidance Action the protection, preservation and distribution of the Avoidance Action Trust Assets; and

- (iii) to satisfy any other obligations of the Trust expressly set forth in the Plan, the Confirmation Order or this Trust Agreement to be satisfied out of the Avoidance Action Trust Administrative Cash.
- (b) (i) If, at any time, the Trust Administrator determines that the Avoidance Action Trust Administrative Cash is not reasonably likely to be adequate to satisfy the current and projected future taxes, fees, costs and expenses (including, without limitation any Avoidance Action Trust SEC Reporting Costs) of the Trust (other than in respect of the Other Debtor Residual Trust Assets), the Trust Administrator may, with the approval of the Trust Monitor, reserve an amount, or increase the amount previously reserved, of Distributable Trust Assets to satisfy such taxes, fees, costs and expenses (the "Holdback"). If at any time, the Trust Administrator determines that the Holdback is materially greater than the amount of the current and projected future taxes, fees, costs and expenses as aforesaid, the Trust Administrator shall, with the approval of the Trust Monitor, release from the Holdback the amount of such excess.
- (ii) To the extent necessary to satisfy the taxes, fees, costs and expenses on account of which the Holdback may be reserved, the Trust Administrator may, in consultation with the Trust Monitor, and upon approval by the Bankruptcy Court in accordance with the provisions of Section 6.1(b)(iii), apply all or a portion of the Holdback to the satisfaction of such taxes, fees, costs and expenses.
- (iii) The application of the Trust Administrator seeking Bankruptcy Court approval to utilize Distributable Trust Assets shall include the position of the Trust Monitor in respect thereof. The Trust Administrator shall provide at least twenty (20) days' notice to the Trust Monitor, the holders of Units and the holders of Disputed General Unsecured Claims prior to a hearing on a motion to use any Distributable Trust Assets.
- (c) If, at any time, the Trust Administrator determines that the Avoidance Action Trust Administrative Cash, together with the Holdback (if Distributable Trust Assets are, at such time, available for the purposes of creating a Holdback), is not reasonably likely to be adequate to satisfy the current and projected future taxes, fees, costs and expenses (including, without limitation any Avoidance Action Trust SEC Reporting Costs and the fees and expenses of Trust Professionals) of the Trust (other than in respect of the Other Debtor Residual Trust Assets), the Trust Administrator may utilize the GUC Trust Supplemental Cash, without the need to seek or obtain approval of the Bankruptcy Court, to satisfy such fees and expenses.
- (d) (i) If, at any time, the Trust Administrator determines that the Avoidance Action Trust Administrative Cash, together with the Holdback (if Distributable Trust Assets are, at such time, available for the purposes of creating a Holdback) and the GUC Trust Supplemental Cash, if any, is not reasonably likely to be adequate to satisfy the current and projected future fees, costs and expenses (including, without limitation any Avoidance Action Trust SEC Reporting Costs, the fees and expenses of Trust Professionals, and any tax liabilities), the Trust Administrator may, in consultation with the Trust Monitor, and upon approval by the Bankruptcy Court in accordance with the provisions of Section 6.1(d)(ii), sell or grant liens on the

Term Loan Avoidance Action or any other property of the Trust (other than the Other Debtor Residual Trust Assets) and apply all or a portion of the proceeds of such sale or grant to the satisfaction of such fees, costs and expenses; provided that, except for the Trust's incurrence of expenses in the ordinary course of the Trust's business without any lien or right of setoff attached thereto, the Trust may not incur additional indebtedness, whether unsecured or secured by any interest (including a security interest) in the Term Loan Avoidance Action or Avoidance Action Proceeds without (x) the prior written consent of the DIP Lenders, (y) the prior written consent of the CE Capital Providers, if such incurrence would (1) create a lien senior to or pari passu with the lien securing the CE Capital Providers' Entitlement, or (2) except as set forth in Section 2.7 of the CE Subordination Agreement, result in the subordination of any portion of the payment of the CE Capital Providers' Entitlement pursuant to Section 5.1(d) of this Trust Agreement, and (z) the prior written consent of the LW Capital Provider, if such incurrence would (1) create a lien senior to or pari passu with the lien securing the LW Capital Provider's Entitlement, or (2) except as set forth in Section 2.7 of the LW Subordination Agreement, result in the subordination of any portion of the payment of the LW Capital Provider's Entitlement pursuant to Section 5.1(d) of this Trust Agreement. In the event that the Trust seeks other litigation funding, (A) the DIP Lenders shall be provided a right of first refusal to provide such funding, which right of first refusal must be exercised within thirty (30) days after being presented in writing the opportunity to provide such funding, and (B) such funding (even if provided by the DIP Lenders) shall be junior and subordinate to the Litigation Cost Advance, the amounts owed to the DIP Lenders on account of prior funding of the Trust, and the Supplemental Capital, and shall be subject to a form of subordination that is acceptable to the DIP Lenders, the CE Capital Providers and the LW Capital Provider in all respects, with such consent not to be unreasonably withheld. Upon any such sale or grant of liens pursuant to this Section 6.1(d), the resulting proceeds shall be designated as Other Supplemental Cash. Any income earned on the Other Supplemental Cash, net of taxes paid thereon, shall be Other Supplemental Cash.

- (ii) The application of the Trust Administrator seeking Bankruptcy Court approval to sell or grant liens pursuant to Section 6.1(d)(i) hereof shall include the position of the Trust Monitor in respect thereof. The Trust Administrator shall provide at least twenty (20) days' notice to the Trust Monitor, the DIP Lenders, the CE Capital Providers, the LW Capital Provider, the holders of Units and the holders of Disputed General Unsecured Claims prior to a hearing on a motion to sell or grant liens pursuant to Section 6.1(d)(i) hereof.
- (e) Notwithstanding that as a result of the utilization of Distributable Trust Assets pursuant to Section 6.1(b) the amount of GUC Distributable Trust Assets shall be less than the assets required to satisfy Claims in the amount of the Current Total Amount then outstanding, the Trust Administrator shall continue to satisfy Disputed General Unsecured Claims, any Unresolved Term Loan Avoidance Action Claims and any Unresolved Other Avoidance Action Claims that become Allowed General Unsecured Claims in the order they are resolved as otherwise provided in this Trust Agreement.
- 6.2. <u>Payment of Costs, Expenses and Liabilities in respect of the Other Debtor Residual Accepted Assets.</u>
- (a) The Trust Administrator shall not be required to undertake any activity in respect of the Other Debtor Residual Accepted Assets, including for purposes of realizing upon

such assets in order to make distributions of Distributable Other Debtor Residual Trust Assets to the DIP Lenders, unless there shall be available to the Trust Administrator Other Debtor Residual Trust Administrative Cash sufficient for such purposes.

- (b) If sufficient Other Debtor Residual Trust Administrative Cash shall be available to the Trust Administrator, then the Trust Administrator shall, as approved by the Trust Monitor or as directed by a majority in interest of the DIP Lenders:
- (i) pay reasonable costs and expenses of the Trust that are incurred in connection with the Other Debtor Residual Accepted Assets (including any taxes imposed on the Trust, actual reasonable fees and out-of-pocket expenses incurred by Trust Professionals retained by the Trust Administrator in connection with the administration of the Other Debtor Residual Accepted Assets and preservation of books and records);
- (ii) satisfy other obligations or other liabilities incurred or assumed by the Trust in respect of the Other Debtor Residual Accepted Assets (or to which the Other Debtor Residual Accepted Assets are otherwise subject) in accordance with the Plan, the Confirmation Order, or this Trust Agreement, including fees and expenses incurred and in connection with, the prosecution and resolution of any action to realize upon the Other Debtor Residual Accepted Assets and the protection, preservation and distribution of the Other Debtor Residual Accepted Assets; and
- (iii) satisfy any other obligations of the Trust expressly set forth in the Plan, the Confirmation Order or this Trust Agreement to be satisfied out of the Other Debtor Residual Trust Administrative Cash.

6.3. Budget.

(a) The Trust Administrator shall prepare and submit to the Trust Monitor, the DIP Lenders, the CE Capital Providers and the LW Capital Provider for approval a reasonably detailed annual plan and budget (the "Budget") at least thirty (30) days prior to the commencement of each calendar year. Such annual plan and Budget shall set forth (on a quarterly basis) in reasonable detail: (A) the Trust Administrator's anticipated actions to administer the Avoidance Action Trust Assets; and (B) the anticipated fees and expenses, including professional fees, associated with the administration of the Trust, a separate amount representing the anticipated fees and expenses of the Trust Monitor and detail as to how the Trust will budget and spend the Avoidance Action Trust Administrative Cash. Such Budget shall be updated and submitted to the Trust Monitor, the DIP Lenders, the CE Capital Providers and the LW Capital Provider for review on a quarterly basis, and each such quarterly update shall reflect the variances (with explanations) between (x) the Budget, (y) any updated Budget, and (z) the actual results for the same period. If the Trust identifies a need for or the incurrence of actual expenditures that materially differ from the relevant Budget, the Trust will notify the DIP Lenders, the CE Capital Providers and the LW Capital Provider as promptly as practicable, and no later than fifteen (15) days after identifying the issue. For the avoidance of doubt, the DIP Lenders, the CE Capital Providers and the LW Capital Provider may object in the Bankruptcy Court with respect to any quarterly update that materially changes the Budget and the Bankruptcy Court shall resolve such dispute. All actions by the Trust Administrator shall be consistent with the Budget (as updated). The Trust

Administrator may obtain any required approval of the Budget on reasonable negative notice (which shall be not less than 15 days after receipt of the Budget) and approval of the Budget shall not be unreasonably withheld. In the event of any dispute concerning the Budget (or the taking of actions consistent with the Budget), the Trust Administrator or the Trust Monitor may petition the Bankruptcy Court to resolve such dispute.

- (b) The Trust Administrator, with the approval of the Trust Monitor and the DIP Lenders may agree on a budget for activities in respect of the Other Debtor Residual Accepted Assets.
- (c) Notwithstanding any other provision of this Trust Agreement, the approval of the DIP Lenders shall not be required for any use of the Avoidance Action Trust SEC Reporting Cash, the GUC Trust Supplemental Cash or the Other Supplemental Cash.

6.4. [Intentionally omitted.]

- 6.5. <u>Compliance with Laws</u>. Any and all distributions of Avoidance Action Trust Assets shall be in compliance with applicable laws, including applicable federal and state tax and securities laws.
- 6.6. <u>Fiscal Year</u>. Except for the first and last years of the Trust, the fiscal year of the Trust shall be the calendar year. For the first and last years of the Trust, the fiscal year of the Trust shall be such portion of the calendar year that the Trust is in existence.

6.7. Books and Records.

- (a) The Trust Administrator shall maintain and preserve the Debtors' books, records and files that shall have been delivered to or created by the Trust Administrator.
- (b) The Trust Administrator shall maintain books and records relating to the assets, liabilities, income and expense of the Trust, all distributions made by the Trust and the payment of fees and expenses of, and satisfaction of claims against or assumed by, the Trust and the Trust Administrator, in such detail and for such period of time as may be necessary to enable it to make full and proper reports in respect thereof in accordance with the provisions of this Trust Agreement and otherwise to comply with applicable provisions of law, including tax law.
- (c) The Trust Administrator shall maintain, or cause to be maintained, a register of holders of Units, from time to time outstanding, to the extent any Units are issued hereunder, in customary form.
- 6.8. <u>Cash Payments</u>. All distributions of Distributable Trust Cash required to be made by the Trust Administrator may be made in Cash denominated in U.S. dollars by checks drawn on a United States domestic bank selected by the Trust Administrator or, at the option of the Trust Administrator, by wire transfer from a United States domestic bank selected by the Trust Administrator or as otherwise required or provided in applicable agreements; <u>provided</u>, <u>however</u>, that cash payments to foreign persons may be made, at the option of the Trust Administrator, in such funds as and by such means as are necessary or customary in a particular foreign jurisdiction.

6.9. <u>Insurance</u>. The Trust shall maintain customary insurance coverage for the protection of the Trust Administrator Parties and the Trust Monitor Parties and any such other persons serving as administrators and overseers of the Trust, on and after the Avoidance Action Trust Transfer Date, in all cases in accordance with the Budget. The Trust Administrator may also obtain such insurance coverage as it deems necessary and appropriate with respect to real and personal property which may become Avoidance Action Trust Assets, if any, in accordance with such Budget. To the extent that there is any incremental cost for customary insurance coverage covering the activities of the Trust Administrator Parties and the Trust Monitor Parties in respect of the Other Debtor Residual Accepted Assets, the Trust Administrator and the Trust Monitor shall not be required to undertake any such activities unless there is available sufficient Other Debtor Residual Trust Administrative Cash to fund such incremental cost.

ARTICLE VII TAX MATTERS

7.1. Tax Treatment.

- (a) For all U.S. federal and applicable state and local income tax purposes, all parties (including the Debtors, the Trust Administrator, the holders of the DIP Credit Agreement Claims, and the holders of Allowed General Unsecured Claims) shall treat the Trust and the transfer of the Avoidance Action Trust Assets and the Other Debtor Residual Assets to the Trust in a manner consistent with the remainder of this Section 7.1.
- (b) If no Other Debtor Residual Assets are transferred to the Trust upon the dissolution of MLC and the DIP Lender Distributable Trust Assets have not been determined (either as a percentage or as a fixed amount of Distributable Trust Assets or on some other basis) on or prior to the Avoidance Action Trust Transfer Date, then (subject to clause (c) below) the Trust Administrator shall treat the Trust as (A) a "disputed ownership fund" governed by Treasury Regulation section 1.468B-9 (including, if required, timely so electing), (B) if permitted under applicable law and at the election of the Trust Administrator, as a "complex trust," or (C) as otherwise permitted pursuant to a private letter ruling from the IRS.
- (c) If Other Debtor Residual Assets are transferred to the Trust upon the dissolution of MLC or the DIP Lender Distributable Trust Assets have been determined (either as a percentage or as a fixed amount of Distributable Trust Assets or on some other basis) on or prior to the Avoidance Action Trust Transfer Date, or otherwise upon determination of the DIP Lender Distributable Trust Assets (either as a percentage or as a fixed amount of Distributable Trust Assets or on some other basis) after the Avoidance Action Trust Transfer Date, the Trust (other than the Avoidance Action Trust Claims Reserve) shall be treated as a liquidating trust that is treated as a grantor trust and the Avoidance Action Trust Assets (upon the determination of the DIP Lender Distributable Trust Assets) and Other Debtor Residual Trust Assets (upon the transfer of Other Debtor Residual Assets to the Trust) shall be treated as (i) being transferred directly to the Trust Beneficiaries; provided, however, that to the extent Avoidance Action Trust Assets are allocable to Disputed General Unsecured Claims, Unresolved Term Loan Avoidance Action Claims or Unresolved Other Avoidance Action Claims, such Avoidance Action Trust Assets shall be treated as being transferred to the Avoidance Action Trust Claims Reserve, followed by (ii) the transfer by such Trust Beneficiaries of the Avoidance Action Trust Assets (other than the Avoidance

Action Trust Assets allocable to the Avoidance Action Trust Claims Reserve) and the Other Debtor Residual Trust Assets, as applicable, to the Trust in exchange for beneficial interests in the Trust. Accordingly, Trust Beneficiaries receiving beneficial interests in the Trust shall be treated as the grantors and owners of their respective share of the Avoidance Action Trust Assets (other than any Avoidance Action Trust Assets allocable to the Avoidance Action Trust Claims Reserve) and Other Debtor Residual Trust Assets, as applicable.

- (d) Any determination made pursuant to this <u>Section 7.1</u> shall be conclusive and binding on all parties (including the Debtors, the Trust Administrator, the holders of the DIP Credit Agreement Claims, and the holders of Allowed General Unsecured Claims) for U.S. federal, and (to the extent permitted by applicable law) state and local, income tax purposes. Accordingly, to the extent permitted by applicable law, all parties shall report consistently with the U.S. federal income tax treatment of the Trust by the Trust Administrator for state and local income tax purposes.
- 7.2. <u>Valuation of Assets</u>. As soon as practicable after the Avoidance Action Trust Transfer Date, the Trust Administrator shall make a good-faith valuation of the Avoidance Action Trust Assets and Other Debtor Residual Trust Assets, and such valuation shall be made available from time to time, to the extent relevant, and shall be used consistently by all parties (including the Debtors, the Trust Administrator, the holders of the DIP Credit Agreement Claims, and the holders of Allowed General Unsecured Claims) for all U.S. federal and applicable state and local income tax purposes.
- Payment of Taxes. The Trust Administrator shall be responsible for payment, out 7.3. of the Avoidance Action Trust Assets, of any taxes imposed on the Trust (other than in respect of the Other Debtor Residual Assets) or the Avoidance Action Trust Assets, including the Avoidance Action Trust Claims Reserve. The Trust Administrator shall be responsible for payment, out of the Other Debtor Residual Assets of any taxes imposed on the Trust in respect of the Other Debtor Residual Assets or on the Other Debtor Residual Assets. In the event, and to the extent, any Cash retained on account of Disputed General Unsecured Claims, Unresolved Term Loan Avoidance Action Claims or Unresolved Other Avoidance Action Claims in the Avoidance Action Trust Claims Reserve is insufficient to pay the portion of any such taxes attributable to the taxable income arising from the assets allocable to, or retained on account of, Disputed General Unsecured Claims, Unresolved Term Loan Avoidance Action Claims or Unresolved Other Avoidance Action Claims, such taxes shall be (i) reimbursed from any subsequent Cash amounts retained on account of Disputed General Unsecured Claims, Unresolved Term Loan Avoidance Action Claims or Unresolved Other Avoidance Action Claims, or (ii) to the extent such Disputed General Unsecured Claims, Unresolved Term Loan Avoidance Action Claims or Unresolved Other Avoidance Action Claims subsequently have been resolved, deducted from any amounts otherwise distributable by the Trust Administrator as a result of the resolution of such Disputed General Unsecured Claims, Unresolved Term Loan Avoidance Action Claims or Unresolved Other Avoidance Action Claims.

7.4. <u>Tax Reporting</u>.

(a) The Trust Administrator shall file (or cause to be filed) Tax Returns for the Trust treating the Trust (except the Avoidance Action Trust Claims Reserve or as otherwise provided in Section 7.1(b) above) as a grantor trust pursuant to Treasury Regulation section 1.671-

- 4(a) and in accordance with the applicable provisions of this Section 7.4. The Trust Administrator also shall annually send to each Trust Beneficiary a separate statement setting forth such Trust Beneficiary's share of items of income, gain, loss, deduction, or credit of the Trust (including, for the avoidance of doubt, earnings on the Avoidance Action Trust Administrative Cash, the Avoidance Action Trust SEC Reporting Cash, the GUC Trust Supplemental Cash and the Other Supplemental Cash) and shall instruct all Trust Beneficiaries to report such items on their respective U.S. federal income Tax Returns or to forward the appropriate information to their respective beneficial holders with instructions to report such items on their U.S. federal income Tax Returns. The Trust Administrator also shall file (or cause to be filed) any other statements, returns, or disclosures relating to the Trust that are required by any governmental unit.
- Allocations of the Trust's taxable income among the Trust Beneficiaries shall be determined by reference to the manner in which an amount of Cash equal to such taxable income would be distributed (without regard to any restrictions on distributions described herein) if, immediately prior to such deemed distribution, the Trust had distributed all of its other assets (valued at their tax book value and other than assets attributable to the Avoidance Action Trust Claims Reserve) to the Trust Beneficiaries, in each case up to the tax book value of the assets treated as contributed by such Trust Beneficiaries, adjusted for prior taxable income and loss and taking into account all prior and concurrent distributions from the Trust. Similarly, taxable loss of the Trust shall be allocated by reference to the manner in which an economic loss would be borne immediately after a liquidating distribution of the remaining Avoidance Action Trust Assets and Other Debtor Residual Trust Assets. The tax book value of the Avoidance Action Trust Assets and Other Debtor Residual Trust Assets for this purpose shall equal their fair market value on the Avoidance Action Trust Transfer Date and Other Debtor Residual Trust Assets Transfer Date, as applicable, adjusted in accordance with tax accounting principles prescribed by the Tax Code, applicable Treasury Regulations, and other applicable administrative and judicial authorities and pronouncements.
- (c) The Trust Administrator shall (x) treat the Avoidance Action Trust Claims Reserve for U.S. federal income tax purposes as either (i) a "disputed ownership fund" governed by Treasury Regulation section 1.468B-9 by timely so electing or (ii) a "complex trust," provided, however, that if the Trust is treated as a "disputed ownership fund" or as a "complex trust" pursuant to Section 7.1(b) above, then the Avoidance Action Trust Claims Reserve shall be treated in the same manner, and (y) to the extent permitted by applicable law, report consistently with the foregoing for state and local income tax purposes. Any determination made pursuant to this Section 7.4 shall be conclusive and binding on all parties (including the Debtors, the Trust Administrator, the holders of the DIP Credit Agreement Claims, and the holders of Allowed General Unsecured Claims) for U.S. federal, state, and local income tax purposes.
- 7.5. <u>Tax Withholdings</u>. The Trust Administrator shall withhold and pay to the appropriate taxing authority all amounts required to be withheld pursuant to the Tax Code, Treasury Regulations or other applicable requirements, including any provision of any foreign, state or local tax law, with respect to any payment or distribution to the Trust Beneficiaries. All such amounts withheld, and paid to the appropriate taxing authority, shall be treated as amounts distributed to such Trust Beneficiaries for all purposes of this Trust Agreement. The Trust Administrator shall be authorized to collect such tax information from the Trust Beneficiaries (including social security numbers or other tax identification numbers) as it in its sole discretion

deems necessary to effectuate the Plan, the Confirmation Order and this Trust Agreement, or to comply with any applicable withholding or reporting requirement. The Trust Administrator may refuse to make a distribution to any Trust Beneficiary that fails to furnish such information in a timely fashion, until such information is furnished; *provided*, *however*, that upon a Trust Beneficiary furnishing such information, the Trust Administrator shall make such distribution to which such Trust Beneficiary is entitled, without interest.

7.6. Expedited Determination of Taxes. The Trust Administrator may request an expedited determination of taxes of the Trust, including the Avoidance Action Trust Claims Reserve, under Section 505(b) of the Bankruptcy Code for any or all Tax Returns filed for, or on behalf of, the Trust for any or all taxable periods (or part thereof) through the dissolution of the Trust.

7.7. [Intentionally omitted.]

- 7.8. <u>Delivery of Statement of Transfers</u>. If the Trust Administrator elects to treat (i) the Trust, pursuant to and to the extent provided in <u>Section 7.1</u> above and/or (ii) the Avoidance Action Trust Claims Reserve as a disputed ownership fund within the meaning of Treasury Regulation section 1.468B-9, then following the Avoidance Action Trust Transfer Date (but in no event later than February 15th of the calendar year following the Avoidance Action Trust Transfer Date), MLC shall provide a "§ 1.468B-9 Statement" to the Trust Administrator in accordance with Treasury Regulation section 1.468B-9(g).
- 7.9. <u>Allocation of Distributions Between Principal and Interest</u>. All deemed distributions (including deemed transfers pursuant to <u>Section 7.1(b)(i)</u>) in connection with the allowance of any Allowed General Unsecured Claim shall be allocated first to the principal amount of such Allowed General Unsecured Claim, as determined for federal income tax purposes, and thereafter, to the remaining portion of such Allowed General Unsecured Claim, if any.

ARTICLE VIII POWERS OF AND LIMITATIONS ON THE TRUST ADMINISTRATOR

8.1. <u>Powers of the Trust Administrator</u>.

- (I) Other Than in Respect of the Other Debtor Residual Accepted Assets
- (a) Pursuant to the terms of the Plan and the Confirmation Order, the Trust Administrator shall have various powers, duties and responsibilities concerning the prosecution of and resolution of the Term Loan Avoidance Action, maximizing the property of the Trust, the disposition of the Avoidance Action Trust Assets and the administration of the Trust. In addition, the Trust Administrator shall coordinate with the GUC Trust Administrator to maximize efficiency in distributions to general unsecured creditors in any situation where such coordination would be beneficial.
- (b) The Trust Administrator shall have only such rights, powers and privileges expressly set forth in the Plan, the Confirmation Order or this Trust Agreement and as otherwise provided by applicable law. Subject to the Plan, the Confirmation Order and other provisions herein, including the provisions relating to approvals of the Trust Monitor, the Trust Administrator

shall be expressly authorized to undertake the following actions, in the Trust Administrator's good faith judgment, in the best interests of the Trust Beneficiaries and in furtherance of the purpose of the Trust:

- (i) hold and manage the Avoidance Action Trust Assets;
- (ii) hold legal title to any and all rights of the Trust Beneficiaries in, to or arising from the Avoidance Action Trust Assets, for the benefit of the Trust Beneficiaries that are entitled to distributions therefrom under the Plan, whether, in the case of GUC Beneficiaries, their General Unsecured Claims are Allowed on or after the Avoidance Action Trust Transfer Date;
- (iii) prosecute and, if appropriate, sell, grant liens upon (subject to Section 6.1(d) hereof), settle and resolve, abandon and/or dismiss the Term Loan Avoidance Action;
- (iv) execute all agreements, instruments and other documents (including, without limitation, any loan agreements or sale agreements for the purposes of funding the fees and expenses of the Trust), and effect all other actions necessary or appropriate to dispose of the Avoidance Action Trust Assets;
- (v) monitor and enforce the implementation of the Plan insofar as relating to this Trust Agreement, the Avoidance Action Trust Assets or the Trust;
- (vi) calculate and implement distributions of the GUC Distributable Trust Assets obtained through the exercise of its power and authority as contemplated by the Plan, the Confirmation Order and this Trust Agreement and in accordance with the interests of the holders of Allowed General Unsecured Claims;
- (vii) retain, pay, oversee and direct the services of, and terminate Trust Professionals in accordance with <u>Section 8.3</u> hereof to carry out its duties and obligations hereunder, in all cases in accordance with the Budget;
- (viii) pay the reasonable fees and expenses of the Trust Administrator and Trust Monitor, in all cases in accordance with the Budget;
- (ix) incur and pay all reasonable expenses, satisfy ordinary course liabilities and make all other payments reasonable and necessary to administer and dispose of the Avoidance Action Trust Assets, in all cases in accordance with the Budget;
- (x) invest monies received by the Trust, the Trust Administrator or otherwise held by the Trust or the Trust Administrator in accordance with <u>Section 8.4</u> hereof;
- (xi) protect and enforce the rights to the Avoidance Action Trust Assets vested in the Trust Administrator by this Trust Agreement by any method deemed reasonably appropriate, including by judicial proceedings or pursuant to any applicable bankruptcy, insolvency, moratorium or similar law and general principles of equity;

- (xii) vote any claim or interest held by the Trust in a case under the Bankruptcy Code and receive any distribution therefrom for the benefit of the Trust;
- (xiii) make all necessary filings in accordance with any applicable law, statute or regulation;
- (xiv) purchase customary insurance coverage in accordance with <u>Section</u> 6.9 hereof;
- (xv) assert and/or waive any applicable privileges (legal or otherwise) on behalf of the Trust, or with respect to the Avoidance Action Trust Assets held by the Debtors at any time (prepetition or postpetition);
 - (xvi) maintain the books and records of the Trust;
- (xvii) open, maintain and close any bank, securities or other accounts that are necessary and appropriate to manage the Avoidance Action Trust Assets, including but not limited to the accounts listed on Exhibit A hereto;
- (xviii) receive from the Debtors and administer the Avoidance Action Trust SEC Reporting Cash in accordance with <u>Section 2.3(e)</u> hereof and file such reports as may be required pursuant to the applicable rules, regulations and interpretations of the SEC;
- (xix) receive from the GUC Trust and administer and utilize the GUC Trust Supplemental Cash in accordance with Section 2.3(f) hereof; and
- (xx) perform such functions and take such actions as are provided for or permitted in the Plan, the Confirmation Order, this Trust Agreement, the Settlement Agreement, the Litigation Cost Advance Agreement, or any other agreement executed pursuant to the Plan and take any other actions as it may deem to be reasonably necessary or appropriate to realize, preserve and dispose of the Avoidance Action Trust Assets.
 - (c) [Intentionally omitted.]
- (d) In all circumstances, the Trust Administrator shall act in the best interests of all Trust Beneficiaries and in furtherance of the purpose of the Trust, and in a manner not inconsistent with the best interests of the Trust Beneficiaries and consistent with the Budget. The Trust Administrator shall not take any action inconsistent with the purpose of the Trust, or take (or fail to take) any action that would cause the Trust (other than the Avoidance Action Trust Claims Reserve) to fail to qualify as a liquidating trust within the meaning of Treasury Regulation section 301.7701-4(d) that is treated as a grantor trust.
- (e) Notwithstanding any provision herein to the contrary, the Trust Administrator shall not serve on the board of directors, management committee or any similar governing body of any non-Debtor subsidiary of MLC, where the charter, limited liability company agreement, partnership agreement or other similar constituent document of such subsidiary does not provide for a liquidating purpose for such subsidiary. Except as otherwise provided in this Trust Agreement, the Trust Administrator will not be required to obtain the order

or approval of the Bankruptcy Court, or any other court of competent jurisdiction in, or account to the Bankruptcy Court or any other court of competent jurisdiction for, the exercise of any right, power or privilege conferred hereunder. Notwithstanding the foregoing, where the Trust Administrator determines, in its reasonable discretion, that it is necessary, appropriate or desirable, the Trust Administrator will have the right to submit to the Bankruptcy Court or any other court of competent jurisdiction any question or questions regarding any specific action proposed to be taken by the Trust Administrator with respect to this Trust Agreement, the Trust, or the Avoidance Action Trust Assets, including the administration and distribution of the Avoidance Action Trust Assets and the termination of the Trust. Pursuant to the Plan, the Bankruptcy Court has retained jurisdiction for such purposes and may approve or disapprove any such proposed action upon motion by the Trust Administrator.

(II) In Respect of the Other Debtor Residual Accepted Assets

The Trust Administrator shall have the rights, powers and privileges to act in respect of the Other Debtor Residual Accepted Assets, if any, in the manner set forth in Section 8.1(I), mutatis mutandis. In all such circumstances, the Trust Administrator shall act in the best interests of DIP Lenders and in furtherance of the purpose of the Trust, and in a manner not inconsistent with the best interests of the DIP Lenders. For the avoidance of doubt, the Trust Administrator shall not be obligated to undertake any activities in respect of the Other Debtor Residual Trust Assets unless there shall be available sufficient Other Debtor Residual Trust Administrative Cash to pay in full its fees, costs and expenses in respect thereof.

Limitations on the Trust Administrator. The Trust Administrator shall not be authorized to engage, in its capacity as Trust Administrator, in any trade or business with respect to the Avoidance Action Trust Assets or to take (or fail to take) any action that would cause the Trust (other than the Avoidance Action Trust Claims Reserve) to fail to qualify as a liquidating trust within the meaning of Treasury Regulation section 301.7701-4(d) that is treated as a grantor The Trust Administrator shall take such actions consistent with the prompt orderly disposition of the Avoidance Action Trust Assets and the Other Debtor Residual Accepted Assets, if any, as required by applicable law and consistent with the treatment of the Trust (other than the Avoidance Action Trust Claims Reserve) as a liquidating trust within the meaning of Treasury Regulation section 301.7701-4(d) that is treated as a grantor trust, to the extent such actions are permitted by this Trust Agreement. The Trust Administrator shall, in its capacity as Trust Administrator, be restricted to (x) the liquidation of the Trust on behalf, and for the benefit, of the Trust Beneficiaries and the distribution and application of Avoidance Action Trust Assets for the purposes set forth in, and the conservation and protection of the Avoidance Action Trust Assets and the administration thereof, and (y) the liquidation of the Trust on behalf, and for the benefit, of the DIP Lenders and the distribution and application of Other Debtor Residual Trust Assets for the purposes set forth in, and the conservation and protection of the Other Debtor Residual Trust Assets and the administration thereof, in each case in accordance with, the provisions of the Plan, the Confirmation Order and this Trust Agreement.

8.3. Agents and Professionals.

(a) The Trust Administrator on behalf of the Trust may, but shall not be required to, from time to time enter into contracts with, consult with and retain Trust Professionals,

on such terms as the Trust Administrator deems appropriate in accordance with the terms hereof and (other than in respect of the Other Debtor Residual Accepted Assets) in accordance with the Budget. None of the professionals that represented parties-in-interest in the Chapter 11 Cases shall be precluded from being engaged by the Trust Administrator solely on account of their service as a professional for such parties-in-interest prior to the Avoidance Action Trust Transfer Date or the Other Debtor Residual Assets Transfer Date, as the case may be.

- After the Avoidance Action Trust Transfer Date, Trust Professionals shall (b) be required to submit reasonably detailed invoices on a monthly basis to the Trust Administrator, the Trust Monitor, the DIP Lenders, the CE Capital Providers and the LW Capital Provider, including in such invoices a description of the work performed, the individuals who performed such work, the hourly rate of such person, plus an itemized statement of expenses for which payment is sought. Invoices submitted to the DIP Lenders, the CE Capital Providers and the LW Capital Provider may be in a form reasonably necessary to protect the attorney-client privilege. For the avoidance of doubt, any and all payments by the Trust from funds that (i) originated from the Litigation Cost Advance or the Supplemental Capital, or (ii) that are contained in the Funding Account, the CE Capital Providers Funding Account or the LW Capital Provider Funding Account shall be subject to this invoice disclosure requirement. No payment on account of such invoices shall be made until 15 days after presentation of the invoices to the DIP Lenders, the CE Capital Providers and the LW Capital Provider. After such period has passed, the Trust Administrator shall timely pay all such invoices that are not disputed by the Trust Administrator or the Trust Monitor. If any of the DIP Lenders, the CE Capital Providers or the LW Capital Provider have questions or concerns about any invoice submitted to the Trust, or about any other issue relating to the Trust's finances or administration, they shall be entitled to discuss the matter with the Trust Monitor, as well as with the Trust Administrator and/or counsel for the Trust, as appropriate; and the Trust Monitor will consult with the DIP Lenders, the CE Capital Providers or the LW Capital Provider, as applicable, and give good faith consideration to any objection that the DIP Lenders, the CE Capital Providers and/or the LW Capital Provider raise before approval of payment by the Trust Administrator or non-objection by the Trust Monitor. The Trust Monitor, Trust Administrator, and/or counsel for the Trust, as appropriate, will engage in good faith discussions and attempt in good faith to provide information reasonably requested by any of the DIP Lenders, the CE Capital Providers or the LW Capital Provider. In the event of any dispute concerning the entitlement to, or the reasonableness of any compensation and/or expenses of any Trust Professionals, either the Trust Administrator or the affected Trust Professional may petition the Bankruptcy Court to resolve the dispute.
- (c) Except as permitted by Section 6.1(b), (c), and (d), and Section 2.3(e) and (f), all payments to Trust Professionals (other than in respect of Other Debtor Residual Trust Assets) shall be paid out of the Avoidance Action Trust Administrative Cash. Payments to Trust Professionals for activities in respect of the Other Debtor Residual Trust Assets shall be paid out of the Other Debtor Residual Trust Administrative Cash.

8.4. Investment of Trust Cash.

(a) The Trust Administrator shall establish segregated accounts for the Trust Cash as follows: (i) Distributable Trust Cash which shall be held in trust for the benefit of the Trust Beneficiaries; (ii) Distributable Other Debtor Residual Cash, which shall be held in trust for the

benefit of the DIP Lenders and on which the DIP Lenders shall have a lien; (iii) Avoidance Action Trust Administrative Cash (excluding the Litigation Cost Advance) and Other Debtor Residual Trust Administrative Cash which shall be used to pay the administrative expenses of the Trust, on which the DIP Lenders shall have a lien; (iv) Avoidance Action Trust SEC Reporting Cash which shall be used to satisfy Avoidance Action Trust SEC Reporting Costs in accordance with Section 2.3(e) hereof; (v) GUC Trust Supplemental Cash which shall be used to satisfy current and projected fees and expenses of the Trust (including for the payment of Trust Professionals and any tax liabilities) in accordance with Section 2.3(f) hereof; (vi) Other Supplemental Cash, which shall be used to satisfy the current and projected future fees, costs and expenses (including, without limitation any Avoidance Action Trust SEC Reporting Costs, the fees and expenses of Trust Professionals, and any tax liabilities) in accordance with Section 6.1(d) hereof; (vii) Litigation Cost Advance, which shall be held in the Funding Account, which Funding Account shall be subject to a deposit account control agreement acceptable to the DIP Lenders and the Trust and on which the DIP Lenders shall have a first priority lien; (viii) the CE Supplemental Capital, which shall be held in the CE Capital Provider Funding Account, which upon the request of the CE Capital Providers shall be subject to a deposit account control agreement acceptable to the CE Capital Providers and the Trust; and (ix) the LW Supplemental Capital, which shall be held in the LW Capital Provider Funding Account, which upon the request of the LW Capital Provider shall be subject to a deposit account control agreement acceptable to the LW Capital Provider and the Trust.

- (b) The Trust Administrator shall invest the Trust Cash (including any earnings thereon or proceeds thereof) in the manner set forth in this Section 8.4, but shall otherwise be under no liability for interest or income on any monies received by the Trust hereunder and held for distribution or payment to the Trust Beneficiaries, except as such interest shall actually be received. Investment of any Trust Cash shall be administered in accordance with the general duties and obligations hereunder. The right and power of the Trust Administrator to invest the Trust Cash and the proceeds thereof, or any income earned by the Trust, shall be limited to investing such Trust Cash (pending distribution or disbursement in accordance with the Plan or this Trust Agreement) in Permissible Investments; provided, however, that such Permissible Investments shall be limited to include only those investments that a liquidating trust, within the meaning of Treasury Regulations section 301.7701-4(d), may be permitted to hold, pursuant to the Treasury Regulations, or any modification in the IRS guidelines, whether set forth in IRS rulings, other IRS pronouncements or otherwise.
- (c) For the avoidance of doubt, the Trust is not, and will not hold itself out as, an "investment company" as such term is understood under the Investment Company Act of 1940, and is prohibited from investing, reinvesting or trading in securities (other than making any Permissible Investments as contemplated by the Plan, the Confirmation Order and this Trust Agreement) or conducting any trade or business other than implementing the Plan and distributing Distributable Trust Assets under the Plan and this Trust Agreement.
- 8.5. <u>Communication with the GUC Trust Administrator</u>. The Trust Administrator shall communicate with the GUC Trust Administrator to obtain such information regarding, as of a given date, (A) the holders and amounts of General Unsecured Claims, Disputed General Unsecured Claims, Unresolved Term Loan Avoidance Action Claims, Unresolved Other Avoidance Action Claims and Resolved Allowed General Unsecured Claims, (B) the respective

Maximum Amounts of all Disputed General Unsecured Claims, Unresolved Term Loan Avoidance Action Claims and Unresolved Other Avoidance Action Claims, (C) the Current Total Amount, (D) the Aggregate Maximum Amount, (E) any components of the foregoing and (E) any other information within the custody or control of the GUC Trust Administrator which shall be necessary or desirable in order for the Trust Administrator to timely make any calculation or determination hereunder to identify and make distributions to the GUC Beneficiaries and to maintain any books and records required to be maintained, or necessary or desirable for the Trust Administrator or the Trust Monitor to fulfill their respective functions, hereunder; provided, however, that the provision of such information shall be under appropriate arrangements of confidentiality to the extent such information has at the time not been publicly disclosed.

8.6. <u>Termination</u>. The duties, responsibilities and powers of the Trust Administrator will terminate when the Trust is dissolved and terminated pursuant to <u>Article IV</u> hereof and the Trust Administrator has performed all of its obligations under <u>Section 4.3</u>, by an order of the Bankruptcy Court or by entry of a final decree closing the Debtors' cases before the Bankruptcy Court; <u>provided</u>, <u>however</u>, that <u>Sections 9.4, 9.5</u> and <u>9.6</u> hereof shall survive such termination, dissolution and entry.

ARTICLE IX ADDITIONAL MATTERS CONCERNING THE TRUST ADMINISTRATOR

- 9.1. Reliance by Trust Administrator. Except as otherwise provided in the Plan, the Confirmation Order or this Trust Agreement, the Trust Administrator may rely and shall be protected in acting upon any resolution, statement, instrument, opinion, report, notice, request, consent, order or other paper or document reasonably believed by the Trust Administrator to be genuine and to have been signed or presented by the proper party or parties.
- 9.2. <u>Liability to Third Persons</u>. To the fullest extent permitted by applicable law, the Trust Administrator Parties shall not be subject to any personal liability whatsoever, in tort, contract or otherwise, to any person (including, in the case of the Trust Administrator, to any Trust Professionals retained by the Trust Administrator in accordance with this Trust Agreement) in connection with the Avoidance Action Trust Assets, the Other Debtor Residual Trust Assets or the affairs of the Trust and shall not be liable with respect to any action taken or omitted to be taken in good faith, except for actions and omissions determined by a Final Order of the Bankruptcy Court to be due to their respective willful misconduct (including, but not limited to, conduct that results in a personal profit at the expense of the Trust), gross negligence, fraud, malpractice, criminal conduct, unauthorized use of confidential information that causes damages, breach of fiduciary duty (to the extent applicable), or ultra vires acts, and all such persons shall look solely to the Avoidance Action Trust Assets (other than in respect of the Other Debtor Residual Accepted Assets and activities related thereto) or the Other Debtor Residual Trust Assets (in respect of the Other Debtor Residual Accepted Assets and activities related thereto) for satisfaction of claims of any nature arising in connection with affairs of the Trust.
- 9.3. <u>Non-liability of Trust Administrator for Acts of Others</u>. Except as provided herein, nothing contained in the Plan, the Confirmation Order or this Trust Agreement shall be deemed to be an assumption by the Trust Administrator of any of the liabilities, obligations or duties of the Debtors or shall be deemed to be or contain a covenant or agreement by the Trust

Administrator to assume or accept any such liability, obligation or duty. Any successor Trust Administrator may accept and rely upon any accounting made by or on behalf of any predecessor Trust Administrator hereunder, and any statement or representation made as to the assets comprising the Avoidance Action Trust Assets or the Other Debtor Residual Trust Assets, or as to any other fact bearing upon the prior administration of the Trust, so long as it has a good faith basis to do so. The Trust Administrator shall not be liable for having accepted and relied in good faith upon any such accounting, statement or representation if it is later proved to be incomplete, inaccurate or untrue. Neither the Trust Administrator nor any successor Trust Administrator shall be liable for any act or omission of any predecessor Trust Administrator, nor have a duty to enforce any claims against any predecessor Trust Administrator on account of any such act or omission, unless directed in good faith to do so by the Trust Monitor.

- 9.4. Exculpation. As of the earlier of the Avoidance Action Trust Transfer Date and the Other Debtor Residual Assets Transfer Date, to the fullest extent permitted by applicable law, the Trust Administrator Parties shall be and hereby are exculpated by all Persons, including holders of DIP Credit Agreement Claims, General Unsecured Claims and Units and other parties-ininterest, from any and all claims, causes of action and other assertions of liability arising out of the discharge of their respective powers and duties conferred by the Plan, the Confirmation Order, this Trust Agreement or any Order of the Bankruptcy Court entered pursuant to or in furtherance of the Plan, or applicable law or otherwise, except for actions or omissions to act that are determined by Final Order of the Bankruptcy Court to have arisen out of each such Trust Administrator Party's own respective willful misconduct (including, but not limited to, conduct that results in a personal profit at the expense of the Trust), gross negligence, fraud, malpractice, criminal conduct, unauthorized use of confidential information that causes damages, breach of fiduciary duty (to the extent applicable), or ultra vires acts. No holder of a DIP Credit Agreement Claim, General Unsecured Claim or other party-in-interest will have or be permitted to pursue any claim or cause of action against the Trust Administrator Parties or the Trust, for making payments and distributions in accordance with the Plan, the Confirmation Order or this Trust Agreement or for implementing the provisions thereof. Any action taken or omitted to be taken with the express approval of the Bankruptcy Court and, in the case of action taken in respect of the Other Debtor Residual Accepted Assets, with the approval or at the direction of the DIP Lenders will conclusively be deemed not to constitute willful misconduct, gross negligence, fraud, malpractice, criminal conduct, unauthorized use of confidential information that causes damages, breach of fiduciary duty, or ultra vires acts; provided, however, that notwithstanding any provision herein to the contrary, the Trust Administrator shall not be obligated to comply with a direction of the Trust Monitor, whether or not express, which would result in a change to the distribution provisions of the Plan, the Confirmation Order or this Trust Agreement.
- 9.5. <u>Limitation of Liability</u>. In no event shall the Trust Administrator Parties be liable for punitive, exemplary, consequential, special or other damages for a breach of, or otherwise in connection with, this Trust Agreement under any circumstances.

9.6. Indemnity.

(a) To the fullest extent permitted by applicable law, the Trust Administrator Parties shall be indemnified by the Trust from the Avoidance Action Trust Assets (other than in respect of the Other Debtor Residual Accepted Assets and activities related thereto) or the Other

Debtor Residual Trust Assets (in respect of the Other Debtor Residual Accepted Assets and activities related thereto) for any losses, claims, damages, liabilities and expenses occurring after the earlier of the Avoidance Action Trust Transfer Date and the Other Debtor Residual Assets Transfer Date, including reasonable attorneys' fees, disbursements and related expenses which the Trust Administrator Parties may incur or to which the Trust Administrator Parties may become subject in connection with any action, suit, proceeding or investigation brought by or threatened against one or more of the Trust Administrator Parties on account of the acts or omissions in their capacity as, or on behalf of, the Trust Administrator; provided, however, that the Trust shall not be liable to indemnify any Trust Administrator Party for any act or omission arising out of such Trust Administrator Party's respective actions that are determined by a Final Order of the Bankruptcy Court to be willful misconduct (including, but not limited to, conduct that results in a personal profit at the expense of the Trust), gross negligence, fraud, malpractice, criminal conduct, unauthorized use of confidential information that causes damages, breach of fiduciary duty (to the extent applicable), or ultra vires acts. Notwithstanding any provision herein to the contrary, the Trust Administrator Parties shall be entitled to obtain advances from the Trust to cover their reasonable expenses of defending themselves in any action brought against them as a result of the acts or omissions, actual or alleged, of an Trust Administrator Party in its capacity as such; provided, however, that the Trust Administrator Parties receiving such advances shall repay the amounts so advanced to the Trust immediately upon the entry of a final, non-appealable judgment or order finding that such Trust Administrator Parties were not entitled to any indemnity under the provisions of this Section 9.6. Any amounts payable to any Trust Administrator Party pursuant to this Section 9.6 (other than in respect of the Other Debtor Residual Accepted Assets and activities related thereto) shall be satisfied as follows: (i) first from the Avoidance Action Trust Administrative Cash, (ii) second from the Distributable Trust Cash, if any; provided, however, that the use of Distributable Trust Cash as contemplated in clause (ii) of the foregoing shall be subject to the prior approval by the Bankruptcy Court, as provided in Section 6.1(b); (iii) third from the GUC Trust Supplemental Cash, if any; and (iv) fourth from the Other Supplemental Cash, if any. Any amounts payable to any Trust Administrator Party pursuant to this Section 9.6 in respect of the Other Debtor Residual Accepted Assets and activities related thereto shall be satisfied as follows: (i) first from the Other Debtor Residual Trust Administrative Cash, and (ii) second from the Distributable Other Debtor Residual Trust Cash, if any.

(b) Anything to the contrary in this Trust Agreement or in any other agreement notwithstanding, to the extent that the Avoidance Action Trust Administrative Cash, Distributable Trust Cash, GUC Trust Supplemental Cash or Other Supplemental Cash, or the Other Debtor Residual Trust Administrative Cash or Distributable Other Debtor Residual Trust Cash, as the case may be, shall be insufficient to fully indemnify the Trust Administrator Parties or to provide advances to the Trust Administrator Parties in accordance with Section 9.6(a), the Trust Administrator Parties shall be indemnified and shall be entitled to obtain advances, first from the Other GUC Trust Administrative Cash (as defined in the GUC Trust Agreement), and second from the GUC Trust Distributable Assets (as defined in the GUC Trust Agreement), to the same extent as the GUC Trust Administrator Parties under Section 9.6 of the GUC Trust Agreement or any successor provision thereunder, as provided in Section 9.6 of the GUC Trust Agreement in effect on the date hereof.

(c) The foregoing indemnities in respect of any Trust Administrator Party shall survive the termination of such Trust Administrator Party from the capacity for which they are indemnified.

9.7. <u>Compensation and Expenses</u>.

- (a) The Trust Administrator shall receive fair and reasonable compensation for its services (other than in respect of the Other Debtor Residual Accepted Assets and activities related thereto), to be paid out of the Avoidance Action Trust Administrative Cash in accordance with the approved Budget (or from the Avoidance Action Trust SEC Reporting Cash in accordance with Section 2.3(e) hereof, the GUC Trust Supplemental Cash in accordance with Section 2.3(f) hereof, or the Other Supplemental Cash in accordance with Section 6.1(d) hereof). The Trust Administrator shall be entitled, without the need for approval of the Bankruptcy Court, to reimburse itself on a monthly basis (i) from the Avoidance Action Trust Administrative Cash, the GUC Trust Supplemental Cash or the Other Supplemental Cash, as applicable, for such compensation and all reasonable out-of-pocket expenses actually incurred in the performance of duties in accordance with this Trust Agreement and the Budget and (ii) from the Avoidance Action Trust SEC Reporting Cash for such compensation and all reasonable out-of-pocket expenses actually incurred in the performance of duties in accordance with this Trust Agreement and with the approval of the Trust Monitor.
- (b) The Trust Administrator shall receive compensation for its services in respect of the Other Debtor Residual Accepted Assets and activities related thereto, to be paid out of the Other Debtor Residual Trust Administrative Cash, as the Trust Administrator and a majority in interest of the DIP lenders shall agree; provided that the Trust Administrator shall not be obligated to undertake any activities in respect of the Other Debtor Residual Accepted Assets unless such compensation arrangements shall be acceptable to the Trust Administrator in its sole discretion.
- 9.8. No Personal Financial Liability. No provision of the Plan, Confirmation Order or this Trust Agreement shall be construed as requiring the Trust Administrator to expend or risk its own funds or otherwise to incur any personal financial liability (x) in the performance of any of its duties thereunder or hereunder, including any situation where the Avoidance Action Trust Assets are insufficient to permit the administration of the Trust or distributions as contemplated herein or the payment of fees and expenses of the Trust Professionals, or (y) in the exercise of any of its rights or powers afforded hereunder or thereunder.

ARTICLE X SUCCESSOR TRUST ADMINISTRATORS

10.1. <u>Resignation</u>. The Trust Administrator may resign from the Trust by giving at least sixty (60) days' prior written notice thereof to the Trust Monitor. Such resignation shall become effective on the later to occur of (x) the date specified in such written notice and (y) the effective date of the appointment of a successor Trust Administrator in accordance with <u>Section 10.4</u> hereof and such successor's acceptance of such appointment in accordance with <u>Section 10.5</u> hereof.

- 10.2. <u>Removal</u>. The holders of a majority of the Units or the DIP Lenders may at any time petition the Bankruptcy Court for the removal of the Trust Administrator, but only for good cause shown. Such removal shall become effective on the date ordered by the Bankruptcy Court, provided that such removal shall not become effective until the appointment of a successor Trust Administrator in accordance with <u>Section 10.4</u> hereof and such successor's acceptance of such appointment in accordance with <u>Section 10.5</u> hereof. The services of the Trust Administrator shall also terminate upon its bankruptcy, provided that such termination shall not become effective until the appointment of a successor Trust Administrator in accordance with <u>Section 10.4</u> hereof and such successor's acceptance of such appointment in accordance with <u>Section 10.5</u> hereof.
- 10.3. Effect of Resignation or Removal. The resignation, removal or bankruptcy of the Trust Administrator shall not operate to terminate the Trust or to revoke any existing agency created pursuant to the terms of the Plan, the Confirmation Order or this Trust Agreement or invalidate any action theretofore taken by the Trust Administrator. The exculpation, indemnity and limitation of liability provisions of Article X of this Trust Agreement shall survive the resignation, removal or bankruptcy of the Trust Administrator. All fees and expenses properly incurred by the Trust Administrator prior to the resignation, Incompetency, removal or bankruptcy of the Trust Administrator shall be paid from the Avoidance Action Trust Administrative Cash, the GUC Trust Supplemental Cash or the Other Supplemental Cash (other than in respect of the Other Debtor Residual Accepted Assets and activities related thereto), from the Other Debtor Residual Trust Administrative Cash (in respect of the Other Debtor Residual Accepted Assets and activities related thereto) or from Avoidance Action Trust SEC Reporting Cash (in respect of Avoidance Action Trust SEC Reporting Costs in accordance with Section 2.3(e) hereof), unless such fees and expenses are disputed by (x) the Trust Monitor or (y) the successor Trust Administrator, in which case the Bankruptcy Court shall resolve the dispute and any disputed fees and expenses of the predecessor Trust Administrator that are subsequently allowed by the Bankruptcy Court shall be paid from the Avoidance Action Trust Administrative Cash, the GUC Trust Supplemental Cash or the Other Supplemental Cash (other than in respect of the Other Debtor Residual Accepted Assets and activities related thereto), from the Other Debtor Residual Trust Administrative Cash (in respect of the Other Debtor Residual Accepted Assets and activities related thereto) or from Avoidance Action Trust SEC Reporting Cash (in respect of Avoidance Action Trust SEC Reporting Costs in accordance with Section 2.3(e) hereof). In the event of the resignation, removal or bankruptcy of the Trust Administrator, such Trust Administrator shall:
- (a) promptly execute and deliver such documents, instruments and other writings as may be reasonably requested by the successor Trust Administrator or directed by the Bankruptcy Court to effect the termination of such Trust Administrator's capacity under this Trust Agreement;
- (b) promptly deliver to the successor Trust Administrator all documents, instruments, records and other writings related to the Trust as may be in the possession of such Trust Administrator; and
- (c) otherwise assist and cooperate in effecting the assumption of its obligations and functions by such successor Trust Administrator.

- 10.4. Appointment of Successor. In the event of the resignation, removal, Incompetency or bankruptcy of the Trust Administrator, the Trust Monitor shall promptly appoint a successor Trust Administrator, provided that such appointment shall not take effect unless approved by the Bankruptcy Court upon the petition of the Trust Monitor and until the successor Trust Administrator shall have delivered written acceptance of its appointment as described Section 10.5 below. If a successor Trust Administrator does not take office within thirty (30) days after the resignation, removal, Incompetency or bankruptcy of the retiring Trust Administrator, the Bankruptcy Court, upon its own motion or the motion of the retiring Trust Administrator or any Trust Beneficiary, shall appoint a successor Trust Administrator.
- 10.5. Acceptance of Appointment by Successor Trust Administrator. Any successor Trust Administrator appointed hereunder shall execute an instrument accepting its appointment and shall deliver one counterpart thereof to the Bankruptcy Court for filing and to the Trust Monitor and, in case of the Trust Administrator's resignation, to the resigning Trust Administrator. Thereupon, such successor Trust Administrator shall, without any further act, become vested with all the duties, powers, rights, obligations, title, discretion and privileges of its predecessor in the Trust with like effect as if originally named Trust Administrator and shall be deemed appointed pursuant to Bankruptcy Code Section 1123(b)(3)(B); provided, however, such successor Trust Administrator shall file an amendment to the Certificate of Trust with the Secretary of State as required by the Delaware Act. The predecessor Trust Administrator shall duly assign, transfer and deliver to such successor Trust Administrator all Avoidance Action Trust Assets held by such predecessor Trust Administrator hereunder and shall, as directed by the Bankruptcy Court or reasonably requested by such successor Trust Administrator, execute and deliver an instrument or instruments conveying and transferring to such successor Trust Administrator upon the trusts herein expressed, all the duties, powers, rights, obligations, title, discretion and privileges of the predecessor Trust Administrator.
- 10.6. Successor Entity to Trust Administrator. Any business entity into which the Trust Administrator may be merged or converted or with which it may be consolidated, or any entity resulting from any merger, conversion or consolidation to which the Trust Administrator shall be a party, or any entity succeeding to all or substantially all of the corporate trust business of the Trust Administrator, shall be the successor of the Trust Administrator hereunder, without the execution or filing of any paper or any further act on the part of any of the parties hereto; provided, however, such successor Trust Administrator shall file an amendment to the Certificate of Trust with the Secretary of State as required by the Delaware Act.

ARTICLE XI TRUST MONITOR

11.1. General.

(a) The Trust Monitor shall oversee the activities of the Trust Administrator as set forth in this Trust Agreement. In all circumstances, the Trust Monitor shall act in the best interests of all Trust Beneficiaries, in furtherance of the purpose of the Trust, and in accordance with this Trust Agreement.

- (b) In furtherance of its rights and responsibilities under this Trust Agreement, the Trust Monitor shall have access, on reasonable advance notice and during regular business hours, to all such books and records of the Trust and the Trust Administrator, shall have the right to consult with all such professionals engaged by the Trust Administrator and shall participate in all such meetings of the Trust Administrator and the Trust Professionals as the Trust Monitor deems reasonably necessary or appropriate. Any documents shared between the Trust Administrator and the Trust Monitor shall be subject to joint privilege, and such sharing shall not be deemed to waive any attorney-client or work product privilege in respect of such documents.
 - (c) [Intentionally omitted.]
- (d) Notwithstanding anything in this <u>Section 11.1</u> or <u>Section 11.2</u> hereof, the Trust Monitor shall not take (or fail to take) any action which will cause the Trust (other than the Avoidance Action Trust Claims Reserve) to fail to qualify as a liquidating trust within the meaning of Treasury Regulation section 301.7701-4(d) that is treated as a grantor trust.

11.2. Appointment and Removal of the Trust Monitor.

- (a) Subject to Section 11.2(d), the Trust Monitor shall serve until the earlier of (w) the final distribution of all Distributable Trust Assets and the Distributable Other Debtor Residual Trust Assets, if any, (x) its resignation pursuant to subsection (b) of this Section 11.2, (y) its removal pursuant to subsection (c) of this Section 11.2 or (z) its bankruptcy or insolvency.
- (b) The Trust Monitor may resign at any time by written notice of resignation to the Trust Administrator, a copy of which shall also be filed by the Trust Monitor with the Bankruptcy Court. Such resignation shall be effective no earlier than sixty (60) days from the date of such notice or such earlier time as a successor is appointed in accordance with the provisions of subsection (d) of this Section 11.2.
- (c) The holders of a majority of the Units or the DIP Lenders may at any time petition the Bankruptcy Court for the removal of the Trust Monitor, but only for good cause shown. Such removal shall become effective on the date ordered by the Bankruptcy Court.
- (d) In the event of the resignation, removal, bankruptcy or insolvency of the Trust Monitor, the Trust Administrator shall promptly appoint a successor Trust Monitor, provided that such appointment shall not take effect unless approved by the Bankruptcy Court upon the petition of the Trust Administrator and until the successor Trust Monitor shall have delivered written acceptance of its appointment as described in clause (e) of this Section 11.2 below; and provided further that until a new Trust Monitor's appointment is effective, the resigning Trust Monitor's appointment shall remain in effect, and the resigning Trust Monitor shall fulfill all obligations and duties of the Trust Monitor. If a successor Trust Monitor does not take office within thirty (30) days after the resignation, removal, incompetency, bankruptcy or insolvency of the retiring Trust Monitor, the Bankruptcy Court, upon its own motion or the motion of the retiring Trust Monitor or any Trust Beneficiary, shall appoint a successor Trust Monitor.
- (e) All fees and expenses properly incurred by the Trust Monitor prior to the resignation, Incompetency, removal or bankruptcy of the Trust Monitor shall be paid from the Avoidance Action Trust Administrative Cash, the GUC Trust Supplemental Cash or the Other

Supplemental Cash (other than in respect of the Other Debtor Residual Accepted Assets and activities related thereto), from the Other Debtor Residual Trust Administrative Cash (in respect of the Other Debtor Residual Accepted Assets and activities related thereto) or from Avoidance Action Trust SEC Reporting Cash (in respect of Avoidance Action Trust SEC Reporting Costs in accordance with Section 2.3(e) hereof), unless such fees and expenses are disputed by (x) the Trust Administrator or (y) the successor Trust Monitor, in which case the Bankruptcy Court shall resolve the dispute and any disputed fees and expenses of the predecessor Trust Monitor that are subsequently allowed by the Bankruptcy Court shall be paid from the Avoidance Action Trust Administrative Cash, the GUC Trust Supplemental Cash or the Other Supplemental Cash (other than in respect of the Other Debtor Residual Accepted Assets and activities related thereto) from the Other Debtor Residual Trust Administrative Cash (in respect of the Other Debtor Residual Accepted Assets and activities related thereto) or from Avoidance Action Trust SEC Reporting Cash (in respect of Avoidance Action Trust SEC Reporting Cash (in respect of Avoidance Action Trust SEC Reporting Cash (in respect of Avoidance Action Trust SEC Reporting Cash (in respect of Avoidance Action Trust SEC Reporting Cash (in respect of Avoidance Action Trust SEC Reporting Cash (in respect of Avoidance Action Trust SEC Reporting Cash (in respect of Avoidance Action Trust SEC Reporting Cash (in respect of Avoidance Action Trust SEC Reporting Cash (in respect of Avoidance Action Trust SEC Reporting Cash (in respect of Avoidance Action Trust SEC Reporting Cash (in respect of Avoidance Action Trust SEC Reporting Cash (in respect of Avoidance Action Trust SEC Reporting Cash (in respect of Avoidance Action Trust SEC Reporting Cash (in respect of Avoidance Action Trust SEC Reporting Cash (in respect of Avoidance Action Trust SEC Reporting Cash (in respect of Avoidance Action Trust SEC Reporting Cash

- (f) Any successor Trust Monitor appointed hereunder shall execute an instrument accepting its appointment and shall deliver one counterpart thereof to the Bankruptcy Court for filing and to the Trust Administrator.
- (g) Immediately upon effectiveness of the appointment of a successor Trust Monitor, all rights, powers, duties, authority, and privileges of the predecessor Trust Monitor hereunder will be vested in and undertaken by the successor Trust Monitor without any further act. The successor Trust Monitor shall not be liable personally for any act or omission of the predecessor Trust Monitor.

11.3. Approval of and Consultation with the Trust Monitor.

- (I) Other Than in Respect of the Other Debtor Residual Accepted Assets
- (a) Notwithstanding anything in this Trust Agreement to the contrary, the Trust Administrator shall submit to the Trust Monitor for its review and prior approval the following matters, in addition to any other matters that expressly require the approval of the Trust Monitor pursuant to the terms of the Plan, the Confirmation Order or this Trust Agreement:
- (i) Any decision to settle or otherwise resolve the Term Loan Avoidance Action;
- (ii) Any decision to refrain from making any distributions to the holders of Allowed General Unsecured Claims or Units, as the case may be, in accordance with this Trust Agreement, except as expressly permitted herein;
- (iii) Any decision to retain and/or to terminate the retention of Trust Professionals (other than legal counsel retained to represent the Trust Administrator in connection with its role as Trust Administrator, which shall be in the Trust Administrator's sole discretion);
- (iv) The incurrence of any cost or expense of the Trust in excess of 10% of any individual line item therefor in the approved Budget, measured on a quarterly basis;

<u>provided</u>, <u>however</u>, that approval of the Trust Monitor shall not be required in the case of any cost or expense authorized by further order of the Bankruptcy Court;

- (v) The Budget described in <u>Section 6.3</u> hereof and any changes thereto;
- (vi) Any amendment of this Trust Agreement as provided in $\underline{\text{Section}}$ $\underline{13.13}$ hereof; and
- (vii) Any distribution that is not made in accordance with the provisions of <u>Article V</u> as contemplated by <u>Section 5.7</u>; <u>provided</u>, <u>however</u>, that any deviation from the provisions of <u>Article V</u> other than as contemplated by <u>Section 5.7</u> shall also require approval of the Bankruptcy Court.
- (b) In addition to any other matters that expressly require consultation with the Trust Monitor pursuant to the terms of the Plan, the Confirmation Order or this Trust Agreement, the Trust Administrator shall consult with the Trust Monitor in advance of an application to the Bankruptcy Court to use, or to sell or borrow against, the Term Loan Avoidance Action or the Distributable Trust Assets in order to satisfy expenses of the Trust, as contemplated by Section 6.1(b) and Section 6.1(d) hereof.
- (c) In the event of any disagreement between the Trust Administrator and the Trust Monitor regarding any matter requiring the approval or direction of the Trust Monitor under this Trust Agreement, the Trust Administrator and the Trust Monitor shall consult and negotiate diligently and in good faith to resolve such disagreement. If despite their good faith efforts, the Trust Administrator and the Trust Monitor are unable to resolve any disagreement, or the Trust Administrator cannot otherwise obtain approval or direction from the Trust Monitor as required by this Trust Agreement, the Trust Administrator may petition the Bankruptcy Court, with a copy to the Trust Monitor, requesting such approval or direction.
 - (II) In Respect of the Other Debtor Residual Accepted Assets

The Trust Monitor shall have the rights, powers and privileges to act in respect of the Other Debtor Residual Accepted Assets, if any, in the manner set forth in Section 11.3 (I), mutatis mutandis and to the extent applicable. For the avoidance of doubt, the Trust Monitor shall not be obligated to undertake any activities in respect of the Other Debtor Residual Trust Assets unless there shall be available sufficient Other Debtor Residual Trust Administrative Cash to pay in full its fees, costs and expenses in respect thereof.

11.4. Exculpation and Indemnification; Limitation of Liability. To the fullest extent permitted by applicable law, the Trust Monitor Parties shall not be subject to personal liability, and shall be exculpated and indemnified, and shall have the right to obtain advances to cover reasonable expenses of defense, to the same extent as the Trust Administrator Parties pursuant to Section 9.2, Section 9.4, Section 9.5, Section 9.6 and Section 10.3. In no event will the Trust Monitor Parties be liable for punitive, exemplary, consequential, special or other damages for a breach of, or otherwise in connection with, this Trust Agreement under any circumstances.

11.5. Compensation and Expenses.

- (a) The Trust Monitor shall receive fair and reasonable compensation for its services (other than in respect of the Other Debtor Residual Accepted Assets and activities related thereto), to be paid out of the Avoidance Action Trust Administrative Cash, in accordance with the approved Budget (or from the Avoidance Action Trust SEC Reporting Cash in accordance with Section 2.3(e) hereof, the GUC Trust Supplemental Cash in accordance with Section 2.3(f) hereof, or the Other Supplemental Cash in accordance with Section 6.1(d) hereof). The Trust Monitor shall be entitled on a monthly basis, without the need for approval of the Bankruptcy Court, to direct the Trust Administrator to reimburse the Trust Monitor (i) from the Avoidance Action Trust Administrative Cash, the GUC Trust Supplemental Cash or the Other Supplemental Cash, as applicable, for all reasonable out-of-pocket expenses actually incurred in the performance of duties in accordance with this Trust Agreement, consistent with the Budget prepared pursuant to Section 6.3 hereof and (ii) from the Avoidance Action Trust SEC Reporting Cash for such compensation and all reasonable out-of-pocket expenses actually incurred in the performance of duties in accordance with this Trust Agreement.
- (b) The Trust Monitor shall receive compensation for its services than in respect of the Other Debtor Residual Accepted Assets and activities related thereto, to be paid out of the Other Debtor Residual Trust Administrative Cash, as the Trust Monitor and a majority in interest of the DIP lenders shall agree; provided that the Trust Monitor shall not be obligated to undertake any activities in respect of the Other Debtor Residual Accepted Assets unless such compensation arrangements shall be acceptable to the Trust Monitor in its sole discretion.

ARTICLE XII ACTION BY MAJORITY OF HOLDERS OF UNITS

Holders of a majority of the Units or the DIP Lenders from time to time outstanding may petition the Bankruptcy Court to remove the Trust Administrator in accordance with Section 10.2 or to remove the Trust Monitor in accordance with Section 11.1, but in each case only for good cause shown. In determining whether the holders of a majority of the Units have concurred in any such petition, Units held by the Trust Administrator or the Trust Monitor or any of their respective Affiliates shall be disregarded.

ARTICLE XIII MISCELLANEOUS PROVISIONS

- 13.1. Actions Taken on Other Than Business Day. In the event that any payment or act under the Plan, the Confirmation Order or this Trust Agreement is required to be made or performed on a date that is not a Business Day, then the making of such payment or the performance of such act may be completed on the next succeeding Business Day, but shall be deemed to have been completed as of the required date.
- 13.2. <u>Governing Law</u>. This Trust Agreement shall be governed by and construed in accordance with the laws of the State of Delaware without giving effect to rules governing conflicts of law.

- 13.3. <u>Jurisdiction</u>. Subject to the proviso below, the parties agree that the Bankruptcy Court shall have exclusive and continuing jurisdiction over the Trust and the Trust Administrator, including the administration and activities of the Trust and the Trust Administrator; <u>provided</u>, <u>however</u>, that notwithstanding the foregoing, the Trust Administrator shall have power and authority to bring any action in any court of competent jurisdiction to prosecute any claims or Causes of Action assigned to the Trust, including the Delaware Chancery Court, the Delaware Superior Court and the Delaware Supreme Court.
- 13.4. Third Party Beneficiary. Trust Beneficiaries (including the DIP Lenders in their capacities as such) are third party beneficiaries of this Trust Agreement. The Trust Administrator Parties (other than the Trust Administrator) are third party beneficiaries of the provisions of Section 9.2, Section 9.4 and Section 9.6 of this Trust Agreement. The Trust Monitor Parties (other than the Trust Monitor) are third party beneficiaries of the provisions of Section 11.4 of this Trust Agreement, and, to the extent incorporated therein, Section 9.2, Section 9.4, Section 9.5 and Section 9.6 of this Trust Agreement. Except as aforesaid, there are no other third party beneficiaries of this Trust Agreement.
- 13.5. Severability. In the event any provision of this Trust Agreement or the application thereof to any person or circumstances shall be determined by a final, non-appealable judgment or order to be invalid or unenforceable to any extent, the remainder of this Trust Agreement or the application of such provision to persons or circumstances or in jurisdictions other than those as to or in which it is held invalid or unenforceable, shall not be affected thereby, and each provision of this Trust Agreement shall be valid and enforceable to the fullest extent permitted by law.
- 13.6. <u>Notices</u>. Any notice or other communication required or permitted to be made under this Trust Agreement shall be in writing and shall be deemed to have been sufficiently given, for all purposes, if delivered personally, by email, facsimile, sent by nationally recognized overnight delivery service or mailed by first-class mail:
 - (A) if to the Trust Administrator, to:

Wilmington Trust Company Rodney Square North 1100 North Market Street Wilmington, Delaware, 19890-1615

Phone: (302) 636-6000 Fax: (302) 636-4140

Attn: Corporate Trust Administration

With a copy to:

Binder & Schwartz LLP 366 Madison Avenue, 6th Floor New York, NY 10017 Phone: (212) 510-7008

Attn: Eric B. Fisher

(B) if to the Trust Monitor, to:

Arthur J. Gonzalez New York University School of Law 40 Washington Square South New York, NY 10012

- (C) if to any Trust Beneficiary, to:
 - (1) in the case of a DIP Lender,
 - a. if to the U.S. Treasury, to:

United States Department of the Treasury 1500 Pennsylvania Avenue, NW Washington, D.C. 20220 Attn: Chief Counsel, Office of Financial Stability Telecopier: (202) 927-9225

with a copy to:

OFSChiefCounselNotices@treasury.gov

b. if to Export Development Canada, to:

Export Development Canada 151 Slater Street Ottawa, Ontario Canada K1A 1K3 Attention: Loans Services Telecopy: 613-598-2514;

with a copy to:

Export Development Canada
151 Slater Street
Ottawa, Ontario
Canada K1A 1K3
Attention: Asset Management/Covenants

Officer

Telecopy: 613-598-3186

(2) if to the CE Capital Providers, to:

Cynthiana LLC 2711 Centerville Road Wilmington, DE 19808 Fax Number: (302) 636-5454

Email Address: info@litfinsolutions.com

Attn: Cynthiana LLC Manager

Earlham LLC 2711 Centerville Road Wilmington, DE 19808 Fax Number: (302) 636-5454

Email Address: info@litfinsolutions.com

Attn: Earlham LLC Manager

(3) if to the LW Capital Provider, to:

LW Holdco VI LLC 1350 Avenue of the Americas, 2nd Floor New York, New York 10019 Fax Number: (646) 389-1032 Email Address: weinstein@lakewhillans.com

Attn: Boaz Weinstein

- (4) in the case of a holder of an Allowed General Unsecured Claim, to the last known address of such holder according to the Debtors' Schedules and/or such holder's proof of claim; and
- (5) in the case of holder of Units, to such address as appears on the books and records of the Trust Administrator, or such other address as may be designated from time to time by notice given in accordance with the provisions of this Section 13.6.
- 13.7. <u>Headings</u>. The headings contained in this Trust Agreement are solely for convenience of reference and shall not affect the meaning or interpretation of this Trust Agreement or of any term or provision hereof.
- 13.8. <u>Plan</u>. The terms of this Trust Agreement are intended to supplement the terms provided by the Plan and the Confirmation Order. To the extent that the terms of sections 5.6 and 6.5 of the Plan are inconsistent with the terms set forth in this Trust Agreement with respect to the Trust, then the terms of the Trust Agreement shall govern. All other provisions of the Plan shall supersede the provisions of this Trust Agreement, including section 6.15 of the Plan, which provides that the restrictions set forth in paragraph 20 of the Final Order approving the DIP Credit Agreement (ECF No. 2529) shall continue to apply.

13.9. Ambiguities and Construction.

- (a) The Trust created by this Trust Agreement (other than the Avoidance Action Trust Claims Reserve) is intended to qualify as a liquidating trust under Treasury Regulation section 301.7701-4(d) for U.S. federal and applicable state and local income tax purposes and, to the extent provided by law, shall be governed and construed in all respects as such a trust and any ambiguity herein shall be construed consistent herewith and, if necessary, this Trust Agreement may be amended to comply with such U.S. federal and applicable state and local income tax laws, which amendments may apply retroactively.
 - (b) Unless the context otherwise requires:
 - (i) a term has the meaning assigned to it;
 - (ii) "or" is not exclusive;
- (iii) words in the singular include the plural, and in the plural include the singular;
- (iv) all references herein to Articles, Sections and other subsections, unless referring specifically to the Plan or provisions of the Bankruptcy Code, the Bankruptcy Rules, or other law, statute or regulation, refer to the corresponding Articles, Sections and other subsections of this Trust Agreement;
- (v) the words "hereof," "herein," "hereunder" and similar words refer to this Trust Agreement as a whole and not to any particular provision, Article, Section or subsection of this Trust Agreement unless otherwise specified;
- (vi) words importing persons shall include firms, associations, corporations and other entities;
- (vii) any pronoun shall include the corresponding masculine, feminine and neuter forms; and
 - (viii) "including" means including without limitation.
- 13.10. <u>Entire Trust Agreement</u>. This Trust Agreement contains the entire agreement between the parties and supersedes all prior and contemporaneous agreements or understandings between the parties with respect to the subject matter hereof.
- 13.11. <u>Cooperation</u>. The Debtors shall turn over or otherwise make available to the Trust Administrator at no cost to the Trust or the Trust Administrator, all books and records reasonably required by the Trust Administrator to carry out its duties hereunder, and agree to otherwise reasonably cooperate with the Trust Administrator in carrying out its duties hereunder, subject to the obligation to preserve the confidential nature of the Debtors' books and records, as provided in Section 13.12.

- 13.12. <u>Confidentiality</u>. The Trust Administrator and the Trust Monitor, and their respective employees, members, agents, professionals and advisors, including the Trust Professionals (each a "<u>Confidential Party</u>" and collectively the "<u>Confidential Parties</u>") shall hold strictly confidential and not use for personal gain any material, non-public information of which they have become aware in their capacity as a Confidential Party, of or pertaining to any Debtor to which any of the Avoidance Action Trust Assets relates or which is otherwise received from the Debtors by the Trust; <u>provided</u>, <u>however</u>, that such information may be disclosed if
- (i) it is now or in the future becomes generally available to the public other than as a result of a disclosure by the Confidential Parties; or
- (ii) such disclosure is required of the Confidential Parties pursuant to legal process, including subpoena or other court order or other applicable laws or regulations.

In the event that any Confidential Party is requested to divulge confidential information pursuant to clause (ii), such Confidential Party shall promptly, in advance of making such disclosure, provide reasonable notice of such required disclosure to the Trust Administrator (or the Trust Monitor in case the Trust Administrator is the disclosing party) to allow sufficient time to object to or prevent such disclosure through judicial or other means and shall cooperate reasonably with the Trust Administrator (or the Trust Monitor, as applicable) in making any such objection, including appearing in any judicial or administrative proceeding in support of any objection to such disclosure.

13.13. Amendment and Waiver.

- (a) The Trust Administrator, with the approval of the Trust Monitor, may amend or supplement this Trust Agreement without notice to or consent of the Bankruptcy Court or any Trust Beneficiary for the purpose of (x) curing any ambiguity, omission, inconsistency or correcting or supplementing any defective provision; (y) evidencing and providing for the acceptance of the appointment of a successor Trust Administrator or Trust Monitor; or (z) making any other changes to this Trust Agreement that does not adversely affect the interests of the Trust Beneficiaries in any material respect.
- (b) The Trust Administrator may amend or supplement this Trust Agreement for any other purpose, but only on petition to, and with the approval of, the Bankruptcy Court; provided that (x) no amendment or supplement to this Trust Agreement shall be inconsistent with the purpose and intent of the Trust to dispose of in an expeditious but orderly manner the Avoidance Action Trust Assets in accordance with the terms of the Plan, the Confirmation Order and this Trust Agreement, and (y) this Trust Agreement shall not be amended in a manner that is inconsistent with the Plan in the form confirmed by the Bankruptcy Court, subject to any post-confirmation modifications to the Plan pursuant to Section 1127 of the Bankruptcy Code.
- (c) Any amendment to this Trust Agreement shall be filed with the Bankruptcy Court.
- (d) No amendment shall be made to any provision of this Trust Agreement that materially and adversely affects the rights of the DIP Lenders, the CE Capital Providers or the LW

Capital Provider without the written consent of the DIP Lenders, the CE Capital Providers or the LW Capital Provider, as applicable.

- (e) The Trust Administrator shall file any amendment to the Certificate of Trust with the Secretary of State as may be required or permitted by the Delaware Act.
- 13.14. <u>Counterparts</u>. This Trust Agreement may be executed in any number of counterparts, each of which shall be deemed an original, but all such counterparts shall together constitute but one and the same instrument. A facsimile or portable document file (PDF) signature of any party shall be considered to have the same binding legal effect as an original signature.

IN WITNESS WHEREOF, the parties hereto have executed this Trust Agreement or caused this Trust Agreement to be duly executed by their respective officers, representatives or agents, effective as of the date first above written.

By:	
Dy.	Name:
	Title:
— a	and —
	THUR J. GONZALEZ, Trust Monitor
By:	
Dy.	Name:
	Title:

WILMINGTON TRUST COMPANY,

Exhibit A

Cash Accounts:

AAT Approved Budget Cash Account
AAT Distributable Trust Assets Account
Avoidance Action Budget Sub Account
Avoidance Action Assets Sub Account
Avoidance Action Trust SEC Reporting Costs Account
CE Capital Providers Funding Account
LW Capital Provider Funding Account
Litigation Cost Advance Funding Account
Other Supplemental Cash Account
Segregated Account

Exhibit B

FORM OF CERTIFICATE OF TRUST OF MOTORS LIQUIDATION COMPANY AVOIDANCE ACTION TRUST

THIS Certificate of Trust of Motors Liquidation Company Avoidance Action Trust (the "Trust") is being duly executed and filed on behalf of the Trust by the undersigned, as trustee, to form a statutory trust under the Delaware Statutory Trust Act (12 <u>Del. C.</u> § 3801 et m.) (the "Act").

- 1. <u>Name</u>. The name of the statutory trust formed by this Certificate of Trust is Motors Liquidation Company Avoidance Action Trust.
- 2. <u>Delaware Trustee.</u> The name and business address of the trustee of the Trust with a principal place of business in the State of Delaware are Wilmington Trust Company, 1100 North Market Street, Wilmington, Delaware 19890, Attn: Corporate Trust Administration.
 - 3. <u>Effective Date.</u> This Certificate of Trust shall be effective upon filing.

IN WITNESS WHEREOF, the undersigned has duly executed this Certificate of Trust in accordance with Section 3811(a)(1) of the Act.

as T	rust Administrator and trustee
By:	
,	Name:
	Title:

WILMINGTON TRUST COMPANY,

Exhibit D

THIRD AMENDED AND RESTATED MOTORS LIQUIDATION COMPANY AVOIDANCE ACTION TRUST AGREEMENT

This THIRD AMENDED AND RESTATED MOTORS LIQUIDATION COMPANY AVOIDANCE ACTION TRUST AGREEMENT, dated as of June 29, 2017 (as it may be amended from time to time, this "Trust Agreement"), by and among Wilmington Trust Company, as trust administrator and trustee (together with any successor appointed under the terms hereof, the "Trust Administrator") of the Motors Liquidation Company Avoidance Action Trust (the "Trust") for the benefit of the Trust Beneficiaries (as defined below), and Arthur J. Gonzalez, as trust monitor (together with any predecessor or successor appointed under the terms hereof, the "Trust Monitor") of the Trust, amends and restates in its entirety the Second Amended and Restated Trust Agreement (as defined below). Capitalized terms used herein and not otherwise defined herein shall have the meanings ascribed to such terms in the Debtors' Second Amended Joint Chapter 11 Plan of liquidation pursuant to chapter 11 of Title 11 of the United States Code, 11 U.S.C. §§ 101 et seq., as amended (the "Bankruptcy Code"), dated March 18, 2011, as confirmed (including all exhibits thereto, as the same may be further amended, modified, or supplemented from time to time, the "Plan").

WITNESSETH:

WHEREAS, the Trust Administrator and the Trust Monitor are party to the Motors Liquidation Company Avoidance Action Trust Agreement, dated as of March 30, 2011, by and among Motors Liquidation Company ("MLC"), MLC of Harlem, Inc., MLCS, LLC, MLCS Distribution Corporation, Remediation and Liability Management Company, Inc., and Environmental Corporate Remediation Company, Inc. (collectively, the "Debtors"), as debtors and debtors-in-possession, Wilmington Trust Company, as Trust Administrator, and FTI Consulting, Inc., as Trust Monitor (the "Original Trust Agreement"); and

WHEREAS, each of the Debtors has, prior to the date hereof, ceased to operate and dissolved; and

WHEREAS, the Original Trust Agreement was amended and restated in its entirety, with the approval of the Bankruptcy Court (as defined below), pursuant to that certain Amended and Restated Motors Liquidation Company Avoidance Action Trust Agreement dated as of May 11, 2012 by and between the Trust Administrator and the Trust Monitor (the "First Amended and Restated Trust Agreement"); and

WHEREAS, there was a dispute between the Official Committee of Unsecured Creditors of Motors Liquidation Company (the "Committee") and the DIP Lenders (defined below) as to entitlements to Avoidance Action Proceeds, which dispute was resolved by mutual agreement (the "Litigation Settlement") pursuant to which (a) the DIP Lenders provided the Litigation Cost Advance (as defined below), (b) the DIP Lenders shall be entitled to be repaid the Litigation Cost Advance out of Distributable Trust Assets (as defined below) and (c) the DIP Lenders shall be entitled to receive thirty percent (30%) of all remaining Distributable Trust Assets and,

subject to the terms of this Trust Agreement, the holders of Allowed General Unsecured Claims (or Units) shall be entitled to receive seventy percent (70%) of all remaining Distributable Trust Assets, with each such distribution to be made at or about the same time and on a *pari passu* basis, as set forth more fully in that certain Stipulation and Agreed Order (I) Settling Disputed Entitlements of Debtor-in-Possession Lenders and Official Committee of Unsecured Creditors to Potential Term Loan Avoidance Action Proceeds, and (II) Modifying Avoidance Action Trust Agreement to Implement Settlement, executed by the Committee, the DIP Lenders and the Trust and dated July 14, 2016 (the "Settlement Agreement"); and

WHEREAS, the Trust Administrator and the Trust Monitor determined that the Avoidance Action Trust Administrative Cash and Supplemental Avoidance Action Trust Cash (each defined below) held at that time by the Trust was insufficient to satisfy projected fees, costs and expenses of the Trust, and, as an integral part of the Litigation Settlement, the DIP Lenders agreed to provide the Litigation Cost Advance to the Trust on the terms set forth in that certain agreement, dated July 14, 2016, executed by the Trust and the DIP Lenders (the "Litigation Cost Advance Agreement"), which included, among other things, that the DIP Lenders would provide the Litigation Cost Advance to fund the prosecution of the Avoidance Action; and

WHEREAS, on July 15, 2016, the Trust, jointly with the Official Committee of Unsecured Creditors (the "Committee"), filed a motion with the Bankruptcy Court (the "9019 Motion") seeking approval of the Settlement Agreement and the Litigation Cost Advance Agreement; and

WHEREAS, on August 24, 2016, the Bankruptcy Court entered an order approving the 9019 Motion (the "9019 Order"); and

WHEREAS, pursuant Section 13.13(b) of the First Amended and Restated Trust Agreement, the 9019 Motion and the 9019 Order, the Trust Administrator petitioned and received from the Bankruptcy Court approval to amend and restate in its entirety the First Amended and Restated Trust Agreement; and

WHEREAS, the First Amended and Restated Trust Agreement was amended and restated in its entirety pursuant to that certain Second Amended and Restated Motors Liquidation Company Avoidance Action Trust Agreement dated as of August 25, 2016, by and between the Trust Administrator and the Trust Monitor (the "Second Amended and Restated Trust Agreement"); and

WHEREAS, one of the preconditions to the DIP Lenders' provision of the Litigation Cost Advance was the Trust's agreement, as reflected in Section 6.1(d)(1) of the Second Amended and Restated Trust Agreement, that any additional litigation funding provided to the Trust would be (i) junior and subordinate to the Litigation Cost Advance and any other amounts owed to the DIP Lenders on account of prior funding of the Trust and (ii) subject to a form of subordination acceptable to the DIP Lenders in all respects; and

WHEREAS, on or about the date hereof, the Trust Administrator and the Trust Monitor have determined that the Avoidance Action Trust Administrative Cash and Supplemental

Avoidance Action Trust Cash currently held by the Trust remain insufficient to satisfy projected fees, costs and expenses of the Trust; and

WHEREAS, with the consent of the DIP Lenders, the Trust Administrator and Trust Monitor entered into a Capital Provision Agreement, dated June 2, 2017 (the "<u>Capital Provision Agreement</u>"), with Delaware limited liability companies Cynthiana LLC and Earlham LLC (each a "<u>Capital Provider</u>," and, together, the "<u>Capital Providers</u>"), pursuant to which the Capital Providers will provide up to \$15,000,000 in additional funding to the Trust (the "<u>Supplemental Capital</u>") as Other Supplemental Cash; and

WHEREAS, pursuant to Section 6.1(d)(1) of the Second Amended and Restated Trust Agreement, entry into that certain Subordination Agreement (the "Subordination Agreement"), dated as of June 2, 2017, by and among the DIP Lenders, the Capital Providers, and the Trust, was a precondition of the DIP Lenders' consent to the Capital Providers' provision of the Supplemental Capital to the Trust; and

WHEREAS, on June 7, 2017, the Trust moved for entry of an order, pursuant to sections 105 and 1142 of title 11 of the United States Code: (1) approving the amendment and restatement of the Second Amended Avoidance Action Trust Agreement, (2) authorizing the Avoidance Action Trust to enter into the Capital Provision Agreement and to grant a lien to the Capital Providers, and (3) authorizing the Avoidance Action Trust and Avoidance Action Trust Administrator to use \$1,750,759.93 of the Holdback to fund necessary fees, costs and expenses of the Avoidance Action Trust (the "Capital Provision Motion"); and

WHEREAS, on June 29, 2017, the Bankruptcy Court entered an order granting the Capital Provision Motion (the "Capital Provision Order"); and

WHEREAS, pursuant Section 13.13(b) of the Second Amended and Restated Trust Agreement, the Capital Provision Motion and the Capital Provision Order, the Trust Administrator petitioned and received from the Bankruptcy Court approval to amend and restate in its entirety the Second Amended and Restated Trust Agreement.

NOW, THEREFORE, in accordance with Section 13.13(b) of the Second Amended and Restated Trust Agreement and the Capital Provision Agreement, the Second Amended and Restated Trust Agreement is hereby amended and restated as follows:

Background

- A. Beginning on June 1, 2009, the Debtors filed in the United States Bankruptcy Court for the Southern District of New York (the "Bankruptcy Court") voluntary petitions for relief under chapter 11 of the Bankruptcy Code (the "Chapter 11 Cases").
- B. On or about August 31, 2010, the Debtors filed their Plan and Disclosure Statement in the Bankruptcy Court. The Debtors filed an amended Plan and Disclosure Statement on December 7, 2010. The Debtors filed a second amended Plan on March 18, 2011.

- C. The Disclosure Statement was approved by the Bankruptcy Court on December 8, 2010.
- D. On or about March 29, 2011, the Bankruptcy Court entered an order (the "Confirmation Order") confirming the Plan.
- E. The Plan provides for the creation of the Trust as a post-confirmation successor to MLC within the meaning of Section 1145(a) of the Bankruptcy Code, to hold and administer the Avoidance Action Trust Assets for the benefit of the Trust Beneficiaries and, to the extent received by the Trust, to distribute the Distributable Trust Assets to the Trust Beneficiaries in accordance with the terms of the Plan, the Confirmation Order and this Trust Agreement.
- F. The Plan also provides that the Trust Administrator may determine to hold and administer Other Debtor Residual Trust Assets, if any, for the benefit of the DIP Lenders.
- G. The Trust was created, with respect to the Avoidance Action Trust Assets, on behalf of, and for the benefit of, the Trust Beneficiaries, and, with respect to the Other Debtor Residual Trust Assets, if any, on behalf of, and for the benefit of, the DIP Lenders.
- The Trust Administrator shall have all powers necessary to implement the provisions of this Trust Agreement and administer the Trust in respect of the Avoidance Action Trust Assets, including the power to: (i) prosecute for the benefit of the Trust Beneficiaries, through counsel and other professionals selected by the Trust Administrator, the Term Loan Avoidance Action and any other causes of action that may from time to time be held by the Trust in respect thereof; (ii) preserve and maintain the Avoidance Action Trust Assets; (iii) distribute the Distributable Trust Assets, if any, to the Trust Beneficiaries in accordance with the Plan, the Confirmation Order and this Trust Agreement; (iv) expend the Avoidance Action Trust Administrative Cash, the Avoidance Action Trust SEC Reporting Cash (if applicable), and the Supplemental Avoidance Action Trust Cash to cover fees and expenses of the Trust, including payment of taxes and the fees and expenses of the Trust Professionals (other than in respect thereof of the Other Debtor Residual Trust Assets); (v) enter into any contracts or agreements necessary or desirable to facilitate the implementation of the provisions of this Trust Agreement, including but not limited to loan agreements or sale agreements for the purposes of funding the fees and expenses of the Trust; and (vi) sell or grant liens on the Term Loan Avoidance Action or any other property of the Trust (other than the Other Debtor Residual Trust Assets) pursuant to the terms of this Trust Agreement for the purposes of implementing the provisions of this Trust Agreement and the 9019 Order, including but not limited to funding the fees and expenses of the Trust.
- I. The Trust Administrator shall also have all powers necessary to implement the provisions of this Trust Agreement and administer the Trust in respect of the Other Debtor Residual Trust Assets, if any, including the power to: (i) prosecute for the benefit of the DIP Lenders, through counsel and other professionals selected by the Trust Administrator, any causes of action that may from time to time be held by the Trust in respect thereof; (ii) preserve and maintain the Other Debtor Residual Trust Assets; (iii) distribute the Distributable Other Debtor Residual Trust Assets, if any, to the DIP Lenders in accordance with the Plan, the Confirmation Order and this Trust Agreement; and (iv) expend the Other Debtor Residual Trust Administrative

Cash to cover fees and expenses of the Trust, including payment of the fees and expenses of the Trust Professionals, in respect thereof.

- J. The Trust Administrator shall otherwise perform the functions and take the actions provided for in this Trust Agreement or permitted in the Plan and/or the Confirmation Order, or in any other agreement executed pursuant to the Plan, in each case subject to the provisions of <u>Articles VI</u>, <u>VIII</u> and <u>XI</u> hereof regarding the rights and powers of the Trust Monitor.
- K. The Trust is subject to the continuing jurisdiction of the Bankruptcy Court, whose approval is required to pay or distribute money or property to, or on behalf of, a Trust Beneficiary, except as expressly provided in this Trust Agreement.
- L. The Trust (other than the Avoidance Action Trust Claims Reserve) is intended to qualify as a liquidating trust under Treasury Regulation section 301.7701-4(d) that is treated as a "grantor trust" for federal and applicable state and local income tax purposes.

Agreement

NOW, THEREFORE, in consideration of the promises and the mutual covenants contained herein, the Trust Administrator and the Trust Monitor agree as follows:

ARTICLE I DEFINED TERMS

- 1.1. <u>Definitions</u>. Whenever used in this Trust Agreement, unless the context otherwise requires, the following words and phrases shall have the respective meanings ascribed to them as follows:
- (a) "9019 Motion" has the meaning set forth in the preamble to this Trust Agreement.
- (b) "9019 Order" has the meaning set forth in the preamble to this Trust Agreement.
- (c) "<u>Affiliates</u>" means, with respect to any Person, any other Person which directly or indirectly controls, is controlled by or is under common control with such Person. For purposes of this definition "control" means, with respect to any Person, the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of such Person, whether through ownership of voting securities, by contract or otherwise.
- (d) "<u>Aggregate Maximum Amount</u>" means the sum of the Maximum Amounts of all Disputed General Unsecured Claims, Unresolved Term Loan Avoidance Action Claims and Unresolved Other Avoidance Action Claims, as set forth in the applicable GUC Trust Report as the Aggregate Maximum Amount as of a given date.

- (e) "Agreed Allocation" means the DIP Lenders shall be entitled to receive thirty percent (30%), and holders of Allowed General Unsecured Claims (or Units) shall be entitled to receive seventy percent (70%), with each such distribution to be made at or about the same time and on a *pari passu* basis.
- (f) "<u>Allowed General Unsecured Claims</u>" means, collectively, (i) the Initial Allowed General Unsecured Claims and (ii) the Resolved Allowed General Unsecured Claims.
- (g) "<u>Avoidance Action Proceeds</u>" means the proceeds of the Term Loan Avoidance Action.
- (h) "Avoidance Action Trust Administrative Cash" means the Trust Cash (other than the Supplemental Avoidance Action Trust Cash and the Avoidance Action Trust SEC Reporting Cash), including the fifteen million (\$15,000,000) Litigation Cost Advance and the \$1.6 million initially received from the DIP Lenders by the Debtors on the Avoidance Action Trust Transfer Date, held and maintained by the Trust Administrator for the purpose of paying the fees and expenses incurred by the Trust Administrator (including fees and expenses for Trust Professionals) in connection with the Trust and any obligations imposed on the Trust Administrator or the Trust, including fees and expenses relating to the performance of the Trust Administrator's obligations under this Trust Agreement and the Plan, other than in respect of the Other Debtor Residual Trust Assets.
- (i) "Avoidance Action Trust Assets" means, collectively, (i) the Term Loan Avoidance Action transferred to the Trust, (ii) the Avoidance Action Proceeds, (iii) the Avoidance Action Trust Administrative Cash, (iv) the Avoidance Action Trust SEC Reporting Cash; (v) the Supplemental Avoidance Action Trust Cash; (vi) the Litigation Cost Advance; and (vii) the Supplemental Capital provided by the Capital Providers.
- (j) "<u>Avoidance Action Trust SEC Reporting Costs</u>" means any costs, fees or expenses incurred by the Trust that are directly or indirectly related to reports that may be required to be filed by the Trust with the SEC pursuant to applicable rules, regulations and interpretations of the SEC (including, without limitation, any legal, accounting or registration fees, costs and expenses incurred by the Trust with respect thereto).
- (k) "<u>Avoidance Action Trust SEC Reporting Cash</u>" has the meaning set forth in Section 2.3(e) of this Trust Agreement.
- (l) "<u>Avoidance Action Trust Transfer Date</u>" means the date selected by the Debtors on which the Avoidance Action Trust Assets (other than the Supplemental Avoidance Action Trust Cash and the Avoidance Action Proceeds) are transferred to the Trust, which transfer shall occur on or before December 15, 2011.
- (m) "<u>Bankruptcy Code</u>" has the meaning set forth in the preamble to this Trust Agreement.

- (n) "<u>Bankruptcy Court</u>" has the meaning set forth in Background paragraph A.
 - (o) "Budget" has the meaning set forth in Section 6.3.
- (p) "<u>calendar quarter</u>" means the relevant three-month period ending on the last day of March, June, September or December, as applicable, of each calendar year; <u>provided</u>, <u>however</u>, that the calendar quarter that contains the Avoidance Action Trust Transfer Date shall be the period commencing on the Avoidance Action Trust Transfer Date and concluding on the date on which the relevant calendar quarter would naturally end in accordance with the foregoing.
- (q) "<u>Capital Provider</u>" and "<u>Capital Providers</u>" have the meanings set forth in the preamble to this Trust Agreement.
- (r) "Capital Provider Funding Account" means that segregated account. in the name of the Trust, established at Wilmington Trust Company, in which the Trust shall hold the Supplemental Capital. For the avoidance of doubt, the Capital Provider Funding Account shall contain Other Supplemental Cash as defined in this Trust Agreement.
- (s) "<u>Capital Providers' Entitlement</u>" means the funds to which the Capital Providers are entitled pursuant to the Capital Provision Agreement.
- (t) "Capital Provision Agreement" has the meaning set forth in the preamble to this Trust Agreement, as the same may be amended, restated, supplemented or modified from time to time after the date hereof.
- (u) "Capital Provision Motion" has the meaning set forth in the preamble to this Trust Agreement.
- (v) "<u>Capital Provision Order</u>" has the meaning set forth in the preamble to this Trust Agreement.
- (w) "<u>Certificate of Trust</u>" means the certificate of trust of the Trust as required by Section 3810 of the Delaware Act.
- (x) "<u>Chapter 11 Cases</u>" has the meaning set forth in Background paragraph A.
 - (y) "Claim Conflict Resolution" has the meaning set forth in Section 3.6.
- (z) "Confidential Party" and "Confidential Parties" have the meanings set forth in Section 13.12.
- (aa) "<u>Confirmation Order</u>" has the meaning set forth in Background paragraph D.

- (bb) "<u>Current Total Amount</u>" means as of a given date, the sum of (A) the Total Allowed Amount as of such date and (B) the Aggregate Maximum Amount as of such date, as set forth in the applicable GUC Trust Report as the Current Total Amount as of a given date.
- (cc) "<u>Debtors</u>" has the meaning set forth in the preamble to this Trust Agreement.
- (dd) "Delaware Act" means the Delaware Statutory Trust Act, 12 Del. C. \S 3801 et seq.
- (ee) "<u>DIP Credit Agreement Claims</u>" means all Claims arising under the DIP Credit Agreement and Orders approving the DIP Credit Agreement dated June 25, 2009 and July 5, 2009.
- (ff) "<u>DIP Lender Advances</u>" means an amount equal to (i) the amounts of Cash advanced directly or indirectly by MLC and the DIP Lenders to fund the costs and expenses associated with realizing the proceeds of the Term Loan Avoidance Action, including, without limitation, any such amounts expended to fund the costs and expenses of professionals retained by the defendants in the Term Loan Avoidance Action, (ii) without duplication, the amount of the Avoidance Action Trust Administrative Cash, and (iii) without duplication, the Litigation Cost Advance.
- (gg) "<u>DIP Lender Distributable Trust Assets</u>" means the assets distributable to the DIP Lenders, in accordance with the Agreed Allocation and the Settlement Agreement.
- (hh) "<u>DIP Lenders</u>" means the U.S. Treasury and Export Development Canada, as lenders under the DIP Credit Agreement.
- (ii) "<u>Disputed General Unsecured Claims</u>" means the General Unsecured Claims against the Debtors that are Disputed (as defined in the Plan) as of the Initial GUC Record Date, until a time when such claims become Resolved General Unsecured Claims or are otherwise resolved pursuant to the claims resolution procedures contained in the Plan.
- (jj) "<u>Distributable Other Debtor Residual Trust Assets</u>" means the Other Debtor Residual Assets Proceeds, if any, together with any earnings (including interest) thereon.
- (kk) "<u>Distributable Other Debtor Residual Trust Cash</u>" means any Cash or cash equivalents included in the Distributable Other Debtor Residual Trust Assets.
- (ll) "<u>Distributable Trust Assets</u>" means the Avoidance Action Proceeds together with any earnings (including interest) thereon.
- (mm) "<u>Distributable Trust Cash</u>" means any Cash or cash equivalents included in the Distributable Trust Assets.

- (nn) "<u>Distribution Date</u>" means the date of any distribution made by the Trust Administrator to the Trust Beneficiaries pursuant to this Trust Agreement, whether to the DIP Lenders pursuant to <u>Section 5.1(d)</u> or on account of Allowed General Unsecured Claims and/or Units.
 - (oo) "<u>Distribution Threshold</u>" means \$10,000,000.
- (pp) "<u>Excess Distributable Trust Assets</u>" means Distributable Trust Assets that the Trust Administrator, with the approval of the Trust Monitor, determines shall be distributed following the Initial GUC Distribution Date.
- (qq) "Excess GUC Distributable Trust Assets Determination Date" has the meaning set forth in Section 5.4(a).
- (rr) "Excess GUC Distributable Trust Assets" has the meaning set forth in Section 5.4(c).
 - (ss) "Final Recovery Date" has the meaning set forth in Section 5.1(a).
- (tt) "<u>First Amended and Restated Trust Agreement</u>" has the meaning set forth in the preamble to this Trust Agreement.
- (uu) "<u>Funding Account</u>" means that segregated account, in the name of the Trust, established at a bank reasonably acceptable to the Trust and the DIP Lenders, in which the Trust shall hold the Litigation Cost Advance.
- (vv) "<u>GUC Beneficiaries</u>" means the holders of Allowed General Unsecured Claims or Units received in respect of such claims."
- (ww) "GUC Distributable Trust Assets" means: (i) the Segregated Account Assets; and (ii) the assets distributable to the holders of Allowed General Unsecured Claims in accordance with the Agreed Allocation and the Settlement Agreement.
- (xx) "GUC Trust Advances" means the Distributable Trust Assets distributable to the holders of Allowed General Unsecured Claims, in an amount equal to the aggregate amount of GUC Trust Supplemental Cash received from the GUC Trust pursuant to Section 2.3(f)(i) hereof, together with any earnings (including interest) thereon, minus the aggregate amount of Cash held in the Segregated Account on the date of measurement.
- (yy) "<u>GUC Trust Reports</u>" means the reports prepared by the GUC Trust Administrator each quarter as provided in the GUC Trust Agreement, which shall be delivered to the Trust Administrator pursuant to the terms of the GUC Trust Agreement.
- (zz) " $\underline{GUC\ Trust\ Supplemental\ Cash}$ " has the meaning set forth in $\underline{Section}$ 2.3(f)(i).
 - (aaa) "Holdback" has the meaning set forth in Section 6.1(b).

- (bbb) "<u>Incompetency</u>" means, with respect to any Person, the incompetency of such Person is a natural person.
- (ccc) "<u>Initial Allowed General Unsecured Claims</u>" has the meaning set forth in <u>Section 5.2(a)</u>.
- (ddd) "<u>Initial GUC Distribution Date</u>" has the meaning set forth in <u>Section 5.2(a)</u>.
 - (eee) "Initial GUC Record Date" has the meaning set forth in Section 5.2(a).
 - (fff) "IRS" means the Internal Revenue Service.
- (ggg) "<u>Litigation Cost Advance</u>" means the fifteen million dollars (\$15,000,000) to be provided by the DIP Lenders pursuant to the Litigation Cost Advance Agreement and the Settlement Agreement and subject to a deposit account control agreement acceptable to the DIP Lenders and the Trust.
- (hhh) "<u>Litigation Cost Advance Agreement</u>" has the meaning set forth in the preamble to this Trust Agreement.
- (iii) "<u>Litigation Settlement</u>" has the meaning set forth in the preamble to this Trust Agreement.
- (jjj) "Maximum Amount" means the maximum amount of any Disputed General Unsecured Claim, Unresolved Term Loan Avoidance Action Claim or Unresolved Other Avoidance Action Claim, calculated in accordance with the GUC Trust Agreement and as set forth in the applicable GUC Trust Report as the Maximum Amount of any Claim or group of Claims as of a given date.
- (kkk) "<u>MLC</u>" has the meaning set forth in the preamble to this Trust Agreement.
- (lll) "Original Trust Agreement" has the meaning set forth in the preamble to this Trust Agreement.
- (mmm) "Other Avoidance Action Claims" means the additional General Unsecured Claims that have arisen as a result of recovery of proceeds of the Avoidance Actions other than the Term Loan Avoidance Action (and any related unsecured claims).
- (nnn) "Other Debtor Residual Assets" means any assets of MLC remaining at such time as the Debtors shall be liquidated, other than the Term Loan Avoidance Action and the Term Loan Avoidance Action Administrative Cash and any other assets of MLC whose disposition is specifically provided for under the Plan or the Confirmation Order.
- (000) Other Debtor Residual Accepted Assets" means Other Debtor Residual Assets accepted by the Trust Administrator for transfer to the Trust pursuant to Section 2.3(b).

- (ppp) "Other Debtor Residual Assets Proceeds" means any proceeds realized in respect of the Other Debtor Residual Accepted Assets.
- (qqq) "Other Debtor Residual Trust Administrative Cash" means the Cash, if any, held and maintained by the Trust Administrator for the purpose of paying the fees and expenses incurred by the Trust Administrator (including fees and expenses for Trust Professionals) in connection with the Trust and any obligations imposed on the Trust Administrator or the Trust, including fees and expenses relating to the performance of the Trust Administrator's obligations under this Trust Agreement and the Plan, but only in respect of the Other Debtor Residual Trust Assets, which Cash may be obtained by transfer to the Trust by the Debtors, from the DIP Lenders (in their sole discretion) or from the proceeds of the Other Debtor Residual Accepted Assets.
- (rrr) "Other Debtor Residual Trust Assets" means, if any, collectively, (i) the Other Debtor Residual Accepted Assets transferred to the Trust, (ii) the Other Debtor Residual Assets Proceeds and (iii) the Other Debtor Residual Trust Administrative Cash.
- (sss) "Other Debtor Residual Assets Transfer Date" means the date selected by the Debtors on which the Other Debtor Residual Assets Transfer, if any, are transferred to the Trust, which transfer shall occur on or before December 15, 2011.
- (ttt) "Other Supplemental Cash" means any Cash received by the Trust pursuant to Section 6.1(d) hereof from the sale of, or granting of liens on, all or a portion of the Term Loan Avoidance Action or any other property held by the Trust (other than the Other Debtor Residual Trust Assets), or from any source other than the GUC Trust, including the Supplemental Capital.
 - (uuu) "Permissible Investments" means investments in any of the following:
- (i) Marketable securities issued by the U.S. Government and supported by the full faith and credit of the U.S. Treasury, either by statute or an opinion of the Attorney General of the United States;
- (ii) Marketable debt securities, rated Aaa by Moody's and/ or AAA by S&P, issued by U. S. Government-sponsored enterprises, U. S. Federal agencies, U. S. Federal financing banks, and international institutions whose capital stock has been subscribed for by the United States;
- (iii) Certificates of deposit, time deposits, and bankers acceptances of any bank or trust company incorporated under the laws of the United States or any state, provided that, at the date of acquisition, such investment, and/or the commercial paper or other short term debt obligation of such bank or trust company has a short-term credit rating or ratings from Moody's and/or S&P, each at least P-1 or A-1;
- (iv) Commercial paper of any corporation incorporated under the laws of the United States or any state thereof which on the date of acquisition is rated by Moody's and/or S&P, provided each such credit rating is least P-1 and/or A-1;

- (v) Money market mutual funds that are registered with the Securities and Exchange Commission under the Investment Company Act of 1940, as amended, and operated in accordance with Rule 2a-7 and that at the time of such investment are rated Aaa by Moody's and/or AAAm by S&P, including such funds for which the Trust Administrator or an Affiliate provides investment advice or other services;
- (vi) Tax-exempt variable rate commercial paper, tax-exempt adjustable rate option tender bonds, and other tax-exempt bonds or notes issued by municipalities in the United States, having a short-term rating of "MIG-1" or "VMIG-1" or a long term rating of "AA" (Moody's), or a short-term rating of "A-1" or a long term rating of "AA" (S&P); and
- (vii) Repurchase obligations with a term of not more than thirty days, 102 percent collateralized, for underlying securities of the types described in clauses (i) and (ii) above, entered into with any bank or trust company or its respective affiliate meeting the requirements specified in clause (iii) above.
- (vvv) "<u>Plan</u>" has the meaning set forth in the preamble to this Trust Agreement.
- (www) "Resolved Allowed General Unsecured Claims" means, collectively, (I) the Disputed General Unsecured Claims that are allowed after the Initial GUC Record Date in accordance with the claims resolution procedures administered under the Plan (to the extent so resolved); (II) the Term Loan Avoidance Action Claims, to the extent and in the amount collected by the Trust against the respective defendants in the underlying litigation (including by way of settlement); and (III) the Other Avoidance Action Claims, to the extent and in the amount collected against the respective defendants in the underlying litigations (including by way of settlement). For the avoidance of doubt, unless and until a Disputed General Unsecured Claim, Unresolved Term Loan Avoidance Action Claim or Unresolved Other Avoidance Action Claim becomes a Resolved Allowed General Unsecured Claim, the holders of such claim shall not receive any distribution from the Trust.
- (xxx) "Resolved Allowed General Unsecured Claims Determination Date" has the meaning set forth in Section 5.3(a).
 - (yyy) RESERVED
 - (zzz) RESERVED
 - (aaaa) "SEC" means the Securities and Exchange Commission.
- (bbbb) "Second Amended and Restated Trust Agreement" has the meaning set forth in the preamble to this Trust Agreement.
- (cccc) "<u>Secretary of State</u>" means the Office of the Secretary of State of the State of Delaware.
 - (dddd) "Segregated Account" has the meaning set forth in Section 2.3(f)(i).

- (eeee) "Segregated Account Assets" means the Cash held in the Segregated Account.
- (ffff) "<u>Settlement Agreement</u>" has the meaning set forth in the preamble to this Trust Agreement.
- (gggg)"Supplemental Avoidance Action Trust Cash" means the GUC Trust Supplemental Cash and the Other Supplemental Cash.
- (hhhh)"Supplemental Capital" has the meaning set forth in the preamble to this Trust Agreement.
- (iiii) "<u>Tax Returns</u>" means all tax returns, reports, certificates, forms or similar statements or documents.
- (jjjj) "Term Loan Avoidance Action" means the Avoidance Action commenced by the Creditors' Committee against JPMorgan Chase Bank, N.A., individually and as Administrative Agent, and various lenders party to a term loan agreement, dated as of November 29, 2006, between General Motors Corporation, as borrower, JPMorgan Chase Bank, N.A., as agent, and various institutions as lenders and agents, styled *Official Committee of Unsecured Creditors of Motors Liquidation Co. v. JPMorgan Chase Bank, N.A. et al.*, Adv. Pro. No. 09-00504 (Bankr. S.D.N.Y. July 31, 2009).
- (kkkk)"<u>Term Loan Avoidance Action Claims</u>" means the additional General Unsecured Claims that have arisen as a result of recovery of proceeds of the Term Loan Avoidance Action (or any related unsecured claims).
- (Illl) "<u>Total Allowed Amount</u>" means the sum of the amount of all Initial Allowed General Unsecured Claims plus the amount of all Resolved Allowed General Unsecured Claims, as set forth in the applicable GUC Trust Report.
- (mmmm) "<u>Treasury Regulations</u>" means the income tax regulations promulgated under the Tax Code, including any amended or successor income tax regulations thereto.
- (nnnn)"<u>Trust</u>" has the meaning set forth in the preamble to this Trust Agreement.
- (0000)"<u>Trust Administrator</u>" has the meaning set forth in the preamble to this Trust Agreement.
- (pppp)"<u>Trust Administrator Parties</u>" means the Trust Administrator and its principals, directors, officers, employees, agents, representatives, attorneys, accountants, advisors and other professionals (including the Trust Professionals).
- (qqqq)"<u>Trust Agreement</u>" has the meaning set forth in the preamble to this Trust Agreement.

- (rrrr) "<u>Trust Beneficiaries</u>" means the holders of the DIP Credit Agreement Claims and the holders of Allowed General Unsecured Claims (or Units received in respect of such claims).
- (ssss) "<u>Trust Cash</u>" means the Cash or cash equivalents included in the Avoidance Action Trust Assets or the Other Debtor Residual Trust Assets, if any.
- (tttt) "<u>Trust Monitor</u>" has the meaning set forth in the preamble to this Trust Agreement.
- (uuuu) "<u>Trust Monitor Parties</u>" means the Trust Monitor and its principals, directors, officers, employees, agents, representatives, attorneys, accountants, advisors and other professionals.
- (vvvv) "<u>Trust Professionals</u>" means, collectively, independent contractors, including attorneys, accountants, appraisers, disbursing agents or other parties deemed by the Trust Administrator to have the qualifications necessary or desirable to assist in the proper administration of the Trust and that are employed or retained by the Trust in such capacities.
- (wwww) "<u>Unit Issuance Ratio</u>" means the ratio of one Unit for each \$1,000 in amount of Allowed General Unsecured Claims.
- (xxxx)"<u>Units</u>" means the units of beneficial interest issued by the Trust to holders of Allowed General Unsecured Claims.
- (yyyy) "<u>Unresolved Other Avoidance Action Claim</u>" means an Other Avoidance Action Claim that has not or has not yet arisen because no determination (including by way of settlement) has been made in the respective Avoidance Action against the respective defendant who would be entitled to such claim in the event of such determination (or if a determination has been made against the defendant, the proceeds related to such resolution have not been recovered in full).
- (zzzz) "<u>Unresolved Term Loan Avoidance Action Claim</u>" means a Term Loan Avoidance Action Claim that has not or has not yet arisen because no determination (including by way of settlement) has been made in the Term Loan Avoidance Action against the respective defendant who would be entitled to such claim in the event of such determination (or if a determination has been made against the defendant, the proceeds related to such resolution have not been recovered in full).
- (aaaaa) "<u>Wind-Down Facility</u>" means the \$1.175 billion wind-down facility provided to the Debtors pursuant to the DIP Credit Agreement.

ARTICLE II DECLARATION OF TRUST

2.1. <u>Creation of Trust</u>. The Debtors and the Trust Administrator, pursuant to the Plan and the Confirmation Order and in accordance with the applicable provisions of chapter 11 of the Bankruptcy Code, hereby constitute and create the Trust, in the form of a

statutory trust under the Delaware Act, which shall bear the name "Motors Liquidation Company Avoidance Action Trust." In connection with the exercise of the Trust Administrator's power hereunder, the Trust Administrator may use this name or such variation thereof as the Trust Administrator sees fit. The Trust Administrator, as trustee of the Trust, is hereby authorized and directed to execute and file a Certificate of Trust for the Trust in the form attached hereto as Exhibit B.

2.2. <u>Purpose of Trust</u>. The sole purpose of the Trust is to liquidate and distribute its assets pursuant to the Plan in accordance with Treasury Regulation section 301.7701-4(d), with no objective to continue or engage in the conduct of a trade or business.

2.3. Transfer of Avoidance Action Trust Assets to the Trust.

- (a) Effective upon the Avoidance Action Trust Transfer Date, the Debtors hereby transfer to the Trust, pursuant to Bankruptcy Code Sections 1123(a)(5)(B) and 1123(b)(3)(B), and in accordance with the Plan and the Confirmation Order, the Avoidance Action Trust Assets (other than the Avoidance Action Trust SEC Reporting Cash, and the Supplemental Avoidance Action Trust Cash), as they exist on the Avoidance Action Trust Transfer Date, free and clear of any and all liens, claims, encumbrances and interests (legal, beneficial or otherwise) of all other persons to the maximum extent contemplated by and permissible under Bankruptcy Code Section 1141(c); provided, however that notwithstanding anything to the contrary in the Plan, Disclosure Statement, Confirmation Order, this Trust Agreement or any other agreement, the DIP Lenders shall maintain their liens on the Avoidance Action Trust Administrative Cash, provided that for the avoidance of doubt, the DIP Lenders shall not demand acceleration of their liens on the Avoidance Action Trust Administrative Cash except in accordance with the provisions of section 7.2 of the DIP Credit Agreement. Such transfers shall be exempt from any stamp, real estate transfer, mortgage reporting, sales, use or other similar tax pursuant to Bankruptcy Code Section 1146. The Debtors and their successors and assigns shall be released from any and all liability with respect to the transfer of such Avoidance Action Trust Assets to the Trust as aforesaid. Nothing in this Trust Agreement is intended to, or shall be construed to, effect a release, extinguishment or compromise of any claim or cause of action transferred to the Trust pursuant to this Trust Agreement. The Avoidance Action Trust Assets and all other property held from time to time by the Trust under this Trust Agreement (other than the Other Debtor Residual Trust Assets) and any earnings (including interest) thereon are to be managed, applied and disposed of by the Trust Administrator in accordance with the terms hereof, the Plan and the Confirmation Order for the benefit of the Trust Beneficiaries and the Capital Providers, and for no other party, subject to the further covenants, conditions and terms hereinafter set forth, including the provisions of Section 2.6.
- (b) To the extent any Avoidance Action Trust Assets (other than the Supplemental Avoidance Action Trust Cash) cannot be transferred to the Trust, because of a restriction on transferability under applicable non-bankruptcy law that is not superseded by Bankruptcy Code Section 1123 or any other provision of the Bankruptcy Code, such assets shall be retained by the Debtors or any successor thereto including, without limitation, the GUC Trust. The proceeds of the sale of any such assets retained by the Debtors (or any

successor thereto) shall be allocated to the Trust pursuant to the Plan as if such transfer had not been restricted under applicable non-bankruptcy law. The Trust Administrator may commence an action in the Bankruptcy Court to resolve any dispute regarding the allocation of the proceeds of any assets retained by the Debtors (or any successor thereto) pursuant to the Plan and Confirmation Order.

- (c) On the Avoidance Action Trust Transfer Date, the Debtors shall also deliver, or cause to be delivered, to the Trust a complete list of all General Unsecured Claims, both Allowed and Disputed, reflected on the claims registry as of the Avoidance Action Trust Transfer Date, including the names and addresses of all holders of such General Unsecured Claims, whether such claims have been Allowed or are Disputed, and the details of all objections in respect of Disputed General Unsecured Claims.
- (d) Effective upon the Other Debtor Residual Assets Transfer Date, the Debtors hereby transfer to the Trust, pursuant to Bankruptcy Code Sections 1123(a)(5)(B) and 1123(b)(3)(B), and in accordance with the Plan and the Confirmation Order, such of the Other Debtor Residual Assets, as they exist on the Other Debtor Residual Assets Transfer Date, as the Trust Administrator, in its sole discretion but with the approval of the Trust Monitor, shall determine to accept, free and clear of any and all liens, claims, encumbrances and interests of all other persons to the maximum extent contemplated by and permissible under Bankruptcy Code Section 1141(c); provided that, for the avoidance of doubt, the Trust Administrator may determine not to accept the transfer to the Trust of any or all of the Other Debtor Residual Assets for any reason or for no reason. Any such transfer shall be exempt from any stamp, real estate transfer, mortgage reporting, sales, use or other similar tax pursuant to Bankruptcy Code Section 1146. The Debtors and their successors and assigns shall be released from any and all liability with respect to the transfer of the Other Debtor Residual Accepted Assets to the Trust as aforesaid. Nothing in this Trust Agreement is intended to, or shall be construed to, effect a release, extinguishment or compromise of any claim or cause of action transferred to the Trust pursuant to this Trust Agreement, and notwithstanding anything to the contrary in the Plan, Disclosure Statement, Confirmation Order, this Trust Agreement or any other agreement, the DIP Lenders shall maintain their liens on the Other Debtor Residual Accepted Assets. The Other Debtor Residual Trust Assets and all other property held from time to time by the Trust under this Trust Agreement in respect thereof, and any earnings (including interest) thereon, are to be managed, applied and disposed of by the Trust Administrator in accordance with the terms hereof, the Plan and the Confirmation Order for the benefit of the DIP Lenders, and for no other party, subject to the further covenants, conditions and terms hereinafter set forth, including the provisions of Section 2.6.
- (e) (i) On the Avoidance Action Trust Transfer Date, the Debtors shall, pursuant to Section 2.3(e) of the GUC Trust Agreement, transfer Cash to the Trust in an amount of \$500,000 (the "Avoidance Action Trust SEC Reporting Cash"). The Avoidance Action Trust SEC Reporting Cash shall be held by the Trust in a segregated account and shall be used solely for the satisfaction of Avoidance Action Trust SEC Reporting Costs. Any taxes imposed on the Trust in respect of the Avoidance Action Trust SEC Reporting Cash shall be satisfied from the income realized thereon.

- (ii) The Trust Administrator shall only use Avoidance Action Trust SEC Reporting Cash to satisfy Avoidance Action Trust SEC Reporting Costs to extent there is no other available source of funds to satisfy such expenses (other than the Supplemental Avoidance Action Trust Cash), including, without limitation, any funds obtained through the reservation and application of all or a portion of the Holdback pursuant to Section 6.1(b) hereof.
- (iii) If the Trust Administrator determines that (x) reports are not, and at no time will be, required to be filed by the Trust with the SEC pursuant to applicable rules, regulations and interpretations of the SEC or (y) the Trust has other available funds (as set forth in Section 2.3(e)(ii) hereof) which are sufficient to satisfy any current or future projected, fees, costs or expenses that are directly or indirectly related to reports that may be required to be filed by the Trust with the SEC pursuant to applicable rules, regulations and interpretations of the SEC (including, without limitation, any legal, accounting or registration fees, costs and expenses incurred by the Trust with respect thereto), then the Trust shall transfer to the GUC Trust any Avoidance Action Trust SEC Reporting Cash that has not been applied as provided in this Section 2.3(e).
- (iv) Any income earned on the Avoidance Action Trust SEC Reporting Cash, net of taxes paid thereon, shall be Avoidance Action Trust Administrative Cash.
- (f) (i) From time to time, in accordance with a Final Order of the Bankruptcy Court if so required, the GUC Trust may deliver Cash to the Trust to be used solely for the satisfaction of fees and expenses of the Trust (including for the payment of the fees and expenses of Trust Professionals) or the satisfaction of any federal, state or local taxes incurred by the Trust. Any Cash so delivered to the Trust shall be designated as "GUC Trust Supplemental Cash." GUC Trust Supplemental Cash shall be held by the Trust in a segregated account (the "Segregated Account"), and shall be used solely for the satisfaction of fees and expenses of the Trust (including for the payment of the fees and expenses of Trust Professionals), for the satisfaction of taxes incurred by the Trust or for distribution to holders of Allowed General Unsecured Claims pursuant to Section 5.1(d)(iv) hereof.
- (ii) The Trust Administrator shall only use the GUC Trust Supplemental Cash to satisfy fees and expenses of the Trust and tax liabilities of the Trust to the extent that there is no other source of funds to satisfy such expenses (other than the Avoidance Action Trust SEC Reporting Cash, and other than through the sale or encumbrance of the Term Loan Avoidance Action or any other property of the Trust pursuant to Section 6.1(d) hereof), including, without limitation, Avoidance Action Trust Administrative Cash, and any funds obtained through the reservation and application of all or a portion of the Holdback pursuant to Section 6.1(b) hereof. Any income earned on the GUC Trust Supplemental Cash, net of taxes paid thereon, shall be GUC Trust Supplemental Cash.
 - 2.4. <u>Appointment and Acceptance of Trust Administrator</u>. The Trust Administrator shall be deemed to be appointed pursuant to Bankruptcy Code Section 1123(b)(3)(B) and is hereby appointed trustee of the Trust under the Delaware Act. The Trust Administrator hereby accepts such appointments, including trusteeship of the Trust created by this Trust Agreement and the grant, assignment, transfer, conveyance and

delivery by the Debtors to the Trust Administrator, (i) on behalf of the Trust, and for the benefit of the Trust Beneficiaries, of all of their respective right, title and interest in the Distributable Trust Assets, and (ii) on behalf of the Trust, and for the benefit of the DIP Lenders, of all of their respective right, title and interest in the Other Debtor Residual Trust Assets, upon and subject to the terms and conditions set forth in the Plan, the Confirmation Order and this Trust Agreement. The Trust Administrator's powers are exercisable solely in a fiduciary capacity consistent with, and in furtherance of, the purpose of the Trust and not otherwise, and in accordance with applicable law, including the Delaware Act. The Trust Administrator shall have the authority to bind the Trust within the limitations set forth herein, but shall for all purposes hereunder be acting in the capacity as Trust Administrator, and not individually.

2.5. <u>Distribution of Distributable Trust Assets</u>. The Trust Administrator shall, in an expeditious but orderly manner and subject to the provisions of the Plan, the Confirmation Order and this Trust Agreement, make timely distributions of the Distributable Trust Assets and the Distributable Other Debtor Residual Assets in accordance with the terms hereof and not unduly prolong the existence of the Trust. The Trust Administrator may incur and pay any reasonable and necessary expenses in connection with the administration of the Trust, including the fees and expenses of the Trust Professionals *provided*, *however*, that all such expenditures (other than in respect of the Other Debtor Residual Trust Assets) shall be made in accordance with the Budget.

2.6. No Reversion to Debtors.

- (a) In no event shall any part of the Avoidance Action Trust Assets or the Other Debtor Residual Trust Assets revert to or be distributed to or for the benefit of any Debtor. All Distributable Trust Assets shall be applied as provided in Section 5.1(d), including to the satisfaction of Allowed General Unsecured Claims, including through distributions made in respect of the Units. All Distributable Other Debtor Residual Trust Assets shall be applied as provided in Article 5A.
- (b) To the extent that after satisfaction in full of all of the costs and expenses of the administration of the Trust, after all Allowed General Unsecured Claims have been paid pursuant to the Plan, after satisfaction of all other obligations or liabilities of the Trust incurred or assumed in accordance with the Plan, Confirmation Order or this Trust Agreement (or to which the Avoidance Action Trust Assets are otherwise subject), and after the affairs of the Trust have been finally wound up and concluded in accordance with the provisions of Section 4.3 hereof and Section 3808 of the Delaware Act, there shall remain any Avoidance Action Trust Administrative Cash, the Trust Administrator is authorized to and shall distribute any such remaining Avoidance Action Trust Administrative Cash to the DIP Lenders in accordance with the terms of the DIP Credit Agreement and the Plan. To the extent any portion of such residue is not accepted by the respective DIP Lenders, the Trust Administrator shall (i) be authorized to distribute up to \$100,000 of such remaining Avoidance Action Trust Administrative Cash to an organization described in Section 501(c)(3) of the Tax Code and exempt from U.S. federal income tax under section 501(a) of the Tax Code that is unrelated to the Debtors, the Trust, any Trust Administrator Parties or any Trust Monitor Parties, or (ii) request an order from the Bankruptcy Court authorizing the Trust Administrator to distribute

any such remaining Avoidance Action Trust Administrative Cash to such an organization, or authorizing such other disposition as recommended by the Trust Administrator and approved by the Bankruptcy Court.

(c) To the extent that after satisfaction in full of all of the costs and expenses of the administration of the Trust, after all Allowed General Unsecured Claims have been paid pursuant to the Plan, after satisfaction of all other obligations or liabilities of the Trust incurred or assumed in accordance with the Plan, Confirmation Order or this Trust Agreement (or to which the Avoidance Action Trust Assets are otherwise subject), and after the affairs of the Trust have been finally wound up and concluded in accordance with the provisions of Section 4.3 hereof and Section 3808 of the Delaware Act, there shall remain any GUC Trust Supplemental Cash, the Trust Administrator is authorized to and shall distribute any such remaining GUC Trust Supplemental Cash to the holders of Allowed General Unsecured Claims (as defined in the GUC Trust Agreement), regardless of whether such amount of GUC Trust Supplemental Cash is less than the Distribution Threshold, provided that, if the remaining GUC Trust Supplemental Cash is less than \$100,000, the Trust Administrator shall (i) be authorized to distribute such remaining GUC Trust Supplemental Cash to an organization described in Section 501(c)(3) of the Tax Code and exempt from U.S. federal income tax under section 501(a) of the Tax Code that is unrelated to the Debtors, the Trust, any Trust Administrator Parties or any Trust Monitor Parties, or (ii) request an order from the Bankruptcy Court authorizing the Trust Administrator to distribute any such remaining GUC Trust Supplemental Cash to such an organization, or authorizing such other disposition as recommended by the Trust Administrator and approved by the Bankruptcy Court.

ARTICLE III TRUST BENEFICIARIES; UNITS

3.1. Rights of Beneficiaries.

- (a) Except as provided in <u>Section 2.6</u> hereof, the Trust Beneficiaries shall be the sole beneficiaries of the Trust (to the extent of the Avoidance Action Trust Assets) and the Distributable Trust Assets, and the Trust Administrator shall retain only such incidents of ownership as are necessary to undertake the actions and transactions authorized in the Plan, the Confirmation Order and this Trust Agreement, including those powers set forth in <u>Articles VI and VIII</u> hereof.
- (b) The beneficial interest of a Trust Beneficiary in the Trust is hereby declared and shall be in all respects and for all purposes intangible personal property.
- (c) Except as expressly provided herein, a Trust Beneficiary shall have no title or right to, or possession, management or control of, the Trust, or the Avoidance Action Trust Assets, or to any right to demand a partition or division of such assets or to require an accounting of the Trust Administrator or the Trust Monitor. The whole legal title to the Avoidance Action Trust Assets shall be vested in the Trust as a separate legal entity under the Delaware Act or, if necessary, in the Trust Administrator on behalf of the Trust and the sole beneficial interest of the Trust Beneficiaries shall be as set forth in this Trust Agreement.

- 3.2. <u>Limited Liability</u>. No provision of the Plan, the Confirmation Order or this Trust Agreement, and no mere enumeration herein of the rights or privileges of any Trust Beneficiary, shall give rise to any liability of such Trust Beneficiary solely in its capacity as such, whether such liability is asserted by any Debtor, by creditors or employees of any Debtor, or by any other Person. GUC Beneficiaries are deemed to receive the GUC Distributable Trust Assets in accordance with the provisions of the Plan, the Confirmation Order and this Trust Agreement in exchange for their Allowed General Unsecured Claims or on account of their Units, as applicable, without further obligation or liability of any kind, but subject to the provisions of this Trust Agreement.
- 3.3. No Control by Trust Beneficiaries. A Trust Beneficiary shall have no title to, or any right to possess, manage or control, the Avoidance Action Trust Assets, or any portion thereof or interest therein, except as expressly provided herein. No surviving spouse, heir or devisee of any deceased Trust Beneficiary shall have any right of dower, homestead or inheritance, or of partition, or any other right, statutory or otherwise, in the Avoidance Action Trust Assets, but the whole title to all the Avoidance Action Trust Assets shall be vested in the Trust Administrator and the sole interest of the Trust Beneficiaries shall be the rights and benefits provided to such persons under this Trust Agreement.

3.4. Issuance of Units.

- (a) The Trust shall issue Units to holders of Allowed General Unsecured Claims as provided in this Trust Agreement. On the Initial GUC Distribution Date, holders of Initial Allowed General Unsecured Claims shall receive the number of Units equal to the amount of such Initial Allowed General Unsecured Claims multiplied by the Unit Issuance Ratio, rounded up or down to the nearest whole Unit. Following the Initial GUC Distribution Date, holders of Resolved Allowed General Unsecured Claims shall receive the number of Units equal to the amount of such Resolved Allowed General Unsecured Claims multiplied by the Unit Issuance Ratio, rounded up or down to the nearest whole Unit. Units will represent the contingent right to receive, on a pro rata basis as provided in the Plan, the Confirmation Order and this Trust Agreement, Excess GUC Distributable Trust Assets. The Units shall be issued subject to all the terms and conditions of the Plan, the Confirmation Order and this Trust Agreement. References in this Trust Agreement to holders of Units shall be to the record holders of such Units.
- (b) As provided in <u>Section 7.5</u> hereof, the Trust Administrator may retain Units otherwise issuable pursuant to this section with respect to Allowed General Unsecured Claims that are subject to withholding, and the Trust Administrator shall apply amounts distributed in respect of such retained Units to satisfy such withholding obligations.
- (c) Notwithstanding the foregoing, if (i) as of the Initial GUC Distribution Date, the total amount of the Disputed General Unsecured Claims in the aggregate is less than 0.5% of the Current Total Amount, or (ii) the Initial GUC Distribution Date is determined by the Trust Administrator to also be the final Distribution Date, no Units shall be distributed and any GUC Distributable Trust Assets remaining after satisfaction of all Initial Allowed General

Unsecured Claims and any other obligations of the Trust shall be disposed of as set forth in the last sentence of Section 2.6(c).

3.5. Ownership of Units; Transfers of Units.

- (a) The interest of a Trust Beneficiary is in all respects personal property, and upon the death, insolvency or incapacity of an individual GUC Beneficiary, such GUC Beneficiary's Units shall pass to the legal representative of such GUC Beneficiary.
- (b) The Units will be issued and evidenced by appropriate notation on the books and records of the Trust Administrator. The Units shall not be certificated and shall not be transferable, assignable, pledged or hypothecated in whole or in part, except by applicable laws of descent and distribution (in the case of a deceased individual GUC Beneficiary); by operation of law; in accordance with applicable Bankruptcy law; or as otherwise approved by the Bankruptcy Court. The Trust Administrator shall not be required to recognize any equitable or other claims to such interest by the transferee thereof, and the named GUC Beneficiary shall remain as such for all purposes hereunder.
 - 3.6. Conflicting Claims to Units. If the Trust Administrator has actual knowledge of any conflicting claims or demands that have been made or asserted with respect to a Unit, or a beneficial interest therein, the Trust Administrator shall be entitled, at its sole election, to refuse to comply with any such conflicting claims or demands. In so refusing, the Trust Administrator may elect to make no payment or distribution with respect to the Unit subject to the claims or demands involved, or any part thereof, and the Trust Administrator shall be entitled to refer such conflicting claims or demands to the Bankruptcy Court, which shall have exclusive and continuing jurisdiction over resolution of such conflicting claims or demands. The Trust Administrator shall not be or become liable to any party for either (i) its election to continue making distributions pursuant to its books and records, without regard to the conflicting claims or demands; or (ii) its election to cease payments or distributions with respect to the subject Unit. In the event that the Trust Administrator elects to cease payments, it shall be entitled to refuse to act until either (x) the rights of the adverse claimants have been adjudicated by a Final Order of the Bankruptcy Court (or such other court of proper jurisdiction) or (y) all differences have been resolved by a written agreement among all of such parties and the Trust Administrator, which agreement shall include a complete release of the Trust, the Trust Administrator Parties and the Trust Monitor Parties in form and substance reasonably satisfactory to the Trust Administrator and Trust Monitor (the occurrence of either (x) or (y), a "Claim Conflict Resolution"). Until a Claim Conflict Resolution is reached with respect to such conflicting claims or demands, the Trust Administrator shall hold in a segregated account any payments or distributions from the Trust to be made with respect to the Unit(s) at issue. Promptly after a Claim Conflict Resolution is reached, the Trust Administrator shall transfer the payments and distributions, if any, held in the segregated account, together with interest and income earned thereon, if any, in accordance with the terms of such Claim Conflict Resolution.
 - 3.7. <u>Distributions Relating to Note Claims and Eurobond Claims</u>. The Trust shall distribute GUC Distributable Trust Assets and Units to the Indenture Trustees and

Fiscal and Paying Agents, to the extent necessary to provide each beneficial holder of debt securities arising out of or relating to the Note Claims and Eurobond Claims with an amount of GUC Distributable Trust Assets and Units equal to the amount of GUC Distributable Trust Assets and Units such holder would receive had its claim been treated as an Initial Allowed General Unsecured Claim hereunder. For the avoidance of doubt, Units will be issued and evidenced by appropriate notation on the books and records of the Trust Administrator in the names of the Indenture Trustees and the Fiscal and Paying Agents, as applicable, and not in the individual names of the beneficial holders of debt securities arising out of or relating to the Note Claims and Eurobond Claims.

ARTICLE IV DURATION AND TERMINATION OF THE TRUST

4.1. Duration. The Trust shall become effective upon (x) the earlier to occur of (i) the Avoidance Action Trust Transfer Date and (ii) the Other Debtor Residual Assets Transfer Date, if any, and (y) the execution of this Trust Agreement and the filing of the Certificate of Trust with the Secretary of State and shall remain and continue in full force and effect until (a) all of the Distributable Trust Assets and all Distributable Other Debtor Residual Trust Assets, if any, have been distributed pursuant to the Plan and this Trust Agreement, (b) the Trust Administrator determines, in its discretion and with the approval of the Trust Monitor, that the administration of the Avoidance Action Trust Assets is not likely to yield sufficient additional Distributable Trust Assets or Distributable Other Debtor Residual Trust Assets to justify further pursuit, and (c) all other distributions required to be made by the Trust Administrator under the Plan and this Trust Agreement have been made, but in no event shall the Trust be dissolved later than three (3) years from the earlier of (i) the Avoidance Action Trust Transfer Date and (ii) the Other Debtor Residual Assets Transfer Date, unless the Bankruptcy Court, upon motion within the six (6) month period prior to the third (3rd) anniversary of the earlier of (i) the Avoidance Action Trust Transfer Date and (ii) the Other Debtor Residual Assets Transfer Date (or at least six (6) months prior to the end of an extension period), determines that a fixed period extension (such that, together with any prior extensions, the dissolution of the Trust shall occur no later than five (5) years from the date on which the Trust became effective, without a favorable private letter ruling from the IRS that any further extension would not adversely affect the status of the Trust as a liquidating trust for U.S. federal income tax purposes) is necessary to facilitate or complete the recovery and liquidation of the Avoidance Action Trust Assets or the Other Debtor Residual Trust Assets, as the case may be. If at any time the Trust Administrator determines, in reliance upon such professionals as the Trust Administrator may retain and with the approval of the Trust Monitor, that (x) the expense of administering the Trust so as to make a final distribution to the Trust Beneficiaries is likely to exceed the value of the Avoidance Action Trust Assets remaining in the Trust and (y) the expense of administering the Trust so as to make a final distribution to the DIP Lenders is likely to exceed the value of the Other Debtor Residual Trust Assets, if any, remaining in the Trust, the Trust Administrator may apply to the Bankruptcy Court for authority to (i) reserve any amounts necessary to dissolve the Trust, (ii) transfer the balance to the DIP Lenders and the holders of Allowed General Unsecured Claims (as defined in the GUC Trust Agreement), in accordance with the Agreed

Allocation and the Settlement Agreement, or donate any balance to a charitable organization described in section 501(c)(3) of the Tax Code and exempt from U.S. federal income tax under section 501(a) of the Tax Code that is unrelated to the Debtors, the Trust, any Trust Administrator Parties or any Trust Monitor Parties, and (iii) dissolve the Trust.

- 4.2. <u>Dissolution of the Trust</u>. Notwithstanding anything to the contrary in this Trust Agreement, in no event shall the Trust Administrator unduly prolong the duration of the Trust, and the Trust Administrator shall, in the exercise of its reasonable business judgment and in the interests of all Trust Beneficiaries, at all times endeavor to terminate the Trust as soon as practicable in accordance with the purposes and provisions of this Trust Agreement and the Plan. Upon final dissolution and wind-up of the Trust, the Trust Administrator shall file a certificate of cancellation for the Trust with the Secretary of State.
- 4.3. Continuance of Trust for Purposes of Winding Up. After the dissolution of the Trust and solely for the purpose of liquidating and winding up its affairs, the Trust Administrator shall continue to act in such capacity until its duties hereunder have been fully performed. The Trust Administrator shall retain the books, records and files that shall have been delivered to or created by the Trust Administrator until distribution or resolution of all the Avoidance Action Trust Assets and Other Debtor Residual Trust Assets, if any. At the Trust Administrator's discretion, all of such books, records and files may be destroyed at any time following the later of (x) final distribution of the Avoidance Action Trust Assets and Other Debtor Residual Trust Assets, if any, unless such books, records and files are necessary to fulfill the Trust Administrator's obligations pursuant to Articles VI and VIII hereof and subject to any joint prosecution and common interests agreement(s) to which the Trust Administrator may be party, and (y) the date until which the Trust Administrator is required by applicable law to retain such books, records and files.

ARTICLE V DISTRIBUTIONS TO TRUST BENEFICIARIES

5.1. General.

(a) Until such date as the Term Loan Avoidance Action shall have been completely and finally resolved in full against all defendants (including by way of settlement) and the Trust Administrator determines that all Avoidance Action Proceeds have been collected in respect thereof (such date, the "Final Recovery Date"), the Trust Administrator shall establish Distribution Dates no less frequently than once each calendar year and no more frequently than once a calendar quarter for the distribution of Distributable Trust Assets as provided in this Article V; provided that distributions need not be made in any calendar year to the extent (A) there are no Distributable Trust Assets held by the Trust or (B) the Trust Administrator, with the approval of the Trust Monitor, determines (i) that it is reasonably necessary to retain the Distributable Trust Assets to meet contingent liabilities and maintain the value of the Avoidance Action Trust Assets (such as, for example, in the event that the Trust Administrator determines that the Distributable Trust Assets are so small in amount as not to justify making a distribution, taking into account the costs that would be incurred in

making the distribution, the anticipated total amount of Distributable Trust Assets expected to be available for distribution over time and the timing of the distribution or distributions thereof), or (ii) that it is necessary to retain the Distributable Trust Assets to pay reasonable incurred and anticipated expenses (including any taxes imposed on the Trust or in respect of the Avoidance Action Trust Assets) or to satisfy liabilities incurred and anticipated by the Trust in accordance with the Plan, the Confirmation Order and this Trust Agreement. Following the Final Recovery Date, distributions shall be made on a quarterly basis, as provided in this Article V.

- (b) Except as otherwise set forth herein, no distributions shall be made with respect to any portion of a Disputed General Unsecured Claim, Unresolved Term Loan Avoidance Action Claim or Unresolved Other Avoidance Action Claim unless and until such Disputed General Unsecured Claim, Unresolved Term Loan Avoidance Action Claim or Unresolved Other Avoidance Action Claim has become an Allowed General Unsecured Claim.
- (c) To the extent that a Disputed General Unsecured Claim, Unresolved Term Loan Avoidance Action Claim or Unresolved Other Avoidance Action Claim has become an Allowed General Unsecured Claim, distributions (if any) shall be made to the holder of such Allowed General Unsecured Claim in accordance with the provisions of the Plan, the Confirmation Order and this Trust Agreement.
 - (d) The Distributable Trust Assets shall be distributed
- (i) first, to the DIP Lenders, each in their relative proportion pursuant to the terms of the DIP Credit Agreement, in the amount of the DIP Lender Advances;
- (ii) second, to the Capital Providers, in the amount of the Capital Providers' Entitlement;
- (iii) third, to the Segregated Account in the amount of the GUC Trust Advances to be distributed to the holders of Allowed General Unsecured Claims; and
- (iv) fourth, to: the DIP Lenders, each in their relative proportion pursuant to the terms of the DIP Credit Agreement, and the holders of Allowed General Unsecured Claims, in accordance with the Agreed Allocation.
- (e) On any Distribution Date following the Final Recovery Date on which the Trust does not hold sufficient Distributable Trust Assets, after taking into account any amounts necessary to satisfy the current and projected future fees and expenses of the Trust (including those of any Trust Professionals) pursuant to Section 6.1(b), to satisfy all Disputed General Unsecured Claims or other Claims that became Resolved Allowed General Unsecured Claims Determination Date following which there was a distribution pursuant to Section 5.3(c), the Trust Administrator shall distribute all Distributable Trust Assets that remain in the Trust to the DIP Lenders and the holders of such Resolved Allowed General Unsecured Claims pro rata by Claim amount, in accordance with the Agreed Allocation. Following such distribution, any

remaining unsatisfied portion of such Resolved Allowed General Unsecured Claims, together with all remaining Disputed General Unsecured Claims and other Claims (including any Term Loan Avoidance Action Claims and any Other Avoidance Action Claims) shall be forever barred from assertion against the Trust.

- (f) Anything to the contrary herein notwithstanding but subject to <u>Section 5.1(a)</u>, the Trust Administrator shall not make a distribution of Distributable Trust Assets on any Distribution Date pursuant to <u>Sections 5.2 or 5.4</u>, if the amount to be distributed pursuant thereto does not exceed the Distribution Threshold. In such case, any Distributable Trust Assets then available for distribution shall be held by the Trust and distributed on a subsequent Distribution Date when the amount of the Distributable Trust Assets to be distributed shall exceed the Distribution Threshold; *provided* that if on the date determined by the Trust Administrator to be the final Distribution Date the Distributable Trust Assets do not exceed the Distribution Threshold, then such Distributable Trust Assets shall be disposed of as provided in the final sentence of <u>Section 2.6(b)</u> and in <u>Section 2.6(c)</u>.
- (g) For the avoidance of doubt, if the Trust fails to recover any Avoidance Action Proceeds or if the Avoidance Action Proceeds recovered by the Trust through the Final Recovery Date do not exceed the amount of DIP Lender Advances then no distributions shall be made hereunder in respect of any Allowed General Unsecured Claims.
- (h) For the avoidance of doubt, if the Avoidance Action Proceeds recovered by the Trust through the Final Recovery Date exceed the amount of DIP Lender Advances but do not exceed the amount of the Segregated Account Assets, the Segregated Account Assets shall be distributed as GUC Distributable Trust Assets pursuant to Section 5.2(a) hereof.
 - 5.2. <u>Distribution to the DIP Lenders and to Holders of Initial Allowed General</u> Unsecured Claims.
- (a) Once the DIP Lender Advances have been indefeasibly paid in full in cash, the Capital Providers' Entitlement has been indefeasibly paid in full in cash, and the Segregated Account repaid up to the amount of the GUC Trust Advances, the Trust Administrator shall make distributions of the DIP Lender Distributable Trust Assets and the GUC Distributable Trust Assets, from time to time in accordance with Section 5.1(a). Prior to the first such distribution, the Trust Administrator shall establish a record date (the "Initial GUC Record Date") for the holders of the General Unsecured Claims that are allowed as of the Initial GUC Record Date (the "Initial Allowed General Unsecured Claims") who shall be entitled to participate in the first distribution of GUC Distributable Trust Assets, which date shall be the last day of the calendar quarter next preceding the date of such distribution (such Distribution Date, the "Initial GUC Distribution Date"). On the Initial GUC Distribution Date, the Trust Administrator shall distribute to each holder of an Initial Allowed General Unsecured Claim a distribution consisting of:
- (i) an amount of GUC Distributable Trust Assets at the time available for distribution, in proportion to the amount of such Initial Allowed General Unsecured Claim as prescribed in subsection (d) below; and

- (ii) a number of Units as provided in Section 3.4.
- (b) During the period, if any, following the Initial GUC Distribution Date and prior to the Final Recovery Date, and as promptly as practicable following the Final Recovery Date, as and to the extent that additional GUC Distributable Trust Assets become available for distribution from time to time as a result of additional recoveries by the Trust in the Term Loan Avoidance Action, the Trust Administrator shall, from time to time in accordance with Section 5.1(a), make additional distributions to the holders of Initial Allowed General Unsecured Claims, on a Distribution Date or Dates designated by the Trust Administrator for such purpose, in the amount prescribed in subsection (d) below. Subject to the proviso in Section 5.1(a), the Trust Administrator shall make such distributions no less frequently than once each calendar year and no more frequently than once each calendar quarter.
- (c) The amount of GUC Distributable Trust Assets that the holder of an Initial Allowed General Unsecured Claim shall be entitled to receive pursuant to Section 5.2 (or, in the case of a holder of a Resolved Allowed General Unsecured Claim being treated as if it were a holder of an Initial Allowed General Unsecured Claim in the amount of its Resolved Allowed General Unsecured Claim pursuant to Section 5.3(c) on any Distribution Date (including the Initial Distribution Date) shall be determined in accordance with the following formula:

$$D = (\frac{A}{C}) \times G$$

Where—

- D is the distribution that the holder of an Initial Allowed General Unsecured Claim (or the holder of a Resolved Allowed General Unsecured Claim pursuant to <u>Section</u> 5.3(c)) will be entitled to receive;
- A is the amount of the Initial Allowed General Unsecured Claim (or the holder of a Resolved Allowed General Unsecured Claim pursuant to Section 5.3(c));
- C is the Current Total Amount as of the last day of the calendar quarter next preceding the respective Distribution Date; and
- G is the amount available for distribution determined as of the last day of the calendar quarter next preceding the respective Distribution Date, after taking account of any Holdback, or the application of such Holdback, pursuant to Section 6.1(b).
 - 5.3. Distributions to Holders of Resolved Allowed General Unsecured Claims.
- (a) Following the Initial GUC Distribution Date, the Trust Administrator, with the approval of the Trust Monitor, shall periodically make a determination (the date of any such determination, which shall in all cases be made as of the last day of a calendar quarter, being a "Resolved Allowed General Unsecured Claims Determination Date") whether any holders of Disputed General Unsecured Claims or other Claims have become holders of Resolved Allowed General Unsecured Claims since the next preceding Resolved Allowed General Unsecured Claims Determination Date or, in the case of the first such determination,

since the Initial GUC Record Date. During the period, if any, following the Initial GUC Distribution Date and prior to the Final Recovery Date, the Trust Administrator shall make such determination no less frequently than once each calendar year and no more frequently than once each calendar quarter, and following the Final Recovery Date such determination shall be made once each calendar quarter.

- (b) On a Distribution Date scheduled as soon as practicable following each Resolved Allowed General Unsecured Claims Determination Date, the Trust Administrator shall, subject to the proviso in Section 5.1(a), distribute to each holder of a Resolved Allowed General Unsecured Claim identified on such Resolved Allowed General Unsecured Claims Determination Date, if any:
- (i) the pro rata amount of GUC Distributable Assets that the holder of such Resolved Allowed General Unsecured Claim would have received had such Resolved Allowed General Unsecured Claim been an Initial Allowed General Unsecured Claim, including the aggregate amount of Excess GUC Distributable Trust Assets that the holder would have received had it been the holder of Units referred to in clause (ii) below on each Excess GUC Distributable Trust Assets Determination Date occurring on or prior to the date of such distribution; provided that a holder of a Resolved Allowed General Unsecured Claim shall not receive pursuant to this clause (i) an amount of Excess GUC Distributable Assets distributed in respect of any prior Excess GUC Distributable Trust Assets Determination Date to the extent that it will be receiving such Excess GUC Distributable Assets as a distribution on the Units to be received by such holder pursuant to clause (ii) below; and
 - (ii) a number of Units as provided in <u>Section 3.4</u>.
- (c) Once a holder of a Resolved Allowed General Unsecured Claim has received the distribution prescribed in the preceding subsection (b), such holder shall be treated as if it were a holder of an Initial Allowed General Unsecured Claim in the amount of its Resolved Allowed General Unsecured Claim on all subsequent Distribution Dates established for purposes of Section 5.2(c).
- (d) For the avoidance of doubt, it is intended that the distributions to be made to holders of Resolved Allowed General Unsecured Claims in accordance with this <u>Section 5.3</u> shall provide such holders, as nearly as possible, with the exact same amount of distributions as if such holders had been holders of Initial Allowed General Unsecured Claims.

5.4. Distribution of Excess Distributable Trust Assets.

(a) Following the Initial GUC Distribution Date, the Trust Administrator, with the approval of the Trust Monitor, shall periodically make a determination (the date of any such determination, which shall in all cases be as of the last day of a calendar quarter, being an "Excess Distributable Trust Assets Determination Date") of the Excess Distributable Trust Assets as of such date, taking into account the extent to which Disputed General Unsecured Claims are disallowed or the Term Loan Avoidance Action or other Avoidance Actions are resolved in favor of the defendants therein. During the period, if any, following the Initial GUC Distribution Date and prior to the Final Recovery Date, the Trust

Administrator shall make such determination no less frequently than once each calendar year and no more frequently than once each calendar quarter, and following the Final Recovery Date such determination shall be made once each calendar quarter.

- (b) On a Distribution Date scheduled as soon as practicable following each Excess Distributable Trust Assets Determination Date, the Trust Administrator shall, subject to the proviso of Section 5.1(a), distribute the Excess Distributable Trust Assets, in each case determined as of the respective Excess Distributable Trust Assets Determination Date, to the DIP Lenders and the holders of Units outstanding on such Excess Distributable Trust Assets Determination Date (including Units distributed or to be distributed to holders of Resolved Allowed General Unsecured Claims pursuant to Section 5.3(b)(ii) on such Distribution Date) in accordance with the Agreed Allocation.
- (c) Any Excess Distributable Trust Assets allocable to the holders of Units outstanding on such Excess Distributable Trust Assets Determination Date (the "Excess GUC Distributable Trust Assets"), shall be distributed pro rata according to the following formula:

$$D_{U} = \left(\frac{U_{H}}{U_{O}}\right) \quad x \left(\Sigma G - H\right) x \left[\frac{T_{U}}{C_{U}} - \frac{T_{U}}{\left(C_{U} + L\right)}\right]$$

Where-

- D_U is the distribution of Excess GUC Distributable Trust Assets that a holder of Units will be entitled to receive;
- U_H is the number of Units held by the holder;
- U_O is the total number of Units outstanding (including Units distributed to holders of Resolved Allowed General Unsecured Claims pursuant to Section 5.3(b)(ii) on such Distribution Date);
- Σ G is the sum of the amounts that are or were available for distribution to holders of Initial Allowed General Unsecured Claims (and holders of Resolved Allowed General Unsecured Claims pursuant to Section 5.3(c)) on the relevant Distribution Date and each prior Distribution Date;
- T_U is the Total Allowed Amount as of the Excess GUC Distributable Trust Assets Determination Date;
- C_U is the Current Total Amount as of the Excess GUC Distributable Trust Assets Determination Date;
- L is the aggregate amount of all (i) Disputed General Unsecured Claims disallowed since the next preceding Excess GUC Distributable Trust Assets Determination Date (or, in the case of the first Excess GUC Distributable Trust Assets Determination Date, since the Initial GUC Record Date), (ii) Unresolved Term Loan Avoidance Action Claims to the extent resolved (including by way of settlement) in favor of the respective defendants since the next preceding Excess GUC Distributable Trust Assets Determination Date (or, in the case of the first Excess GUC Distributable Trust Assets Determination Date, since the Initial GUC Record Date); and (iii) all Unresolved Other

Avoidance Action Claims to the extent resolved (including by way of settlement) in favor of the respective defendants since the next preceding Excess GUC Distributable Trust Assets Determination Date (or, in the case of the first Excess GUC Distributable Trust Assets Determination Date, since the Initial GUC Record Date); and

- H is the amount, if any, of any holdback pursuant to <u>Section 6.1</u> that was not otherwise deducted from the amounts available for distribution on a Distribution Date.
- (d) Notwithstanding the foregoing, if the Trust Administrator becomes aware of previously unknown potential Allowed General Unsecured Claims, the Trust Administrator may, with the approval of the Trust Monitor, withhold distribution of Excess GUC Distributable Trust Assets to the holders of Units in an amount that the Trust Administrator, with the approval of the Trust Monitor, estimates to be the maximum amount reasonably allowable in respect of such previously unknown claims.
 - 5.5. <u>Retention of Avoidance Action Trust Assets</u>. Notwithstanding anything in this Trust Agreement to the contrary, the Trust Administrator shall at all times, to the extent practicable, retain
- (a) sufficient GUC Distributable Trust Assets as the Trust Administrator shall determine, with the approval of the Trust Monitor, as would be distributable (I) to all holders of Disputed General Unsecured Claims at the time outstanding as if all Disputed General Unsecured Claims were allowed at the Maximum Amount, but only until such Disputed General Unsecured Claims are resolved, (II) to the holders of all Resolved Allowed General Unsecured Claims at the time outstanding, to the extent not previously distributed, (III) in respect of any Unresolved Term Loan Avoidance Action Claims at the Maximum Amount thereof but only until the Term Loan Avoidance Action is dismissed by Final Order or such claims become Resolved Allowed General Unsecured Claims, and (IV) in respect of any Unresolved Other Avoidance Action Claims at the Maximum Amount thereof but only until such claims become Resolved Allowed General Unsecured Claims or the related other Avoidance Actions are dismissed by Final Order; and
- (b) sufficient Avoidance Action Trust Administrative Cash and Supplemental Avoidance Action Trust Cash as the Trust Administrator shall determine, with the approval of the Trust Monitor and subject to the Budget, is necessary (x) to pay the reasonable incurred or anticipated fees and expenses of the Trust (including any taxes imposed on the Trust or in respect of the Avoidance Action Trust Assets) and (y) to satisfy other liabilities incurred or anticipated by the Trust in accordance with the Plan, the Confirmation Order and this Trust Agreement.
 - 5.6. <u>Minimum Distributions</u>. Notwithstanding anything to the contrary contained herein, no Cash payment in an amount less than \$25 shall be made by the Trust Administrator to any holder of an Allowed General Unsecured Claim or Unit under any circumstance; <u>provided</u> that the Trust Administrator shall carry the entitlement of such holder of an Allowed General Unsecured Claim or Unit to such amount on its books and records, shall aggregate such amount with any subsequent amount to which such holder shall become entitled and shall make payment of such amount to such holder at such time

as the amounts due such holder in the aggregate shall equal \$25 or more; <u>provided further</u> that if any such amount shall be owing to a holder of an Allowed General Unsecured Claim or Unit as of the date determined by the Trust Administrator to be the final Distribution Date, such amount shall be disposed of as provided in the final sentence of Section 2.6(c).

- 5.7. Distributions Not in Compliance with this Article. Subject to Section 5.3(d), in the event that the Trust Administrator determines in good faith that it is necessary or desirable in order to carry out the intent and purposes of the Plan, the Confirmation Order and this Trust Agreement to receive any assets or make any distribution in a manner that is not in technical compliance with this Trust Agreement, the Trust Administrator shall be permitted to receive assets or make, or cause to be made, distributions in such manner, but only with the approval of the Trust Monitor; provided, however, that no such distribution shall result in any holder of an Allowed General Unsecured Claim receiving a distribution in excess of the distribution that such holder would have received had such claim been an Initial Allowed General Unsecured Claim or shall discriminate among the holders of Units. Except as aforesaid, no payment or distribution of Avoidance Action Trust Assets shall be made to, or on behalf of, a Trust Beneficiary or any other person except in strict accordance with the terms of this Trust Agreement, the Plan, and the Confirmation Order, unless such payment or distribution shall have been approved by the Bankruptcy Court.
- 5.8. No Accounting. Except as otherwise provided in the Plan, the Confirmation Order or this Trust Agreement, nothing shall require the Trust Administrator to file any accounting or seek approval of any court with respect to the administration of the Trust or as a condition for making any payment or distribution out of the Avoidance Action Trust Assets.

ARTICLE VA DISTRIBUTIONS TO DIP LENDERS

If any Other Debtor Residual Accepted Assets shall be transferred to the Trust, the Trust Administrator shall make distributions to the DIP Lenders of Distributable Other Debtor Residual Assets, if any, from time to time (but no less frequently than once each calendar year), pro rata as their interests appear, as the Trust Administrator shall determine with the approval of the Trust Monitor or as the Trust Administrator shall be directed by a majority in interest of the DIP Lenders; provided that distributions need not be made in any calendar year to the extent (A) there are no Distributable Other Debtor Residual Assets held by the Trust or (B) the Trust Administrator, with the approval of the Trust Monitor, determines that it is necessary to retain the Distributable Other Debtor Residual Assets to (i) meet any contingent liabilities of the Trust or maintain the value of the Other Debtor Residual Trust Assets (such as for example, in the event that the Trust Administrator determines that the Distributable Other Debtor Residual Assets are so small in amount as not to justify making a distribution, taking into account the costs that would be incurred in making the distribution, the anticipated total amount of Distributable Other Debtor Residual Assets expected to be available for distribution over time and the timing of the distribution or distributions thereof), or (ii) pay reasonable incurred and/or

anticipated expenses of the Trust (including any taxes imposed on the Trust or in respect of the Other Debtor Residual Trust Assets) or to satisfy liabilities incurred and/or anticipated by the Trust in accordance with the Plan, the Confirmation Order and this Trust Agreement.

ARTICLE VI ADMINISTRATION OF THE TRUST

- 6.1. Payment of Costs, Expenses and Liabilities (other than in respect of the Other Debtor Residual Accepted Assets).
- (a) Subject to the Budget, the Trust Administrator shall use the Avoidance Action Trust Administrative Cash:
- (i) to pay reasonable costs and expenses of the Trust that are incurred in connection with the administration thereof (including any taxes imposed on the Trust, actual reasonable fees and out-of-pocket expenses incurred by Trust Professionals retained by the Trust Administrator in connection with the administration of the Avoidance Action Trust Assets and preservation of books and records);
- (ii) to satisfy other obligations or other liabilities incurred or assumed by the Trust (or to which the Avoidance Action Trust Assets are otherwise subject) in accordance with the Plan, the Confirmation Order, or this Trust Agreement, including fees and expenses incurred and in connection with, the prosecution and resolution of the Term Loan Avoidance Action the protection, preservation and distribution of the Avoidance Action Trust Assets; and
- (iii) to satisfy any other obligations of the Trust expressly set forth in the Plan, the Confirmation Order or this Trust Agreement to be satisfied out of the Avoidance Action Trust Administrative Cash.
- (b) (i) If, at any time, the Trust Administrator determines that the Avoidance Action Trust Administrative Cash is not reasonably likely to be adequate to satisfy the current and projected future taxes, fees, costs and expenses (including, without limitation any Avoidance Action Trust SEC Reporting Costs) of the Trust (other than in respect of the Other Debtor Residual Trust Assets), the Trust Administrator may, with the approval of the Trust Monitor, reserve an amount, or increase the amount previously reserved, of Distributable Trust Assets to satisfy such taxes, fees, costs and expenses (the "Holdback"). If at any time, the Trust Administrator determines that the Holdback is materially greater than the amount of the current and projected future taxes, fees, costs and expenses as aforesaid, the Trust Administrator shall, with the approval of the Trust Monitor, release from the Holdback the amount of such excess.
- (ii) To the extent necessary to satisfy the taxes, fees, costs and expenses on account of which the Holdback may be reserved, the Trust Administrator may, in consultation with the Trust Monitor, and upon approval by the Bankruptcy Court in accordance with the provisions of Section 6.1(b)(iii), apply all or a portion of the Holdback to the satisfaction of such taxes, fees, costs and expenses.

- (iii) The application of the Trust Administrator seeking Bankruptcy Court approval to utilize Distributable Trust Assets shall include the position of the Trust Monitor in respect thereof. The Trust Administrator shall provide at least twenty (20) days' notice to the Trust Monitor, the holders of Units and the holders of Disputed General Unsecured Claims prior to a hearing on a motion to use any Distributable Trust Assets.
 - (c) If, at any time, the Trust Administrator determines that the Avoidance Action Trust Administrative Cash, together with the Holdback (if Distributable Trust Assets are, at such time, available for the purposes of creating a Holdback), is not reasonably likely to be adequate to satisfy the current and projected future taxes, fees, costs and expenses (including, without limitation any Avoidance Action Trust SEC Reporting Costs and the fees and expenses of Trust Professionals) of the Trust (other than in respect of the Other Debtor Residual Trust Assets), the Trust Administrator may utilize the GUC Trust Supplemental Cash, without the need to seek or obtain approval of the Bankruptcy Court, to satisfy such fees and expenses.
 - (d) (i) If, at any time, the Trust Administrator determines that the Avoidance Action Trust Administrative Cash, together with the Holdback (if Distributable Trust Assets are, at such time, available for the purposes of creating a Holdback) and the GUC Trust Supplemental Cash, if any, is not reasonably likely to be adequate to satisfy the current and projected future fees, costs and expenses (including, without limitation any Avoidance Action Trust SEC Reporting Costs, the fees and expenses of Trust Professionals, and any tax liabilities), the Trust Administrator may, in consultation with the Trust Monitor, and upon approval by the Bankruptcy Court in accordance with the provisions of Section 6.1(d)(ii), sell or grant liens on the Term Loan Avoidance Action or any other property of the Trust (other than the Other Debtor Residual Trust Assets) and apply all or a portion of the proceeds of such sale or grant to the satisfaction of such fees, costs and expenses; provided that, except for the Trust's incurrence of expenses in the ordinary course of the Trust's business without any lien or right of setoff attached thereto, the Trust may not incur additional indebtedness, whether unsecured or secured by any interest (including a security interest) in the Term Loan Avoidance Action or Avoidance Action Proceeds without (x) the prior written consent of the DIP Lenders and (y) if such incurrence would (1) create a lien senior to or pari passu with the lien securing the Capital Providers' Entitlement or, (2) except as set forth in Section 2.7 of the Subordination Agreement, result in the subordination of any portion of the payment of the Capital Providers' Entitlement pursuant to Section 5.1(d) of this Trust Agreement, the prior written consent of the Capital Providers. In the event that the Trust seeks other litigation funding, (A) the DIP Lenders shall be provided a right of first refusal to provide such funding, which right of first refusal must be exercised within thirty (30) days after being presented in writing the opportunity to provide such funding, and (B) such funding (even if provided by the DIP Lenders) shall be junior and subordinate to the Litigation Cost Advance, the amounts owed to the DIP Lenders on account of prior funding of the Trust, and the Supplemental Capital, and shall be subject to a form of subordination that is acceptable to the DIP Lenders and the Capital Providers in all respects, with such consent not to be unreasonably withheld. Upon any such sale or grant of liens pursuant to this Section 6.1(d), the resulting proceeds shall be designated as Other Supplemental Cash. Any income earned on the Other Supplemental Cash, net of taxes paid thereon, shall be Other Supplemental Cash.

- (ii) The application of the Trust Administrator seeking Bankruptcy Court approval to sell or grant liens pursuant to Section 6.1(d)(i) hereof shall include the position of the Trust Monitor in respect thereof. The Trust Administrator shall provide at least twenty (20) days' notice to the Trust Monitor, the DIP Lenders, the Capital Providers, the holders of Units and the holders of Disputed General Unsecured Claims prior to a hearing on a motion to sell or grant liens pursuant to Section 6.1(d)(i) hereof.
- (e) Notwithstanding that as a result of the utilization of Distributable Trust Assets pursuant to Section 6.1(b) the amount of GUC Distributable Trust Assets shall be less than the assets required to satisfy Claims in the amount of the Current Total Amount then outstanding, the Trust Administrator shall continue to satisfy Disputed General Unsecured Claims, any Unresolved Term Loan Avoidance Action Claims and any Unresolved Other Avoidance Action Claims that become Allowed General Unsecured Claims in the order they are resolved as otherwise provided in this Trust Agreement.
 - 6.2. <u>Payment of Costs, Expenses and Liabilities in respect of the Other Debtor Residual Accepted Assets.</u>
- (a) The Trust Administrator shall not be required to undertake any activity in respect of the Other Debtor Residual Accepted Assets, including for purposes of realizing upon such assets in order to make distributions of Distributable Other Debtor Residual Trust Assets to the DIP Lenders, unless there shall be available to the Trust Administrator Other Debtor Residual Trust Administrative Cash sufficient for such purposes.
- (b) If sufficient Other Debtor Residual Trust Administrative Cash shall be available to the Trust Administrator, then the Trust Administrator shall, as approved by the Trust Monitor or as directed by a majority in interest of the DIP Lenders:
- (i) pay reasonable costs and expenses of the Trust that are incurred in connection with the Other Debtor Residual Accepted Assets (including any taxes imposed on the Trust, actual reasonable fees and out-of-pocket expenses incurred by Trust Professionals retained by the Trust Administrator in connection with the administration of the Other Debtor Residual Accepted Assets and preservation of books and records);
- (ii) satisfy other obligations or other liabilities incurred or assumed by the Trust in respect of the Other Debtor Residual Accepted Assets (or to which the Other Debtor Residual Accepted Assets are otherwise subject) in accordance with the Plan, the Confirmation Order, or this Trust Agreement, including fees and expenses incurred and in connection with, the prosecution and resolution of any action to realize upon the Other Debtor Residual Accepted Assets and the protection, preservation and distribution of the Other Debtor Residual Accepted Assets; and
- (iii) satisfy any other obligations of the Trust expressly set forth in the Plan, the Confirmation Order or this Trust Agreement to be satisfied out of the Other Debtor Residual Trust Administrative Cash.

6.3. Budget.

- (a) The Trust Administrator shall prepare and submit to the Trust Monitor, the DIP Lenders and the Capital Providers for approval a reasonably detailed annual plan and budget (the "Budget") at least thirty (30) days prior to the commencement of each calendar year. Such annual plan and Budget shall set forth (on a quarterly basis) in reasonable detail: (A) the Trust Administrator's anticipated actions to administer the Avoidance Action Trust Assets; and (B) the anticipated fees and expenses, including professional fees, associated with the administration of the Trust, a separate amount representing the anticipated fees and expenses of the Trust Monitor and detail as to how the Trust will budget and spend the Avoidance Action Trust Administrative Cash. Such Budget shall be updated and submitted to the Trust Monitor, the DIP Lenders, and the Capital Providers for review on a quarterly basis, and each such quarterly update shall reflect the variances (with explanations) between (x) the Budget, (y) any updated Budget, and (z) the actual results for the same period. If the Trust identifies a need for or the incurrence of actual expenditures that materially differ from the relevant Budget, the Trust will notify the DIP Lenders and the Capital Providers as promptly as practicable, and no later than fifteen (15) days after identifying the issue. For the avoidance of doubt, the DIP Lenders and the Capital Providers may object in the Bankruptcy Court with respect to any quarterly update that materially changes the Budget and the Bankruptcy Court shall resolve such dispute. All actions by the Trust Administrator shall be consistent with the Budget (as updated). The Trust Administrator may obtain any required approval of the Budget on reasonable negative notice (which shall be not less than 15 days after receipt of the Budget) and approval of the Budget shall not be unreasonably withheld. In the event of any dispute concerning the Budget (or the taking of actions consistent with the Budget), the Trust Administrator or the Trust Monitor may petition the Bankruptcy Court to resolve such dispute.
- (b) The Trust Administrator, with the approval of the Trust Monitor and the DIP Lenders may agree on a budget for activities in respect of the Other Debtor Residual Accepted Assets.
- (c) Notwithstanding any other provision of this Trust Agreement, the approval of the DIP Lenders shall not be required for any use of the Avoidance Action Trust SEC Reporting Cash, the GUC Trust Supplemental Cash or the Other Supplemental Cash.

6.4. [Intentionally omitted.]

- 6.5. <u>Compliance with Laws</u>. Any and all distributions of Avoidance Action Trust Assets shall be in compliance with applicable laws, including applicable federal and state tax and securities laws.
- 6.6. <u>Fiscal Year</u>. Except for the first and last years of the Trust, the fiscal year of the Trust shall be the calendar year. For the first and last years of the Trust, the fiscal year of the Trust shall be such portion of the calendar year that the Trust is in existence.

6.7. Books and Records.

- (a) The Trust Administrator shall maintain and preserve the Debtors' books, records and files that shall have been delivered to or created by the Trust Administrator.
- (b) The Trust Administrator shall maintain books and records relating to the assets, liabilities, income and expense of the Trust, all distributions made by the Trust and the payment of fees and expenses of, and satisfaction of claims against or assumed by, the Trust and the Trust Administrator, in such detail and for such period of time as may be necessary to enable it to make full and proper reports in respect thereof in accordance with the provisions of this Trust Agreement and otherwise to comply with applicable provisions of law, including tax law.
- (c) The Trust Administrator shall maintain, or cause to be maintained, a register of holders of Units, from time to time outstanding, to the extent any Units are issued hereunder, in customary form.
 - 6.8. <u>Cash Payments</u>. All distributions of Distributable Trust Cash required to be made by the Trust Administrator may be made in Cash denominated in U.S. dollars by checks drawn on a United States domestic bank selected by the Trust Administrator or, at the option of the Trust Administrator, by wire transfer from a United States domestic bank selected by the Trust Administrator or as otherwise required or provided in applicable agreements; <u>provided</u>, <u>however</u>, that cash payments to foreign persons may be made, at the option of the Trust Administrator, in such funds as and by such means as are necessary or customary in a particular foreign jurisdiction.
 - 6.9. <u>Insurance</u>. The Trust shall maintain customary insurance coverage for the protection of the Trust Administrator Parties and the Trust Monitor Parties and any such other persons serving as administrators and overseers of the Trust, on and after the Avoidance Action Trust Transfer Date, in all cases in accordance with the Budget. The Trust Administrator may also obtain such insurance coverage as it deems necessary and appropriate with respect to real and personal property which may become Avoidance Action Trust Assets, if any, in accordance with such Budget. To the extent that there is any incremental cost for customary insurance coverage covering the activities of the Trust Administrator Parties and the Trust Monitor Parties in respect of the Other Debtor Residual Accepted Assets, the Trust Administrator and the Trust Monitor shall not be required to undertake any such activities unless there is available sufficient Other Debtor Residual Trust Administrative Cash to fund such incremental cost.

ARTICLE VII TAX MATTERS

7.1. Tax Treatment.

(a) For all U.S. federal and applicable state and local income tax purposes, all parties (including the Debtors, the Trust Administrator, the holders of the DIP Credit Agreement Claims, and the holders of Allowed General Unsecured Claims) shall treat the

Trust and the transfer of the Avoidance Action Trust Assets and the Other Debtor Residual Assets to the Trust in a manner consistent with the remainder of this Section 7.1.

- (b) If no Other Debtor Residual Assets are transferred to the Trust upon the dissolution of MLC and the DIP Lender Distributable Trust Assets have not been determined (either as a percentage or as a fixed amount of Distributable Trust Assets or on some other basis) on or prior to the Avoidance Action Trust Transfer Date, then (subject to clause (c) below) the Trust Administrator shall treat the Trust as (A) a "disputed ownership fund" governed by Treasury Regulation section 1.468B-9 (including, if required, timely so electing), (B) if permitted under applicable law and at the election of the Trust Administrator, as a "complex trust," or (C) as otherwise permitted pursuant to a private letter ruling from the IRS.
- (c) If Other Debtor Residual Assets are transferred to the Trust upon the dissolution of MLC or the DIP Lender Distributable Trust Assets have been determined (either as a percentage or as a fixed amount of Distributable Trust Assets or on some other basis) on or prior to the Avoidance Action Trust Transfer Date, or otherwise upon determination of the DIP Lender Distributable Trust Assets (either as a percentage or as a fixed amount of Distributable Trust Assets or on some other basis) after the Avoidance Action Trust Transfer Date, the Trust (other than the Avoidance Action Trust Claims Reserve) shall be treated as a liquidating trust that is treated as a grantor trust and the Avoidance Action Trust Assets (upon the determination of the DIP Lender Distributable Trust Assets) and Other Debtor Residual Trust Assets (upon the transfer of Other Debtor Residual Assets to the Trust) shall be treated as (i) being transferred directly to the Trust Beneficiaries; provided, however, that to the extent Avoidance Action Trust Assets are allocable to Disputed General Unsecured Claims, Unresolved Term Loan Avoidance Action Claims or Unresolved Other Avoidance Action Claims, such Avoidance Action Trust Assets shall be treated as being transferred to the Avoidance Action Trust Claims Reserve, followed by (ii) the transfer by such Trust Beneficiaries of the Avoidance Action Trust Assets (other than the Avoidance Action Trust Assets allocable to the Avoidance Action Trust Claims Reserve) and the Other Debtor Residual Trust Assets, as applicable, to the Trust in exchange for beneficial interests in the Trust. Accordingly, Trust Beneficiaries receiving beneficial interests in the Trust shall be treated as the grantors and owners of their respective share of the Avoidance Action Trust Assets (other than any Avoidance Action Trust Assets allocable to the Avoidance Action Trust Claims Reserve) and Other Debtor Residual Trust Assets, as applicable.
- (d) Any determination made pursuant to this <u>Section 7.1</u> shall be conclusive and binding on all parties (including the Debtors, the Trust Administrator, the holders of the DIP Credit Agreement Claims, and the holders of Allowed General Unsecured Claims) for U.S. federal, and (to the extent permitted by applicable law) state and local, income tax purposes. Accordingly, to the extent permitted by applicable law, all parties shall report consistently with the U.S. federal income tax treatment of the Trust by the Trust Administrator for state and local income tax purposes.
 - 7.2. <u>Valuation of Assets</u>. As soon as practicable after the Avoidance Action Trust Transfer Date, the Trust Administrator shall make a good-faith valuation of the Avoidance Action Trust Assets and Other Debtor Residual Trust Assets, and such valuation shall be made available from time to time, to the extent relevant, and shall be

used consistently by all parties (including the Debtors, the Trust Administrator, the holders of the DIP Credit Agreement Claims, and the holders of Allowed General Unsecured Claims) for all U.S. federal and applicable state and local income tax purposes.

7.3. Payment of Taxes. The Trust Administrator shall be responsible for payment, out of the Avoidance Action Trust Assets, of any taxes imposed on the Trust (other than in respect of the Other Debtor Residual Assets) or the Avoidance Action Trust Assets, including the Avoidance Action Trust Claims Reserve. The Trust Administrator shall be responsible for payment, out of the Other Debtor Residual Assets of any taxes imposed on the Trust in respect of the Other Debtor Residual Assets or on the Other Debtor Residual Assets. In the event, and to the extent, any Cash retained on account of Disputed General Unsecured Claims, Unresolved Term Loan Avoidance Action Claims or Unresolved Other Avoidance Action Claims in the Avoidance Action Trust Claims Reserve is insufficient to pay the portion of any such taxes attributable to the taxable income arising from the assets allocable to, or retained on account of, Disputed General Unsecured Claims, Unresolved Term Loan Avoidance Action Claims or Unresolved Other Avoidance Action Claims, such taxes shall be (i) reimbursed from any subsequent Cash amounts retained on account of Disputed General Unsecured Claims, Unresolved Term Loan Avoidance Action Claims or Unresolved Other Avoidance Action Claims, or (ii) to the extent such Disputed General Unsecured Claims, Unresolved Term Loan Avoidance Action Claims or Unresolved Other Avoidance Action Claims subsequently have been resolved, deducted from any amounts otherwise distributable by the Trust Administrator as a result of the resolution of such Disputed General Unsecured Claims, Unresolved Term Loan Avoidance Action Claims or Unresolved Other Avoidance Action Claims.

7.4. <u>Tax Reporting</u>.

- (a) The Trust Administrator shall file (or cause to be filed) Tax Returns for the Trust treating the Trust (except the Avoidance Action Trust Claims Reserve or as otherwise provided in Section 7.1(b) above) as a grantor trust pursuant to Treasury Regulation section 1.671-4(a) and in accordance with the applicable provisions of this Section 7.4. The Trust Administrator also shall annually send to each Trust Beneficiary a separate statement setting forth such Trust Beneficiary's share of items of income, gain, loss, deduction, or credit of the Trust (including, for the avoidance of doubt, earnings on the Avoidance Action Trust Administrative Cash, the Avoidance Action Trust SEC Reporting Cash, the GUC Trust Supplemental Cash and the Other Supplemental Cash) and shall instruct all Trust Beneficiaries to report such items on their respective U.S. federal income Tax Returns or to forward the appropriate information to their respective beneficial holders with instructions to report such items on their U.S. federal income Tax Returns. The Trust Administrator also shall file (or cause to be filed) any other statements, returns, or disclosures relating to the Trust that are required by any governmental unit.
- (b) Allocations of the Trust's taxable income among the Trust Beneficiaries shall be determined by reference to the manner in which an amount of Cash equal to such taxable income would be distributed (without regard to any restrictions on distributions described herein) if, immediately prior to such deemed distribution, the Trust had distributed all of its other assets (valued at their tax book value and other than assets attributable to the

Avoidance Action Trust Claims Reserve) to the Trust Beneficiaries, in each case up to the tax book value of the assets treated as contributed by such Trust Beneficiaries, adjusted for prior taxable income and loss and taking into account all prior and concurrent distributions from the Trust. Similarly, taxable loss of the Trust shall be allocated by reference to the manner in which an economic loss would be borne immediately after a liquidating distribution of the remaining Avoidance Action Trust Assets and Other Debtor Residual Trust Assets. The tax book value of the Avoidance Action Trust Assets and Other Debtor Residual Trust Assets for this purpose shall equal their fair market value on the Avoidance Action Trust Transfer Date and Other Debtor Residual Trust Assets Transfer Date, as applicable, adjusted in accordance with tax accounting principles prescribed by the Tax Code, applicable Treasury Regulations, and other applicable administrative and judicial authorities and pronouncements.

- (c) The Trust Administrator shall (x) treat the Avoidance Action Trust Claims Reserve for U.S. federal income tax purposes as either (i) a "disputed ownership fund" governed by Treasury Regulation section 1.468B-9 by timely so electing or (ii) a "complex trust," provided, however, that if the Trust is treated as a "disputed ownership fund" or as a "complex trust" pursuant to Section 7.1(b) above, then the Avoidance Action Trust Claims Reserve shall be treated in the same manner, and (y) to the extent permitted by applicable law, report consistently with the foregoing for state and local income tax purposes. Any determination made pursuant to this Section 7.4 shall be conclusive and binding on all parties (including the Debtors, the Trust Administrator, the holders of the DIP Credit Agreement Claims, and the holders of Allowed General Unsecured Claims) for U.S. federal, state, and local income tax purposes.
 - 7.5. Tax Withholdings. The Trust Administrator shall withhold and pay to the appropriate taxing authority all amounts required to be withheld pursuant to the Tax Code, Treasury Regulations or other applicable requirements, including any provision of any foreign, state or local tax law, with respect to any payment or distribution to the Trust Beneficiaries. All such amounts withheld, and paid to the appropriate taxing authority, shall be treated as amounts distributed to such Trust Beneficiaries for all purposes of this Trust Agreement. The Trust Administrator shall be authorized to collect such tax information from the Trust Beneficiaries (including social security numbers or other tax identification numbers) as it in its sole discretion deems necessary to effectuate the Plan, the Confirmation Order and this Trust Agreement, or to comply with any applicable withholding or reporting requirement. The Trust Administrator may refuse to make a distribution to any Trust Beneficiary that fails to furnish such information in a timely fashion, until such information is furnished; provided, however, that upon a Trust Beneficiary furnishing such information, the Trust Administrator shall make such distribution to which such Trust Beneficiary is entitled, without interest.
 - 7.6. Expedited Determination of Taxes. The Trust Administrator may request an expedited determination of taxes of the Trust, including the Avoidance Action Trust Claims Reserve, under Section 505(b) of the Bankruptcy Code for any or all Tax Returns filed for, or on behalf of, the Trust for any or all taxable periods (or part thereof) through the dissolution of the Trust.

7.7. [Intentionally omitted.]

- 7.8. <u>Delivery of Statement of Transfers</u>. If the Trust Administrator elects to treat (i) the Trust, pursuant to and to the extent provided in <u>Section 7.1</u> above and/or (ii) the Avoidance Action Trust Claims Reserve as a disputed ownership fund within the meaning of Treasury Regulation section 1.468B-9, then following the Avoidance Action Trust Transfer Date (but in no event later than February 15th of the calendar year following the Avoidance Action Trust Transfer Date), MLC shall provide a "§ 1.468B-9 Statement" to the Trust Administrator in accordance with Treasury Regulation section 1.468B-9(g).
- 7.9. <u>Allocation of Distributions Between Principal and Interest</u>. All deemed distributions (including deemed transfers pursuant to <u>Section 7.1(b)(i)</u>) in connection with the allowance of any Allowed General Unsecured Claim shall be allocated first to the principal amount of such Allowed General Unsecured Claim, as determined for federal income tax purposes, and thereafter, to the remaining portion of such Allowed General Unsecured Claim, if any.

ARTICLE VIII POWERS OF AND LIMITATIONS ON THE TRUST ADMINISTRATOR

- 8.1. Powers of the Trust Administrator.
- (I) Other Than in Respect of the Other Debtor Residual Accepted Assets
- (a) Pursuant to the terms of the Plan and the Confirmation Order, the Trust Administrator shall have various powers, duties and responsibilities concerning the prosecution of and resolution of the Term Loan Avoidance Action, maximizing the property of the Trust, the disposition of the Avoidance Action Trust Assets and the administration of the Trust. In addition, the Trust Administrator shall coordinate with the GUC Trust Administrator to maximize efficiency in distributions to general unsecured creditors in any situation where such coordination would be beneficial.
- (b) The Trust Administrator shall have only such rights, powers and privileges expressly set forth in the Plan, the Confirmation Order or this Trust Agreement and as otherwise provided by applicable law. Subject to the Plan, the Confirmation Order and other provisions herein, including the provisions relating to approvals of the Trust Monitor, the Trust Administrator shall be expressly authorized to undertake the following actions, in the Trust Administrator's good faith judgment, in the best interests of the Trust Beneficiaries and in furtherance of the purpose of the Trust:
 - (i) hold and manage the Avoidance Action Trust Assets;
- (ii) hold legal title to any and all rights of the Trust Beneficiaries in, to or arising from the Avoidance Action Trust Assets, for the benefit of the Trust Beneficiaries that are entitled to distributions therefrom under the Plan, whether, in the case of GUC

Beneficiaries, their General Unsecured Claims are Allowed on or after the Avoidance Action Trust Transfer Date;

- (iii) prosecute and, if appropriate, sell, grant liens upon (subject to Section 6.1(d) hereof), settle and resolve, abandon and/or dismiss the Term Loan Avoidance Action;
- (iv) execute all agreements, instruments and other documents (including, without limitation, any loan agreements or sale agreements for the purposes of funding the fees and expenses of the Trust), and effect all other actions necessary or appropriate to dispose of the Avoidance Action Trust Assets;
- (v) monitor and enforce the implementation of the Plan insofar as relating to this Trust Agreement, the Avoidance Action Trust Assets or the Trust;
- (vi) calculate and implement distributions of the GUC Distributable Trust Assets obtained through the exercise of its power and authority as contemplated by the Plan, the Confirmation Order and this Trust Agreement and in accordance with the interests of the holders of Allowed General Unsecured Claims;
- (vii) retain, pay, oversee and direct the services of, and terminate Trust Professionals in accordance with <u>Section 8.3</u> hereof to carry out its duties and obligations hereunder, in all cases in accordance with the Budget;
- (viii) pay the reasonable fees and expenses of the Trust Administrator and Trust Monitor, in all cases in accordance with the Budget;
- (ix) incur and pay all reasonable expenses, satisfy ordinary course liabilities and make all other payments reasonable and necessary to administer and dispose of the Avoidance Action Trust Assets, in all cases in accordance with the Budget;
- (x) invest monies received by the Trust, the Trust Administrator or otherwise held by the Trust or the Trust Administrator in accordance with Section 8.4 hereof;
- (xi) protect and enforce the rights to the Avoidance Action Trust Assets vested in the Trust Administrator by this Trust Agreement by any method deemed reasonably appropriate, including by judicial proceedings or pursuant to any applicable bankruptcy, insolvency, moratorium or similar law and general principles of equity;
- (xii) vote any claim or interest held by the Trust in a case under the Bankruptcy Code and receive any distribution therefrom for the benefit of the Trust;
- (xiii) make all necessary filings in accordance with any applicable law, statute or regulation;
- (xiv) purchase customary insurance coverage in accordance with <u>Section</u> 6.9 hereof;

- (xv) assert and/or waive any applicable privileges (legal or otherwise) on behalf of the Trust, or with respect to the Avoidance Action Trust Assets held by the Debtors at any time (prepetition or postpetition);
 - (xvi) maintain the books and records of the Trust;
- (xvii) open, maintain and close any bank, securities or other accounts that are necessary and appropriate to manage the Avoidance Action Trust Assets, including but not limited to the accounts listed on <u>Exhibit A</u> hereto;
- (xviii) receive from the Debtors and administer the Avoidance Action Trust SEC Reporting Cash in accordance with <u>Section 2.3(e)</u> hereof and file such reports as may be required pursuant to the applicable rules, regulations and interpretations of the SEC;
- (xix) receive from the GUC Trust and administer and utilize the GUC Trust Supplemental Cash in accordance with Section 2.3(f) hereof; and
- (xx) perform such functions and take such actions as are provided for or permitted in the Plan, the Confirmation Order, this Trust Agreement, the Settlement Agreement, the Litigation Cost Advance Agreement, or any other agreement executed pursuant to the Plan and take any other actions as it may deem to be reasonably necessary or appropriate to realize, preserve and dispose of the Avoidance Action Trust Assets.
 - (c) [Intentionally omitted.]
- (d) In all circumstances, the Trust Administrator shall act in the best interests of all Trust Beneficiaries and in furtherance of the purpose of the Trust, and in a manner not inconsistent with the best interests of the Trust Beneficiaries and consistent with the Budget. The Trust Administrator shall not take any action inconsistent with the purpose of the Trust, or take (or fail to take) any action that would cause the Trust (other than the Avoidance Action Trust Claims Reserve) to fail to qualify as a liquidating trust within the meaning of Treasury Regulation section 301.7701-4(d) that is treated as a grantor trust.
- (e) Notwithstanding any provision herein to the contrary, the Trust Administrator shall not serve on the board of directors, management committee or any similar governing body of any non-Debtor subsidiary of MLC, where the charter, limited liability company agreement, partnership agreement or other similar constituent document of such subsidiary does not provide for a liquidating purpose for such subsidiary. Except as otherwise provided in this Trust Agreement, the Trust Administrator will not be required to obtain the order or approval of the Bankruptcy Court, or any other court of competent jurisdiction in, or account to the Bankruptcy Court or any other court of competent jurisdiction for, the exercise of any right, power or privilege conferred hereunder. Notwithstanding the foregoing, where the Trust Administrator determines, in its reasonable discretion, that it is necessary, appropriate or desirable, the Trust Administrator will have the right to submit to the Bankruptcy Court or any other court of competent jurisdiction any question or questions regarding any specific action proposed to be taken by the Trust Administrator with respect to this Trust Agreement, the Trust, or the Avoidance Action Trust Assets, including the

administration and distribution of the Avoidance Action Trust Assets and the termination of the Trust. Pursuant to the Plan, the Bankruptcy Court has retained jurisdiction for such purposes and may approve or disapprove any such proposed action upon motion by the Trust Administrator.

(II) <u>In Respect of the Other Debtor Residual Accepted Assets</u>

The Trust Administrator shall have the rights, powers and privileges to act in respect of the Other Debtor Residual Accepted Assets, if any, in the manner set forth in Section 8.1(I), mutatis mutandis. In all such circumstances, the Trust Administrator shall act in the best interests of DIP Lenders and in furtherance of the purpose of the Trust, and in a manner not inconsistent with the best interests of the DIP Lenders. For the avoidance of doubt, the Trust Administrator shall not be obligated to undertake any activities in respect of the Other Debtor Residual Trust Assets unless there shall be available sufficient Other Debtor Residual Trust Administrative Cash to pay in full its fees, costs and expenses in respect thereof.

Limitations on the Trust Administrator. The Trust Administrator shall not 8.2. be authorized to engage, in its capacity as Trust Administrator, in any trade or business with respect to the Avoidance Action Trust Assets or to take (or fail to take) any action that would cause the Trust (other than the Avoidance Action Trust Claims Reserve) to fail to qualify as a liquidating trust within the meaning of Treasury Regulation section 301.7701-4(d) that is treated as a grantor trust. The Trust Administrator shall take such actions consistent with the prompt orderly disposition of the Avoidance Action Trust Assets and the Other Debtor Residual Accepted Assets, if any, as required by applicable law and consistent with the treatment of the Trust (other than the Avoidance Action Trust Claims Reserve) as a liquidating trust within the meaning of Treasury Regulation section 301.7701-4(d) that is treated as a grantor trust, to the extent such actions are permitted by this Trust Agreement. The Trust Administrator shall, in its capacity as Trust Administrator, be restricted to (x) the liquidation of the Trust on behalf, and for the benefit, of the Trust Beneficiaries and the distribution and application of Avoidance Action Trust Assets for the purposes set forth in, and the conservation and protection of the Avoidance Action Trust Assets and the administration thereof, and (y) the liquidation of the Trust on behalf, and for the benefit, of the DIP Lenders and the distribution and application of Other Debtor Residual Trust Assets for the purposes set forth in, and the conservation and protection of the Other Debtor Residual Trust Assets and the administration thereof, in each case in accordance with, the provisions of the Plan, the Confirmation Order and this Trust Agreement.

8.3. Agents and Professionals.

(a) The Trust Administrator on behalf of the Trust may, but shall not be required to, from time to time enter into contracts with, consult with and retain Trust Professionals, on such terms as the Trust Administrator deems appropriate in accordance with the terms hereof and (other than in respect of the Other Debtor Residual Accepted Assets) in accordance with the Budget. None of the professionals that represented parties-in-interest in the Chapter 11 Cases shall be precluded from being engaged by the Trust Administrator solely on account of their service as a professional for such parties-in-interest prior to the Avoidance

Action Trust Transfer Date or the Other Debtor Residual Assets Transfer Date, as the case may be.

- (b) After the Avoidance Action Trust Transfer Date, Trust Professionals shall be required to submit reasonably detailed invoices on a monthly basis to the Trust Administrator, the Trust Monitor, the DIP Lenders and the Capital Providers, including in such invoices a description of the work performed, the individuals who performed such work, the hourly rate of such person, plus an itemized statement of expenses for which payment is sought. Invoices submitted to the DIP Lenders and Capital Providers may be in a form reasonably necessary to protect the attorney-client privilege. For the avoidance of doubt, any and all payments by the Trust from funds that (i) originated from the Litigation Cost Advance or the Supplemental Capital, or (ii) that are contained in the Funding Account or the Capital Provider Funding Account shall be subject to this invoice disclosure requirement. No payment on account of such invoices shall be made until 15 days after presentation of the invoices to the DIP Lenders and the Capital Providers. After such period has passed, the Trust Administrator shall timely pay all such invoices that are not disputed by the Trust Administrator or the Trust Monitor. If either or both of the DIP Lenders or the Capital Providers have questions or concerns about any invoice submitted to the Trust, or about any other issue relating to the Trust's finances or administration, they shall be entitled to discuss the matter with the Trust Monitor, as well as with the Trust Administrator and/or counsel for the Trust, as appropriate; and the Trust Monitor will consult with the DIP Lenders or Capital Providers, as applicable, and give good faith consideration to any objection that the DIP Lenders or Capital Providers raise before approval of payment by the Trust Administrator or non-objection by the Trust Monitor. The Trust Monitor, Trust Administrator, and/or counsel for the Trust, as appropriate, will engage in good faith discussions and attempt in good faith to provide information reasonably requested by either or both DIP Lenders or the Capital Providers. In the event of any dispute concerning the entitlement to, or the reasonableness of any compensation and/or expenses of any Trust Professionals, either the Trust Administrator or the affected Trust Professional may petition the Bankruptcy Court to resolve the dispute.
- (c) Except as permitted by Section 6.1(b), (c), and (d), and Section 2.3(e) and (f), all payments to Trust Professionals (other than in respect of Other Debtor Residual Trust Assets) shall be paid out of the Avoidance Action Trust Administrative Cash. Payments to Trust Professionals for activities in respect of the Other Debtor Residual Trust Assets shall be paid out of the Other Debtor Residual Trust Administrative Cash.

8.4. Investment of Trust Cash.

(a) The Trust Administrator shall establish segregated accounts for the Trust Cash as follows: (i) Distributable Trust Cash which shall be held in trust for the benefit of the Trust Beneficiaries; (ii) Distributable Other Debtor Residual Cash, which shall be held in trust for the benefit of the DIP Lenders and on which the DIP Lenders shall have a lien; (iii) Avoidance Action Trust Administrative Cash (excluding the Litigation Cost Advance) and Other Debtor Residual Trust Administrative Cash which shall be used to pay the administrative expenses of the Trust, on which the DIP Lenders shall have a lien; (iv) Avoidance Action Trust SEC Reporting Cash which shall be used to satisfy Avoidance Action Trust SEC Reporting Costs in accordance with Section 2.3(e) hereof; (v) GUC Trust

Supplemental Cash which shall be used to satisfy current and projected fees and expenses of the Trust (including for the payment of Trust Professionals and any tax liabilities) in accordance with Section 2.3(f) hereof; (vi) Other Supplemental Cash, which shall be used to satisfy the current and projected future fees, costs and expenses (including, without limitation any Avoidance Action Trust SEC Reporting Costs, the fees and expenses of Trust Professionals, and any tax liabilities) in accordance with Section 6.1(d) hereof; (v) Litigation Cost Advance, which shall be held in the Funding Account, which Funding Account shall be subject to a deposit account control agreement acceptable to the DIP Lenders and the Trust and on which the DIP Lenders shall have a first priority lien; and (vi) the Supplemental Capital, which shall be held in the Capital Provider Funding Account, which upon the request of the Capital Providers shall be subject to a deposit account control agreement acceptable to the Capital Providers and the Trust.

- (b) The Trust Administrator shall invest the Trust Cash (including any earnings thereon or proceeds thereof) in the manner set forth in this Section 8.4, but shall otherwise be under no liability for interest or income on any monies received by the Trust hereunder and held for distribution or payment to the Trust Beneficiaries, except as such interest shall actually be received. Investment of any Trust Cash shall be administered in accordance with the general duties and obligations hereunder. The right and power of the Trust Administrator to invest the Trust Cash and the proceeds thereof, or any income earned by the Trust, shall be limited to investing such Trust Cash (pending distribution or disbursement in accordance with the Plan or this Trust Agreement) in Permissible Investments; provided, however, that such Permissible Investments shall be limited to include only those investments that a liquidating trust, within the meaning of Treasury Regulations section 301.7701-4(d), may be permitted to hold, pursuant to the Treasury Regulations, or any modification in the IRS guidelines, whether set forth in IRS rulings, other IRS pronouncements or otherwise.
- (c) For the avoidance of doubt, the Trust is not, and will not hold itself out as, an "investment company" as such term is understood under the Investment Company Act of 1940, and is prohibited from investing, reinvesting or trading in securities (other than making any Permissible Investments as contemplated by the Plan, the Confirmation Order and this Trust Agreement) or conducting any trade or business other than implementing the Plan and distributing Distributable Trust Assets under the Plan and this Trust Agreement.
 - 8.5. Communication with the GUC Trust Administrator. The Trust Administrator shall communicate with the GUC Trust Administrator to obtain such information regarding, as of a given date, (A) the holders and amounts of General Unsecured Claims, Disputed General Unsecured Claims, Unresolved Term Loan Avoidance Action Claims, Unresolved Other Avoidance Action Claims and Resolved Allowed General Unsecured Claims, (B) the respective Maximum Amounts of all Disputed General Unsecured Claims, Unresolved Term Loan Avoidance Action Claims and Unresolved Other Avoidance Action Claims, (C) the Current Total Amount, (D) the Aggregate Maximum Amount, (E) any components of the foregoing and (E) any other information within the custody or control of the GUC Trust Administrator which shall be necessary or desirable in order for the Trust Administrator to timely make any calculation

or determination hereunder to identify and make distributions to the GUC Beneficiaries and to maintain any books and records required to be maintained, or necessary or desirable for the Trust Administrator or the Trust Monitor to fulfill their respective functions, hereunder; provided, however, that the provision of such information shall be under appropriate arrangements of confidentiality to the extent such information has at the time not been publicly disclosed.

8.6. <u>Termination</u>. The duties, responsibilities and powers of the Trust Administrator will terminate when the Trust is dissolved and terminated pursuant to <u>Article IV</u> hereof and the Trust Administrator has performed all of its obligations under <u>Section 4.3</u>, by an order of the Bankruptcy Court or by entry of a final decree closing the Debtors' cases before the Bankruptcy Court; <u>provided</u>, <u>however</u>, that <u>Sections 9.4, 9.5</u> and <u>9.6</u> hereof shall survive such termination, dissolution and entry.

ARTICLE IX ADDITIONAL MATTERS CONCERNING THE TRUST ADMINISTRATOR

- 9.1. Reliance by Trust Administrator. Except as otherwise provided in the Plan, the Confirmation Order or this Trust Agreement, the Trust Administrator may rely and shall be protected in acting upon any resolution, statement, instrument, opinion, report, notice, request, consent, order or other paper or document reasonably believed by the Trust Administrator to be genuine and to have been signed or presented by the proper party or parties.
- 9.2. Liability to Third Persons. To the fullest extent permitted by applicable law, the Trust Administrator Parties shall not be subject to any personal liability whatsoever, in tort, contract or otherwise, to any person (including, in the case of the Trust Administrator, to any Trust Professionals retained by the Trust Administrator in accordance with this Trust Agreement) in connection with the Avoidance Action Trust Assets, the Other Debtor Residual Trust Assets or the affairs of the Trust and shall not be liable with respect to any action taken or omitted to be taken in good faith, except for actions and omissions determined by a Final Order of the Bankruptcy Court to be due to their respective willful misconduct (including, but not limited to, conduct that results in a personal profit at the expense of the Trust), gross negligence, fraud, malpractice, criminal conduct, unauthorized use of confidential information that causes damages, breach of fiduciary duty (to the extent applicable), or *ultra vires* acts, and all such persons shall look solely to the Avoidance Action Trust Assets (other than in respect of the Other Debtor Residual Accepted Assets and activities related thereto) or the Other Debtor Residual Trust Assets (in respect of the Other Debtor Residual Accepted Assets and activities related thereto) for satisfaction of claims of any nature arising in connection with affairs of the Trust.
- 9.3. <u>Non-liability of Trust Administrator for Acts of Others</u>. Except as provided herein, nothing contained in the Plan, the Confirmation Order or this Trust Agreement shall be deemed to be an assumption by the Trust Administrator of any of the liabilities, obligations or duties of the Debtors or shall be deemed to be or contain a covenant or agreement by the Trust Administrator to assume or accept any such liability,

obligation or duty. Any successor Trust Administrator may accept and rely upon any accounting made by or on behalf of any predecessor Trust Administrator hereunder, and any statement or representation made as to the assets comprising the Avoidance Action Trust Assets or the Other Debtor Residual Trust Assets, or as to any other fact bearing upon the prior administration of the Trust, so long as it has a good faith basis to do so. The Trust Administrator shall not be liable for having accepted and relied in good faith upon any such accounting, statement or representation if it is later proved to be incomplete, inaccurate or untrue. Neither the Trust Administrator nor any successor Trust Administrator shall be liable for any act or omission of any predecessor Trust Administrator, nor have a duty to enforce any claims against any predecessor Trust Administrator on account of any such act or omission, unless directed in good faith to do so by the Trust Monitor.

- 9.4. Exculpation. As of the earlier of the Avoidance Action Trust Transfer Date and the Other Debtor Residual Assets Transfer Date, to the fullest extent permitted by applicable law, the Trust Administrator Parties shall be and hereby are exculpated by all Persons, including holders of DIP Credit Agreement Claims, General Unsecured Claims and Units and other parties-in-interest, from any and all claims, causes of action and other assertions of liability arising out of the discharge of their respective powers and duties conferred by the Plan, the Confirmation Order, this Trust Agreement or any Order of the Bankruptcy Court entered pursuant to or in furtherance of the Plan, or applicable law or otherwise, except for actions or omissions to act that are determined by Final Order of the Bankruptcy Court to have arisen out of each such Trust Administrator Party's own respective willful misconduct (including, but not limited to, conduct that results in a personal profit at the expense of the Trust), gross negligence, fraud, malpractice, criminal conduct, unauthorized use of confidential information that causes damages, breach of fiduciary duty (to the extent applicable), or ultra vires acts. No holder of a DIP Credit Agreement Claim, General Unsecured Claim or other party-in-interest will have or be permitted to pursue any claim or cause of action against the Trust Administrator Parties or the Trust, for making payments and distributions in accordance with the Plan, the Confirmation Order or the this Trust Agreement or for implementing the provisions thereof. Any action taken or omitted to be taken with the express approval of the Bankruptcy Court and, in the case of action taken in respect of the Other Debtor Residual Accepted Assets, with the approval or at the direction of the DIP Lenders will conclusively be deemed not to constitute willful misconduct, gross negligence, fraud, malpractice, criminal conduct, unauthorized use of confidential information that causes damages, breach of fiduciary duty, or *ultra vires* acts; *provided*, *however*, that notwithstanding any provision herein to the contrary, the Trust Administrator shall not be obligated to comply with a direction of the Trust Monitor, whether or not express, which would result in a change to the distribution provisions of the Plan, the Confirmation Order or this Trust Agreement.
- 9.5. <u>Limitation of Liability</u>. In no event shall the Trust Administrator Parties be liable for punitive, exemplary, consequential, special or other damages for a breach of, or otherwise in connection with, this Trust Agreement under any circumstances.

9.6. Indemnity.

- (a) To the fullest extent permitted by applicable law, the Trust Administrator Parties shall be indemnified by the Trust from the Avoidance Action Trust Assets (other than in respect of the Other Debtor Residual Accepted Assets and activities related thereto) or the Other Debtor Residual Trust Assets (in respect of the Other Debtor Residual Accepted Assets and activities related thereto) for any losses, claims, damages, liabilities and expenses occurring after the earlier of the Avoidance Action Trust Transfer Date and the Other Debtor Residual Assets Transfer Date, including reasonable attorneys' fees, disbursements and related expenses which the Trust Administrator Parties may incur or to which the Trust Administrator Parties may become subject in connection with any action, suit, proceeding or investigation brought by or threatened against one or more of the Trust Administrator Parties on account of the acts or omissions in their capacity as, or on behalf of, the Trust Administrator; provided, however, that the Trust shall not be liable to indemnify any Trust Administrator Party for any act or omission arising out of such Trust Administrator Party's respective actions that are determined by a Final Order of the Bankruptcy Court to be willful misconduct (including, but not limited to, conduct that results in a personal profit at the expense of the Trust), gross negligence, fraud, malpractice, criminal conduct, unauthorized use of confidential information that causes damages, breach of fiduciary duty (to the extent applicable), or *ultra vires* acts. Notwithstanding any provision herein to the contrary, the Trust Administrator Parties shall be entitled to obtain advances from the Trust to cover their reasonable expenses of defending themselves in any action brought against them as a result of the acts or omissions, actual or alleged, of an Trust Administrator Party in its capacity as such; provided, however, that the Trust Administrator Parties receiving such advances shall repay the amounts so advanced to the Trust immediately upon the entry of a final, non-appealable judgment or order finding that such Trust Administrator Parties were not entitled to any indemnity under the provisions of this Section 9.6. Any amounts payable to any Trust Administrator Party pursuant to this Section 9.6 (other than in respect of the Other Debtor Residual Accepted Assets and activities related thereto) shall be satisfied as follows: (i) first from the Avoidance Action Trust Administrative Cash, (ii) second from the Distributable Trust Cash, if any; provided, however, that the use of Distributable Trust Cash as contemplated in clause (ii) of the foregoing shall be subject to the prior approval by the Bankruptcy Court, as provided in Section 6.1(b); (iii) third from the GUC Trust Supplemental Cash, if any; and (iv) fourth from the Other Supplemental Cash, if any. Any amounts payable to any Trust Administrator Party pursuant to this Section 9.6 in respect of the Other Debtor Residual Accepted Assets and activities related thereto shall be satisfied as follows: (i) first from the Other Debtor Residual Trust Administrative Cash, and (ii) second from the Distributable Other Debtor Residual Trust Cash, if any.
- (b) Anything to the contrary in this Trust Agreement or in any other agreement notwithstanding, to the extent that the Avoidance Action Trust Administrative Cash, Distributable Trust Cash, GUC Trust Supplemental Cash or Other Supplemental Cash, or the Other Debtor Residual Trust Administrative Cash or Distributable Other Debtor Residual Trust Cash, as the case may be, shall be insufficient to fully indemnify the Trust Administrator Parties or to provide advances to the Trust Administrator Parties in accordance with Section 9.6(a), the Trust Administrator Parties shall be indemnified and shall be entitled to obtain advances, first from the Other GUC Trust Administrative Cash (as defined in the

GUC Trust Agreement), and second from the GUC Trust Distributable Assets (as defined in the GUC Trust Agreement), to the same extent as the GUC Trust Administrator Parties under Section 9.6 of the GUC Trust Agreement or any successor provision thereunder, as provided in Section 9.6 of the GUC Trust Agreement in effect on the date hereof.

(c) The foregoing indemnities in respect of any Trust Administrator Party shall survive the termination of such Trust Administrator Party from the capacity for which they are indemnified.

9.7. Compensation and Expenses.

- (a) The Trust Administrator shall receive fair and reasonable compensation for its services (other than in respect of the Other Debtor Residual Accepted Assets and activities related thereto), to be paid out of the Avoidance Action Trust Administrative Cash in accordance with the approved Budget (or from the Avoidance Action Trust SEC Reporting Cash in accordance with Section 2.3(e) hereof, the GUC Trust Supplemental Cash in accordance with Section 2.3(f) hereof, or the Other Supplemental Cash in accordance with Section 6.1(d) hereof). The Trust Administrator shall be entitled, without the need for approval of the Bankruptcy Court, to reimburse itself on a monthly basis (i) from the Avoidance Action Trust Administrative Cash, the GUC Trust Supplemental Cash or the Other Supplemental Cash, as applicable, for such compensation and all reasonable out-of-pocket expenses actually incurred in the performance of duties in accordance with this Trust Agreement and the Budget and (ii) from the Avoidance Action Trust SEC Reporting Cash for such compensation and all reasonable out-of-pocket expenses actually incurred in the performance of duties in accordance with this Trust Agreement and with the approval of the Trust Monitor.
- (b) The Trust Administrator shall receive compensation for its services in respect of the Other Debtor Residual Accepted Assets and activities related thereto, to be paid out of the Other Debtor Residual Trust Administrative Cash, as the Trust Administrator and a majority in interest of the DIP lenders shall agree; provided that the Trust Administrator shall not be obligated to undertake any activities in respect of the Other Debtor Residual Accepted Assets unless such compensation arrangements shall be acceptable to the Trust Administrator in its sole discretion.
 - 9.8. No Personal Financial Liability. No provision of the Plan, Confirmation Order or this Trust Agreement shall be construed as requiring the Trust Administrator to expend or risk its own funds or otherwise to incur any personal financial liability (x) in the performance of any of its duties thereunder or hereunder, including any situation where the Avoidance Action Trust Assets are insufficient to permit the administration of the Trust or distributions as contemplated herein or the payment of fees and expenses of the Trust Professionals, or (y) in the exercise of any of its rights or powers afforded hereunder or thereunder.

ARTICLE X SUCCESSOR TRUST ADMINISTRATORS

- 10.1. <u>Resignation</u>. The Trust Administrator may resign from the Trust by giving at least sixty (60) days' prior written notice thereof to the Trust Monitor. Such resignation shall become effective on the later to occur of (x) the date specified in such written notice and (y) the effective date of the appointment of a successor Trust Administrator in accordance with <u>Section 10.4</u> hereof and such successor's acceptance of such appointment in accordance with <u>Section 10.5</u> hereof.
- at any time petition the Bankruptcy Court for the removal of the Trust Administrator, but only for good cause shown. Such removal shall become effective on the date ordered by the Bankruptcy Court, provided that such removal shall not become effective until the appointment of a successor Trust Administrator in accordance with Section 10.4 hereof and such successor's acceptance of such appointment in accordance with Section 10.5 hereof. The services of the Trust Administrator shall also terminate upon its bankruptcy, provided that such termination shall not become effective until the appointment of a successor Trust Administrator in accordance with Section 10.4 hereof and such successor's acceptance of such appointment in accordance with Section 10.5 hereof.
- 10.3. Effect of Resignation or Removal. The resignation, removal or bankruptcy of the Trust Administrator shall not operate to terminate the Trust or to revoke any existing agency created pursuant to the terms of the Plan, the Confirmation Order or this Trust Agreement or invalidate any action theretofore taken by the Trust Administrator. The exculpation, indemnity and limitation of liability provisions of Article X of this Trust Agreement shall survive the resignation, removal or bankruptcy of the Trust Administrator. All fees and expenses properly incurred by the Trust Administrator prior to the resignation, Incompetency, removal or bankruptcy of the Trust Administrator shall be paid from the Avoidance Action Trust Administrative Cash, the GUC Trust Supplemental Cash or the Other Supplemental Cash (other than in respect of the Other Debtor Residual Accepted Assets and activities related thereto), from the Other Debtor Residual Trust Administrative Cash (in respect of the Other Debtor Residual Accepted Assets and activities related thereto) or from Avoidance Action Trust SEC Reporting Cash (in respect of Avoidance Action Trust SEC Reporting Costs in accordance with Section 2.3(e) hereof), unless such fees and expenses are disputed by (x) the Trust Monitor or (y) the successor Trust Administrator, in which case the Bankruptcy Court shall resolve the dispute and any disputed fees and expenses of the predecessor Trust Administrator that are subsequently allowed by the Bankruptcy Court shall be paid from the Avoidance Action Trust Administrative Cash, the GUC Trust Supplemental Cash or the Other Supplemental Cash (other than in respect of the Other Debtor Residual Accepted Assets and activities related thereto), from the Other Debtor Residual Trust Administrative Cash (in respect of the Other Debtor Residual Accepted Assets and activities related thereto) or from Avoidance Action Trust SEC Reporting Cash (in respect of Avoidance Action Trust SEC Reporting Costs in accordance with <u>Section 2.3(e)</u> hereof). In the event of the resignation, removal or bankruptcy of the Trust Administrator, such Trust Administrator shall:

- (a) promptly execute and deliver such documents, instruments and other writings as may be reasonably requested by the successor Trust Administrator or directed by the Bankruptcy Court to effect the termination of such Trust Administrator's capacity under this Trust Agreement;
- (b) promptly deliver to the successor Trust Administrator all documents, instruments, records and other writings related to the Trust as may be in the possession of such Trust Administrator; and
- (c) otherwise assist and cooperate in effecting the assumption of its obligations and functions by such successor Trust Administrator.
 - 10.4. Appointment of Successor. In the event of the resignation, removal, Incompetency or bankruptcy of the Trust Administrator, the Trust Monitor shall promptly appoint a successor Trust Administrator, provided that such appointment shall not take effect unless approved by the Bankruptcy Court upon the petition of the Trust Monitor and until the successor Trust Administrator shall have delivered written acceptance of its appointment as described Section 10.5 below. If a successor Trust Administrator does not take office within thirty (30) days after the resignation, removal, Incompetency or bankruptcy of the retiring Trust Administrator, the Bankruptcy Court, upon its own motion or the motion of the retiring Trust Administrator or any Trust Beneficiary, shall appoint a successor Trust Administrator.
 - Acceptance of Appointment by Successor Trust Administrator. Any 10.5. successor Trust Administrator appointed hereunder shall execute an instrument accepting its appointment and shall deliver one counterpart thereof to the Bankruptcy Court for filing and to the Trust Monitor and, in case of the Trust Administrator's resignation, to the resigning Trust Administrator. Thereupon, such successor Trust Administrator shall, without any further act, become vested with all the duties, powers, rights, obligations, title, discretion and privileges of its predecessor in the Trust with like effect as if originally named Trust Administrator and shall be deemed appointed pursuant to Bankruptcy Code Section 1123(b)(3)(B); provided, however, such successor Trust Administrator shall file an amendment to the Certificate of Trust with the Secretary of State as required by the Delaware Act. The predecessor Trust Administrator shall duly assign, transfer and deliver to such successor Trust Administrator all Avoidance Action Trust Assets held by such predecessor Trust Administrator hereunder and shall, as directed by the Bankruptcy Court or reasonably requested by such successor Trust Administrator, execute and deliver an instrument or instruments conveying and transferring to such successor Trust Administrator upon the trusts herein expressed, all the duties, powers, rights, obligations, title, discretion and privileges of the predecessor Trust Administrator.
 - 10.6. <u>Successor Entity to Trust Administrator</u>. Any business entity into which the Trust Administrator may be merged or converted or with which it may be consolidated, or any entity resulting from any merger, conversion or consolidation to which the Trust Administrator shall be a party, or any entity succeeding to all or substantially all of the corporate trust business of the Trust Administrator, shall be the successor of the Trust Administrator hereunder, without the execution or filing of any paper or any further act on

the part of any of the parties hereto; *provided*, *however*, such successor Trust Administrator shall file an amendment to the Certificate of Trust with the Secretary of State as required by the Delaware Act.

ARTICLE XI TRUST MONITOR

11.1. General.

- (a) The Trust Monitor shall oversee the activities of the Trust Administrator as set forth in this Trust Agreement. In all circumstances, the Trust Monitor shall act in the best interests of all Trust Beneficiaries, in furtherance of the purpose of the Trust, and in accordance with this Trust Agreement.
- (b) In furtherance of its rights and responsibilities under this Trust Agreement, the Trust Monitor shall have access, on reasonable advance notice and during regular business hours, to all such books and records of the Trust and the Trust Administrator, shall have the right to consult with all such professionals engaged by the Trust Administrator and shall participate in all such meetings of the Trust Administrator and the Trust Professionals as the Trust Monitor deems reasonably necessary or appropriate. Any documents shared between the Trust Administrator and the Trust Monitor shall be subject to joint privilege, and such sharing shall not be deemed to waive any attorney-client or work product privilege in respect of such documents.

(c) [Intentionally omitted.]

(d) Notwithstanding anything in this <u>Section 11.1</u> or <u>Section 11.2</u> hereof, the Trust Monitor shall not take (or fail to take) any action which will cause the Trust (other than the Avoidance Action Trust Claims Reserve) to fail to qualify as a liquidating trust within the meaning of Treasury Regulation section 301.7701-4(d) that is treated as a grantor trust.

11.2. Appointment and Removal of the Trust Monitor.

- (a) Subject to Section 11.2(d), the Trust Monitor shall serve until the earlier of (w) the final distribution of all Distributable Trust Assets and the Distributable Other Debtor Residual Trust Assets, if any, (x) its resignation pursuant to subsection (b) of this Section 11.2, (y) its removal pursuant to subsection (c) of this Section 11.2 or (z) its bankruptcy or insolvency.
- (b) The Trust Monitor may resign at any time by written notice of resignation to the Trust Administrator, a copy of which shall also be filed by the Trust Monitor with the Bankruptcy Court. Such resignation shall be effective no earlier than sixty (60) days from the date of such notice or such earlier time as a successor is appointed in accordance with the provisions of subsection (d) of this Section 11.2.

- (c) The holders of a majority of the Units or the DIP Lenders may at any time petition the Bankruptcy Court for the removal of the Trust Monitor, but only for good cause shown. Such removal shall become effective on the date ordered by the Bankruptcy Court.
- (d) In the event of the resignation, removal, bankruptcy or insolvency of the Trust Monitor, the Trust Administrator shall promptly appoint a successor Trust Monitor, provided that such appointment shall not take effect unless approved by the Bankruptcy Court upon the petition of the Trust Administrator and until the successor Trust Monitor shall have delivered written acceptance of its appointment as described in clause (e) of this Section 11.2 below; and provided further that until a new Trust Monitor's appointment is effective, the resigning Trust Monitor's appointment shall remain in effect, and the resigning Trust Monitor shall fulfill all obligations and duties of the Trust Monitor. If a successor Trust Monitor does not take office within thirty (30) days after the resignation, removal, incompetency, bankruptcy or insolvency of the retiring Trust Monitor, the Bankruptcy Court, upon its own motion or the motion of the retiring Trust Monitor or any Trust Beneficiary, shall appoint a successor Trust Monitor.
- (e) All fees and expenses properly incurred by the Trust Monitor prior to the resignation, Incompetency, removal or bankruptcy of the Trust Monitor shall be paid from the Avoidance Action Trust Administrative Cash, the GUC Trust Supplemental Cash or the Other Supplemental Cash (other than in respect of the Other Debtor Residual Accepted Assets and activities related thereto), from the Other Debtor Residual Trust Administrative Cash (in respect of the Other Debtor Residual Accepted Assets and activities related thereto) or from Avoidance Action Trust SEC Reporting Cash (in respect of Avoidance Action Trust SEC Reporting Costs in accordance with Section 2.3(e) hereof), unless such fees and expenses are disputed by (x) the Trust Administrator or (y) the successor Trust Monitor, in which case the Bankruptcy Court shall resolve the dispute and any disputed fees and expenses of the predecessor Trust Monitor that are subsequently allowed by the Bankruptcy Court shall be paid from the Avoidance Action Trust Administrative Cash, the GUC Trust Supplemental Cash or the Other Supplemental Cash (other than in respect of the Other Debtor Residual Accepted Assets and activities related thereto) from the Other Debtor Residual Trust Administrative Cash (in respect of the Other Debtor Residual Accepted Assets and activities related thereto) or from Avoidance Action Trust SEC Reporting Cash (in respect of Avoidance Action Trust SEC Reporting Costs in accordance with Section 2.3(e) hereof).
- (f) Any successor Trust Monitor appointed hereunder shall execute an instrument accepting its appointment and shall deliver one counterpart thereof to the Bankruptcy Court for filing and to the Trust Administrator.
- (g) Immediately upon effectiveness of the appointment of a successor Trust Monitor, all rights, powers, duties, authority, and privileges of the predecessor Trust Monitor hereunder will be vested in and undertaken by the successor Trust Monitor without any further act. The successor Trust Monitor shall not be liable personally for any act or omission of the predecessor Trust Monitor.

- 11.3. <u>Approval of and Consultation with the Trust Monitor.</u>
- (I) Other Than in Respect of the Other Debtor Residual Accepted Assets
- (a) Notwithstanding anything in this Trust Agreement to the contrary, the Trust Administrator shall submit to the Trust Monitor for its review and prior approval the following matters, in addition to any other matters that expressly require the approval of the Trust Monitor pursuant to the terms of the Plan, the Confirmation Order or this Trust Agreement:
- (i) Any decision to settle or otherwise resolve the Term Loan Avoidance Action;
- (ii) Any decision to refrain from making any distributions to the holders of Allowed General Unsecured Claims or Units, as the case may be, in accordance with this Trust Agreement, except as expressly permitted herein;
- (iii) Any decision to retain and/or to terminate the retention of Trust Professionals (other than legal counsel retained to represent the Trust Administrator in connection with its role as Trust Administrator, which shall be in the Trust Administrator's sole discretion);
- (iv) The incurrence of any cost or expense of the Trust in excess of 10% of any individual line item therefor in the approved Budget, measured on a quarterly basis; *provided*, *however*, that approval of the Trust Monitor shall not be required in the case of any cost or expense authorized by further order of the Bankruptcy Court;
- (v) The Budget described in <u>Section 6.3</u> hereof and any changes thereto;
- (vi) Any amendment of this Trust Agreement as provided in $\underline{\text{Section}}$ $\underline{13.13}$ hereof; and
- (vii) Any distribution that is not made in accordance with the provisions of <u>Article V</u> as contemplated by <u>Section 5.7</u>; <u>provided</u>, <u>however</u>, that any deviation from the provisions of <u>Article V</u> other than as contemplated by <u>Section 5.7</u> shall also require approval of the Bankruptcy Court.
- (b) In addition to any other matters that expressly require consultation with the Trust Monitor pursuant to the terms of the Plan, the Confirmation Order or this Trust Agreement, the Trust Administrator shall consult with the Trust Monitor in advance of an application to the Bankruptcy Court to use, or to sell or borrow against, the Term Loan Avoidance Action or the Distributable Trust Assets in order to satisfy expenses of the Trust, as contemplated by Section 6.1(b) and Section 6.1(d) hereof.
- (c) In the event of any disagreement between the Trust Administrator and the Trust Monitor regarding any matter requiring the approval or direction of the Trust Monitor under this Trust Agreement, the Trust Administrator and the Trust Monitor shall consult and

negotiate diligently and in good faith to resolve such disagreement. If despite their good faith efforts, the Trust Administrator and the Trust Monitor are unable to resolve any disagreement, or the Trust Administrator cannot otherwise obtain approval or direction from the Trust Monitor as required by this Trust Agreement, the Trust Administrator may petition the Bankruptcy Court, with a copy to the Trust Monitor, requesting such approval or direction.

(II) <u>In Respect of the Other Debtor Residual Accepted Assets</u>

The Trust Monitor shall have the rights, powers and privileges to act in respect of the Other Debtor Residual Accepted Assets, if any, in the manner set forth in Section 11.3 (I), mutatis mutandis and to the extent applicable. For the avoidance of doubt, the Trust Monitor shall no be obligated to undertake any activities in respect of the Other Debtor Residual Trust Assets unless there shall be available sufficient Other Debtor Residual Trust Administrative Cash to pay in full its fees, costs and expenses in respect thereof.

11.4. Exculpation and Indemnification; Limitation of Liability. To the fullest extent permitted by applicable law, the Trust Monitor Parties shall not be subject to personal liability, and shall be exculpated and indemnified, and shall have the right to obtain advances to cover reasonable expenses of defense, to the same extent as the Trust Administrator Parties pursuant to Section 9.2, Section 9.4, Section 9.5, Section 9.6 and Section 10.3. In no event will the Trust Monitor Parties be liable for punitive, exemplary, consequential, special or other damages for a breach of, or otherwise in connection with, this Trust Agreement under any circumstances.

11.5. Compensation and Expenses.

- (a) The Trust Monitor shall receive fair and reasonable compensation for its services (other than in respect of the Other Debtor Residual Accepted Assets and activities related thereto), to be paid out of the Avoidance Action Trust Administrative Cash, in accordance with the approved Budget (or from the Avoidance Action Trust SEC Reporting Cash in accordance with Section 2.3(e) hereof, the GUC Trust Supplemental Cash in accordance with Section 2.3(f) hereof, or the Other Supplemental Cash in accordance with Section 6.1(d) hereof). The Trust Monitor shall be entitled on a monthly basis, without the need for approval of the Bankruptcy Court, to direct the Trust Administrator to reimburse the Trust Monitor (i) from the Avoidance Action Trust Administrative Cash, the GUC Trust Supplemental Cash or the Other Supplemental Cash, as applicable, for all reasonable out-of-pocket expenses actually incurred in the performance of duties in accordance with this Trust Agreement, consistent with the Budget prepared pursuant to Section 6.3 hereof and (ii) from the Avoidance Action Trust SEC Reporting Cash for such compensation and all reasonable out-of-pocket expenses actually incurred in the performance of duties in accordance with this Trust Agreement.
- (b) The Trust Monitor shall receive compensation for its services than in respect of the Other Debtor Residual Accepted Assets and activities related thereto, to be paid out of the Other Debtor Residual Trust Administrative Cash, as the Trust Monitor and a majority in interest of the DIP lenders shall agree; provided that the Trust Monitor shall not be obligated to undertake any activities in respect of the Other Debtor Residual Accepted Assets

unless such compensation arrangements shall be acceptable to the Trust Monitor in its sole discretion.

ARTICLE XII ACTION BY MAJORITY OF HOLDERS OF UNITS

Holders of a majority of the Units or the DIP Lenders from time to time outstanding may petition the Bankruptcy Court to remove the Trust Administrator in accordance with Section 10.2 or to remove the Trust Monitor in accordance with Section 11.1, but in each case only for good cause shown. In determining whether the holders of a majority of the Units have concurred in any such petition, Units held by the Trust Administrator or the Trust Monitor or any of their respective Affiliates shall be disregarded.

ARTICLE XIII MISCELLANEOUS PROVISIONS

- 13.1. <u>Actions Taken on Other Than Business Day</u>. In the event that any payment or act under the Plan, the Confirmation Order or this Trust Agreement is required to be made or performed on a date that is not a Business Day, then the making of such payment or the performance of such act may be completed on the next succeeding Business Day, but shall be deemed to have been completed as of the required date.
- 13.2. <u>Governing Law</u>. This Trust Agreement shall be governed by and construed in accordance with the laws of the State of Delaware without giving effect to rules governing conflicts of law.
- 13.3. <u>Jurisdiction</u>. Subject to the proviso below, the parties agree that the Bankruptcy Court shall have exclusive and continuing jurisdiction over the Trust and the Trust Administrator, including the administration and activities of the Trust and the Trust Administrator; <u>provided</u>, <u>however</u>, that notwithstanding the foregoing, the Trust Administrator shall have power and authority to bring any action in any court of competent jurisdiction to prosecute any claims or Causes of Action assigned to the Trust, including the Delaware Chancery Court, the Delaware Superior Court and the Delaware Supreme Court.
- 13.4. <u>Third Party Beneficiary</u>. Trust Beneficiaries (including the DIP Lenders in their capacities as such) are third party beneficiaries of this Trust Agreement. The Trust Administrator Parties (other than the Trust Administrator) are third party beneficiaries of the provisions of <u>Section 9.2</u>, <u>Section 9.4</u> and <u>Section 9.6</u> of this Trust Agreement. The Trust Monitor Parties (other than the Trust Monitor) are third party beneficiaries of the provisions of <u>Section 11.4</u> of this Trust Agreement, and, to the extent incorporated therein, <u>Section 9.2</u>, <u>Section 9.4</u>, <u>Section 9.5</u> and <u>Section 9.6</u> of this Trust Agreement. Except as aforesaid, there are no other third party beneficiaries of this Trust Agreement.
- 13.5. <u>Severability</u>. In the event any provision of this Trust Agreement or the application thereof to any person or circumstances shall be determined by a final, non-appealable judgment or order to be invalid or unenforceable to any extent, the remainder

of this Trust Agreement or the application of such provision to persons or circumstances or in jurisdictions other than those as to or in which it is held invalid or unenforceable, shall not be affected thereby, and each provision of this Trust Agreement shall be valid and enforceable to the fullest extent permitted by law.

- 13.6. <u>Notices</u>. Any notice or other communication required or permitted to be made under this Trust Agreement shall be in writing and shall be deemed to have been sufficiently given, for all purposes, if delivered personally, by email, facsimile, sent by nationally recognized overnight delivery service or mailed by first-class mail:
 - (A) if to the Trust Administrator, to:

Wilmington Trust Company Rodney Square North 1100 North Market Street Wilmington, Delaware, 19890-1615

Phone: (302) 636-6000 Fax: (302) 636-4140

Attn: Corporate Trust Administration

With a copy to:

Binder & Schwartz LLP 366 Madison Avenue, 6th Floor New York, NY 10017 Phone: (212) 510-7008 Attn: Eric B. Fisher

(B) if to the Trust Monitor, to:

Arthur J. Gonzalez New York University School of Law 40 Washington Square South New York, NY 10012

- (C) if to any Trust Beneficiary, to:
 - (1) in the case of a DIP Lender,
 - a. if to the U.S. Treasury, to:

United States Department of the Treasury 1500 Pennsylvania Avenue, NW Washington, D.C. 20220 Attn: Chief Counsel, Office of Financial Stability Telecopier: (202) 927-9225

with a copy to:

OFSChiefCounselNotices@treasury.gov

b. if to Export Development Canada, to:

Export Development Canada 151 Slater Street Ottawa, Ontario Canada K1A 1K3 Attention: Loans Services Telecopy: 613-598-2514;

with a copy to:

Export Development Canada 151 Slater Street Ottawa, Ontario Canada K1A 1K3 Attention: Asset Management/Covenants Officer Telecopy: 613-598-3186

(2) if to the Capital Providers, to:

Cynthiana LLC 2711 Centerville Road Wilmington, DE 19808 Fax Number: (302) 636-5454 Email Address: info@litfinsolutions.com

Eman Address. mio@mmsordnons.

Attn: Cynthiana LLC Manager

Earlham LLC 2711 Centerville Road Wilmington, DE 19808 Fax Number: (302) 636-5454

Email Address: info@litfinsolutions.com

Attn: Earlham LLC Manager

(3) in the case of a holder of an Allowed General
Unsecured Claim, to the last known address of such
holder according to the Debtors' Schedules and/or
such holder's proof of claim; and

- (4) in the case of holder of Units, to such address as appears on the books and records of the Trust Administrator, or such other address as may be designated from time to time by notice given in accordance with the provisions of this Section 13.6.
- 13.7. <u>Headings</u>. The headings contained in this Trust Agreement are solely for convenience of reference and shall not affect the meaning or interpretation of this Trust Agreement or of any term or provision hereof.
- 13.8. <u>Plan</u>. The terms of this Trust Agreement are intended to supplement the terms provided by the Plan and the Confirmation Order. To the extent that the terms of sections 5.6 and 6.5 of the Plan are inconsistent with the terms set forth in this Trust Agreement with respect to the Trust, then the terms of the Trust Agreement shall govern. All other provisions of the Plan shall supersede the provisions of this Trust Agreement, including section 6.15 of the Plan, which provides that the restrictions set forth in paragraph 20 of the Final Order approving the DIP Credit Agreement (ECF No. 2529) shall continue to apply.

13.9. Ambiguities and Construction.

- (a) The Trust created by this Trust Agreement (other than the Avoidance Action Trust Claims Reserve) is intended to qualify as a liquidating trust under Treasury Regulation section 301.7701-4(d) for U.S. federal and applicable state and local income tax purposes and, to the extent provided by law, shall be governed and construed in all respects as such a trust and any ambiguity herein shall be construed consistent herewith and, if necessary, this Trust Agreement may be amended to comply with such U.S. federal and applicable state and local income tax laws, which amendments may apply retroactively.
 - (b) Unless the context otherwise requires:
 - (i) a term has the meaning assigned to it;
 - (ii) "or" is not exclusive;
- (iii) words in the singular include the plural, and in the plural include the singular;
- (iv) all references herein to Articles, Sections and other subsections, unless referring specifically to the Plan or provisions of the Bankruptcy Code, the Bankruptcy Rules, or other law, statute or regulation, refer to the corresponding Articles, Sections and other subsections of this Trust Agreement;
- (v) the words "hereof," "herein," "hereunder" and similar words refer to this Trust Agreement as a whole and not to any particular provision, Article, Section or subsection of this Trust Agreement unless otherwise specified;

- (vi) words importing persons shall include firms, associations, corporations and other entities;
- (vii) any pronoun shall include the corresponding masculine, feminine and neuter forms; and
 - (viii) "including" means including without limitation.
 - 13.10. <u>Entire Trust Agreement</u>. This Trust Agreement contains the entire agreement between the parties and supersedes all prior and contemporaneous agreements or understandings between the parties with respect to the subject matter hereof.
 - 13.11. <u>Cooperation</u>. The Debtors shall turn over or otherwise make available to the Trust Administrator at no cost to the Trust or the Trust Administrator, all books and records reasonably required by the Trust Administrator to carry out its duties hereunder, and agree to otherwise reasonably cooperate with the Trust Administrator in carrying out its duties hereunder, subject to the obligation to preserve the confidential nature of the Debtors' books and records, as provided in <u>Section 13.12</u>.
 - 13.12. <u>Confidentiality</u>. The Trust Administrator and the Trust Monitor, and their respective employees, members, agents, professionals and advisors, including the Trust Professionals (each a "<u>Confidential Party</u>" and collectively the "<u>Confidential Parties</u>") shall hold strictly confidential and not use for personal gain any material, non-public information of which they have become aware in their capacity as a Confidential Party, of or pertaining to any Debtor to which any of the Avoidance Action Trust Assets relates or which is otherwise received from the Debtors by the Trust; <u>provided</u>, <u>however</u>, that such information may be disclosed if
- (i) it is now or in the future becomes generally available to the public other than as a result of a disclosure by the Confidential Parties; or
- (ii) such disclosure is required of the Confidential Parties pursuant to legal process, including subpoena or other court order or other applicable laws or regulations.

In the event that any Confidential Party is requested to divulge confidential information pursuant to clause (ii), such Confidential Party shall promptly, in advance of making such disclosure, provide reasonable notice of such required disclosure to the Trust Administrator (or the Trust Monitor in case the Trust Administrator is the disclosing party) to allow sufficient time to object to or prevent such disclosure through judicial or other means and shall cooperate reasonably with the Trust Administrator (or the Trust Monitor, as applicable) in making any such objection, including appearing in any judicial or administrative proceeding in support of any objection to such disclosure.

13.13. Amendment and Waiver.

(a) The Trust Administrator, with the approval of the Trust Monitor, may amend or supplement this Trust Agreement without notice to or consent of the Bankruptcy

Court or any Trust Beneficiary for the purpose of (x) curing any ambiguity, omission, inconsistency or correcting or supplementing any defective provision; (y) evidencing and providing for the acceptance of the appointment of a successor Trust Administrator or Trust Monitor; or (z) making any other changes to this Trust Agreement that does not adversely affect the interests of the Trust Beneficiaries in any material respect.

- (b) The Trust Administrator may amend or supplement this Trust Agreement for any other purpose, but only on petition to, and with the approval of, the Bankruptcy Court; provided that (x) no amendment or supplement to this Trust Agreement shall be inconsistent with the purpose and intent of the Trust to dispose of in an expeditious but orderly manner the Avoidance Action Trust Assets in accordance with the terms of the Plan, the Confirmation Order and this Trust Agreement, and (y) this Trust Agreement shall not be amended in a manner that is inconsistent with the Plan in the form confirmed by the Bankruptcy Court, subject to any post-confirmation modifications to the Plan pursuant to Section 1127 of the Bankruptcy Code.
- (c) Any amendment to this Trust Agreement shall be filed with the Bankruptcy Court.
- (d) No amendment shall be made to any provision of this Trust Agreement that materially and adversely affects the rights of the DIP Lenders or the Capital Providers without the written consent of the DIP Lenders and the Capital Providers, as applicable.
- (e) The Trust Administrator shall file any amendment to the Certificate of Trust with the Secretary of State as may be required or permitted by the Delaware Act.
 - 13.14. <u>Counterparts</u>. This Trust Agreement may be executed in any number of counterparts, each of which shall be deemed an original, but all such counterparts shall together constitute but one and the same instrument. A facsimile or portable document file (PDF) signature of any party shall be considered to have the same binding legal effect as an original signature.

IN WITNESS WHEREOF, the parties hereto have executed this Trust Agreement or caused this Trust Agreement to be duly executed by their respective officers, representatives or agents, effective as of the date first above written.

Title:

WILMINGTON TRUST COMPANY, as Trust Administrator and trustee

By:

Name: Son Johnson

Title: Vill Hunden

- and
ARTHUR J. GONZALEZ, as Trust Monitor

By:

Name:

IN WITNESS WHEREOF, the parties hereto have executed this Trust Agreement or caused this Trust Agreement to be duly executed by their respective officers, representatives or agents, effective as of the date first above written.

WILMINGTON TRUST COMPANY, as Trust Administrator and trustee

By:	
Name:	
Title:	

- and -

ARTHUR J. GONZALEZ, as Trust Monitor

By:<u>//</u>

Name

le: Trust-Monito

Exhibit A

Cash Accounts:

AAT Approved Budget Cash Account
AAT Distributable Trust Assets Account
Avoidance Action Budget Sub Account
Avoidance Action Assets Sub Account
Avoidance Action Trust SEC Reporting Costs Account
Capital Provider Funding Account
Litigation Cost Advance Funding Account
Other Supplemental Cash Account
Segregated Account

Exhibit B

FORM OF CERTIFICATE OF TRUST OF

MOTORS LIQUIDATION COMPANY AVOIDANCE ACTION TRUST

THIS Certificate of Trust of Motors Liquidation Company Avoidance Action Trust (the "Trust") is being duly executed and filed on behalf of the Trust by the undersigned, as trustee, to form a statutory trust under the Delaware Statutory Trust Act (12 <u>Del. C.</u> § 3801 <u>et seq.</u>) (the "Act").

- 1. Name. The name of the statutory trust formed by this Certificate of Trust is Motors Liquidation Company Avoidance Action Trust.
- 2. <u>Delaware Trustee</u>. The name and business address of the trustee of the Trust with a principal place of business in the State of Delaware are Wilmington Trust Company, 1100 North Market Street, Wilmington, Delaware 19890, Attn: Corporate Trust Administration.
 - 3. <u>Effective Date</u>. This Certificate of Trust shall be effective upon filing.

IN WITNESS WHEREOF, the undersigned has duly executed this Certificate of Trust in accordance with Section 3811(a)(1) of the Act.

WILMINGTON TRUST COMPANY, not in its individual capacity but solely as Trust administrator and trustee

By:___ Name:

Title:

Vill Periden

Exhibit E

THIRDFOURTH AMENDED AND RESTATED MOTORS LIQUIDATION COMPANY AVOIDANCE ACTION TRUST AGREEMENT

WITNESSETH:

WHEREAS, the Trust Administrator and the Trust Monitor are party to the Motors Liquidation Company Avoidance Action Trust Agreement, dated as of March 30, 2011, by and among Motors Liquidation Company ("MLC"), MLC of Harlem, Inc., MLCS, LLC, MLCS Distribution Corporation, Remediation and Liability Management Company, Inc., and Environmental Corporate Remediation Company, Inc. (collectively, the "Debtors"), as debtors and debtors-in-possession, Wilmington Trust Company, as Trust Administrator, and FTI Consulting, Inc., as Trust Monitor (the "Original Trust Agreement"); and

WHEREAS, each of the Debtors has, prior to the date hereof, ceased to operate and dissolved; and

WHEREAS, the Original Trust Agreement was amended and restated in its entirety, with the approval of the Bankruptcy Court (as defined below), pursuant to that certain Amended and Restated Motors Liquidation Company Avoidance Action Trust Agreement dated as of May 11, 2012 by and between the Trust Administrator and the Trust Monitor (the "First Amended and Restated Trust Agreement"); and

WHEREAS, there was a dispute between the Official Committee of Unsecured Creditors of Motors Liquidation Company (the "Committee") and the DIP Lenders (defined below) as to entitlements to Avoidance Action Proceeds, which dispute was resolved by mutual agreement (the "Litigation Settlement") pursuant to which (a) the DIP Lenders provided the Litigation Cost Advance (as defined below), (b) the DIP Lenders shall be entitled to be repaid the Litigation Cost Advance out of Distributable Trust Assets (as defined below) and (c) the DIP Lenders shall be entitled to receive thirty percent (30%) of all remaining Distributable Trust Assets and, subject to the terms of this Trust Agreement, the holders of Allowed General Unsecured Claims (or Units) shall be entitled to receive seventy percent (70%) of all remaining Distributable Trust Assets, with each such distribution to be made at or about the same time and on a pari passu basis, as set forth

more fully in that certain Stipulation and Agreed Order (I) Settling Disputed Entitlements of Debtor-in-Possession Lenders and Official Committee of Unsecured Creditors to Potential Term Loan Avoidance Action Proceeds, and (II) Modifying Avoidance Action Trust Agreement to Implement Settlement, executed by the Committee, the DIP Lenders and the Trust and dated July 14, 2016 (the "Settlement Agreement"); and

WHEREAS, the Trust Administrator and the Trust Monitor determined that the Avoidance Action Trust Administrative Cash and Supplemental Avoidance Action Trust Cash (each defined below) held at that time by the Trust was insufficient to satisfy projected fees, costs and expenses of the Trust, and, as an integral part of the Litigation Settlement, the DIP Lenders agreed to provide the Litigation Cost Advance to the Trust on the terms set forth in that certain agreement, dated July 14, 2016, executed by the Trust and the DIP Lenders (the "Litigation Cost Advance Agreement"), which included, among other things, that the DIP Lenders would provide the Litigation Cost Advance to fund the prosecution of the Avoidance Action; and

WHEREAS, on July 15, 2016, the Trust, jointly with the Official Committee of Unsecured Creditors (the "Committee"), filed a motion with the Bankruptcy Court (the "9019 Motion") seeking approval of the Settlement Agreement and the Litigation Cost Advance Agreement; and

WHEREAS, on August 24, 2016, the Bankruptcy Court entered an order approving the 9019 Motion (the "9019 Order"); and

WHEREAS, pursuant Section 13.13(b) of the First Amended and Restated Trust Agreement, the 9019 Motion and the 9019 Order, the Trust Administrator petitioned and received from the Bankruptcy Court approval to amend and restate in its entirety the First Amended and Restated Trust Agreement; and

WHEREAS, the First Amended and Restated Trust Agreement was amended and restated in its entirety pursuant to that certain Second Amended and Restated Motors Liquidation Company Avoidance Action Trust Agreement dated as of August 25, 2016, by and between the Trust Administrator and the Trust Monitor (the "Second Amended and Restated Trust –Agreement"); and

WHEREAS, one of the preconditions to the DIP Lenders' provision of the Litigation Cost Advance was the Trust's agreement, as reflected in Section 6.1(d)(1) of the Second Amended and Restated Trust Agreement, that any additional litigation funding provided to the Trust would be (i) junior and subordinate to the Litigation Cost Advance and any other amounts owed to the DIP Lenders on account of prior funding of the Trust and (ii) subject to a form of subordination acceptable to the DIP Lenders in all respects; and

WHEREAS, on or about the date hereofJune 29, 2017, the Trust Administrator and the Trust Monitor have determined that the Avoidance Action Trust Administrative Cash and Supplemental Avoidance Action Trust Cash currently held by the Trust remainremained insufficient to satisfy projected fees, costs and expenses of the Trust; and

WHEREAS, with the consent of the DIP Lenders, the Trust Administrator and Trust Monitor entered into a Capital Provision Agreement, dated June 2, 2017 (the "CE Capital Provision –Agreement"), with Delaware limited liability companies Cynthiana LLC and Earlham

LLC (each a "Capital Provider," and, together, the "CE Capital Providers"), pursuant to which the CE Capital Providers willagreed to provide up to \$15,000,000 in additional funding to the Trust (the "CE Supplemental—Capital") as Other Supplemental Cash; and

WHEREAS, pursuant to Section 6.1(d)(11) of the Second Amended and Restated Trust Agreement, entry into that certain Subordination Agreement (the "CE Subordination Agreement"), dated as of June 2, 2017, by and among the DIP Lenders, the CE Capital Providers, and the Trust, was a precondition of the DIP Lenders' consent to the CE Capital Providers' provision of the CE Supplemental Capital to the Trust; and

WHEREAS, on June 7, 2017, the Trust moved for entry of an order, pursuant to sections 105 and 1142 of title 11 of the United States Code: (1) approving the amendment and restatement of the Second Amended Avoidance Action Trust Agreement, (2) authorizing the Avoidance Action Trust to enter into the <u>CE</u> Capital Provision Agreement and to grant a lien to the <u>CE</u> Capital Providers, and (3) authorizing the Avoidance Action Trust and Avoidance Action Trust Administrator to use \$1,750,759.93 of the Holdback to fund necessary fees, costs and expenses of the Avoidance Action Trust (the "<u>CE Capital Provision Motion</u>"); and

WHEREAS, on June 29, 2017, the Bankruptcy Court entered an order granting the <u>CE</u> Capital Provision Motion (the "<u>CE Capital Provision Order</u>"); and

WHEREAS, pursuant Section 13.13(b) of the Second Amended and Restated Trust Agreement, the <u>CE</u> Capital Provision Motion and the <u>CE</u> Capital Provision Order, the Trust Administrator petitioned and received from the Bankruptcy Court approval to amend and restate in its entirety the Second Amended and Restated Trust Agreement.; and

NOW, THEREFORE, in accordance with Section 13.13(b) of the WHEREAS, the Second Amended and Restated Trust Agreement was amended and the restated in its entirety pursuant to that certain Third Amended and Restated Motors Liquidation Company Avoidance Action Trust Agreement dated as of June 29, 2017, by and between the Trust Administrator and the Trust Monitor (the "Third Amended and Restated Trust Agreement"); and

WHEREAS, on or about the date hereof, for all of the reasons set forth in the LW Capital Provision Agreement, the Second Motion (as defined below), the Trust Administrator and the Trust Monitor have determined that the Avoidance Action Trust Administrative Cash and Supplemental Avoidance Action Trust Cash currently held by the Trust are likely to be insufficient to satisfy projected fees, costs and expenses of the Trust; and

WHEREAS, with the consent of the DIP Lenders and the CE Capital Providers, the Trust Administrator and Trust Monitor entered into a Capital Provision Agreement, dated January 21, 2019 (the "LW Capital Provision Agreement"), with LW Holdco VI LLC, a Delaware limited liability company (the "LW Capital Provider," and together with the DIP Lenders and the CE Capital Providers, the "Capital Providers"), pursuant to which the LW Capital Provider agreed to provide up to \$10,000,000 in additional funding to the Trust (the "LW Supplemental Capital," and together with the CE Supplemental Capital, the "Supplemental Capital") as Other Supplemental Cash; and

WHEREAS, pursuant to Section 6.1(d)(i) of the Third Amended and Restated Trust Agreement, entry into that certain Subordination Agreement, dated as of January 24, 2019 (the "LW Subordination Agreement"), by and among the DIP Lenders, the LW Capital Provider and the Trust, the form of which was approved by the CE Capital Providers, is a precondition of the DIP Lenders' consent to the LW Capital Provider's provision of the LW Supplemental Capital to the Trust; and

WHEREAS, on February 4, 2019, the Trust moved for entry of an order, pursuant to sections 105 and 1142 of title 11 of the United States Code: (1) approving the amendment and restatement of the Third Amended Avoidance Action Trust Agreement, and (2) authorizing the Avoidance Action Trust to enter into the LW Capital Provision Agreement and to grant a lien to the LW Capital Provider (the "LW Capital Provision Motion"); and

WHEREAS, on [], 2019, the Bankruptcy Court entered an order granting the LW Capital Provision Motion (the "LW Capital Provision Order"); and

WHEREAS, pursuant Section 13.13(b) of the Third Amended and Restated Trust Agreement, the LW Capital Provision Motion and the LW Capital Provision Order, the Trust Administrator petitioned and received from the Bankruptcy Court approval to amend and restate in its entirety the Third Amended and Restated Trust Agreement.

NOW, THEREFORE, in accordance with Section 13.13(b) of the Third Amended and Restated Trust Agreement and the LW Capital Provision Agreement, the Third Amended and Restated Trust Agreement is hereby amended and restated as follows:

Background

- A. Beginning on June 1, 2009, the Debtors filed in the United States Bankruptcy Court for the Southern District of New York (the "Bankruptcy Court") voluntary petitions for relief under chapter 11 of the Bankruptcy Code (the "Chapter 11 Cases").
- B. On or about August 31, 2010, the Debtors filed their Plan and Disclosure Statement in the Bankruptcy Court. The Debtors filed an amended Plan and Disclosure Statement on December 7, 2010. The Debtors filed a second amended Plan on March 18, 2011.
- C. The Disclosure Statement was approved by the Bankruptcy Court on December 8, 2010.
- D. On or about March 29, 2011, the Bankruptcy Court entered an order (the "Confirmation Order") confirming the Plan.
- E. The Plan provides for the creation of the Trust as a post-confirmation successor to MLC within the meaning of Section 1145(a) of the Bankruptcy Code, to hold and administer the Avoidance Action Trust Assets for the benefit of the Trust Beneficiaries and, to the extent received by the Trust, to distribute the Distributable Trust Assets to the Trust Beneficiaries in accordance with the terms of the Plan, the Confirmation Order and this Trust Agreement.

- F. The Plan also provides that the Trust Administrator may determine to hold and administer Other Debtor Residual Trust Assets, if any, for the benefit of the DIP Lenders.
- G. The Trust was created, with respect to the Avoidance Action Trust Assets, on behalf of, and for the benefit of, the Trust Beneficiaries, and, with respect to the Other Debtor Residual Trust Assets, if any, on behalf of, and for the benefit of, the DIP Lenders.
- The Trust Administrator shall have all powers necessary to implement the H. provisions of this Trust Agreement and administer the Trust in respect of the Avoidance Action Trust Assets, including the power to: (i) prosecute for the benefit of the Trust Beneficiaries, through counsel and other professionals selected by the Trust Administrator, the Term Loan Avoidance Action and any other causes of action that may from time to time be held by the Trust in respect thereof; (ii) preserve and maintain the Avoidance Action Trust Assets; (iii) distribute the Distributable Trust Assets, if any, to the Trust Beneficiaries in accordance with the Plan, the Confirmation Order and this Trust Agreement; (iv) expend the Avoidance Action Trust Administrative Cash, the Avoidance Action Trust SEC Reporting Cash (if applicable), and the Supplemental Avoidance Action Trust Cash to cover fees and expenses of the Trust, including payment of taxes and the fees and expenses of the Trust Professionals (other than in respect thereof of the Other Debtor Residual Trust Assets); (v) enter into any contracts or agreements necessary or desirable to facilitate the implementation of the provisions of this Trust Agreement, including but not limited to loan agreements or sale agreements for the purposes of funding the fees and expenses of the Trust; and (vi) sell or grant liens on the Term Loan Avoidance Action or any other property of the Trust (other than the Other Debtor Residual Trust Assets) pursuant to the terms of this Trust Agreement for the purposes of implementing the provisions of this Trust Agreement and the 9019 Order, including but not limited to funding the fees and expenses of the Trust.
- I. The Trust Administrator shall also have all powers necessary to implement the provisions of this Trust Agreement and administer the Trust in respect of the Other Debtor Residual Trust Assets, if any, including the power to: (i) prosecute for the benefit of the DIP Lenders, through counsel and other professionals selected by the Trust Administrator, any causes of action that may from time to time be held by the Trust in respect thereof; (ii) preserve and maintain the Other Debtor Residual Trust Assets; (iii) distribute the Distributable Other Debtor Residual Trust Assets, if any, to the DIP Lenders in accordance with the Plan, the Confirmation Order and this Trust Agreement; and (iv) expend the Other Debtor Residual Trust Administrative Cash to cover fees and expenses of the Trust, including payment of the fees and expenses of the Trust Professionals, in respect thereof.
- J. The Trust Administrator shall otherwise perform the functions and take the actions provided for in this Trust Agreement or permitted in the Plan and/or the Confirmation Order, or in any other agreement executed pursuant to the Plan, in each case subject to the provisions of Articles VI, VIII and XI hereof regarding the rights and powers of the Trust Monitor.
- K. The Trust is subject to the continuing jurisdiction of the Bankruptcy Court, whose approval is required to pay or distribute money or property to, or on behalf of, a Trust Beneficiary, except as expressly provided in this Trust Agreement.

L. The Trust (other than the Avoidance Action Trust Claims Reserve) is intended to qualify as a liquidating trust under Treasury Regulation section 301.7701-4(d) that is treated as a "grantor trust" for federal and applicable state and local income tax purposes.

Agreement

NOW, **THEREFORE**, in consideration of the promises and the mutual covenants contained herein, the Trust Administrator and the Trust Monitor agree as follows:

ARTICLE I DEFINED TERMS

- 1.1. <u>Definitions</u>. Whenever used in this Trust Agreement, unless the context otherwise requires, the following words and phrases shall have the respective meanings ascribed to them as follows:
- (a) "9019 Motion" has the meaning set forth in the preamble to this Trust Agreement.
- (b) "9019 Order" has the meaning set forth in the preamble to this Trust Agreement.
- (c) "<u>Affiliates</u>" means, with respect to any Person, any other Person which directly or indirectly controls, is controlled by or is under common control with such Person. For purposes of this definition "control" means, with respect to any Person, the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of such Person, whether through ownership of voting securities, by contract or otherwise.
- (d) "<u>Aggregate Maximum Amount</u>" means the sum of the Maximum Amounts of all Disputed General Unsecured Claims, Unresolved Term Loan Avoidance Action Claims and Unresolved Other Avoidance Action Claims, as set forth in the applicable GUC Trust Report as the Aggregate Maximum Amount as of a given date.
- (e) "Agreed Allocation" means the DIP Lenders shall be entitled to receive thirty percent (30%), and holders of Allowed General Unsecured Claims (or Units) shall be entitled to receive seventy percent (70%), with each such distribution to be made at or about the same time and on a pari passu basis.
- (f) "Allowed General Unsecured Claims" means, collectively, (i) the Initial Allowed General Unsecured Claims and (ii) the Resolved Allowed General Unsecured Claims.
- (g) "Avoidance Action Proceeds" means the proceeds of the Term Loan Avoidance Action.
- (h) "Avoidance Action Trust Administrative Cash" means the Trust Cash (other than the Supplemental Avoidance Action Trust Cash and the Avoidance Action Trust SEC Reporting Cash), including the fifteen million (\$15,000,000) Litigation Cost Advance and the \$1.6 million initially received from the DIP Lenders by the Debtors on the Avoidance Action Trust

Transfer Date, held and maintained by the Trust Administrator for the purpose of paying the fees and expenses incurred by the Trust Administrator (including fees and expenses for Trust Professionals) in connection with the Trust and any obligations imposed on the Trust Administrator or the Trust, including fees and expenses relating to the performance of the Trust Administrator's obligations under this Trust Agreement and the Plan, other than in respect of the Other Debtor Residual Trust Assets.

- (i) "Avoidance Action Trust Assets" means, collectively, (i) the Term Loan Avoidance Action transferred to the Trust, (ii) the Avoidance Action Proceeds, (iii) the Avoidance Action Trust Administrative Cash, (iv) the Avoidance Action Trust SEC Reporting Cash; (v) the Supplemental Avoidance Action Trust Cash; (vi) the Litigation Cost Advance; and (vii) the CE Supplemental Capital provided by and (viii) the LW Supplemental Capital Providers.
- (j) "Avoidance Action Trust SEC Reporting Costs" means any costs, fees or expenses incurred by the Trust that are directly or indirectly related to reports that may be required to be filed by the Trust with the SEC pursuant to applicable rules, regulations and interpretations of the SEC (including, without limitation, any legal, accounting or registration fees, costs and expenses incurred by the Trust with respect thereto).
- (k) "<u>Avoidance Action Trust SEC Reporting Cash</u>" has the meaning set forth in Section 2.3(e) of this Trust Agreement.
- (l) "Avoidance Action Trust Transfer Date" means the date selected by the Debtors on which the Avoidance Action Trust Assets (other than the Supplemental Avoidance Action Trust Cash and the Avoidance Action Proceeds) are transferred to the Trust, which transfer shall occur on or before December 15, 2011.
- (m) "<u>Bankruptcy Code</u>" has the meaning set forth in the preamble to this Trust Agreement.
 - (n) "Bankruptcy Court" has the meaning set forth in Background paragraph A.
 - (o) "Budget" has the meaning set forth in Section 6.3.
- (p) "<u>calendar quarter</u>" means the relevant three-month period ending on the last day of March, June, September or December, as applicable, of each calendar year; <u>provided</u>, <u>however</u>, that the calendar quarter that contains the Avoidance Action Trust Transfer Date shall be the period commencing on the Avoidance Action Trust Transfer Date and concluding on the date on which the relevant calendar quarter would naturally end in accordance with the foregoing.
- (q) "CE Capital Providers" has the meaning set forth in the preamble to this Trust Agreement.
- (r) "CE Capital Providers' Entitlement" means the funds to which the CE Capital Providers are entitled pursuant to the Capital Provision Agreement.
- (q)(a)—"CE Capital Providers" and "Capital Providers" have the meanings set forth in the preamble to this Trust Agreement.

- (r)(s) "Capital Provider Funding Account" means that segregated account, in the name of the Trust, established at Wilmington Trust Company, in which the Trust shall hold the CE Supplemental Capital. For the avoidance of doubt, the CE Capital Provider Funding Account shall contain Other Supplemental Cash as defined in this Trust Agreement.
- (s) "Capital Providers' Entitlement" means the funds to which the Capital Providers are entitled pursuant to the Capital Provision Agreement.
- (t) <u>""CE Capital Provision Agreement"</u> has the meaning set forth in the preamble to this Trust Agreement, as the same may be amended, restated, supplemented or modified from time to time after the date hereof.
- (u) "<u>CE Capital Provision Motion</u>" has the meaning set forth in the preamble to this Trust Agreement.
- (v) "<u>CE Capital Provision Order</u>" has the meaning set forth in the preamble to this Trust Agreement.
- (w) "CE Supplemental Capital" has the meaning set forth in the preamble to this Trust Agreement.
- (x) "Capital Provider" and "Capital Providers" have the meanings set forth in the preamble to this Trust Agreement.
- (w)(y) "Certificate of Trust" means the certificate of trust of the Trust as required by Section 3810 of the Delaware Act.
 - $\frac{(x)(z)}{(x)}$ "Chapter 11 Cases" has the meaning set forth in Background paragraph A.
 - (y)(aa) "Claim Conflict Resolution" has the meaning set forth in Section 3.6.
- (z)(bb) "Confidential Party" and "Confidential Parties" have the meanings set forth in Section 13.12.
- (aa)(cc) "Confirmation Order" has the meaning set forth in Background paragraph D.
- (bb)(dd) "Current Total Amount" means as of a given date, the sum of (A) the Total Allowed Amount as of such date and (B) the Aggregate Maximum Amount as of such date, as set forth in the applicable GUC Trust Report as the Current Total Amount as of a given date.
- (ce)(ce) "Debtors" has the meaning set forth in the preamble to this Trust Agreement.
- (dd)(ff)"Delaware Act" means the Delaware Statutory Trust Act, 12 <u>Del</u>. <u>C</u>. §3801 <u>et seq</u>.

- (ee)(gg) "DIP Credit Agreement Claims" means all Claims arising under the DIP Credit Agreement and Orders approving the DIP Credit Agreement dated June 25, 2009 and July 5, 2009.
- (ff)(hh) "DIP Lender Advances" means an amount equal to (i) the amounts of Cash advanced directly or indirectly by MLC and the DIP Lenders to fund the costs and expenses associated with realizing the proceeds of the Term Loan Avoidance Action, including, without limitation, any such amounts expended to fund the costs and expenses of professionals retained by the defendants in the Term Loan Avoidance Action, (ii) without duplication, the amount of the Avoidance Action Trust Administrative Cash, and (iii) without duplication, the Litigation Cost Advance.
- (gg)(ii) "DIP Lender Distributable Trust Assets" means the assets distributable to the DIP Lenders, in accordance with the Agreed Allocation and the Settlement Agreement.
- (hh)(jj) "<u>DIP Lenders</u>" means the U.S. Treasury and Export Development Canada, as lenders under the DIP Credit Agreement.
- (ii)(kk) "Disputed General Unsecured Claims" means the General Unsecured Claims against the Debtors that are Disputed (as defined in the Plan) as of the Initial GUC Record Date, until a time when such claims become Resolved General Unsecured Claims or are otherwise resolved pursuant to the claims resolution procedures contained in the Plan.
- (jj)(11) "Distributable Other Debtor Residual Trust Assets" means the Other Debtor Residual Assets Proceeds, if any, together with any earnings (including interest) thereon.
- (kk)(mm) "Distributable Other Debtor Residual Trust Cash" means any Cash or cash equivalents included in the Distributable Other Debtor Residual Trust Assets.
- (II)(nn) "Distributable Trust Assets" means the Avoidance Action Proceeds together with any earnings (including interest) thereon.
- (mm)(00) "Distributable Trust Cash" means any Cash or cash equivalents included in the Distributable Trust Assets.
- (nn)(pp) "Distribution Date" means the date of any distribution made by the Trust Administrator to the Trust Beneficiaries pursuant to this Trust Agreement, whether to the DIP Lenders pursuant to Section 5.1(d) or on account of Allowed General Unsecured Claims and/or Units.
 - (oo)(qq) "Distribution Threshold" means \$10,000,000.
- (pp)(rr)"Excess Distributable Trust Assets" means Distributable Trust Assets that the Trust Administrator, with the approval of the Trust Monitor, determines shall be distributed following the Initial GUC Distribution Date.
- (qq)(ss) "Excess GUC Distributable Trust Assets Determination Date" has the meaning set forth in Section 5.4(a).

(rr)(tt) "Excess GUC Distributable Trust Assets" has the meaning set forth in Section 5.4(c). "Final Recovery Date" has the meaning set forth in Section 5.1(a). (ss)(uu) (tt)(vv) "First Amended and Restated Trust Agreement" has the meaning set forth in the preamble to this Trust Agreement. "Funding Account" means that segregated account, in the name of (uu)(ww) the Trust, established at a bank reasonably acceptable to the Trust and the DIP Lenders, in which the Trust shall hold the Litigation Cost Advance. "GUC Beneficiaries" means the holders of Allowed General (vv)(xx)Unsecured Claims or Units received in respect of such claims." "GUC Distributable Trust Assets" means: (i) the Segregated (ww)(vv) Account Assets; and (ii) the assets distributable to the holders of Allowed General Unsecured Claims in accordance with the Agreed Allocation and the Settlement Agreement. "GUC Trust Advances" means the Distributable Trust Assets distributable to the holders of Allowed General Unsecured Claims, in an amount equal to the aggregate amount of GUC Trust Supplemental Cash received from the GUC Trust pursuant to Section 2.3(f)(i) hereof, together with any earnings (including interest) thereon, minus the aggregate amount of Cash held in the Segregated Account on the date of measurement. "GUC Trust Reports" means the reports prepared by the GUC Trust (yy)(aaa) Administrator each quarter as provided in the GUC Trust Agreement, which shall be delivered to the Trust Administrator pursuant to the terms of the GUC Trust Agreement. (zz)(bbb) "GUC Trust Supplemental Cash" has the meaning set forth in Section 2.3(f)(i). "Holdback" has the meaning set forth in Section 6.1(b). (aaa)(ccc) "Incompetency" means, with respect to any Person, the (bbb)(ddd) incompetency of such Person if such Person is a natural person. (eee) "Initial Allowed General Unsecured Claims" has the meaning set forth in Section 5.2(a). "Initial GUC Distribution Date" has the meaning set forth in Section (ddd)(fff) 5.2(a). "Initial GUC Record Date" has the meaning set forth in Section (eee)(ggg) 5.2(a).

(fff)(hhh)

"IRS" means the Internal Revenue Service.

- (iii) "LW Capital Provider" has the meaning set forth in the preamble to this Trust Agreement.
- (jjj) "LW Capital Provider's Entitlement" means the funds to which the LW Capital Provider is entitled pursuant to the LW Provider Capital Provision Agreement.
- (kkk) "LW Capital Provider Funding Account" means that segregated account, in the name of the Trust, established at Wilmington Trust Company, in which the Trust shall hold the LW Supplemental Capital.
- (lll) "LW Capital Provision Agreement" has the meaning set forth in the preamble to this Trust Agreement, as the same may be amended, restated, supplemented or modified from time to time after the date hereof.
- (mmm)"LW Capital Provision Motion" has the meaning set forth in the preamble to this Trust Agreement.
- (nnn) "LW Capital Provision Order" has the meaning set forth in the preamble to this Trust Agreement.
- (000) "LW Supplemental Capital" has the meaning set forth in the preamble to this Trust Agreement.
- (\$15,000,000) to be provided by the DIP Lenders pursuant to the Litigation Cost Advance Agreement and the Settlement Agreement and subject to a deposit account control agreement acceptable to the DIP Lenders and the Trust.
- (hhh)(qqq) "Litigation Cost Advance Agreement" has the meaning set forth in the preamble to this Trust Agreement.
- "<u>Litigation Settlement</u>" has the meaning set forth in the preamble to this Trust Agreement.
- (jjj)(sss) "Maximum Amount" means the maximum amount of any Disputed General Unsecured Claim, Unresolved Term Loan Avoidance Action Claim or Unresolved Other Avoidance Action Claim, calculated in accordance with the GUC Trust Agreement and as set forth in the applicable GUC Trust Report as the Maximum Amount of any Claim or group of Claims as of a given date.
- (kkk)(ttt) "MLC" has the meaning set forth in the preamble to this Trust Agreement.
- (III)(uuu) "Original Trust Agreement" has the meaning set forth in the preamble to this Trust Agreement.

(mmm)(vvv) "Other Avoidance Action Claims" means the additional General Unsecured Claims that have arisen as a result of recovery of proceeds of the Avoidance Actions other than the Term Loan Avoidance Action (and any related unsecured claims).

(nnn)(www) "Other Debtor Residual Assets" means any assets of MLC remaining at such time as the Debtors shall be liquidated, other than the Term Loan Avoidance Action and the Term Loan Avoidance Action Administrative Cash and any other assets of MLC whose disposition is specifically provided for under the Plan or the Confirmation Order.

(000)(xxx) "Other Debtor Residual Accepted Assets" means Other Debtor Residual Assets accepted by the Trust Administrator for transfer to the Trust pursuant to Section 2.3(b).

(ppp)(yyy) "Other Debtor Residual Assets Proceeds" means any proceeds realized in respect of the Other Debtor Residual Accepted Assets.

(qqq)(ZZZ) "Other Debtor Residual Trust Administrative Cash" means the Cash, if any, held and maintained by the Trust Administrator for the purpose of paying the fees and expenses incurred by the Trust Administrator (including fees and expenses for Trust Professionals) in connection with the Trust and any obligations imposed on the Trust Administrator or the Trust, including fees and expenses relating to the performance of the Trust Administrator's obligations under this Trust Agreement and the Plan, but only in respect of the Other Debtor Residual Trust Assets, which Cash may be obtained by transfer to the Trust by the Debtors, from the DIP Lenders (in their sole discretion) or from the proceeds of the Other Debtor Residual Accepted Assets.

(rrr)(aaaa) "Other Debtor Residual Trust Assets" means, if any, collectively, (i) the Other Debtor Residual Accepted Assets transferred to the Trust, (ii) the Other Debtor Residual Assets Proceeds and (iii) the Other Debtor Residual Trust Administrative Cash.

(sss)(bbb) "Other Debtor Residual Assets Transfer Date" means the date selected by the Debtors on which the Other Debtor Residual Assets Transfer, if any, are transferred to the Trust, which transfer shall occur on or before December 15, 2011.

(ttt)(ccc) "Other Supplemental Cash" means any Cash received by the Trust pursuant to Section 6.1(d) hereof from the sale of, or granting of liens on, all or a portion of the Term Loan Avoidance Action or any other property held by the Trust (other than the Other Debtor Residual Trust Assets), or from any source other than the GUC Trust, including the CE Supplemental Capital and the LW Supplemental Capital.

(uuu)(dddd) "Permissible Investments" means investments in any of the following:

(i) Marketable securities issued by the U.S. Government and supported by the full faith and credit of the U.S. Treasury, either by statute or an opinion of the Attorney General of the United States;

- (ii) Marketable debt securities, rated Aaa by Moody's and/ or AAA by S&P, issued by U.S. Government-sponsored enterprises, U.S. Federal agencies, U.S. Federal financing banks, and international institutions whose capital stock has been subscribed for by the United States;
- (iii) Certificates of deposit, time deposits, and bankers acceptances of any bank or trust company incorporated under the laws of the United States or any state, provided that, at the date of acquisition, such investment, and/or the commercial paper or other short term debt obligation of such bank or trust company has a short-term credit rating or ratings from Moody's and/or S&P, each at least P-1 or A-1;
- (iv) Commercial paper of any corporation incorporated under the laws of the United States or any state thereof which on the date of acquisition is rated by Moody's and/or S&P, provided each such credit rating is least P-1 and/or A-1;
- (v) Money market mutual funds that are registered with the Securities and Exchange Commission under the Investment Company Act of 1940, as amended, and operated in accordance with Rule 2a-7 and that at the time of such investment are rated Aaa by Moody's and/or AAAm by S&P, including such funds for which the Trust Administrator or an Affiliate provides investment advice or other services;
- (vi) Tax-exempt variable rate commercial paper, tax-exempt adjustable rate option tender bonds, and other tax-exempt bonds or notes issued by municipalities in the United States, having a short-term rating of "MIG-1" or "VMIG-1" or a long term rating of "AA" (Moody's), or a short-term rating of "A-1" or a long term rating of "AA" (S&P); and
- (vii) Repurchase obligations with a term of not more than thirty days, 102 percent collateralized, for underlying securities of the types described in clauses (i) and (ii) above, entered into with any bank or trust company or its respective affiliate meeting the requirements specified in clause (iii) above.
- (vvv)(eeee) "Plan" has the meaning set forth in the preamble to this Trust Agreement.
- (www)(ffff) "Resolved Allowed General Unsecured Claims" means, collectively, (I) the Disputed General Unsecured Claims that are allowed after the Initial GUC Record Date in accordance with the claims resolution procedures administered under the Plan (to the extent so resolved); (II) the Term Loan Avoidance Action Claims, to the extent and in the amount collected by the Trust against the respective defendants in the underlying litigation (including by way of settlement); and (III) the Other Avoidance Action Claims, to the extent and in the amount collected against the respective defendants in the underlying litigations (including by way of settlement). For the avoidance of doubt, unless and until a Disputed General Unsecured Claim, Unresolved Term Loan Avoidance Action Claim or Unresolved Other Avoidance Action Claim becomes a Resolved Allowed General Unsecured Claim, the holders of such claim shall not receive any distribution from the Trust.

(xxx)(gggg) "Resolved Allowed General Unsecured Claims Determination Date" has the meaning set forth in Section 5.3(a).

(yyy)(hhhh) RESERVED

(ZZZ)(iiii) RESERVED

(aaaa)(jiji) "SEC" means the Securities and Exchange Commission.

(bbbb)(kkkk) "Second Amended and Restated Trust Agreement" has the meaning set forth in the preamble to this Trust Agreement.

(cccc)(1111) "Secretary of State" means the Office of the Secretary of State of the State of Delaware.

(dddd)(mmmm) "Segregated Account" has the meaning set forth in Section 2.3(f)(i).

(eeee)(nnnn) "Segregated Account Assets" means the Cash held in the Segregated Account.

(ffff)(0000) "Settlement Agreement" has the meaning set forth in the preamble to this Trust Agreement.

(gggg)(pppp) "Supplemental Avoidance Action Trust Cash" means the GUC Trust Supplemental Cash and the Other Supplemental Cash.

(hhhh)(qqqq) "Supplemental Capital" has the meaning set forth in the preamble to this Trust Agreement.

<u>(iiii)(rrrr)</u> "<u>Tax Returns</u>" means all tax returns, reports, certificates, forms or similar statements or documents.

(jjjj)(ssss) "Term Loan Avoidance Action" means the Avoidance Action commenced by the Creditors' Committee against JPMorgan Chase Bank, N.A., individually and as Administrative Agent, and various lenders party to a term loan agreement, dated as of November 29, 2006, between General Motors Corporation, as borrower, JPMorgan Chase Bank, N.A., as agent, and various institutions as lenders and agents, styled *Official Committee of Unsecured Creditors of Motors Liquidation Co. v. JPMorgan Chase Bank, N.A. et al.*, Adv. Pro. No. 09-00504 (Bankr. S.D.N.Y. July 31, 2009).

(kkkk)(tttt) "Term Loan Avoidance Action Claims" means the additional General Unsecured Claims that have arisen as a result of recovery of proceeds of the Term Loan Avoidance Action (or any related unsecured claims).

(uuuu) "Third Amended and Restated Trust Agreement" has the meaning set forth in the preamble to this Trust Agreement.

(IIII)(vvvv) "<u>Total Allowed Amount</u>" means the sum of the amount of all Initial Allowed General Unsecured Claims plus the amount of all Resolved Allowed General Unsecured Claims, as set forth in the applicable GUC Trust Report.

(mmmm)(www) "Treasury Regulations" means the income tax regulations promulgated under the Tax Code, including any amended or successor income tax regulations thereto.

 $\frac{\text{(nnnn)}(xxxx)}{\text{``Trust''}}$ has the meaning set forth in the preamble to this Trust Agreement.

(0000)(yyyy) "Trust Administrator" has the meaning set forth in the preamble to this Trust Agreement.

(pppp)(zzzz) "Trust Administrator Parties" means the Trust Administrator and its principals, directors, officers, employees, agents, representatives, attorneys, accountants, advisors and other professionals (including the Trust Professionals).

(qqqq)(aaaaa) "Trust Agreement" has the meaning set forth in the preamble to this Trust Agreement.

(rrrr)(bbbb) "Trust Beneficiaries" means the holders of the DIP Credit Agreement Claims and the holders of Allowed General Unsecured Claims (or Units received in respect of such claims).

(ssss)(cccc) "Trust Cash" means the Cash or cash equivalents included in the Avoidance Action Trust Assets or the Other Debtor Residual Trust Assets, if any.

(tttt)(ddddd) "Trust Monitor" has the meaning set forth in the preamble to this Trust Agreement.

(uuuu)(eeeee) "Trust Monitor Parties" means the Trust Monitor and its principals, directors, officers, employees, agents, representatives, attorneys, accountants, advisors and other professionals.

(vvvv)(fffff) "Trust Professionals" means, collectively, independent contractors, including attorneys, accountants, appraisers, disbursing agents or other parties deemed by the Trust Administrator to have the qualifications necessary or desirable to assist in the proper administration of the Trust and that are employed or retained by the Trust in such capacities.

(www)(ggggg) "Unit Issuance Ratio" means the ratio of one Unit for each \$1,000 in amount of Allowed General Unsecured Claims.

(xxxx)(hhhhh) "Units" means the units of beneficial interest issued by the Trust to holders of Allowed General Unsecured Claims.

(yyyy)(iiiii) "Unresolved Other Avoidance Action Claim" means an Other Avoidance Action Claim that has not or has not yet arisen because no determination (including by way of settlement) has been made in the respective Avoidance Action against the respective defendant who would be entitled to such claim in the event of such determination (or if a determination has been made against the defendant, the proceeds related to such resolution have not been recovered in full).

(2222)(jijji) "Unresolved Term Loan Avoidance Action Claim" means a Term Loan Avoidance Action Claim that has not or has not yet arisen because no determination (including by way of settlement) has been made in the Term Loan Avoidance Action against the respective defendant who would be entitled to such claim in the event of such determination (or if a determination has been made against the defendant, the proceeds related to such resolution have not been recovered in full).

(aaaaa)(kkkk) "Wind-Down Facility" means the \$1.175 billion wind-down facility provided to the Debtors pursuant to the DIP Credit Agreement.

ARTICLE II DECLARATION OF TRUST

- 2.1. <u>Creation of Trust</u>. The Debtors and the Trust Administrator, pursuant to the Plan and the Confirmation Order and in accordance with the applicable provisions of chapter 11 of the Bankruptcy Code, hereby constitute and create the Trust, in the form of a statutory trust under the Delaware Act, which shall bear the name "Motors Liquidation Company Avoidance Action Trust." In connection with the exercise of the Trust Administrator's power hereunder, the Trust Administrator may use this name or such variation thereof as the Trust Administrator sees fit. The Trust Administrator, as trustee of the Trust, is hereby authorized and directed to execute and file a Certificate of Trust for the Trust in the form attached hereto as Exhibit B.
- 2.2. <u>Purpose of Trust</u>. The sole purpose of the Trust is to liquidate and distribute its assets pursuant to the Plan in accordance with Treasury Regulation section 301.7701-4(d), with no objective to continue or engage in the conduct of a trade or business.

2.3. Transfer of Avoidance Action Trust Assets to the Trust.

Effective upon the Avoidance Action Trust Transfer Date, the Debtors (a) hereby transfer to the Trust, pursuant to Bankruptcy Code Sections 1123(a)(5)(B) and 1123(b)(3)(B), and in accordance with the Plan and the Confirmation Order, the Avoidance Action Trust Assets (other than the Avoidance Action Trust SEC Reporting Cash, and the Supplemental Avoidance Action Trust Cash), as they exist on the Avoidance Action Trust Transfer Date, free and clear of any and all liens, claims, encumbrances and interests (legal, beneficial or otherwise) of all other persons to the maximum extent contemplated by and permissible under Bankruptcy Code Section 1141(c); provided, however that notwithstanding anything to the contrary in the Plan, Disclosure Statement, Confirmation Order, this Trust Agreement or any other agreement, the DIP Lenders shall maintain their liens on the Avoidance Action Trust Administrative Cash, provided that for the avoidance of doubt, the DIP Lenders shall not demand acceleration of their liens on the Avoidance Action Trust Administrative Cash except in accordance with the provisions of section 7.2 of the DIP Credit Agreement. Such transfers shall be exempt from any stamp, real estate transfer, mortgage reporting, sales, use or other similar tax pursuant to Bankruptcy Code Section 1146. The Debtors and their successors and assigns shall be released from any and all liability with respect to the transfer of such Avoidance Action Trust Assets to the Trust as aforesaid. Nothing in this Trust Agreement is intended to, or shall be construed to, effect a release, extinguishment or compromise of any claim or cause of action transferred to the Trust pursuant to this Trust Agreement. The Avoidance Action Trust Assets and all other property held

from time to time by the Trust under this Trust Agreement (other than the Other Debtor Residual Trust Assets) and any earnings (including interest) thereon are to be managed, applied and disposed of by the Trust Administrator in accordance with the terms hereof, the Plan and the Confirmation Order for the benefit of the Trust Beneficiaries and the Capital Providers, and for no other party, subject to the further covenants, conditions and terms hereinafter set forth, including the provisions of Section 2.6.

- (b) To the extent any Avoidance Action Trust Assets (other than the Supplemental Avoidance Action Trust Cash) cannot be transferred to the Trust, because of a restriction on transferability under applicable non-bankruptcy law that is not superseded by Bankruptcy Code Section 1123 or any other provision of the Bankruptcy Code, such assets shall be retained by the Debtors or any successor thereto including, without limitation, the GUC Trust. The proceeds of the sale of any such assets retained by the Debtors (or any successor thereto) shall be allocated to the Trust pursuant to the Plan as if such transfer had not been restricted under applicable non-bankruptcy law. The Trust Administrator may commence an action in the Bankruptcy Court to resolve any dispute regarding the allocation of the proceeds of any assets retained by the Debtors (or any successor thereto) pursuant to the Plan and Confirmation Order.
- (c) On the Avoidance Action Trust Transfer Date, the Debtors shall also deliver, or cause to be delivered, to the Trust a complete list of all General Unsecured Claims, both Allowed and Disputed, reflected on the claims registry as of the Avoidance Action Trust Transfer Date, including the names and addresses of all holders of such General Unsecured Claims, whether such claims have been Allowed or are Disputed, and the details of all objections in respect of Disputed General Unsecured Claims.
- Effective upon the Other Debtor Residual Assets Transfer Date, the Debtors hereby transfer to the Trust, pursuant to Bankruptcy Code Sections 1123(a)(5)(B) and 1123(b)(3)(B), and in accordance with the Plan and the Confirmation Order, such of the Other Debtor Residual Assets, as they exist on the Other Debtor Residual Assets Transfer Date, as the Trust Administrator, in its sole discretion but with the approval of the Trust Monitor, shall determine to accept, free and clear of any and all liens, claims, encumbrances and interests of all other persons to the maximum extent contemplated by and permissible under Bankruptcy Code Section 1141(c); provided that, for the avoidance of doubt, the Trust Administrator may determine not to accept the transfer to the Trust of any or all of the Other Debtor Residual Assets for any reason or for no reason. Any such transfer shall be exempt from any stamp, real estate transfer, mortgage reporting, sales, use or other similar tax pursuant to Bankruptcy Code Section 1146. The Debtors and their successors and assigns shall be released from any and all liability with respect to the transfer of the Other Debtor Residual Accepted Assets to the Trust as aforesaid. Nothing in this Trust Agreement is intended to, or shall be construed to, effect a release, extinguishment or compromise of any claim or cause of action transferred to the Trust pursuant to this Trust Agreement, and notwithstanding anything to the contrary in the Plan, Disclosure Statement, Confirmation Order, this Trust Agreement or any other agreement, the DIP Lenders shall maintain their liens on the Other Debtor Residual Accepted Assets. The Other Debtor Residual Trust Assets and all other property held from time to time by the Trust under this Trust Agreement in respect thereof, and any earnings (including interest) thereon, are to be managed, applied and disposed of by the Trust Administrator in accordance with the terms hereof, the Plan and the Confirmation Order for the benefit of the DIP Lenders, and for no other party, subject to

the further covenants, conditions and terms hereinafter set forth, including the provisions of Section 2.6.

- (e) (i) On the Avoidance Action Trust Transfer Date, the Debtors shall, pursuant to Section 2.3(e) of the GUC Trust Agreement, transfer Cash to the Trust in an amount of \$500,000 (the "Avoidance Action Trust SEC Reporting Cash"). The Avoidance Action Trust SEC Reporting Cash shall be held by the Trust in a segregated account and shall be used solely for the satisfaction of Avoidance Action Trust SEC Reporting Costs. Any taxes imposed on the Trust in respect of the Avoidance Action Trust SEC Reporting Cash shall be satisfied from the income realized thereon.
- (ii) The Trust Administrator shall only use Avoidance Action Trust SEC Reporting Cash to satisfy Avoidance Action Trust SEC Reporting Costs to extent there is no other available source of funds to satisfy such expenses (other than the Supplemental Avoidance Action Trust Cash), including, without limitation, any funds obtained through the reservation and application of all or a portion of the Holdback pursuant to Section 6.1(b) hereof.
- (iii) If the Trust Administrator determines that (x) reports are not, and at no time will be, required to be filed by the Trust with the SEC pursuant to applicable rules, regulations and interpretations of the SEC or (y) the Trust has other available funds (as set forth in Section 2.3(e)(ii) hereof) which are sufficient to satisfy any current or future projected, fees, costs or expenses that are directly or indirectly related to reports that may be required to be filed by the Trust with the SEC pursuant to applicable rules, regulations and interpretations of the SEC (including, without limitation, any legal, accounting or registration fees, costs and expenses incurred by the Trust with respect thereto), then the Trust shall transfer to the GUC Trust any Avoidance Action Trust SEC Reporting Cash that has not been applied as provided in this Section 2.3(e).
- (iv) Any income earned on the Avoidance Action Trust SEC Reporting Cash, net of taxes paid thereon, shall be Avoidance Action Trust Administrative Cash.
- (f) (i) From time to time, in accordance with a Final Order of the Bankruptcy Court if so required, the GUC Trust may deliver Cash to the Trust to be used solely for the satisfaction of fees and expenses of the Trust (including for the payment of the fees and expenses of Trust Professionals) or the satisfaction of any federal, state or local taxes incurred by the Trust. Any Cash so delivered to the Trust shall be designated as "GUC Trust –Supplemental Cash." GUC Trust Supplemental Cash shall be held by the Trust in a segregated account (the "Segregated Account"), and shall be used solely for the satisfaction of fees and expenses of the Trust (including for the payment of the fees and expenses of Trust Professionals), for the satisfaction of taxes incurred by the Trust or for distribution to holders of Allowed General Unsecured Claims pursuant to Section 5.1(d)(iv) hereof.
- (ii) The Trust Administrator shall only use the GUC Trust Supplemental Cash to satisfy fees and expenses of the Trust and tax liabilities of the Trust to the extent that there is no other source of funds to satisfy such expenses (other than the Avoidance Action Trust SEC Reporting Cash, and other than through the sale or encumbrance of the Term Loan Avoidance Action or any other property of the Trust pursuant to Section 6.1(d) hereof), including, without

limitation, Avoidance Action Trust Administrative Cash, and any funds obtained through the reservation and application of all or a portion of the Holdback pursuant to Section 6.1(b) hereof. Any income earned on the GUC Trust Supplemental Cash, net of taxes paid thereon, shall be GUC Trust Supplemental Cash.

- 2.4. Appointment and Acceptance of Trust Administrator. The Trust Administrator shall be deemed to be appointed pursuant to Bankruptcy Code Section 1123(b)(3)(B) and is hereby appointed trustee of the Trust under the Delaware Act. The Trust Administrator hereby accepts such appointments, including trusteeship of the Trust created by this Trust Agreement and the grant, assignment, transfer, conveyance and delivery by the Debtors to the Trust Administrator, (i) on behalf of the Trust, and for the benefit of the Trust Beneficiaries, of all of their respective right, title and interest in the Distributable Trust Assets, and (ii) on behalf of the Trust, and for the benefit of the DIP Lenders, of all of their respective right, title and interest in the Other Debtor Residual Trust Assets, upon and subject to the terms and conditions set forth in the Plan, the Confirmation Order and this Trust Agreement. The Trust Administrator's powers are exercisable solely in a fiduciary capacity consistent with, and in furtherance of, the purpose of the Trust and not otherwise, and in accordance with applicable law, including the Delaware Act. The Trust Administrator shall have the authority to bind the Trust within the limitations set forth herein, but shall for all purposes hereunder be acting in the capacity as Trust Administrator, and not individually.
- 2.5. <u>Distribution of Distributable Trust Assets</u>. The Trust Administrator shall, in an expeditious but orderly manner and subject to the provisions of the Plan, the Confirmation Order and this Trust Agreement, make timely distributions of the Distributable Trust Assets and the Distributable Other Debtor Residual Assets in accordance with the terms hereof and not unduly prolong the existence of the Trust. The Trust Administrator may incur and pay any reasonable and necessary expenses in connection with the administration of the Trust, including the fees and expenses of the Trust Professionals *provided*, *however*, that all such expenditures (other than in respect of the Other Debtor Residual Trust Assets) shall be made in accordance with the Budget.

2.6. No Reversion to Debtors.

- (a) In no event shall any part of the Avoidance Action Trust Assets or the Other Debtor Residual Trust Assets revert to or be distributed to or for the benefit of any Debtor. All Distributable Trust Assets shall be applied as provided in Section 5.1(d), including to the satisfaction of Allowed General Unsecured Claims, including through distributions made in respect of the Units. All Distributable Other Debtor Residual Trust Assets shall be applied as provided in Article 5A.
- (b) To the extent that after satisfaction in full of all of the costs and expenses of the administration of the Trust, after all Allowed General Unsecured Claims have been paid pursuant to the Plan, after satisfaction of all other obligations or liabilities of the Trust incurred or assumed in accordance with the Plan, Confirmation Order or this Trust Agreement (or to which the Avoidance Action Trust Assets are otherwise subject), and after the affairs of the Trust have been finally wound up and concluded in accordance with the provisions of Section 4.3 hereof and Section 3808 of the Delaware Act, there shall remain any Avoidance Action Trust Administrative Cash, the Trust Administrator is authorized to and shall distribute any such

remaining Avoidance Action Trust Administrative Cash to the DIP Lenders in accordance with the terms of the DIP Credit Agreement and the Plan. To the extent any portion of such residue is not accepted by the respective DIP Lenders, the Trust Administrator shall (i) be authorized to distribute up to \$100,000 of such remaining Avoidance Action Trust Administrative Cash to an organization described in Section 501(c)(3) of the Tax Code and exempt from U.S. federal income tax under section 501(a) of the Tax Code that is unrelated to the Debtors, the Trust, any Trust Administrator Parties or any Trust Monitor Parties, or (ii) request an order from the Bankruptcy Court authorizing the Trust Administrator to distribute any such remaining Avoidance Action Trust Administrative Cash to such an organization, or authorizing such other disposition as recommended by the Trust Administrator and approved by the Bankruptcy Court.

To the extent that after satisfaction in full of all of the costs and expenses of the administration of the Trust, after all Allowed General Unsecured Claims have been paid pursuant to the Plan, after satisfaction of all other obligations or liabilities of the Trust incurred or assumed in accordance with the Plan, Confirmation Order or this Trust Agreement (or to which the Avoidance Action Trust Assets are otherwise subject), and after the affairs of the Trust have been finally wound up and concluded in accordance with the provisions of Section 4.3 hereof and Section 3808 of the Delaware Act, there shall remain any GUC Trust Supplemental Cash, the Trust Administrator is authorized to and shall distribute any such remaining GUC Trust Supplemental Cash to the holders of Allowed General Unsecured Claims (as defined in the GUC Trust Agreement), regardless of whether such amount of GUC Trust Supplemental Cash is less than the Distribution Threshold, provided that, if the remaining GUC Trust Supplemental Cash is less than \$100,000, the Trust Administrator shall (i) be authorized to distribute such remaining GUC Trust Supplemental Cash to an organization described in Section 501(c)(3) of the Tax Code and exempt from U.S. federal income tax under section 501(a) of the Tax Code that is unrelated to the Debtors, the Trust, any Trust Administrator Parties or any Trust Monitor Parties, or (ii) request an order from the Bankruptcy Court authorizing the Trust Administrator to distribute any such remaining GUC Trust Supplemental Cash to such an organization, or authorizing such other disposition as recommended by the Trust Administrator and approved by the Bankruptcy Court.

ARTICLE III TRUST BENEFICIARIES; UNITS

3.1. Rights of Beneficiaries.

- (a) Except as provided in <u>Section 2.6</u> hereof, the Trust Beneficiaries shall be the sole beneficiaries of the Trust (to the extent of the Avoidance Action Trust Assets) and the Distributable Trust Assets, and the Trust Administrator shall retain only such incidents of ownership as are necessary to undertake the actions and transactions authorized in the Plan, the Confirmation Order and this Trust Agreement, including those powers set forth in <u>Articles VI</u> and VIII hereof.
- (b) The beneficial interest of a Trust Beneficiary in the Trust is hereby declared and shall be in all respects and for all purposes intangible personal property.
- (c) Except as expressly provided herein, a Trust Beneficiary shall have no title or right to, or possession, management or control of, the Trust, or the Avoidance Action Trust

Assets, or to any right to demand a partition or division of such assets or to require an accounting of the Trust Administrator or the Trust Monitor. The whole legal title to the Avoidance Action Trust Assets shall be vested in the Trust as a separate legal entity under the Delaware Act or, if necessary, in the Trust Administrator on behalf of the Trust and the sole beneficial interest of the Trust Beneficiaries shall be as set forth in this Trust Agreement.

- 3.2. <u>Limited Liability</u>. No provision of the Plan, the Confirmation Order or this Trust Agreement, and no mere enumeration herein of the rights or privileges of any Trust Beneficiary, shall give rise to any liability of such Trust Beneficiary solely in its capacity as such, whether such liability is asserted by any Debtor, by creditors or employees of any Debtor, or by any other Person. GUC Beneficiaries are deemed to receive the GUC Distributable Trust Assets in accordance with the provisions of the Plan, the Confirmation Order and this Trust Agreement in exchange for their Allowed General Unsecured Claims or on account of their Units, as applicable, without further obligation or liability of any kind, but subject to the provisions of this Trust Agreement.
- 3.3. <u>No Control by Trust Beneficiaries</u>. A Trust Beneficiary shall have no title to, or any right to possess, manage or control, the Avoidance Action Trust Assets, or any portion thereof or interest therein, except as expressly provided herein. No surviving spouse, heir or devisee of any deceased Trust Beneficiary shall have any right of dower, homestead or inheritance, or of partition, or any other right, statutory or otherwise, in the Avoidance Action Trust Assets, but the whole title to all the Avoidance Action Trust Assets shall be vested in the Trust Administrator and the sole interest of the Trust Beneficiaries shall be the rights and benefits provided to such persons under this Trust Agreement.

3.4. Issuance of Units.

- (a) The Trust shall issue Units to holders of Allowed General Unsecured Claims as provided in this Trust Agreement. On the Initial GUC Distribution Date, holders of Initial Allowed General Unsecured Claims shall receive the number of Units equal to the amount of such Initial Allowed General Unsecured Claims multiplied by the Unit Issuance Ratio, rounded up or down to the nearest whole Unit. Following the Initial GUC Distribution Date, holders of Resolved Allowed General Unsecured Claims shall receive the number of Units equal to the amount of such Resolved Allowed General Unsecured Claims multiplied by the Unit Issuance Ratio, rounded up or down to the nearest whole Unit. Units will represent the contingent right to receive, on a pro rata basis as provided in the Plan, the Confirmation Order and this Trust Agreement, Excess GUC Distributable Trust Assets. The Units shall be issued subject to all the terms and conditions of the Plan, the Confirmation Order and this Trust Agreement. References in this Trust Agreement to holders of Units shall be to the record holders of such Units.
- (b) As provided in <u>Section 7.5</u> hereof, the Trust Administrator may retain Units otherwise issuable pursuant to this section with respect to Allowed General Unsecured Claims that are subject to withholding, and the Trust Administrator shall apply amounts distributed in respect of such retained Units to satisfy such withholding obligations.
- (c) Notwithstanding the foregoing, if (i) as of the Initial GUC Distribution Date, the total amount of the Disputed General Unsecured Claims in the aggregate is less than 0.5% of the Current Total Amount, or (ii) the Initial GUC Distribution Date is determined by the

Trust Administrator to also be the final Distribution Date, no Units shall be distributed and any GUC Distributable Trust Assets remaining after satisfaction of all Initial Allowed General Unsecured Claims and any other obligations of the Trust shall be disposed of as set forth in the last sentence of Section 2.6(c).

3.5. Ownership of Units; Transfers of Units.

- (a) The interest of a Trust Beneficiary is in all respects personal property, and upon the death, insolvency or incapacity of an individual GUC Beneficiary, such GUC Beneficiary's Units shall pass to the legal representative of such GUC Beneficiary.
- (b) The Units will be issued and evidenced by appropriate notation on the books and records of the Trust Administrator. The Units shall not be certificated and shall not be transferable, assignable, pledged or hypothecated in whole or in part, except by applicable laws of descent and distribution (in the case of a deceased individual GUC Beneficiary); by operation of law; in accordance with applicable Bankruptcy law; or as otherwise approved by the Bankruptcy Court. The Trust Administrator shall not be required to recognize any equitable or other claims to such interest by the transferee thereof, and the named GUC Beneficiary shall remain as such for all purposes hereunder.
- Conflicting Claims to Units. If the Trust Administrator has actual knowledge of 3.6. any conflicting claims or demands that have been made or asserted with respect to a Unit, or a beneficial interest therein, the Trust Administrator shall be entitled, at its sole election, to refuse to comply with any such conflicting claims or demands. In so refusing, the Trust Administrator may elect to make no payment or distribution with respect to the Unit subject to the claims or demands involved, or any part thereof, and the Trust Administrator shall be entitled to refer such conflicting claims or demands to the Bankruptcy Court, which shall have exclusive and continuing jurisdiction over resolution of such conflicting claims or demands. The Trust Administrator shall not be or become liable to any party for either (i) its election to continue making distributions pursuant to its books and records, without regard to the conflicting claims or demands; or (ii) its election to cease payments or distributions with respect to the subject Unit. In the event that the Trust Administrator elects to cease payments, it shall be entitled to refuse to act until either (x) the rights of the adverse claimants have been adjudicated by a Final Order of the Bankruptcy Court (or such other court of proper jurisdiction) or (y) all differences have been resolved by a written agreement among all of such parties and the Trust Administrator, which agreement shall include a complete release of the Trust, the Trust Administrator Parties and the Trust Monitor Parties in form and substance reasonably satisfactory to the Trust Administrator and Trust Monitor (the occurrence of either (x) or (y), a "Claim Conflict Resolution"). Until a Claim Conflict Resolution is reached with respect to such conflicting claims or demands, the Trust Administrator shall hold in a segregated account any payments or distributions from the Trust to be made with respect to Promptly after a Claim Conflict Resolution is reached, the Trust the Unit(s) at issue. Administrator shall transfer the payments and distributions, if any, held in the segregated account, together with interest and income earned thereon, if any, in accordance with the terms of such Claim Conflict Resolution.
- 3.7. <u>Distributions Relating to Note Claims and Eurobond Claims</u>. The Trust shall distribute GUC Distributable Trust Assets and Units to the Indenture Trustees and Fiscal and

Paying Agents, to the extent necessary to provide each beneficial holder of debt securities arising out of or relating to the Note Claims and Eurobond Claims with an amount of GUC Distributable Trust Assets and Units equal to the amount of GUC Distributable Trust Assets and Units such holder would receive had its claim been treated as an Initial Allowed General Unsecured Claim hereunder. For the avoidance of doubt, Units will be issued and evidenced by appropriate notation on the books and records of the Trust Administrator in the names of the Indenture Trustees and the Fiscal and Paying Agents, as applicable, and not in the individual names of the beneficial holders of debt securities arising out of or relating to the Note Claims and Eurobond Claims.

ARTICLE IV DURATION AND TERMINATION OF THE TRUST

4.1. <u>Duration</u>. The Trust shall become effective upon (x) the earlier to occur of the (i) Avoidance Action Trust Transfer Date and (ii) the Other Debtor Residual Assets Transfer Date, if any, and (y) the execution of this Trust Agreement and the filing of the Certificate of Trust with the Secretary of State and shall remain and continue in full force and effect until (a) all of the Distributable Trust Assets and all Distributable Other Debtor Residual Trust Assets, if any, have been distributed pursuant to the Plan and this Trust Agreement, (b) the Trust Administrator determines, in its discretion and with the approval of the Trust Monitor, that the administration of the Avoidance Action Trust Assets is not likely to yield sufficient additional Distributable Trust Assets or Distributable Other Debtor Residual Trust Assets to justify further pursuit, and (c) all other distributions required to be made by the Trust Administrator under the Plan and this Trust Agreement have been made, but in no event shall the Trust be dissolved later than three (3) years from the earlier of (i) the Avoidance Action Trust Transfer Date and (ii) the Other Debtor Residual Assets Transfer Date, unless the Bankruptcy Court, upon motion within the six (6) month period prior to the third (3rd) anniversary of the earlier of (i) the Avoidance Action Trust Transfer Date and (ii) the Other Debtor Residual Assets Transfer Date (or at least six (6) months prior to the end of an extension period), determines that a fixed period extension (such that, together with any prior extensions, the dissolution of the Trust shall occur no later than five (5) years from the date on which the Trust became effective, without a favorable private letter ruling from the IRS that any further extension would not adversely affect the status of the Trust as a liquidating trust for U.S. federal income tax purposes) is necessary to facilitate or complete the recovery and liquidation of the Avoidance Action Trust Assets or the Other Debtor Residual Trust Assets, as the case may be. If at any time the Trust Administrator determines, in reliance upon such professionals as the Trust Administrator may retain and with the approval of the Trust Monitor, that (x) the expense of administering the Trust so as to make a final distribution to the Trust Beneficiaries is likely to exceed the value of the Avoidance Action Trust Assets remaining in the Trust and (y) the expense of administering the Trust so as to make a final distribution to the DIP Lenders is likely to exceed the value of the Other Debtor Residual Trust Assets, if any, remaining in the Trust, the Trust Administrator may apply to the Bankruptcy Court for authority to (i) reserve any amounts necessary to dissolve the Trust, transfer the balance to the DIP Lenders and the holders of Allowed General Unsecured Claims (as defined in the GUC Trust Agreement), in accordance with the Agreed Allocation and the Settlement Agreement, or donate any balance to a charitable organization described in section 501(c)(3) of the Tax Code and exempt from U.S. federal income tax under section 501(a) of the Tax Code that is unrelated to the Debtors, the Trust, any Trust Administrator Parties or any Trust Monitor Parties, and (iii) dissolve the Trust.

- 4.2. <u>Dissolution of the Trust</u>. Notwithstanding anything to the contrary in this Trust Agreement, in no event shall the Trust Administrator unduly prolong the duration of the Trust, and the Trust Administrator shall, in the exercise of its reasonable business judgment and in the interests of all Trust Beneficiaries, at all times endeavor to terminate the Trust as soon as practicable in accordance with the purposes and provisions of this Trust Agreement and the Plan. Upon final dissolution and wind-up of the Trust, the Trust Administrator shall file a certificate of cancellation for the Trust with the Secretary of State.
- 4.3. Continuance of Trust for Purposes of Winding Up. After the dissolution of the Trust and solely for the purpose of liquidating and winding up its affairs, the Trust Administrator shall continue to act in such capacity until its duties hereunder have been fully performed. The Trust Administrator shall retain the books, records and files that shall have been delivered to or created by the Trust Administrator until distribution or resolution of all the Avoidance Action Trust Assets and Other Debtor Residual Trust Assets, if any. At the Trust Administrator's discretion, all of such books, records and files may be destroyed at any time following the later of (x) final distribution of the Avoidance Action Trust Assets and Other Debtor Residual Trust Assets, if any, unless such books, records and files are necessary to fulfill the Trust Administrator's obligations pursuant to Articles VI and VIII hereof and subject to any joint prosecution and common interests agreement(s) to which the Trust Administrator may be party, and (y) the date until which the Trust Administrator is required by applicable law to retain such books, records and files.

ARTICLE V DISTRIBUTIONS TO TRUST BENEFICIARIES

5.1. General.

Until such date as the Term Loan Avoidance Action shall have been (a) completely and finally resolved in full against all defendants (including by way of settlement) and the Trust Administrator determines that all Avoidance Action Proceeds have been collected in respect thereof (such date, the "Final Recovery Date"), the Trust Administrator shall establish Distribution Dates no less frequently than once each calendar year and no more frequently than once a calendar quarter for the distribution of Distributable Trust Assets as provided in this Article V; provided that distributions need not be made in any calendar year to the extent (A) there are no Distributable Trust Assets held by the Trust or (B) the Trust Administrator, with the approval of the Trust Monitor, determines (i) that it is reasonably necessary to retain the Distributable Trust Assets to meet contingent liabilities and maintain the value of the Avoidance Action Trust Assets (such as, for example, in the event that the Trust Administrator determines that the Distributable Trust Assets are so small in amount as not to justify making a distribution, taking into account the costs that would be incurred in making the distribution, the anticipated total amount of Distributable Trust Assets expected to be available for distribution over time and the timing of the distribution or distributions thereof), or (ii) that it is necessary to retain the Distributable Trust Assets to pay reasonable incurred and anticipated expenses (including any taxes imposed on the Trust or in respect of the Avoidance Action Trust Assets) or to satisfy liabilities incurred and anticipated by the Trust in accordance with the Plan, the Confirmation Order and this Trust Agreement. Following the Final Recovery Date, distributions shall be made on a quarterly basis, as provided in this Article V.

- (b) Except as otherwise set forth herein, no distributions shall be made with respect to any portion of a Disputed General Unsecured Claim, Unresolved Term Loan Avoidance Action Claim or Unresolved Other Avoidance Action Claim unless and until such Disputed General Unsecured Claim, Unresolved Term Loan Avoidance Action Claim or Unresolved Other Avoidance Action Claim has become an Allowed General Unsecured Claim.
- (c) To the extent that a Disputed General Unsecured Claim, Unresolved Term Loan Avoidance Action Claim or Unresolved Other Avoidance Action Claim has become an Allowed General Unsecured Claim, distributions (if any) shall be made to the holder of such Allowed General Unsecured Claim in accordance with the provisions of the Plan, the Confirmation Order and this Trust Agreement.
 - (d) The Distributable Trust Assets shall be distributed
- (i) first, to the DIP Lenders, each in their relative proportion pursuant to the terms of the DIP Credit Agreement, in the amount of the DIP Lender Advances;
- (ii) second, to the <u>CE</u> Capital Providers, in the amount of the <u>CE</u> Capital Providers' Entitlement;
- (iii) thirdthird, to the LW Capital Provider, in the amount of the LW Capital Provider's Entitlement;
- (iii)(iv) fourth, to the Segregated Account in the amount of the GUC Trust Advances to be distributed to the holders of Allowed General Unsecured Claims; and
- (iv)(v) fourth fifth, to: the DIP Lenders, each in their relative proportion pursuant to the terms of the DIP Credit Agreement, and the holders of Allowed General Unsecured Claims, in accordance with the Agreed Allocation.
- (e) On any Distribution Date following the Final Recovery Date on which the Trust does not hold sufficient Distributable Trust Assets, after taking into account any amounts necessary to satisfy the current and projected future fees and expenses of the Trust (including those of any Trust Professionals) pursuant to Section 6.1(b), to satisfy all Disputed General Unsecured Claims or other Claims that became Resolved Allowed General Unsecured Claims since the next preceding Resolved Allowed General Unsecured Claims Determination Date following which there was a distribution pursuant to Section 5.3(c), the Trust Administrator shall distribute all Distributable Trust Assets that remain in the Trust to the DIP Lenders and the holders of such Resolved Allowed General Unsecured Claims pro rata by Claim amount, in accordance with the Agreed Allocation. Following such distribution, any remaining unsatisfied portion of such Resolved Allowed General Unsecured Claims, together with all remaining Disputed General Unsecured Claims and other Claims (including any Term Loan Avoidance Action Claims and any Other Avoidance Action Claims) shall be forever barred from assertion against the Trust.
- (f) Anything to the contrary herein notwithstanding but subject to <u>Section 5.1(a)</u>, the Trust Administrator shall not make a distribution of Distributable Trust Assets on any Distribution Date pursuant to <u>Sections 5.2 or 5.4</u>, if the amount to be distributed pursuant thereto does not exceed the Distribution Threshold. In such case, any Distributable Trust Assets then

available for distribution shall be held by the Trust and distributed on a subsequent Distribution Date when the amount of the Distributable Trust Assets to be distributed shall exceed the Distribution Threshold; *provided* that if on the date determined by the Trust Administrator to be the final Distribution Date the Distributable Trust Assets do not exceed the Distribution Threshold, then such Distributable Trust Assets shall be disposed of as provided in the final sentence of Section 2.6(b) and in Section 2.6(c).

- (g) For the avoidance of doubt, if the Trust fails to recover any Avoidance Action Proceeds or if the Avoidance Action Proceeds recovered by the Trust through the Final Recovery Date do not exceed the amount of DIP Lender Advances then no distributions shall be made hereunder in respect of any Allowed General Unsecured Claims.
- (h) For the avoidance of doubt, if the Avoidance Action Proceeds recovered by the Trust through the Final Recovery Date exceed the amount of DIP Lender Advances but do not exceed the amount of the Segregated Account Assets, the Segregated Account Assets shall be distributed as GUC Distributable Trust Assets pursuant to Section 5.2(a) hereof.
- 5.2. <u>Distribution to the DIP Lenders and to Holders of Initial Allowed General Unsecured Claims.</u>
- (a) Once (i) the DIP Lender Advances have been indefeasibly paid in full in cash, (ii) the CE Capital Providers' Entitlement has been indefeasibly paid in full in cash, (iii) the LW Capital Provider's Entitlement has been indefeasibly paid in full in cash, and (iv) the Segregated Account repaid up to the amount of the GUC Trust Advances, the Trust Administrator shall make distributions of the DIP Lender Distributable Trust Assets and the GUC Distributable Trust Assets, from time to time in accordance with Section 5.1(a). Prior to the first such distribution, the Trust Administrator shall establish a record date (the "Initial GUC Record Date") for the holders of the General Unsecured Claims that are allowed as of the Initial GUC Record Date (the "Initial Allowed General Unsecured Claims") who shall be entitled to participate in the first distribution of GUC Distributable Trust Assets, which date shall be the last day of the calendar quarter next preceding the date of such distribution (such Distribution Date, the "Initial GUC Distribution Date"). On the Initial GUC Distribution Date, the Trust Administrator shall distribute to each holder of an Initial Allowed General Unsecured Claim a distribution consisting of:
 - an amount of GUC Distributable Trust Assets at the time available for distribution, in proportion to the amount of such Initial Allowed General Unsecured Claim as prescribed in subsection (d) below; and
 - (2) a number of Units as provided in Section 3.4.
- (b) During the period, if any, following the Initial GUC Distribution Date and prior to the Final Recovery Date, and as promptly as practicable following the Final Recovery Date, as and to the extent that additional GUC Distributable Trust Assets become available for distribution from time to time as a result of additional recoveries by the Trust in the Term Loan Avoidance Action, the Trust Administrator shall, from time to time in accordance with Section 5.1(a), make additional distributions to the holders of Initial Allowed General Unsecured Claims,

on a Distribution Date or Dates designated by the Trust Administrator for such purpose, in the amount prescribed in subsection (d) below. Subject to the proviso in Section 5.1(a), the Trust Administrator shall make such distributions no less frequently than once each calendar year and no more frequently than once each calendar quarter.

(c) The amount of GUC Distributable Trust Assets that the holder of an Initial Allowed General Unsecured Claim shall be entitled to receive pursuant to Section 5.2 (or, in the case of a holder of a Resolved Allowed General Unsecured Claim being treated as if it were a holder of an Initial Allowed General Unsecured Claim in the amount of its Resolved Allowed General Unsecured Claim pursuant to Section 5.3(c)) on any Distribution Date (including the Initial Distribution Date) shall be determined in accordance with the following formula:

$$D = (\underline{A}) \times G$$

Where—

D is the distribution that the holder of an Initial Allowed General Unsecured Claim (or the holder of a Resolved Allowed General Unsecured Claim pursuant to <u>Section 5.3(c)</u>) will be entitled to receive;

A is the amount of the Initial Allowed General Unsecured Claim (or the holder of a Resolved Allowed General Unsecured Claim pursuant to Section 5.3(c));

C is the Current Total Amount as of the last day of the calendar quarter next preceding the respective Distribution Date; and

G is the amount available for distribution determined as of the last day of the calendar quarter next preceding the respective Distribution Date, after taking account of any Holdback, or the application of such Holdback, pursuant to <u>Section 6.1(b)</u>.

5.3. Distributions to Holders of Resolved Allowed General Unsecured Claims.

- (a) Following the Initial GUC Distribution Date, the Trust Administrator, with the approval of the Trust Monitor, shall periodically make a determination (the date of any such determination, which shall in all cases be made as of the last day of a calendar quarter, being a "Resolved Allowed General Unsecured Claims Determination Date") whether any holders of Disputed General Unsecured Claims or other Claims have become holders of Resolved Allowed General Unsecured Claims Determination Date or, in the case of the first such determination, since the Initial GUC Record Date. During the period, if any, following the Initial GUC Distribution Date and prior to the Final Recovery Date, the Trust Administrator shall make such determination no less frequently than once each calendar year and no more frequently than once each calendar quarter, and following the Final Recovery Date such determination shall be made once each calendar quarter.
- (b) On a Distribution Date scheduled as soon as practicable following each Resolved Allowed General Unsecured Claims Determination Date, the Trust Administrator shall, subject to the proviso in Section 5.1(a), distribute to each holder of a Resolved Allowed General

Unsecured Claim identified on such Resolved Allowed General Unsecured Claims Determination Date, if any:

- (i) the pro rata amount of GUC Distributable Assets that the holder of such Resolved Allowed General Unsecured Claim would have received had such Resolved Allowed General Unsecured Claim been an Initial Allowed General Unsecured Claim, including the aggregate amount of Excess GUC Distributable Trust Assets that the holder would have received had it been the holder of Units referred to in clause (ii) below on each Excess GUC Distributable Trust Assets Determination Date occurring on or prior to the date of such distribution; provided that a holder of a Resolved Allowed General Unsecured Claim shall not receive pursuant to this clause (i) an amount of Excess GUC Distributable Assets distributed in respect of any prior Excess GUC Distributable Trust Assets Determination Date to the extent that it will be receiving such Excess GUC Distributable Assets as a distribution on the Units to be received by such holder pursuant to clause (ii) below; and
 - (ii) a number of Units as provided in <u>Section 3.4</u>.
- (c) Once a holder of a Resolved Allowed General Unsecured Claim has received the distribution prescribed in the preceding subsection (b), such holder shall be treated as if it were a holder of an Initial Allowed General Unsecured Claim in the amount of its Resolved Allowed General Unsecured Claim on all subsequent Distribution Dates established for purposes of Section 5.2(c).
- (d) For the avoidance of doubt, it is intended that the distributions to be made to holders of Resolved Allowed General Unsecured Claims in accordance with this <u>Section 5.3</u> shall provide such holders, as nearly as possible, with the exact same amount of distributions as if such holders had been holders of Initial Allowed General Unsecured Claims.

5.4. <u>Distribution of Excess Distributable Trust Assets.</u>

- (a) Following the Initial GUC Distribution Date, the Trust Administrator, with the approval of the Trust Monitor, shall periodically make a determination (the date of any such determination, which shall in all cases be as of the last day of a calendar quarter, being an "Excess Distributable Trust Assets Determination Date") of the Excess Distributable Trust Assets as of such date, taking into account the extent to which Disputed General Unsecured Claims are disallowed or the Term Loan Avoidance Action or other Avoidance Actions are resolved in favor of the defendants therein. During the period, if any, following the Initial GUC Distribution Date and prior to the Final Recovery Date, the Trust Administrator shall make such determination no less frequently than once each calendar year and no more frequently than once each calendar quarter, and following the Final Recovery Date such determination shall be made once each calendar quarter.
- (b) On a Distribution Date scheduled as soon as practicable following each Excess Distributable Trust Assets Determination Date, the Trust Administrator shall, subject to the proviso of Section 5.1(a), distribute the Excess Distributable Trust Assets, in each case determined as of the respective Excess Distributable Trust Assets Determination Date, to the DIP Lenders and the holders of Units outstanding on such Excess Distributable Trust Assets

Determination Date (including Units distributed or to be distributed to holders of Resolved Allowed General Unsecured Claims pursuant to Section 5.3(b)(ii) on such Distribution Date) in accordance with the Agreed Allocation.

(c) Any Excess Distributable Trust Assets allocable to the holders of Units outstanding on such Excess Distributable Trust Assets Determination Date (the "Excess GUC Distributable Trust Assets"), shall be distributed pro rata according to the following formula:

$$DU = (\ \underline{U_H} \) \ x \ (\sum G - H) \ x \ [\ \underline{T_U} \ - \ \underline{T_U} \]$$

$$C_U \ (C_U + L)$$

Where—

D_U is the distribution of Excess GUC Distributable Trust Assets that a holder of Units will be entitled to receive;

U_H is the number of Units held by the holder;

U_O is the total number of Units outstanding (including Units distributed to holders of Resolved Allowed General Unsecured Claims pursuant to Section 5.3(b)(ii) on such Distribution Date);

∑G is the sum of the amounts that are or were available for distribution to holders of Initial Allowed General Unsecured Claims (and holders of Resolved Allowed General Unsecured Claims pursuant to Section 5.3(c)) on the relevant Distribution Date and each prior Distribution Date;

T_U is the Total Allowed Amount as of the Excess GUC Distributable Trust Assets Determination Date:

C_U is the Current Total Amount as of the Excess GUC Distributable Trust Assets Determination Date;

L is the aggregate amount of all (i) Disputed General Unsecured Claims disallowed since the next preceding Excess GUC Distributable Trust Assets Determination Date (or, in the case of the first Excess GUC Distributable Trust Assets Determination Date, since the Initial GUC Record Date), (ii) Unresolved Term Loan Avoidance Action Claims to the extent resolved (including by way of settlement) in favor of the respective defendants since the next preceding Excess GUC Distributable Trust Assets Determination Date (or, in the case of the first Excess GUC Distributable Trust Assets Determination Date, since the Initial GUC Record Date); and (iii) all Unresolved Other Avoidance Action Claims to the extent resolved (including by way of settlement) in favor of the respective defendants since the next preceding Excess GUC Distributable Trust Assets Determination Date (or, in the case of the first Excess GUC Distributable Trust Assets Determination Date, since the Initial GUC Record Date); and

- H is the amount, if any, of any holdback pursuant to <u>Section 6.1</u> that was not otherwise deducted from the amounts available for distribution on a Distribution Date.
- (d) Notwithstanding the foregoing, if the Trust Administrator becomes aware of previously unknown potential Allowed General Unsecured Claims, the Trust Administrator may, with the approval of the Trust Monitor, withhold distribution of Excess GUC Distributable Trust Assets to the holders of Units in an amount that the Trust Administrator, with the approval of the Trust Monitor, estimates to be the maximum amount reasonably allowable in respect of such previously unknown claims.
- 5.5. <u>Retention of Avoidance Action Trust Assets</u>. Notwithstanding anything in this Trust Agreement to the contrary, the Trust Administrator shall at all times, to the extent practicable, retain
- (a) sufficient GUC Distributable Trust Assets as the Trust Administrator shall determine, with the approval of the Trust Monitor, as would be distributable (I) to all holders of Disputed General Unsecured Claims at the time outstanding as if all Disputed General Unsecured Claims were allowed at the Maximum Amount, but only until such Disputed General Unsecured Claims are resolved, (II) to the holders of all Resolved Allowed General Unsecured Claims at the time outstanding, to the extent not previously distributed, (III) in respect of any Unresolved Term Loan Avoidance Action Claims at the Maximum Amount thereof but only until the Term Loan Avoidance Action is dismissed by Final Order or such claims become Resolved Allowed General Unsecured Claims, and (IV) in respect of any Unresolved Other Avoidance Action Claims at the Maximum Amount thereof but only until such claims become Resolved Allowed General Unsecured Claims or the related other Avoidance Actions are dismissed by Final Order; and
- (b) sufficient Avoidance Action Trust Administrative Cash and Supplemental Avoidance Action Trust Cash as the Trust Administrator shall determine, with the approval of the Trust Monitor and subject to the Budget, is necessary (x) to pay the reasonable incurred or anticipated fees and expenses of the Trust (including any taxes imposed on the Trust or in respect of the Avoidance Action Trust Assets) and (y) to satisfy other liabilities incurred or anticipated by the Trust in accordance with the Plan, the Confirmation Order and this Trust Agreement.
- 5.6. Minimum Distributions. Notwithstanding anything to the contrary contained herein, no Cash payment in an amount less than \$25 shall be made by the Trust Administrator to any holder of an Allowed General Unsecured Claim or Unit under any circumstance; *provided* that the Trust Administrator shall carry the entitlement of such holder of an Allowed General Unsecured Claim or Unit to such amount on its books and records, shall aggregate such amount with any subsequent amount to which such holder shall become entitled and shall make payment of such amount to such holder at such time as the amounts due such holder in the aggregate shall equal \$25 or more; *provided further* that if any such amount shall be owing to a holder of an Allowed General Unsecured Claim or Unit as of the date determined by the Trust Administrator to be the final Distribution Date, such amount shall be disposed of as provided in the final sentence of Section 2.6(c).
- 5.7. <u>Distributions Not in Compliance with this Article</u>. Subject to <u>Section 5.3(d)</u>, in the event that the Trust Administrator determines in good faith that it is necessary or desirable in

order to carry out the intent and purposes of the Plan, the Confirmation Order and this Trust Agreement to receive any assets or make any distribution in a manner that is not in technical compliance with this Trust Agreement, the Trust Administrator shall be permitted to receive assets or make, or cause to be made, distributions in such manner, but only with the approval of the Trust Monitor; provided, however, that no such distribution shall result in any holder of an Allowed General Unsecured Claim receiving a distribution in excess of the distribution that such holder would have received had such claim been an Initial Allowed General Unsecured Claim or shall discriminate among the holders of Units. Except as aforesaid, no payment or distribution of Avoidance Action Trust Assets shall be made to, or on behalf of, a Trust Beneficiary or any other person except in strict accordance with the terms of this Trust Agreement, the Plan, and the Confirmation Order, unless such payment or distribution shall have been approved by the Bankruptcy Court.

5.8. <u>No Accounting</u>. Except as otherwise provided in the Plan, the Confirmation Order or this Trust Agreement, nothing shall require the Trust Administrator to file any accounting or seek approval of any court with respect to the administration of the Trust or as a condition for making any payment or distribution out of the Avoidance Action Trust Assets.

ARTICLE VA DISTRIBUTIONS TO DIP LENDERS

If any Other Debtor Residual Accepted Assets shall be transferred to the Trust, the Trust Administrator shall make distributions to the DIP Lenders of Distributable Other Debtor Residual Assets, if any, from time to time (but no less frequently than once each calendar year), pro rata as their interests appear, as the Trust Administrator shall determine with the approval of the Trust Monitor or as the Trust Administrator shall be directed by a majority in interest of the DIP Lenders; provided that distributions need not be made in any calendar year to the extent (A) there are no Distributable Other Debtor Residual Assets held by the Trust or (B) the Trust Administrator, with the approval of the Trust Monitor, determines that it is necessary to retain the Distributable Other Debtor Residual Assets to (i) meet any contingent liabilities of the Trust or maintain the value of the Other Debtor Residual Trust Assets (such as for example, in the event that the Trust Administrator determines that the Distributable Other Debtor Residual Assets are so small in amount as not to justify making a distribution, taking into account the costs that would be incurred in making the distribution, the anticipated total amount of Distributable Other Debtor Residual Assets expected to be available for distribution over time and the timing of the distribution or distributions thereof), or (ii) pay reasonable incurred and/or anticipated expenses of the Trust (including any taxes imposed on the Trust or in respect of the Other Debtor Residual Trust Assets) or to satisfy liabilities incurred and/or anticipated by the Trust in accordance with the Plan, the Confirmation Order and this Trust Agreement.

ARTICLE VI ADMINISTRATION OF THE TRUST

- 6.1. Payment of Costs, Expenses and Liabilities (other than in respect of the Other Debtor Residual Accepted Assets).
- (a) Subject to the Budget, the Trust Administrator shall use the Avoidance Action Trust Administrative Cash:
- (i) to pay reasonable costs and expenses of the Trust that are incurred in connection with the administration thereof (including any taxes imposed on the Trust, actual reasonable fees and out-of-pocket expenses incurred by Trust Professionals retained by the Trust Administrator in connection with the administration of the Avoidance Action Trust Assets and preservation of books and records);
- (ii) to satisfy other obligations or other liabilities incurred or assumed by the Trust (or to which the Avoidance Action Trust Assets are otherwise subject) in accordance with the Plan, the Confirmation Order, or this Trust Agreement, including fees and expenses incurred and in connection with, the prosecution and resolution of the Term Loan Avoidance Action the protection, preservation and distribution of the Avoidance Action Trust Assets; and
- (iii) to satisfy any other obligations of the Trust expressly set forth in the Plan, the Confirmation Order or this Trust Agreement to be satisfied out of the Avoidance Action Trust Administrative Cash.
- (b) (i) If, at any time, the Trust Administrator determines that the Avoidance Action Trust Administrative Cash is not reasonably likely to be adequate to satisfy the current and projected future taxes, fees, costs and expenses (including, without limitation any Avoidance Action Trust SEC Reporting Costs) of the Trust (other than in respect of the Other Debtor Residual Trust Assets), the Trust Administrator may, with the approval of the Trust Monitor, reserve an amount, or increase the amount previously reserved, of Distributable Trust Assets to satisfy such taxes, fees, costs and expenses (the "Holdback"). If at any time, the Trust Administrator determines that the Holdback is materially greater than the amount of the current and projected future taxes, fees, costs and expenses as aforesaid, the Trust Administrator shall, with the approval of the Trust Monitor, release from the Holdback the amount of such excess.
- (ii) To the extent necessary to satisfy the taxes, fees, costs and expenses on account of which the Holdback may be reserved, the Trust Administrator may, in consultation with the Trust Monitor, and upon approval by the Bankruptcy Court in accordance with the provisions of Section 6.1(b)(iii), apply all or a portion of the Holdback to the satisfaction of such taxes, fees, costs and expenses.
- (iii) The application of the Trust Administrator seeking Bankruptcy Court approval to utilize Distributable Trust Assets shall include the position of the Trust Monitor in respect thereof. The Trust Administrator shall provide at least twenty (20) days' notice to the Trust Monitor, the holders of Units and the holders of Disputed General Unsecured Claims prior to a hearing on a motion to use any Distributable Trust Assets.

- (c) If, at any time, the Trust Administrator determines that the Avoidance Action Trust Administrative Cash, together with the Holdback (if Distributable Trust Assets are, at such time, available for the purposes of creating a Holdback), is not reasonably likely to be adequate to satisfy the current and projected future taxes, fees, costs and expenses (including, without limitation any Avoidance Action Trust SEC Reporting Costs and the fees and expenses of Trust Professionals) of the Trust (other than in respect of the Other Debtor Residual Trust Assets), the Trust Administrator may utilize the GUC Trust Supplemental Cash, without the need to seek or obtain approval of the Bankruptcy Court, to satisfy such fees and expenses.
- -(i) If, at any time, the Trust Administrator determines that the (d) Avoidance Action Trust Administrative Cash, together with the Holdback (if Distributable Trust Assets are, at such time, available for the purposes of creating a Holdback) and the GUC Trust Supplemental Cash, if any, is not reasonably likely to be adequate to satisfy the current and projected future fees, costs and expenses (including, without limitation any Avoidance Action Trust SEC Reporting Costs, the fees and expenses of Trust Professionals, and any tax liabilities), the Trust Administrator may, in consultation with the Trust Monitor, and upon approval by the Bankruptcy Court in accordance with the provisions of Section 6.1(d)(ii), sell or grant liens on the Term Loan Avoidance Action or any other property of the Trust (other than the Other Debtor Residual Trust Assets) and apply all or a portion of the proceeds of such sale or grant to the satisfaction of such fees, costs and expenses; provided that, except for the Trust's incurrence of expenses in the ordinary course of the Trust's business without any lien or right of setoff attached thereto, the Trust may not incur additional indebtedness, whether unsecured or secured by any interest (including a security interest) in the Term Loan Avoidance Action or Avoidance Action Proceeds without (x) the prior written consent of the DIP Lenders and (y), (y) the prior written consent of the CE Capital Providers, if such incurrence would (1) create a lien senior to or pari passu with the lien securing the CE Capital Providers' Entitlement, or, (2) except as set forth in Section 2.7 of the CE Subordination Agreement, result in the subordination of any portion of the payment of the CE Capital Providers' Entitlement pursuant to Section 5.1(d) of this Trust Agreement, the prior written consent of the Capital Providers and (z) the prior written consent of the LW Capital Provider, if such incurrence would (1) create a lien senior to or pari passu with the lien securing the LW Capital Provider's Entitlement, or (2) except as set forth in Section 2.7 of the LW Subordination Agreement, result in the subordination of any portion of the payment of the LW Capital Provider's Entitlement pursuant to Section 5.1(d) of this Trust Agreement. In the event that the Trust seeks other litigation funding, (A) the DIP Lenders shall be provided a right of first refusal to provide such funding, which right of first refusal must be exercised within thirty (30) days after being presented in writing the opportunity to provide such funding, and (B) such funding (even if provided by the DIP Lenders) shall be junior and subordinate to the Litigation Cost Advance, the amounts owed to the DIP Lenders on account of prior funding of the Trust, and the Supplemental Capital, and shall be subject to a form of subordination that is acceptable to the DIP Lenders and, the <u>CE</u> Capital Providers and the <u>LW Capital Provider</u> in all respects, with such consent not to be unreasonably withheld. Upon any such sale or grant of liens pursuant to this Section 6.1(d), the resulting proceeds shall be designated as Other Supplemental Cash. Any income earned on the Other Supplemental Cash, net of taxes paid thereon, shall be Other Supplemental Cash.
- (ii) The application of the Trust Administrator seeking Bankruptcy Court approval to sell or grant liens pursuant to Section 6.1(d)(i) hereof shall include the position

of the Trust Monitor in respect thereof. The Trust Administrator shall provide at least twenty (20) days' notice to the Trust Monitor, the DIP Lenders, the <u>CE</u> Capital Providers, the <u>LW Capital Provider</u>, the holders of Units and the holders of Disputed General Unsecured Claims prior to a hearing on a motion to sell or grant liens pursuant to <u>Section 6.1(d)(i)</u> hereof.

(e) Notwithstanding that as a result of the utilization of Distributable Trust Assets pursuant to Section 6.1(b) the amount of GUC Distributable Trust Assets shall be less than the assets required to satisfy Claims in the amount of the Current Total Amount then outstanding, the Trust Administrator shall continue to satisfy Disputed General Unsecured Claims, any Unresolved Term Loan Avoidance Action Claims and any Unresolved Other Avoidance Action Claims that become Allowed General Unsecured Claims in the order they are resolved as otherwise provided in this Trust Agreement.

6.2. <u>Payment of Costs, Expenses and Liabilities in respect of the Other Debtor Residual Accepted Assets.</u>

- (a) The Trust Administrator shall not be required to undertake any activity in respect of the Other Debtor Residual Accepted Assets, including for purposes of realizing upon such assets in order to make distributions of Distributable Other Debtor Residual Trust Assets to the DIP Lenders, unless there shall be available to the Trust Administrator Other Debtor Residual Trust Administrative Cash sufficient for such purposes.
- (b) If sufficient Other Debtor Residual Trust Administrative Cash shall be available to the Trust Administrator, then the Trust Administrator shall, as approved by the Trust Monitor or as directed by a majority in interest of the DIP Lenders:
- (i) pay reasonable costs and expenses of the Trust that are incurred in connection with the Other Debtor Residual Accepted Assets (including any taxes imposed on the Trust, actual reasonable fees and out-of-pocket expenses incurred by Trust Professionals retained by the Trust Administrator in connection with the administration of the Other Debtor Residual Accepted Assets and preservation of books and records);
- (ii) satisfy other obligations or other liabilities incurred or assumed by the Trust in respect of the Other Debtor Residual Accepted Assets (or to which the Other Debtor Residual Accepted Assets are otherwise subject) in accordance with the Plan, the Confirmation Order, or this Trust Agreement, including fees and expenses incurred and in connection with, the prosecution and resolution of any action to realize upon the Other Debtor Residual Accepted Assets and the protection, preservation and distribution of the Other Debtor Residual Accepted Assets; and
- (iii) satisfy any other obligations of the Trust expressly set forth in the Plan, the Confirmation Order or this Trust Agreement to be satisfied out of the Other Debtor Residual Trust Administrative Cash.

6.3. Budget.

(a) The Trust Administrator shall prepare and submit to the Trust Monitor, the DIP Lenders and, the <u>CE</u> Capital Providers and the <u>LW Capital Provider</u> for approval a reasonably

detailed annual plan and budget (the "Budget") at least thirty (30) days prior to the commencement of each calendar year. Such annual plan and Budget shall set forth (on a quarterly basis) in reasonable detail: (A) the Trust Administrator's anticipated actions to administer the Avoidance Action Trust Assets; and (B) the anticipated fees and expenses, including professional fees, associated with the administration of the Trust, a separate amount representing the anticipated fees and expenses of the Trust Monitor and detail as to how the Trust will budget and spend the Avoidance Action Trust Administrative Cash. Such Budget shall be updated and submitted to the Trust Monitor, the DIP Lenders, and, the CE Capital Providers and the LW Capital Provider for review on a quarterly basis, and each such quarterly update shall reflect the variances (with explanations) between (x) the Budget, (y) any updated Budget, and (z) the actual results for the same period. If the Trust identifies a need for or the incurrence of actual expenditures that materially differ from the relevant Budget, the Trust will notify the DIP Lenders-and, the CE Capital Providers and the LW Capital Provider as promptly as practicable, and no later than fifteen (15) days after identifying the issue. For the avoidance of doubt, the DIP Lenders and, the CE Capital Providers and the LW Capital Provider may object in the Bankruptcy Court with respect to any quarterly update that materially changes the Budget and the Bankruptcy Court shall resolve such dispute. All actions by the Trust Administrator shall be consistent with the Budget (as updated). The Trust Administrator may obtain any required approval of the Budget on reasonable negative notice (which shall be not less than 15 days after receipt of the Budget) and approval of the Budget shall not be unreasonably withheld. In the event of any dispute concerning the Budget (or the taking of actions consistent with the Budget), the Trust Administrator or the Trust Monitor may petition the Bankruptcy Court to resolve such dispute.

- (b) The Trust Administrator, with the approval of the Trust Monitor and the DIP Lenders may agree on a budget for activities in respect of the Other Debtor Residual Accepted Assets.
- (c) Notwithstanding any other provision of this Trust Agreement, the approval of the DIP Lenders shall not be required for any use of the Avoidance Action Trust SEC Reporting Cash, the GUC Trust Supplemental Cash or the Other Supplemental Cash.
 - 6.4. [Intentionally omitted.]
- 6.5. <u>Compliance with Laws</u>. Any and all distributions of Avoidance Action Trust Assets shall be in compliance with applicable laws, including applicable federal and state tax and securities laws.
- 6.6. <u>Fiscal Year</u>. Except for the first and last years of the Trust, the fiscal year of the Trust shall be the calendar year. For the first and last years of the Trust, the fiscal year of the Trust shall be such portion of the calendar year that the Trust is in existence.

6.7. Books and Records.

- (a) The Trust Administrator shall maintain and preserve the Debtors' books, records and files that shall have been delivered to or created by the Trust Administrator.
- (b) The Trust Administrator shall maintain books and records relating to the assets, liabilities, income and expense of the Trust, all distributions made by the Trust and the

payment of fees and expenses of, and satisfaction of claims against or assumed by, the Trust and the Trust Administrator, in such detail and for such period of time as may be necessary to enable it to make full and proper reports in respect thereof in accordance with the provisions of this Trust Agreement and otherwise to comply with applicable provisions of law, including tax law.

- (c) The Trust Administrator shall maintain, or cause to be maintained, a register of holders of Units, from time to time outstanding, to the extent any Units are issued hereunder, in customary form.
- 6.8. <u>Cash Payments</u>. All distributions of Distributable Trust Cash required to be made by the Trust Administrator may be made in Cash denominated in U.S. dollars by checks drawn on a United States domestic bank selected by the Trust Administrator or, at the option of the Trust Administrator, by wire transfer from a United States domestic bank selected by the Trust Administrator or as otherwise required or provided in applicable agreements; <u>provided</u>, <u>however</u>, that cash payments to foreign persons may be made, at the option of the Trust Administrator, in such funds as and by such means as are necessary or customary in a particular foreign jurisdiction.
- 6.9. <u>Insurance</u>. The Trust shall maintain customary insurance coverage for the protection of the Trust Administrator Parties and the Trust Monitor Parties and any such other persons serving as administrators and overseers of the Trust, on and after the Avoidance Action Trust Transfer Date, in all cases in accordance with the Budget. The Trust Administrator may also obtain such insurance coverage as it deems necessary and appropriate with respect to real and personal property which may become Avoidance Action Trust Assets, if any, in accordance with such Budget. To the extent that there is any incremental cost for customary insurance coverage covering the activities of the Trust Administrator Parties and the Trust Monitor Parties in respect of the Other Debtor Residual Accepted Assets, the Trust Administrator and the Trust Monitor shall not be required to undertake any such activities unless there is available sufficient Other Debtor Residual Trust Administrative Cash to fund such incremental cost.

ARTICLE VII TAX MATTERS

7.1. Tax Treatment.

- (a) For all U.S. federal and applicable state and local income tax purposes, all parties (including the Debtors, the Trust Administrator, the holders of the DIP Credit Agreement Claims, and the holders of Allowed General Unsecured Claims) shall treat the Trust and the transfer of the Avoidance Action Trust Assets and the Other Debtor Residual Assets to the Trust in a manner consistent with the remainder of this Section 7.1.
- (b) If no Other Debtor Residual Assets are transferred to the Trust upon the dissolution of MLC and the DIP Lender Distributable Trust Assets have not been determined (either as a percentage or as a fixed amount of Distributable Trust Assets or on some other basis) on or prior to the Avoidance Action Trust Transfer Date, then (subject to clause (c) below) the Trust Administrator shall treat the Trust as (A) a "disputed ownership fund" governed by Treasury Regulation section 1.468B-9 (including, if required, timely so electing), (B) if permitted under

applicable law and at the election of the Trust Administrator, as a "complex trust," or (C) as otherwise permitted pursuant to a private letter ruling from the IRS.

- If Other Debtor Residual Assets are transferred to the Trust upon the dissolution of MLC or the DIP Lender Distributable Trust Assets have been determined (either as a percentage or as a fixed amount of Distributable Trust Assets or on some other basis) on or prior to the Avoidance Action Trust Transfer Date, or otherwise upon determination of the DIP Lender Distributable Trust Assets (either as a percentage or as a fixed amount of Distributable Trust Assets or on some other basis) after the Avoidance Action Trust Transfer Date, the Trust (other than the Avoidance Action Trust Claims Reserve) shall be treated as a liquidating trust that is treated as a grantor trust and the Avoidance Action Trust Assets (upon the determination of the DIP Lender Distributable Trust Assets) and Other Debtor Residual Trust Assets (upon the transfer of Other Debtor Residual Assets to the Trust) shall be treated as (i) being transferred directly to the Trust Beneficiaries; provided, however, that to the extent Avoidance Action Trust Assets are allocable to Disputed General Unsecured Claims, Unresolved Term Loan Avoidance Action Claims or Unresolved Other Avoidance Action Claims, such Avoidance Action Trust Assets shall be treated as being transferred to the Avoidance Action Trust Claims Reserve, followed by (ii) the transfer by such Trust Beneficiaries of the Avoidance Action Trust Assets (other than the Avoidance Action Trust Assets allocable to the Avoidance Action Trust Claims Reserve) and the Other Debtor Residual Trust Assets, as applicable, to the Trust in exchange for beneficial interests in the Trust. Accordingly, Trust Beneficiaries receiving beneficial interests in the Trust shall be treated as the grantors and owners of their respective share of the Avoidance Action Trust Assets (other than any Avoidance Action Trust Assets allocable to the Avoidance Action Trust Claims Reserve) and Other Debtor Residual Trust Assets, as applicable.
- (d) Any determination made pursuant to this <u>Section 7.1</u> shall be conclusive and binding on all parties (including the Debtors, the Trust Administrator, the holders of the DIP Credit Agreement Claims, and the holders of Allowed General Unsecured Claims) for U.S. federal, and (to the extent permitted by applicable law) state and local, income tax purposes. Accordingly, to the extent permitted by applicable law, all parties shall report consistently with the U.S. federal income tax treatment of the Trust by the Trust Administrator for state and local income tax purposes.
- 7.2. <u>Valuation of Assets</u>. As soon as practicable after the Avoidance Action Trust Transfer Date, the Trust Administrator shall make a good-faith valuation of the Avoidance Action Trust Assets and Other Debtor Residual Trust Assets, and such valuation shall be made available from time to time, to the extent relevant, and shall be used consistently by all parties (including the Debtors, the Trust Administrator, the holders of the DIP Credit Agreement Claims, and the holders of Allowed General Unsecured Claims) for all U.S. federal and applicable state and local income tax purposes.
- 7.3. Payment of Taxes. The Trust Administrator shall be responsible for payment, out of the Avoidance Action Trust Assets, of any taxes imposed on the Trust (other than in respect of the Other Debtor Residual Assets) or the Avoidance Action Trust Assets, including the Avoidance Action Trust Claims Reserve. The Trust Administrator shall be responsible for payment, out of the Other Debtor Residual Assets of any taxes imposed on the Trust in respect of the Other Debtor Residual Assets or on the Other Debtor Residual Assets. In the event, and to the extent, any Cash

retained on account of Disputed General Unsecured Claims, Unresolved Term Loan Avoidance Action Claims or Unresolved Other Avoidance Action Claims in the Avoidance Action Trust Claims Reserve is insufficient to pay the portion of any such taxes attributable to the taxable income arising from the assets allocable to, or retained on account of, Disputed General Unsecured Claims, Unresolved Term Loan Avoidance Action Claims or Unresolved Other Avoidance Action Claims, such taxes shall be (i) reimbursed from any subsequent Cash amounts retained on account of Disputed General Unsecured Claims, Unresolved Term Loan Avoidance Action Claims or Unresolved Other Avoidance Action Claims, or (ii) to the extent such Disputed General Unsecured Claims, Unresolved Term Loan Avoidance Action Claims or Unresolved Other Avoidance Action Claims, Unresolved Term Loan Avoidance Action Claims, Unresolved Term Loan Avoidance Action Claims or Unresolved Other Avoidance Action Claims, Unresolved Term Loan Avoidance Action Claims or Unresolved Other Avoidance Action Claims, Unresolved Term Loan Avoidance Action Claims or Unresolved Other Avoidance Action Claims.

7.4. Tax Reporting.

- (a) The Trust Administrator shall file (or cause to be filed) Tax Returns for the Trust treating the Trust (except the Avoidance Action Trust Claims Reserve or as otherwise provided in Section 7.1(b) above) as a grantor trust pursuant to Treasury Regulation section 1.671-4(a) and in accordance with the applicable provisions of this Section 7.4. The Trust Administrator also shall annually send to each Trust Beneficiary a separate statement setting forth such Trust Beneficiary's share of items of income, gain, loss, deduction, or credit of the Trust (including, for the avoidance of doubt, earnings on the Avoidance Action Trust Administrative Cash, the Avoidance Action Trust SEC Reporting Cash, the GUC Trust Supplemental Cash and the Other Supplemental Cash) and shall instruct all Trust Beneficiaries to report such items on their respective U.S. federal income Tax Returns or to forward the appropriate information to their respective beneficial holders with instructions to report such items on their U.S. federal income Tax Returns. The Trust Administrator also shall file (or cause to be filed) any other statements, returns, or disclosures relating to the Trust that are required by any governmental unit.
- Allocations of the Trust's taxable income among the Trust Beneficiaries shall be determined by reference to the manner in which an amount of Cash equal to such taxable income would be distributed (without regard to any restrictions on distributions described herein) if, immediately prior to such deemed distribution, the Trust had distributed all of its other assets (valued at their tax book value and other than assets attributable to the Avoidance Action Trust Claims Reserve) to the Trust Beneficiaries, in each case up to the tax book value of the assets treated as contributed by such Trust Beneficiaries, adjusted for prior taxable income and loss and taking into account all prior and concurrent distributions from the Trust. Similarly, taxable loss of the Trust shall be allocated by reference to the manner in which an economic loss would be borne immediately after a liquidating distribution of the remaining Avoidance Action Trust Assets and Other Debtor Residual Trust Assets. The tax book value of the Avoidance Action Trust Assets and Other Debtor Residual Trust Assets for this purpose shall equal their fair market value on the Avoidance Action Trust Transfer Date and Other Debtor Residual Trust Assets Transfer Date, as applicable, adjusted in accordance with tax accounting principles prescribed by the Tax Code, applicable Treasury Regulations, and other applicable administrative and judicial authorities and pronouncements.

- (c) The Trust Administrator shall (x) treat the Avoidance Action Trust Claims Reserve for U.S. federal income tax purposes as either (i) a "disputed ownership fund" governed by Treasury Regulation section 1.468B-9 by timely so electing or (ii) a "complex trust," provided, however, that if the Trust is treated as a "disputed ownership fund" or as a "complex trust" pursuant to Section 7.1(b) above, then the Avoidance Action Trust Claims Reserve shall be treated in the same manner, and (y) to the extent permitted by applicable law, report consistently with the foregoing for state and local income tax purposes. Any determination made pursuant to this Section 7.4 shall be conclusive and binding on all parties (including the Debtors, the Trust Administrator, the holders of the DIP Credit Agreement Claims, and the holders of Allowed General Unsecured Claims) for U.S. federal, state, and local income tax purposes.
- 7.5. Tax Withholdings. The Trust Administrator shall withhold and pay to the appropriate taxing authority all amounts required to be withheld pursuant to the Tax Code, Treasury Regulations or other applicable requirements, including any provision of any foreign, state or local tax law, with respect to any payment or distribution to the Trust Beneficiaries. All such amounts withheld, and paid to the appropriate taxing authority, shall be treated as amounts distributed to such Trust Beneficiaries for all purposes of this Trust Agreement. The Trust Administrator shall be authorized to collect such tax information from the Trust Beneficiaries (including social security numbers or other tax identification numbers) as it in its sole discretion deems necessary to effectuate the Plan, the Confirmation Order and this Trust Agreement, or to comply with any applicable withholding or reporting requirement. The Trust Administrator may refuse to make a distribution to any Trust Beneficiary that fails to furnish such information in a timely fashion, until such information is furnished; *provided*, *however*, that upon a Trust Beneficiary furnishing such information, the Trust Administrator shall make such distribution to which such Trust Beneficiary is entitled, without interest.
- 7.6. Expedited Determination of Taxes. The Trust Administrator may request an expedited determination of taxes of the Trust, including the Avoidance Action Trust Claims Reserve, under Section 505(b) of the Bankruptcy Code for any or all Tax Returns filed for, or on behalf of, the Trust for any or all taxable periods (or part thereof) through the dissolution of the Trust.

7.7. [Intentionally omitted.]

- 7.8. <u>Delivery of Statement of Transfers</u>. If the Trust Administrator elects to treat (i) the Trust, pursuant to and to the extent provided in <u>Section 7.1</u> above and/or (ii) the Avoidance Action Trust Claims Reserve as a disputed ownership fund within the meaning of Treasury Regulation section 1.468B-9, then following the Avoidance Action Trust Transfer Date (but in no event later than February 15th of the calendar year following the Avoidance Action Trust Transfer Date), MLC shall provide a "§ 1.468B-9 Statement" to the Trust Administrator in accordance with Treasury Regulation section 1.468B-9(g).
- 7.9. <u>Allocation of Distributions Between Principal and Interest</u>. All deemed distributions (including deemed transfers pursuant to <u>Section 7.1(b)(i)</u>) in connection with the allowance of any Allowed General Unsecured Claim shall be allocated first to the principal amount of such Allowed General Unsecured Claim, as determined for federal income tax purposes, and thereafter, to the remaining portion of such Allowed General Unsecured Claim, if any.

ARTICLE VIII POWERS OF AND LIMITATIONS ON THE TRUST ADMINISTRATOR

8.1. Powers of the Trust Administrator.

- (I) Other Than in Respect of the Other Debtor Residual Accepted Assets
- (a) Pursuant to the terms of the Plan and the Confirmation Order, the Trust Administrator shall have various powers, duties and responsibilities concerning the prosecution of and resolution of the Term Loan Avoidance Action, maximizing the property of the Trust, the disposition of the Avoidance Action Trust Assets and the administration of the Trust. In addition, the Trust Administrator shall coordinate with the GUC Trust Administrator to maximize efficiency in distributions to general unsecured creditors in any situation where such coordination would be beneficial.
- (b) The Trust Administrator shall have only such rights, powers and privileges expressly set forth in the Plan, the Confirmation Order or this Trust Agreement and as otherwise provided by applicable law. Subject to the Plan, the Confirmation Order and other provisions herein, including the provisions relating to approvals of the Trust Monitor, the Trust Administrator shall be expressly authorized to undertake the following actions, in the Trust Administrator's good faith judgment, in the best interests of the Trust Beneficiaries and in furtherance of the purpose of the Trust:
 - (i) hold and manage the Avoidance Action Trust Assets;
- (ii) hold legal title to any and all rights of the Trust Beneficiaries in, to or arising from the Avoidance Action Trust Assets, for the benefit of the Trust Beneficiaries that are entitled to distributions therefrom under the Plan, whether, in the case of GUC Beneficiaries, their General Unsecured Claims are Allowed on or after the Avoidance Action Trust Transfer Date:
- (iii) prosecute and, if appropriate, sell, grant liens upon (subject to Section 6.1(d) hereof), settle and resolve, abandon and/or dismiss the Term Loan Avoidance Action;
- (iv) execute all agreements, instruments and other documents (including, without limitation, any loan agreements or sale agreements for the purposes of funding the fees and expenses of the Trust), and effect all other actions necessary or appropriate to dispose of the Avoidance Action Trust Assets;
- (v) monitor and enforce the implementation of the Plan insofar as relating to this Trust Agreement, the Avoidance Action Trust Assets or the Trust;
- (vi) calculate and implement distributions of the GUC Distributable Trust Assets obtained through the exercise of its power and authority as contemplated by the Plan, the Confirmation Order and this Trust Agreement and in accordance with the interests of the holders of Allowed General Unsecured Claims;

- (vii) retain, pay, oversee and direct the services of, and terminate Trust Professionals in accordance with <u>Section 8.3</u> hereof to carry out its duties and obligations hereunder, in all cases in accordance with the Budget;
- (viii) pay the reasonable fees and expenses of the Trust Administrator and Trust Monitor, in all cases in accordance with the Budget;
- (ix) incur and pay all reasonable expenses, satisfy ordinary course liabilities and make all other payments reasonable and necessary to administer and dispose of the Avoidance Action Trust Assets, in all cases in accordance with the Budget;
- (x) invest monies received by the Trust, the Trust Administrator or otherwise held by the Trust or the Trust Administrator in accordance with <u>Section 8.4</u> hereof;
- (xi) protect and enforce the rights to the Avoidance Action Trust Assets vested in the Trust Administrator by this Trust Agreement by any method deemed reasonably appropriate, including by judicial proceedings or pursuant to any applicable bankruptcy, insolvency, moratorium or similar law and general principles of equity;
- (xii) vote any claim or interest held by the Trust in a case under the Bankruptcy Code and receive any distribution therefrom for the benefit of the Trust;
- (xiii) make all necessary filings in accordance with any applicable law, statute or regulation;
- (xiv) purchase customary insurance coverage in accordance with <u>Section</u> 6.9 hereof;
- (xv) assert and/or waive any applicable privileges (legal or otherwise) on behalf of the Trust, or with respect to the Avoidance Action Trust Assets held by the Debtors at any time (prepetition or postpetition);
 - (xvi) maintain the books and records of the Trust;
- (xvii) open, maintain and close any bank, securities or other accounts that are necessary and appropriate to manage the Avoidance Action Trust Assets, including but not limited to the accounts listed on <u>Exhibit A</u> hereto;
- (xviii) receive from the Debtors and administer the Avoidance Action Trust SEC Reporting Cash in accordance with <u>Section 2.3(e)</u> hereof and file such reports as may be required pursuant to the applicable rules, regulations and interpretations of the SEC;
- (xix) receive from the GUC Trust and administer and utilize the GUC Trust Supplemental Cash in accordance with Section 2.3(f) hereof; and
- (xx) perform such functions and take such actions as are provided for or permitted in the Plan, the Confirmation Order, this Trust Agreement, the Settlement Agreement, the Litigation Cost Advance Agreement, or any other agreement executed pursuant to the Plan and

take any other actions as it may deem to be reasonably necessary or appropriate to realize, preserve and dispose of the Avoidance Action Trust Assets.

(c) [Intentionally omitted.]

- (d) In all circumstances, the Trust Administrator shall act in the best interests of all Trust Beneficiaries and in furtherance of the purpose of the Trust, and in a manner not inconsistent with the best interests of the Trust Beneficiaries and consistent with the Budget. The Trust Administrator shall not take any action inconsistent with the purpose of the Trust, or take (or fail to take) any action that would cause the Trust (other than the Avoidance Action Trust Claims Reserve) to fail to qualify as a liquidating trust within the meaning of Treasury Regulation section 301.7701-4(d) that is treated as a grantor trust.
- Notwithstanding any provision herein to the contrary, the Trust (e) Administrator shall not serve on the board of directors, management committee or any similar governing body of any non-Debtor subsidiary of MLC, where the charter, limited liability company agreement, partnership agreement or other similar constituent document of such subsidiary does not provide for a liquidating purpose for such subsidiary. Except as otherwise provided in this Trust Agreement, the Trust Administrator will not be required to obtain the order or approval of the Bankruptcy Court, or any other court of competent jurisdiction in, or account to the Bankruptcy Court or any other court of competent jurisdiction for, the exercise of any right, power or privilege conferred hereunder. Notwithstanding the foregoing, where the Trust Administrator determines, in its reasonable discretion, that it is necessary, appropriate or desirable, the Trust Administrator will have the right to submit to the Bankruptcy Court or any other court of competent jurisdiction any question or questions regarding any specific action proposed to be taken by the Trust Administrator with respect to this Trust Agreement, the Trust, or the Avoidance Action Trust Assets, including the administration and distribution of the Avoidance Action Trust Assets and the termination of the Trust. Pursuant to the Plan, the Bankruptcy Court has retained jurisdiction for such purposes and may approve or disapprove any such proposed action upon motion by the Trust Administrator.

(II) In Respect of the Other Debtor Residual Accepted Assets

The Trust Administrator shall have the rights, powers and privileges to act in respect of the Other Debtor Residual Accepted Assets, if any, in the manner set forth in Section 8.1(I), mutatis mutandis. In all such circumstances, the Trust Administrator shall act in the best interests of DIP Lenders and in furtherance of the purpose of the Trust, and in a manner not inconsistent with the best interests of the DIP Lenders. For the avoidance of doubt, the Trust Administrator shall not be obligated to undertake any activities in respect of the Other Debtor Residual Trust Assets unless there shall be available sufficient Other Debtor Residual Trust Administrative Cash to pay in full its fees, costs and expenses in respect thereof.

8.2. <u>Limitations on the Trust Administrator</u>. The Trust Administrator shall not be authorized to engage, in its capacity as Trust Administrator, in any trade or business with respect to the Avoidance Action Trust Assets or to take (or fail to take) any action that would cause the Trust (other than the Avoidance Action Trust Claims Reserve) to fail to qualify as a liquidating trust within the meaning of Treasury Regulation section 301.7701-4(d) that is treated as a grantor

trust. The Trust Administrator shall take such actions consistent with the prompt orderly disposition of the Avoidance Action Trust Assets and the Other Debtor Residual Accepted Assets, if any, as required by applicable law and consistent with the treatment of the Trust (other than the Avoidance Action Trust Claims Reserve) as a liquidating trust within the meaning of Treasury Regulation section 301.7701-4(d) that is treated as a grantor trust, to the extent such actions are permitted by this Trust Agreement. The Trust Administrator shall, in its capacity as Trust Administrator, be restricted to (x) the liquidation of the Trust on behalf, and for the benefit, of the Trust Beneficiaries and the distribution and application of Avoidance Action Trust Assets for the purposes set forth in, and the conservation and protection of the Avoidance Action Trust Assets and the administration thereof, and (y) the liquidation of Other Debtor Residual Trust Assets for the purposes set forth in, and the conservation and protection of the Other Debtor Residual Trust Assets and the administration thereof, in each case in accordance with, the provisions of the Plan, the Confirmation Order and this Trust Agreement.

8.3. Agents and Professionals.

- (a) The Trust Administrator on behalf of the Trust may, but shall not be required to, from time to time enter into contracts with, consult with and retain Trust Professionals, on such terms as the Trust Administrator deems appropriate in accordance with the terms hereof and (other than in respect of the Other Debtor Residual Accepted Assets) in accordance with the Budget. None of the professionals that represented parties-in-interest in the Chapter 11 Cases shall be precluded from being engaged by the Trust Administrator solely on account of their service as a professional for such parties-in-interest prior to the Avoidance Action Trust Transfer Date or the Other Debtor Residual Assets Transfer Date, as the case may be.
- After the Avoidance Action Trust Transfer Date, Trust Professionals shall (b) be required to submit reasonably detailed invoices on a monthly basis to the Trust Administrator, the Trust Monitor, the DIP Lenders and the CE Capital Providers and the LW Capital Provider, including in such invoices a description of the work performed, the individuals who performed such work, the hourly rate of such person, plus an itemized statement of expenses for which payment is sought. Invoices submitted to the DIP Lenders and, the CE Capital Providers and the LW Capital Provider may be in a form reasonably necessary to protect the attorney-client privilege. For the avoidance of doubt, any and all payments by the Trust from funds that (i) originated from the Litigation Cost Advance or the Supplemental Capital, or (ii) that are contained in the Funding Account, the CE Capital Providers Funding Account or the LW Capital Provider Funding Account shall be subject to this invoice disclosure requirement. No payment on account of such invoices shall be made until 15 days after presentation of the invoices to the DIP Lenders and, the CE Capital Providers and the LW Capital Provider. After such period has passed, the Trust Administrator shall timely pay all such invoices that are not disputed by the Trust Administrator or the Trust Monitor. If either or bothany of the DIP Lenders or, the CE Capital Providers or the LW Capital Provider have questions or concerns about any invoice submitted to the Trust, or about any other issue relating to the Trust's finances or administration, they shall be entitled to discuss the matter with the Trust Monitor, as well as with the Trust Administrator and/or counsel for the Trust, as appropriate; and the Trust Monitor will consult with the DIP Lenders-or, the CE Capital Providers or the LW Capital Provider, as applicable, and give good faith consideration to any objection that the DIP Lenders-or, the CE Capital Providers and/or the LW Capital Provider raise

before approval of payment by the Trust Administrator or non-objection by the Trust Monitor. The Trust Monitor, Trust Administrator, and/or counsel for the Trust, as appropriate, will engage in good faith discussions and attempt in good faith to provide information reasonably requested by either or both any of the DIP Lenders or, the CE Capital Providers or the LW Capital Provider. In the event of any dispute concerning the entitlement to, or the reasonableness of any compensation and/or expenses of any Trust Professionals, either the Trust Administrator or the affected Trust Professional may petition the Bankruptcy Court to resolve the dispute.

(c) Except as permitted by Section 6.1(b), (c), and (d), and Section 2.3(e) and (f), all payments to Trust Professionals (other than in respect of Other Debtor Residual Trust Assets) shall be paid out of the Avoidance Action Trust Administrative Cash. Payments to Trust Professionals for activities in respect of the Other Debtor Residual Trust Assets shall be paid out of the Other Debtor Residual Trust Administrative Cash.

8.4. Investment of Trust Cash.

- The Trust Administrator shall establish segregated accounts for the Trust Cash as follows: (i) Distributable Trust Cash which shall be held in trust for the benefit of the Trust Beneficiaries; (ii) Distributable Other Debtor Residual Cash, which shall be held in trust for the benefit of the DIP Lenders and on which the DIP Lenders shall have a lien; (iii) Avoidance Action Trust Administrative Cash (excluding the Litigation Cost Advance) and Other Debtor Residual Trust Administrative Cash which shall be used to pay the administrative expenses of the Trust, on which the DIP Lenders shall have a lien; (iv) Avoidance Action Trust SEC Reporting Cash which shall be used to satisfy Avoidance Action Trust SEC Reporting Costs in accordance with Section 2.3(e) hereof; (v) GUC Trust Supplemental Cash which shall be used to satisfy current and projected fees and expenses of the Trust (including for the payment of Trust Professionals and any tax liabilities) in accordance with Section 2.3(f) hereof; (vi) Other Supplemental Cash, which shall be used to satisfy the current and projected future fees, costs and expenses (including, without limitation any Avoidance Action Trust SEC Reporting Costs, the fees and expenses of Trust Professionals, and any tax liabilities) in accordance with Section 6.1(d) hereof; (vii) Litigation Cost Advance, which shall be held in the Funding Account, which Funding Account shall be subject to a deposit account control agreement acceptable to the DIP Lenders and the Trust and on which the DIP Lenders shall have a first priority lien; and (vi(viii)) the CE Supplemental Capital, which shall be held in the CE Capital Provider Funding Account, which upon the request of the CE Capital Providers shall be subject to a deposit account control agreement acceptable to the CE Capital Providers and the Trust; and (ix) the LW Supplemental Capital, which shall be held in the LW Capital Provider Funding Account, which upon the request of the LW Capital Provider shall be subject to a deposit account control agreement acceptable to the LW Capital Provider and the Trust.
- (b) The Trust Administrator shall invest the Trust Cash (including any earnings thereon or proceeds thereof) in the manner set forth in this Section 8.4, but shall otherwise be under no liability for interest or income on any monies received by the Trust hereunder and held for distribution or payment to the Trust Beneficiaries, except as such interest shall actually be received. Investment of any Trust Cash shall be administered in accordance with the general duties and obligations hereunder. The right and power of the Trust Administrator to invest the Trust Cash and the proceeds thereof, or any income earned by the Trust, shall be limited to

investing such Trust Cash (pending distribution or disbursement in accordance with the Plan or this Trust Agreement) in Permissible Investments; <u>provided</u>, <u>however</u>, that such Permissible Investments shall be limited to include only those investments that a liquidating trust, within the meaning of Treasury Regulations section 301.7701-4(d), may be permitted to hold, pursuant to the Treasury Regulations, or any modification in the IRS guidelines, whether set forth in IRS rulings, other IRS pronouncements or otherwise.

- (c) For the avoidance of doubt, the Trust is not, and will not hold itself out as, an "investment company" as such term is understood under the Investment Company Act of 1940, and is prohibited from investing, reinvesting or trading in securities (other than making any Permissible Investments as contemplated by the Plan, the Confirmation Order and this Trust Agreement) or conducting any trade or business other than implementing the Plan and distributing Distributable Trust Assets under the Plan and this Trust Agreement.
- 8.5. Communication with the GUC Trust Administrator. The Trust Administrator shall communicate with the GUC Trust Administrator to obtain such information regarding, as of a given date, (A) the holders and amounts of General Unsecured Claims, Disputed General Unsecured Claims, Unresolved Other Avoidance Action Claims and Resolved Allowed General Unsecured Claims, (B) the respective Maximum Amounts of all Disputed General Unsecured Claims, Unresolved Term Loan Avoidance Action Claims and Unresolved Other Avoidance Action Claims, (C) the Current Total Amount, (D) the Aggregate Maximum Amount, (E) any components of the foregoing and (E) any other information within the custody or control of the GUC Trust Administrator which shall be necessary or desirable in order for the Trust Administrator to timely make any calculation or determination hereunder to identify and make distributions to the GUC Beneficiaries and to maintain any books and records required to be maintained, or necessary or desirable for the Trust Administrator or the Trust Monitor to fulfill their respective functions, hereunder; provided, however, that the provision of such information shall be under appropriate arrangements of confidentiality to the extent such information has at the time not been publicly disclosed.
- 8.6. <u>Termination</u>. The duties, responsibilities and powers of the Trust Administrator will terminate when the Trust is dissolved and terminated pursuant to <u>Article IV</u> hereof and the Trust Administrator has performed all of its obligations under <u>Section 4.3</u>, by an order of the Bankruptcy Court or by entry of a final decree closing the Debtors' cases before the Bankruptcy Court; <u>provided</u>, <u>however</u>, that <u>Sections 9.4, 9.5</u> and <u>9.6</u> hereof shall survive such termination, dissolution and entry.

ARTICLE IX ADDITIONAL MATTERS CONCERNING THE TRUST ADMINISTRATOR

9.1. <u>Reliance by Trust Administrator</u>. Except as otherwise provided in the Plan, the Confirmation Order or this Trust Agreement, the Trust Administrator may rely and shall be protected in acting upon any resolution, statement, instrument, opinion, report, notice, request, consent, order or other paper or document reasonably believed by the Trust Administrator to be genuine and to have been signed or presented by the proper party or parties.

- 9.2. <u>Liability to Third Persons</u>. To the fullest extent permitted by applicable law, the Trust Administrator Parties shall not be subject to any personal liability whatsoever, in tort, contract or otherwise, to any person (including, in the case of the Trust Administrator, to any Trust Professionals retained by the Trust Administrator in accordance with this Trust Agreement) in connection with the Avoidance Action Trust Assets, the Other Debtor Residual Trust Assets or the affairs of the Trust and shall not be liable with respect to any action taken or omitted to be taken in good faith, except for actions and omissions determined by a Final Order of the Bankruptcy Court to be due to their respective willful misconduct (including, but not limited to, conduct that results in a personal profit at the expense of the Trust), gross negligence, fraud, malpractice, criminal conduct, unauthorized use of confidential information that causes damages, breach of fiduciary duty (to the extent applicable), or ultra vires acts, and all such persons shall look solely to the Avoidance Action Trust Assets (other than in respect of the Other Debtor Residual Accepted Assets and activities related thereto) or the Other Debtor Residual Trust Assets (in respect of the Other Debtor Residual Accepted Assets and activities related thereto) for satisfaction of claims of any nature arising in connection with affairs of the Trust.
- 9.3. Non-liability of Trust Administrator for Acts of Others. Except as provided herein, nothing contained in the Plan, the Confirmation Order or this Trust Agreement shall be deemed to be an assumption by the Trust Administrator of any of the liabilities, obligations or duties of the Debtors or shall be deemed to be or contain a covenant or agreement by the Trust Administrator to assume or accept any such liability, obligation or duty. Any successor Trust Administrator may accept and rely upon any accounting made by or on behalf of any predecessor Trust Administrator hereunder, and any statement or representation made as to the assets comprising the Avoidance Action Trust Assets or the Other Debtor Residual Trust Assets, or as to any other fact bearing upon the prior administration of the Trust, so long as it has a good faith basis to do so. The Trust Administrator shall not be liable for having accepted and relied in good faith upon any such accounting, statement or representation if it is later proved to be incomplete, inaccurate or untrue. Neither the Trust Administrator nor any successor Trust Administrator shall be liable for any act or omission of any predecessor Trust Administrator, nor have a duty to enforce any claims against any predecessor Trust Administrator on account of any such act or omission, unless directed in good faith to do so by the Trust Monitor.
- Exculpation. As of the earlier of the Avoidance Action Trust Transfer Date and 9.4. the Other Debtor Residual Assets Transfer Date, to the fullest extent permitted by applicable law, the Trust Administrator Parties shall be and hereby are exculpated by all Persons, including holders of DIP Credit Agreement Claims, General Unsecured Claims and Units and other parties-ininterest, from any and all claims, causes of action and other assertions of liability arising out of the discharge of their respective powers and duties conferred by the Plan, the Confirmation Order, this Trust Agreement or any Order of the Bankruptcy Court entered pursuant to or in furtherance of the Plan, or applicable law or otherwise, except for actions or omissions to act that are determined by Final Order of the Bankruptcy Court to have arisen out of each such Trust Administrator Party's own respective willful misconduct (including, but not limited to, conduct that results in a personal profit at the expense of the Trust), gross negligence, fraud, malpractice, criminal conduct, unauthorized use of confidential information that causes damages, breach of fiduciary duty (to the extent applicable), or ultra vires acts. No holder of a DIP Credit Agreement Claim, General Unsecured Claim or other party-in-interest will have or be permitted to pursue any claim or cause of action against the Trust Administrator Parties or the Trust, for making payments and

distributions in accordance with the Plan, the Confirmation Order or the this Trust Agreement or for implementing the provisions thereof. Any action taken or omitted to be taken with the express approval of the Bankruptcy Court and, in the case of action taken in respect of the Other Debtor Residual Accepted Assets, with the approval or at the direction of the DIP Lenders will conclusively be deemed not to constitute willful misconduct, gross negligence, fraud, malpractice, criminal conduct, unauthorized use of confidential information that causes damages, breach of fiduciary duty, or ultra vires acts; <u>provided</u>, <u>however</u>, that notwithstanding any provision herein to the contrary, the Trust Administrator shall not be obligated to comply with a direction of the Trust Monitor, whether or not express, which would result in a change to the distribution provisions of the Plan, the Confirmation Order or this Trust Agreement.

9.5. <u>Limitation of Liability</u>. In no event shall the Trust Administrator Parties be liable for punitive, exemplary, consequential, special or other damages for a breach of, or otherwise in connection with, this Trust Agreement under any circumstances.

9.6. <u>Indemnity</u>.

To the fullest extent permitted by applicable law, the Trust Administrator (a) Parties shall be indemnified by the Trust from the Avoidance Action Trust Assets (other than in respect of the Other Debtor Residual Accepted Assets and activities related thereto) or the Other Debtor Residual Trust Assets (in respect of the Other Debtor Residual Accepted Assets and activities related thereto) for any losses, claims, damages, liabilities and expenses occurring after the earlier of the Avoidance Action Trust Transfer Date and the Other Debtor Residual Assets Transfer Date, including reasonable attorneys' fees, disbursements and related expenses which the Trust Administrator Parties may incur or to which the Trust Administrator Parties may become subject in connection with any action, suit, proceeding or investigation brought by or threatened against one or more of the Trust Administrator Parties on account of the acts or omissions in their capacity as, or on behalf of, the Trust Administrator; provided, however, that the Trust shall not be liable to indemnify any Trust Administrator Party for any act or omission arising out of such Trust Administrator Party's respective actions that are determined by a Final Order of the Bankruptcy Court to be willful misconduct (including, but not limited to, conduct that results in a personal profit at the expense of the Trust), gross negligence, fraud, malpractice, criminal conduct, unauthorized use of confidential information that causes damages, breach of fiduciary duty (to the extent applicable), or ultra vires acts. Notwithstanding any provision herein to the contrary, the Trust Administrator Parties shall be entitled to obtain advances from the Trust to cover their reasonable expenses of defending themselves in any action brought against them as a result of the acts or omissions, actual or alleged, of an Trust Administrator Party in its capacity as such; provided, however, that the Trust Administrator Parties receiving such advances shall repay the amounts so advanced to the Trust immediately upon the entry of a final, non-appealable judgment or order finding that such Trust Administrator Parties were not entitled to any indemnity under the provisions of this Section 9.6. Any amounts payable to any Trust Administrator Party pursuant to this Section 9.6 (other than in respect of the Other Debtor Residual Accepted Assets and activities related thereto) shall be satisfied as follows: (i) first from the Avoidance Action Trust Administrative Cash, (ii) second from the Distributable Trust Cash, if any; provided, however, that the use of Distributable Trust Cash as contemplated in clause (ii) of the foregoing shall be subject to the prior approval by the Bankruptcy Court, as provided in Section 6.1(b); (iii) third from the GUC Trust Supplemental Cash, if any; and (iv) fourth from the Other Supplemental Cash, if any.

Any amounts payable to any Trust Administrator Party pursuant to this <u>Section 9.6</u> in respect of the Other Debtor Residual Accepted Assets and activities related thereto shall be satisfied as follows: (i) first from the Other Debtor Residual Trust Administrative Cash, and (ii) second from the Distributable Other Debtor Residual Trust Cash, if any.

- (b) Anything to the contrary in this Trust Agreement or in any other agreement notwithstanding, to the extent that the Avoidance Action Trust Administrative Cash, Distributable Trust Cash, GUC Trust Supplemental Cash or Other Supplemental Cash, or the Other Debtor Residual Trust Cash, as the case may be, shall be insufficient to fully indemnify the Trust Administrator Parties or to provide advances to the Trust Administrator Parties in accordance with Section 9.6(a), the Trust Administrator Parties shall be indemnified and shall be entitled to obtain advances, first from the Other GUC Trust Administrative Cash (as defined in the GUC Trust Agreement), and second from the GUC Trust Distributable Assets (as defined in the GUC Trust Agreement), to the same extent as the GUC Trust Administrator Parties under Section 9.6 of the GUC Trust Agreement or any successor provision thereunder, as provided in Section 9.6 of the GUC Trust Agreement in effect on the date hereof.
- (c) The foregoing indemnities in respect of any Trust Administrator Party shall survive the termination of such Trust Administrator Party from the capacity for which they are indemnified.

9.7. <u>Compensation and Expenses</u>.

- (a) The Trust Administrator shall receive fair and reasonable compensation for its services (other than in respect of the Other Debtor Residual Accepted Assets and activities related thereto), to be paid out of the Avoidance Action Trust Administrative Cash in accordance with the approved Budget (or from the Avoidance Action Trust SEC Reporting Cash in accordance with Section 2.3(e) hereof, the GUC Trust Supplemental Cash in accordance with Section 2.3(f) hereof, or the Other Supplemental Cash in accordance with Section 6.1(d) hereof). The Trust Administrator shall be entitled, without the need for approval of the Bankruptcy Court, to reimburse itself on a monthly basis (i) from the Avoidance Action Trust Administrative Cash, the GUC Trust Supplemental Cash or the Other Supplemental Cash, as applicable, for such compensation and all reasonable out-of-pocket expenses actually incurred in the performance of duties in accordance with this Trust Agreement and the Budget and (ii) from the Avoidance Action Trust SEC Reporting Cash for such compensation and all reasonable out-of-pocket expenses actually incurred in the performance of duties in accordance with this Trust Agreement and with the approval of the Trust Monitor.
- (b) The Trust Administrator shall receive compensation for its services in respect of the Other Debtor Residual Accepted Assets and activities related thereto, to be paid out of the Other Debtor Residual Trust Administrative Cash, as the Trust Administrator and a majority in interest of the DIP lenders shall agree; provided that the Trust Administrator shall not be obligated to undertake any activities in respect of the Other Debtor Residual Accepted Assets unless such compensation arrangements shall be acceptable to the Trust Administrator in its sole discretion.

9.8. No Personal Financial Liability. No provision of the Plan, Confirmation Order or this Trust Agreement shall be construed as requiring the Trust Administrator to expend or risk its own funds or otherwise to incur any personal financial liability (x) in the performance of any of its duties thereunder or hereunder, including any situation where the Avoidance Action Trust Assets are insufficient to permit the administration of the Trust or distributions as contemplated herein or the payment of fees and expenses of the Trust Professionals, or (y) in the exercise of any of its rights or powers afforded hereunder or thereunder.

ARTICLE X SUCCESSOR TRUST ADMINISTRATORS

- 10.1. <u>Resignation</u>. The Trust Administrator may resign from the Trust by giving at least sixty (60) days' prior written notice thereof to the Trust Monitor. Such resignation shall become effective on the later to occur of (x) the date specified in such written notice and (y) the effective date of the appointment of a successor Trust Administrator in accordance with <u>Section 10.4</u> hereof and such successor's acceptance of such appointment in accordance with <u>Section 10.5</u> hereof.
- 10.2. <u>Removal</u>. The holders of a majority of the Units or the DIP Lenders may at any time petition the Bankruptcy Court for the removal of the Trust Administrator, but only for good cause shown. Such removal shall become effective on the date ordered by the Bankruptcy Court, provided that such removal shall not become effective until the appointment of a successor Trust Administrator in accordance with <u>Section 10.4</u> hereof and such successor's acceptance of such appointment in accordance with <u>Section 10.5</u> hereof. The services of the Trust Administrator shall also terminate upon its bankruptcy, provided that such termination shall not become effective until the appointment of a successor Trust Administrator in accordance with <u>Section 10.4</u> hereof and such successor's acceptance of such appointment in accordance with <u>Section 10.5</u> hereof.
- 10.3. Effect of Resignation or Removal. The resignation, removal or bankruptcy of the Trust Administrator shall not operate to terminate the Trust or to revoke any existing agency created pursuant to the terms of the Plan, the Confirmation Order or this Trust Agreement or invalidate any action theretofore taken by the Trust Administrator. The exculpation, indemnity and limitation of liability provisions of Article X of this Trust Agreement shall survive the resignation, removal or bankruptcy of the Trust Administrator. All fees and expenses properly incurred by the Trust Administrator prior to the resignation, Incompetency, removal or bankruptcy of the Trust Administrator shall be paid from the Avoidance Action Trust Administrative Cash, the GUC Trust Supplemental Cash or the Other Supplemental Cash (other than in respect of the Other Debtor Residual Accepted Assets and activities related thereto), from the Other Debtor Residual Trust Administrative Cash (in respect of the Other Debtor Residual Accepted Assets and activities related thereto) or from Avoidance Action Trust SEC Reporting Cash (in respect of Avoidance Action Trust SEC Reporting Costs in accordance with Section 2.3(e) hereof), unless such fees and expenses are disputed by (x) the Trust Monitor or (y) the successor Trust Administrator, in which case the Bankruptcy Court shall resolve the dispute and any disputed fees and expenses of the predecessor Trust Administrator that are subsequently allowed by the Bankruptcy Court shall be paid from the Avoidance Action Trust Administrative Cash, the GUC Trust Supplemental Cash or the Other Supplemental Cash (other than in respect of the Other Debtor Residual Accepted Assets and activities related thereto), from the Other Debtor Residual Trust Administrative Cash (in respect of the Other Debtor Residual Accepted Assets and activities

related thereto) or from Avoidance Action Trust SEC Reporting Cash (in respect of Avoidance Action Trust SEC Reporting Costs in accordance with <u>Section 2.3(e)</u> hereof). In the event of the resignation, removal or bankruptcy of the Trust Administrator, such Trust Administrator shall:

- (a) promptly execute and deliver such documents, instruments and other writings as may be reasonably requested by the successor Trust Administrator or directed by the Bankruptcy Court to effect the termination of such Trust Administrator's capacity under this Trust Agreement;
- (b) promptly deliver to the successor Trust Administrator all documents, instruments, records and other writings related to the Trust as may be in the possession of such Trust Administrator; and
- (c) otherwise assist and cooperate in effecting the assumption of its obligations and functions by such successor Trust Administrator.
- 10.4. Appointment of Successor. In the event of the resignation, removal, Incompetency or bankruptcy of the Trust Administrator, the Trust Monitor shall promptly appoint a successor Trust Administrator, *provided that* such appointment shall not take effect unless approved by the Bankruptcy Court upon the petition of the Trust Monitor and until the successor Trust Administrator shall have delivered written acceptance of its appointment as described Section 10.5 below. If a successor Trust Administrator does not take office within thirty (30) days after the resignation, removal, Incompetency or bankruptcy of the retiring Trust Administrator, the Bankruptcy Court, upon its own motion or the motion of the retiring Trust Administrator or any Trust Beneficiary, shall appoint a successor Trust Administrator.
- 10.5. Acceptance of Appointment by Successor Trust Administrator. Any successor Trust Administrator appointed hereunder shall execute an instrument accepting its appointment and shall deliver one counterpart thereof to the Bankruptcy Court for filing and to the Trust Monitor and, in case of the Trust Administrator's resignation, to the resigning Trust Administrator. Thereupon, such successor Trust Administrator shall, without any further act, become vested with all the duties, powers, rights, obligations, title, discretion and privileges of its predecessor in the Trust with like effect as if originally named Trust Administrator and shall be deemed appointed pursuant to Bankruptcy Code Section 1123(b)(3)(B); provided, however, such successor Trust Administrator shall file an amendment to the Certificate of Trust with the Secretary of State as required by the Delaware Act. The predecessor Trust Administrator shall duly assign, transfer and deliver to such successor Trust Administrator all Avoidance Action Trust Assets held by such predecessor Trust Administrator hereunder and shall, as directed by the Bankruptcy Court or reasonably requested by such successor Trust Administrator, execute and deliver an instrument or instruments conveying and transferring to such successor Trust Administrator upon the trusts herein expressed, all the duties, powers, rights, obligations, title, discretion and privileges of the predecessor Trust Administrator.
- 10.6. <u>Successor Entity to Trust Administrator</u>. Any business entity into which the Trust Administrator may be merged or converted or with which it may be consolidated, or any entity resulting from any merger, conversion or consolidation to which the Trust Administrator shall be a party, or any entity succeeding to all or substantially all of the corporate trust business of the

Trust Administrator, shall be the successor of the Trust Administrator hereunder, without the execution or filing of any paper or any further act on the part of any of the parties hereto; <u>provided</u>, <u>however</u>, such successor Trust Administrator shall file an amendment to the Certificate of Trust with the Secretary of State as required by the Delaware Act.

ARTICLE XI TRUST MONITOR

11.1. General.

- (a) The Trust Monitor shall oversee the activities of the Trust Administrator as set forth in this Trust Agreement. In all circumstances, the Trust Monitor shall act in the best interests of all Trust Beneficiaries, in furtherance of the purpose of the Trust, and in accordance with this Trust Agreement.
- (b) In furtherance of its rights and responsibilities under this Trust Agreement, the Trust Monitor shall have access, on reasonable advance notice and during regular business hours, to all such books and records of the Trust and the Trust Administrator, shall have the right to consult with all such professionals engaged by the Trust Administrator and shall participate in all such meetings of the Trust Administrator and the Trust Professionals as the Trust Monitor deems reasonably necessary or appropriate. Any documents shared between the Trust Administrator and the Trust Monitor shall be subject to joint privilege, and such sharing shall not be deemed to waive any attorney-client or work product privilege in respect of such documents.

(c) [Intentionally omitted.]

(d) Notwithstanding anything in this <u>Section 11.1</u> or <u>Section 11.2</u> hereof, the Trust Monitor shall not take (or fail to take) any action which will cause the Trust (other than the Avoidance Action Trust Claims Reserve) to fail to qualify as a liquidating trust within the meaning of Treasury Regulation section 301.7701-4(d) that is treated as a grantor trust.

11.2. Appointment and Removal of the Trust Monitor.

- (a) Subject to Section 11.2(d), the Trust Monitor shall serve until the earlier of (w) the final distribution of all Distributable Trust Assets and the Distributable Other Debtor Residual Trust Assets, if any, (x) its resignation pursuant to subsection (b) of this Section 11.2, (y) its removal pursuant to subsection (c) of this Section 11.2 or (z) its bankruptcy or insolvency.
- (b) The Trust Monitor may resign at any time by written notice of resignation to the Trust Administrator, a copy of which shall also be filed by the Trust Monitor with the Bankruptcy Court. Such resignation shall be effective no earlier than sixty (60) days from the date of such notice or such earlier time as a successor is appointed in accordance with the provisions of subsection (d) of this <u>Section 11.2</u>.
- (c) The holders of a majority of the Units or the DIP Lenders may at any time petition the Bankruptcy Court for the removal of the Trust Monitor, but only for good cause shown. Such removal shall become effective on the date ordered by the Bankruptcy Court.

- (d) In the event of the resignation, removal, bankruptcy or insolvency of the Trust Monitor, the Trust Administrator shall promptly appoint a successor Trust Monitor, provided that such appointment shall not take effect unless approved by the Bankruptcy Court upon the petition of the Trust Administrator and until the successor Trust Monitor shall have delivered written acceptance of its appointment as described in clause (e) of this Section 11.2 below; and provided further that until a new Trust Monitor's appointment is effective, the resigning Trust Monitor's appointment shall remain in effect, and the resigning Trust Monitor shall fulfill all obligations and duties of the Trust Monitor. If a successor Trust Monitor does not take office within thirty (30) days after the resignation, removal, incompetency, bankruptcy or insolvency of the retiring Trust Monitor, the Bankruptcy Court, upon its own motion or the motion of the retiring Trust Monitor or any Trust Beneficiary, shall appoint a successor Trust Monitor.
- All fees and expenses properly incurred by the Trust Monitor prior to the resignation, Incompetency, removal or bankruptcy of the Trust Monitor shall be paid from the Avoidance Action Trust Administrative Cash, the GUC Trust Supplemental Cash or the Other Supplemental Cash (other than in respect of the Other Debtor Residual Accepted Assets and activities related thereto), from the Other Debtor Residual Trust Administrative Cash (in respect of the Other Debtor Residual Accepted Assets and activities related thereto) or from Avoidance Action Trust SEC Reporting Cash (in respect of Avoidance Action Trust SEC Reporting Costs in accordance with Section 2.3(e) hereof), unless such fees and expenses are disputed by (x) the Trust Administrator or (y) the successor Trust Monitor, in which case the Bankruptcy Court shall resolve the dispute and any disputed fees and expenses of the predecessor Trust Monitor that are subsequently allowed by the Bankruptcy Court shall be paid from the Avoidance Action Trust Administrative Cash, the GUC Trust Supplemental Cash or the Other Supplemental Cash (other than in respect of the Other Debtor Residual Accepted Assets and activities related thereto) from the Other Debtor Residual Trust Administrative Cash (in respect of the Other Debtor Residual Accepted Assets and activities related thereto) or from Avoidance Action Trust SEC Reporting Cash (in respect of Avoidance Action Trust SEC Reporting Costs in accordance with Section 2.3(e) hereof).
- (f) Any successor Trust Monitor appointed hereunder shall execute an instrument accepting its appointment and shall deliver one counterpart thereof to the Bankruptcy Court for filing and to the Trust Administrator.
- (g) Immediately upon effectiveness of the appointment of a successor Trust Monitor, all rights, powers, duties, authority, and privileges of the predecessor Trust Monitor hereunder will be vested in and undertaken by the successor Trust Monitor without any further act. The successor Trust Monitor shall not be liable personally for any act or omission of the predecessor Trust Monitor.
 - 11.3. Approval of and Consultation with the Trust Monitor.
 - (I) Other Than in Respect of the Other Debtor Residual Accepted Assets
- (a) Notwithstanding anything in this Trust Agreement to the contrary, the Trust Administrator shall submit to the Trust Monitor for its review and prior approval the following

matters, in addition to any other matters that expressly require the approval of the Trust Monitor pursuant to the terms of the Plan, the Confirmation Order or this Trust Agreement:

- (i) Any decision to settle or otherwise resolve the Term Loan Avoidance Action;
- (ii) Any decision to refrain from making any distributions to the holders of Allowed General Unsecured Claims or Units, as the case may be, in accordance with this Trust Agreement, except as expressly permitted herein;
- (iii) Any decision to retain and/or to terminate the retention of Trust Professionals (other than legal counsel retained to represent the Trust Administrator in connection with its role as Trust Administrator, which shall be in the Trust Administrator's sole discretion);
- (iv) The incurrence of any cost or expense of the Trust in excess of 10% of any individual line item therefor in the approved Budget, measured on a quarterly basis; provided, however, that approval of the Trust Monitor shall not be required in the case of any cost or expense authorized by further order of the Bankruptcy Court;
 - (v) The Budget described in <u>Section 6.3</u> hereof and any changes thereto;
- (vi) Any amendment of this Trust Agreement as provided in $\underline{\text{Section}}$ $\underline{13.13}$ hereof; and
- (vii) Any distribution that is not made in accordance with the provisions of <u>Article V</u> as contemplated by <u>Section 5.7</u>; <u>provided</u>, <u>however</u>, that any deviation from the provisions of <u>Article V</u> other than as contemplated by <u>Section 5.7</u> shall also require approval of the Bankruptcy Court.
- (b) In addition to any other matters that expressly require consultation with the Trust Monitor pursuant to the terms of the Plan, the Confirmation Order or this Trust Agreement, the Trust Administrator shall consult with the Trust Monitor in advance of an application to the Bankruptcy Court to use, or to sell or borrow against, the Term Loan Avoidance Action or the Distributable Trust Assets in order to satisfy expenses of the Trust, as contemplated by Section 6.1(b) and Section 6.1(d) hereof.
- (c) In the event of any disagreement between the Trust Administrator and the Trust Monitor regarding any matter requiring the approval or direction of the Trust Monitor under this Trust Agreement, the Trust Administrator and the Trust Monitor shall consult and negotiate diligently and in good faith to resolve such disagreement. If despite their good faith efforts, the Trust Administrator and the Trust Monitor are unable to resolve any disagreement, or the Trust Administrator cannot otherwise obtain approval or direction from the Trust Monitor as required by this Trust Agreement, the Trust Administrator may petition the Bankruptcy Court, with a copy to the Trust Monitor, requesting such approval or direction.
 - (II) In Respect of the Other Debtor Residual Accepted Assets

The Trust Monitor shall have the rights, powers and privileges to act in respect of the Other Debtor Residual Accepted Assets, if any, in the manner set forth in Section 11.3 (I), mutatis mutandis and to the extent applicable. For the avoidance of doubt, the Trust Monitor shall not be obligated to undertake any activities in respect of the Other Debtor Residual Trust Assets unless there shall be available sufficient Other Debtor Residual Trust Administrative Cash to pay in full its fees, costs and expenses in respect thereof.

11.4. Exculpation and Indemnification; Limitation of Liability. To the fullest extent permitted by applicable law, the Trust Monitor Parties shall not be subject to personal liability, and shall be exculpated and indemnified, and shall have the right to obtain advances to cover reasonable expenses of defense, to the same extent as the Trust Administrator Parties pursuant to Section 9.2, Section 9.4, Section 9.5, Section 9.6 and Section 10.3. In no event will the Trust Monitor Parties be liable for punitive, exemplary, consequential, special or other damages for a breach of, or otherwise in connection with, this Trust Agreement under any circumstances.

11.5. <u>Compensation and Expenses</u>.

- (a) The Trust Monitor shall receive fair and reasonable compensation for its services (other than in respect of the Other Debtor Residual Accepted Assets and activities related thereto), to be paid out of the Avoidance Action Trust Administrative Cash, in accordance with the approved Budget (or from the Avoidance Action Trust SEC Reporting Cash in accordance with Section 2.3(e) hereof, the GUC Trust Supplemental Cash in accordance with Section 2.3(f) hereof, or the Other Supplemental Cash in accordance with Section 6.1(d) hereof). The Trust Monitor shall be entitled on a monthly basis, without the need for approval of the Bankruptcy Court, to direct the Trust Administrator to reimburse the Trust Monitor (i) from the Avoidance Action Trust Administrative Cash, the GUC Trust Supplemental Cash or the Other Supplemental Cash, as applicable, for all reasonable out-of-pocket expenses actually incurred in the performance of duties in accordance with this Trust Agreement, consistent with the Budget prepared pursuant to Section 6.3 hereof and (ii) from the Avoidance Action Trust SEC Reporting Cash for such compensation and all reasonable out-of-pocket expenses actually incurred in the performance of duties in accordance with this Trust Agreement.
- (b) The Trust Monitor shall receive compensation for its services than in respect of the Other Debtor Residual Accepted Assets and activities related thereto, to be paid out of the Other Debtor Residual Trust Administrative Cash, as the Trust Monitor and a majority in interest of the DIP lenders shall agree; provided that the Trust Monitor shall not be obligated to undertake any activities in respect of the Other Debtor Residual Accepted Assets unless such compensation arrangements shall be acceptable to the Trust Monitor in its sole discretion.

ARTICLE XII ACTION BY MAJORITY OF HOLDERS OF UNITS

Holders of a majority of the Units or the DIP Lenders from time to time outstanding may petition the Bankruptcy Court to remove the Trust Administrator in accordance with Section 10.2 or to remove the Trust Monitor in accordance with Section 11.1, but in each case only for good cause shown. In determining whether the holders of a majority of the Units have concurred in

any such petition, Units held by the Trust Administrator or the Trust Monitor or any of their respective Affiliates shall be disregarded.

ARTICLE XIII MISCELLANEOUS PROVISIONS

- 13.1. Actions Taken on Other Than Business Day. In the event that any payment or act under the Plan, the Confirmation Order or this Trust Agreement is required to be made or performed on a date that is not a Business Day, then the making of such payment or the performance of such act may be completed on the next succeeding Business Day, but shall be deemed to have been completed as of the required date.
- 13.2. <u>Governing Law</u>. This Trust Agreement shall be governed by and construed in accordance with the laws of the State of Delaware without giving effect to rules governing conflicts of law.
- 13.3. <u>Jurisdiction</u>. Subject to the proviso below, the parties agree that the Bankruptcy Court shall have exclusive and continuing jurisdiction over the Trust and the Trust Administrator, including the administration and activities of the Trust and the Trust Administrator; <u>provided</u>, <u>however</u>, that notwithstanding the foregoing, the Trust Administrator shall have power and authority to bring any action in any court of competent jurisdiction to prosecute any claims or Causes of Action assigned to the Trust, including the Delaware Chancery Court, the Delaware Superior Court and the Delaware Supreme Court.
- 13.4. <u>Third Party Beneficiary</u>. Trust Beneficiaries (including the DIP Lenders in their capacities as such) are third party beneficiaries of this Trust Agreement. The Trust Administrator Parties (other than the Trust Administrator) are third party beneficiaries of the provisions of <u>Section 9.2</u>, <u>Section 9.4</u> and <u>Section 9.6</u> of this Trust Agreement. The Trust Monitor Parties (other than the Trust Monitor) are third party beneficiaries of the provisions of <u>Section 11.4</u> of this Trust Agreement, and, to the extent incorporated therein, <u>Section 9.2</u>, <u>Section 9.4</u>, <u>Section 9.5</u> and <u>Section 9.6</u> of this Trust Agreement. Except as aforesaid, there are no other third party beneficiaries of this Trust Agreement.
- 13.5. Severability. In the event any provision of this Trust Agreement or the application thereof to any person or circumstances shall be determined by a final, non-appealable judgment or order to be invalid or unenforceable to any extent, the remainder of this Trust Agreement or the application of such provision to persons or circumstances or in jurisdictions other than those as to or in which it is held invalid or unenforceable, shall not be affected thereby, and each provision of this Trust Agreement shall be valid and enforceable to the fullest extent permitted by law.
- 13.6. <u>Notices</u>. Any notice or other communication required or permitted to be made under this Trust Agreement shall be in writing and shall be deemed to have been sufficiently given, for all purposes, if delivered personally, by email, facsimile, sent by nationally recognized overnight delivery service or mailed by first-class mail:

(A) if to the Trust Administrator, to:

Wilmington Trust Company Rodney Square North 1100 North Market Street Wilmington, Delaware, 19890-1615

Phone: (302) 636-6000 Fax: (302) 636-4140

Attn: Corporate Trust Administration

With a copy to:

Binder & Schwartz LLP 366 Madison Avenue, 6th Floor New York, NY 10017 Phone: (212) 510-7008 Attn: Eric B. Fisher

(B) if to the Trust Monitor, to:

Arthur J. Gonzalez New York University School of Law 40 Washington Square South New York, NY 10012

- (C) if to any Trust Beneficiary, to:
 - (1) in the case of a DIP Lender,
 - a. if to the U.S. Treasury, to:

United States Department of the Treasury 1500 Pennsylvania Avenue, NW Washington, D.C. 20220 Attn: Chief Counsel, Office of Financial Stability Telecopier: (202) 927-9225

with a copy to:

OFSChiefCounselNotices@treasury.gov

b. if to Export Development Canada, to:

Export Development Canada 151 Slater Street Ottawa, Ontario Canada K1A 1K3 Attention: Loans Services Telecopy: 613-598-2514;

with a copy to:

Export Development Canada 151 Slater Street Ottawa, Ontario Canada K1A 1K3 Attention: Asset Management/Covenants Officer Telecopy: 613-598-3186

(2) if to the **CE** Capital Providers, to:

Cynthiana LLC 2711 Centerville Road Wilmington, DE 19808 Fax Number: (302) 636-5454 Email Address: info@litfinsolutions.com Attn: Cynthiana LLC Manager

Earlham LLC 2711 Centerville Road Wilmington, DE 19808 Fax Number: (302) 636-5454

Email Address: info@litfinsolutions.com

Attn: Earlham LLC Manager

if to the LW Capital Provider, to: (3)

LW Holdco VI LLC 1350 Avenue of the Americas, 2nd Floor New York, New York 10019 Fax Number: (646) 389-1032 Email Address: weinstein@lakewhillans.com

Attn: Boaz Weinstein

(3)(4) in the case of a holder of an Allowed General Unsecured Claim, to the last known address of such holder according to the Debtors' Schedules and/or such holder's proof of claim; and

- (4)(5) in the case of holder of Units, to such address as appears on the books and records of the Trust Administrator, or such other address as may be designated from time to time by notice given in accordance with the provisions of this Section 13.6.
- 13.7. <u>Headings</u>. The headings contained in this Trust Agreement are solely for convenience of reference and shall not affect the meaning or interpretation of this Trust Agreement or of any term or provision hereof.
- 13.8. <u>Plan</u>. The terms of this Trust Agreement are intended to supplement the terms provided by the Plan and the Confirmation Order. To the extent that the terms of sections 5.6 and 6.5 of the Plan are inconsistent with the terms set forth in this Trust Agreement with respect to the Trust, then the terms of the Trust Agreement shall govern. All other provisions of the Plan shall supersede the provisions of this Trust Agreement, including section 6.15 of the Plan, which provides that the restrictions set forth in paragraph 20 of the Final Order approving the DIP Credit Agreement (ECF No. 2529) shall continue to apply.

13.9. <u>Ambiguities and Construction</u>.

- (a) The Trust created by this Trust Agreement (other than the Avoidance Action Trust Claims Reserve) is intended to qualify as a liquidating trust under Treasury Regulation section 301.7701-4(d) for U.S. federal and applicable state and local income tax purposes and, to the extent provided by law, shall be governed and construed in all respects as such a trust and any ambiguity herein shall be construed consistent herewith and, if necessary, this Trust Agreement may be amended to comply with such U.S. federal and applicable state and local income tax laws, which amendments may apply retroactively.
 - (b) Unless the context otherwise requires:
 - (i) a term has the meaning assigned to it;
 - (ii) "or" is not exclusive;
- (iii) words in the singular include the plural, and in the plural include the singular;
- (iv) all references herein to Articles, Sections and other subsections, unless referring specifically to the Plan or provisions of the Bankruptcy Code, the Bankruptcy Rules, or other law, statute or regulation, refer to the corresponding Articles, Sections and other subsections of this Trust Agreement;
- (v) the words "hereof," "herein," "hereunder" and similar words refer to this Trust Agreement as a whole and not to any particular provision, Article, Section or subsection of this Trust Agreement unless otherwise specified;
- (vi) words importing persons shall include firms, associations, corporations and other entities;

- (vii) any pronoun shall include the corresponding masculine, feminine and neuter forms; and
 - (viii) "including" means including without limitation.
- 13.10. <u>Entire Trust Agreement</u>. This Trust Agreement contains the entire agreement between the parties and supersedes all prior and contemporaneous agreements or understandings between the parties with respect to the subject matter hereof.
- 13.11. <u>Cooperation</u>. The Debtors shall turn over or otherwise make available to the Trust Administrator at no cost to the Trust or the Trust Administrator, all books and records reasonably required by the Trust Administrator to carry out its duties hereunder, and agree to otherwise reasonably cooperate with the Trust Administrator in carrying out its duties hereunder, subject to the obligation to preserve the confidential nature of the Debtors' books and records, as provided in Section 13.12.
- 13.12. <u>Confidentiality</u>. The Trust Administrator and the Trust Monitor, and their respective employees, members, agents, professionals and advisors, including the Trust Professionals (each a "<u>Confidential Party</u>" and collectively the "<u>Confidential Parties</u>") shall hold strictly confidential and not use for personal gain any material, non-public information of which they have become aware in their capacity as a Confidential Party, of or pertaining to any Debtor to which any of the Avoidance Action Trust Assets relates or which is otherwise received from the Debtors by the Trust; <u>provided</u>, <u>however</u>, that such information may be disclosed if
- (i) it is now or in the future becomes generally available to the public other than as a result of a disclosure by the Confidential Parties; or
- (ii) such disclosure is required of the Confidential Parties pursuant to legal process, including subpoena or other court order or other applicable laws or regulations.

In the event that any Confidential Party is requested to divulge confidential information pursuant to clause (ii), such Confidential Party shall promptly, in advance of making such disclosure, provide reasonable notice of such required disclosure to the Trust Administrator (or the Trust Monitor in case the Trust Administrator is the disclosing party) to allow sufficient time to object to or prevent such disclosure through judicial or other means and shall cooperate reasonably with the Trust Administrator (or the Trust Monitor, as applicable) in making any such objection, including appearing in any judicial or administrative proceeding in support of any objection to such disclosure.

13.13. Amendment and Waiver.

(a) The Trust Administrator, with the approval of the Trust Monitor, may amend or supplement this Trust Agreement without notice to or consent of the Bankruptcy Court or any Trust Beneficiary for the purpose of (x) curing any ambiguity, omission, inconsistency or correcting or supplementing any defective provision; (y) evidencing and providing for the acceptance of the appointment of a successor Trust Administrator or Trust Monitor; or (z) making any other changes to this Trust Agreement that does not adversely affect the interests of the Trust Beneficiaries in any material respect.

- (b) The Trust Administrator may amend or supplement this Trust Agreement for any other purpose, but only on petition to, and with the approval of, the Bankruptcy Court; provided that (x) no amendment or supplement to this Trust Agreement shall be inconsistent with the purpose and intent of the Trust to dispose of in an expeditious but orderly manner the Avoidance Action Trust Assets in accordance with the terms of the Plan, the Confirmation Order and this Trust Agreement, and (y) this Trust Agreement shall not be amended in a manner that is inconsistent with the Plan in the form confirmed by the Bankruptcy Court, subject to any post-confirmation modifications to the Plan pursuant to Section 1127 of the Bankruptcy Code.
- (c) Any amendment to this Trust Agreement shall be filed with the Bankruptcy Court.
- (d) No amendment shall be made to any provision of this Trust Agreement that materially and adversely affects the rights of the DIP Lenders or, the <u>CE</u> Capital Providers or the <u>LW Capital Provider</u> without the written consent of the DIP Lenders and, the <u>CE</u> Capital Providers or the <u>LW Capital Provider</u>, as applicable.
- (e) The Trust Administrator shall file any amendment to the Certificate of Trust with the Secretary of State as may be required or permitted by the Delaware Act.
- 13.14. <u>Counterparts</u>. This Trust Agreement may be executed in any number of counterparts, each of which shall be deemed an original, but all such counterparts shall together constitute but one and the same instrument. A facsimile or portable document file (PDF) signature of any party shall be considered to have the same binding legal effect as an original signature.

09-50026-mg Doc 14413-5 Filed 02/04/19 Entered 02/04/19 15:31:37 Exhibit E Pg 62 of 64

Execution Copy

IN WITNESS WHEREOF, the parties hereto have executed this Trust Agreement or caused this Trust Agreement to be duly executed by their respective officers, representatives or agents, effective as of the date first above written.

By:	N
	Name:
	Title:
	nd — FHUR J. GONZALEZ,
	rust Monitor
as T	TUST IVIOIIITOI
	rust Monitor
as T By:	Name:

WILMINGTON TRUST COMPANY,

Execution Copy

Exhibit A

Cash Accounts:

AAT Approved Budget Cash Account
AAT Distributable Trust Assets Account
Avoidance Action Budget Sub Account
Avoidance Action Assets Sub Account
Avoidance Action Trust SEC Reporting Costs Account
CE Capital Providers Funding Account
LW Capital Provider Funding Account
Litigation Cost Advance Funding Account
Other Supplemental Cash Account
Segregated Account

Execution Copy

Exhibit B

FORM OF CERTIFICATE OF TRUST OF MOTORS LIQUIDATION COMPANY AVOIDANCE ACTION TRUST

THIS Certificate of Trust of Motors Liquidation Company Avoidance Action Trust (the "Trust") is being duly executed and filed on behalf of the Trust by the undersigned, as trustee, to form a statutory trust under the Delaware Statutory Trust Act (12 <u>Del. C.</u> § 3801 et m.) (the "Act").

- 1. <u>Name</u>. The name of the statutory trust formed by this Certificate of Trust is Motors Liquidation Company Avoidance Action Trust.
- 2. <u>Delaware Trustee.</u> The name and business address of the trustee of the Trust with a principal place of business in the State of Delaware are Wilmington Trust Company, 1100 North Market Street, Wilmington, Delaware 19890, Attn: Corporate Trust Administration.
 - 3. <u>Effective Date.</u> This Certificate of Trust shall be effective upon filing.

IN WITNESS WHEREOF, the undersigned has duly executed this Certificate of Trust in accordance with Section 3811(a)(1) of the Act.

D			
By:	Name: Title:		

WILMINGTON TRUST COMPANY,

as Trust Administrator and trustee

Exhibit F

09-50026-mg Doc 14413-6 Filed 02/04/19 Entered 02/04/19 15:31:37 Exhibit F Pg 2 of 29

Execution Version

SECURITY AGREEMENT

TABLE OF CONTENTS

1.	Defin	Definitions			
	1.1	LW Capital Provision Agreement and UCC Terms	1		
	1.2	General Definitions	1		
	1.3	Interpretation	3		
2.	Gran	t of Security	3		
	2.1	Grant of Security – General	3		
3.	Secui	rity for Obligations; Grantor Remains Liable	3		
	3.1	Security for Obligations	3		
	3.2	Continuing Liability Under Collateral	4		
4.	Certa	in perfection requirements, etc.	4		
5.	Repre	esentations and Warranties	5		
	5.1	Grantor Information and Status, Authority, etc	5		
	5.2	Ownership of Collateral and Absence of Other Liens	5		
	5.3	Status of Security Interest	5		
6.	Cove	nants and Agreements	6		
	6.1	Grantor Information and Status	6		
	6.2	Ownership of Collateral, Absence of Other Adverse Claims, and Further Duties	6		
	6.3	Status of Security Interest	7		
	6.4	Reliance	7		
	6.5	Enforcement Costs	7		
7.	Acce	ss; Right of Inspections and Further Assurances	7		
	7.1	Access; Right of Inspection	7		
	7.2	Further Assurances	8		

8.	remed	remedies			
	8.1	Generally	9		
	8.2	Application of Proceeds	11		
	8.3	Cash Proceeds	11		
9.	Conti	nuing Security Interest	11		
10.	Stand	lard of Care; Secured Parties may Perform	11		
11.	Notic	Notices1			
	11.1	Effectiveness of Notices	12		
	11.2	Change of Details	12		
12.	misce	ellaneous	12		
	12.1	Remedies Cumulative	12		
	12.2	Covenants Independent	12		
	12.3	Successors and Assigns	13		
	12.4	No Transfer	13		
	12.5	Amendments	13		
	12.6	Entire Agreement	13		
	12.7	Counterparts	13		
	12.8	No Waiver	13		
	12.9	Severability	14		
13.	Appli	icable Law and Jurisdiction	14		
14.	Intent	Intentionally Omitted1			
15.	Subor	Subordination Agreement			

SCHEDULE 1 TO

SECURITY AGREEMENT

This Schedule forms a part of the Security Agreement between the Parties.

1.	GRANTOR	
	±	Motors Liquidation Company Avoidance Action Trust
1.2	Jurisdiction of formation:	State of Delaware
1.3	Address:	Wilmington Trust Company Rodney Square North 1100 North Market Street Wilmington, Delaware, 19890-1615
Notices	s to the attention of:	Corporate Trust Administration
Fax:		(302) 636-4140
E-mail	:	bandrews@WilmingtonTrust.com
2.	SECURED PARTY	
2.1	Name of Secured Party:	LW Holdco VI LLC ("Secured Party")
2.2	Jurisdiction of formation:	State of Delaware
2.3	Address:	1350 Avenue of the Americas, 2 nd Floor New York, New York 10019
Notices	s to the attention of:	Boaz Weinstein

(646) 389-1032

Fax:

E-mail:

weinstein@lakewhillans.com

3. CLAIM

Any litigation commenced by the Grantor, including In re: Motors Liquidation Company, et al., Chapter 11 Case No. 09-50026 (REG) (Jointly Administered), including but not limited to (i) Motors Liquidation Co. Avoidance Action Trust, by and through the Wilmington Trust Company, solely in its capacity as Trust Administrator and Trustee v. J.P. Morgan Chase Bank, N.A., individually and as administrative Agent for various lenders party to the Term Loan Agreement, et al., Adversary Proceeding Case No. 0900504 (REG), U.S.B.C. S.D.N.Y.; (ii) the action commenced in the Court of Chancery of the State of Delaware against Oaktree Loan Fund, LP, including any related actions; (iii) the action commenced in the Court of Chancery of the State of Delaware against SSS Funding II, LLC; and (iv) any other actions or proceedings brought by the Grantor.

The Claim also includes any variations or expansions of the above claims by the addition of any claims and/or parties from time to time, as well as the following:

- (i) any and all related pre- and post-trial proceedings or processes (or pre- and posthearing proceedings or processes, where applicable) in or in connection with such claim(s), including the pursuit of costs or post-judgment or post-arbitral award remedies:
- (ii) all proceedings seeking to appeal, challenge, confirm, enforce, modify, correct, vacate or annul a judgment or award, as well as proceedings on remand or retrial or rehearing;
- (iii) all ancillary, parallel or alternative dispute resolution proceedings and processes arising out of or related to the acts or occurrences alleged in such claim(s) (including conciliation or mediation or court filings seeking discovery for or filed in aid of a contemplated or pending arbitration);
- (iv) re-filings or parallel filings of such claim(s) and any other legal, diplomatic or administrative proceedings or processes founded on the underlying facts giving rise to or forming a basis for such claim(s);
- (v) ancillary or enforcement proceedings related to the facts or claims alleged from time to time or that could have been alleged in such claim(s) at any time;
- (vi) all arrangements, settlements, negotiations, or compromises made between the Counterparty and any adverse party having the effect of resolving any of the Counterparty's claims against any adverse party that are or could be or could have been brought in such claim(s); and
- (vii) all rights of the Counterparty to collect any damages or awards or otherwise exercise remedies in connection with any of the foregoing.

4. EFFECTIVE DATE OF THIS AGREEMENT

January 24, 2019

SCHEDULE 2 TO

SECURITY AGREEMENT

This Schedule forms a part of the Security Agreement between the Parties.

Complete for each entity Grantor:

(A) Full Legal Name, Type of Organization, Jurisdiction of Organization, Chief Executive Office/Sole Place of Business and Organizational Identification Number of each entity Grantor:

Full Legal Name	Type of Organization	Jurisdiction of Organization	Chief Executive Office/Sole Place of Business	Organization I.D.#
Motors Liquidation Company Avoidance Action Trust	Statutory trust	State of Delaware	Wilmington Trust Company Rodney Square North 1100 North Market Street Wilmington, Delaware, 19890- 1615	Delaware File No. 4961460

(B) Other Names (including any Trade Name or Fictitious Business Name) under which each entity Grantor currently conducts business:

<u>Trade Name or Fictitious Business Name</u>
None.

(C) Changes in Name, Jurisdiction of Organization, Chief Executive Office or Sole Place of Business and Corporate Structure within past five (5) years:

Date of Change	Description of Change
None.	

This **SECURITY AGREEMENT**, dated as of the date set forth on Schedule 1 hereof (as it may be amended, restated, supplemented or otherwise modified from time to time, this "**Agreement**"), between the grantor(s) whose name, organizational details and address are set out in Schedule 1 hereof (if more than one, collectively, and jointly and severally, "**Grantor**"), and the secured party whose name, organizational details and address are set out in Schedule 1 hereof ("**Secured Party**").

RECITALS:

WHEREAS, reference is made to that certain capital provision agreement, dated as of January 21, 2019 (as it may be amended, restated, supplemented or otherwise modified from time to time, the "LW Capital Provision Agreement"), by and between Grantor as the "Counterparty" thereunder and Secured Party as the Capital Provider thereunder;

WHEREAS, in consideration of the payment of the amounts set forth in the LW Capital Provision Agreement and other accommodations of Secured Party, Grantor has agreed to secure its obligations to Secured Party under the LW Capital Provision Agreement as set forth herein; and

NOW, **THEREFORE**, in consideration of the premises and the agreements, provisions and covenants herein contained, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Grantor and Secured Party agree as follows:

1. **DEFINITIONS**

1.1 LW Capital Provision Agreement and UCC Terms

- (a) Capitalized terms used herein but not defined shall have the meaning given to them in the LW Capital Provision Agreement, unless such terms are defined in the UCC, in which case they shall be governed by clause (b) of this Section 1.1.
- (b) Capitalized terms not defined in this Agreement shall have the meaning given to them in the UCC (and, if defined in more than one Article of the UCC, shall have the meaning given in Article 9 thereof).

1.2 General Definitions

In this Agreement, the following terms shall have the following meanings:

[&]quot;Agreement" shall have the meaning set forth in the preamble.

[&]quot;Avoidance Action Proceeds" shall have the meaning set forth in the Trust Agreement.

[&]quot;Bankruptcy Code" means Title 11 of the United States Code entitled

[&]quot;Bankruptcy", as now and hereafter in effect, or any successor statute.

- "Claim" means the claim(s) described on Schedule 1 to this Agreement. If only one claim is set forth on Schedule 1, it shall be referred to herein as "the Claim" or "a Claim". If two or more claims are set forth on Schedule 1, each claim shall be referred to individually as "a Claim" and all such claims together shall be referred to collectively as "the Claim".
- "Claim Proceeds" shall mean "Proceeds" as such term is defined in the LW Capital Provision Agreement.
- "Collateral" shall have the meaning set forth in Section 2.
- "Enforcement Costs" shall have the meaning set forth in Section 6.5.
- "Insolvency Proceedings" shall mean any assignment for the benefit of creditors or other proceedings intended to liquidate or rehabilitate the estate of a Person or any other legal procedure under any law of any jurisdiction having a similar or analogous nature or effect.
- "Intellectual Property" shall mean the collective reference to all rights, priorities and privileges relating to intellectual property, whether arising under the laws of the United States of America, multinational or foreign laws or otherwise, including without limitation, copyrights, copyright licenses, patents, patent licenses, trademarks, trademark licenses, trade secrets, and trade secret licenses, and the right to sue or otherwise recover for any past, present and future infringement, dilution, misappropriation, or other violation or impairment thereof, including the right to receive all Proceeds, including without limitation license fees, royalties, income, payments, claims, damages and proceeds of suit, now or hereafter due and/or payable with respect thereto.
- "LW Capital Provision Agreement" shall have the meaning set forth in the recitals.
- "Material Breach" shall have the meaning set forth in Section 9.1(a).
- "**Proceeds**", when used herein as an <u>un</u>capitalized term, shall have the meaning given in ordinary English usage. (The capitalized terms "Proceeds" and "Cash Proceeds" shall have the meanings given in Article 9 of the UCC, and the capitalized term "Claim Proceeds" shall have the meaning set forth above in this Section 1).
- "Reimbursable Amounts" shall mean all amounts, other than Enforcement Costs, for which Secured Party is entitled to indemnification, reimbursement, recoupment or recompense hereunder in its capacity as Secured Party, and advances made by Secured Party hereunder for the account of Grantor.
- "Secured Obligations" shall have the meaning set forth in Section 3.1.
- "Secured Party" shall have the meaning set forth in the preamble.
- "Subordination Agreement" shall mean the subordination agreement, dated January 24, 2019, among the DIP Lenders, Secured Party and Grantor.
- "Trust Agreement" shall have the meaning set forth in the LW Capital Provision Agreement.

"UCC" shall mean the Uniform Commercial Code as in effect from time to time in the State of New York; *provided*, *however*, that in the event that, by reason of mandatory provisions of law, the rules governing perfection, the effect of perfection or nonperfection and the priority of a security interest in the Collateral are governed by the Uniform Commercial Code as enacted and in effect in a jurisdiction other than the State of New York, the term "UCC" shall mean the Uniform Commercial Code as enacted and in effect in such other jurisdiction solely for purposes of such mandatory provisions.

1.3 Interpretation

Any of the terms defined herein may refer to the singular or plural, as the context may require. Whenever the context may require, any pronoun shall include the corresponding masculine, feminine and neuter forms. References herein to any Section, Appendix, Schedule or Exhibit shall be to a Section, an Appendix, a Schedule or an Exhibit, as the case may be, of this Agreement unless otherwise specifically provided. The use herein of the words "include" or "including", when following any general statement, term or matter, shall not be construed to limit such statement, term or matter to the specific items or matters set forth immediately following such word or to similar items or matters, whether or not non-limiting language (such as "without limitation" or "but not limited to" or words of similar import) is used with reference thereto, but rather shall be deemed to refer to all other items or matters that fall within the broadest possible scope of such general statement, term or matter. The terms "lease" and "license" shall include sub-lease and sub-license, as applicable. This Agreement relates solely to the security interest described herein, and the rights and obligations of the parties with respect thereto, and shall not be interpreted to alter, amend, or modify any of the terms and provisions of the LW Capital Provision Agreement. All references herein to provisions of the UCC shall include all successor provisions under any subsequent version or amendment to any Article of the UCC.

2. GRANT OF SECURITY

2.1 Grant of Security – General

Grantor hereby grants to Secured Party a security interest in and continuing lien on all of Grantor's right, title and interest in the following, in each case whether now or hereafter existing or in which Grantor now has or hereafter acquires an interest and wherever the same may be located: (i) the Avoidance Action Proceeds and (ii) the Capital Provider Funding Account and all Supplemental Capital therein (hereinafter collectively referred to as the "Collateral").

3. SECURITY FOR OBLIGATIONS; GRANTOR REMAINS LIABLE

3.1 Security for Obligations

This Agreement secures, and the Collateral is collateral security for, the prompt and complete payment or performance in full when due, whether at declaration, demand or otherwise (including the payment of amounts that would become due but for the operation of the automatic stay under Section 362(a) of the Bankruptcy Code, 11 U.S.C. § 362(a) (and any successor provision thereof)), of all obligations of Grantor under the LW Capital Provision Agreement and

this Agreement, including any Enforcement Costs and Reimbursable Amounts (collectively, the "Secured Obligations").

3.2 Continuing Liability Under Collateral

Notwithstanding anything herein to the contrary, (i) Grantor shall remain liable for all obligations to third parties in respect of the Collateral, and nothing contained herein is intended or shall be a delegation of duties to Secured Party, (ii) Grantor shall remain liable under each agreement included in the Collateral to perform all of its obligations thereunder in accordance with, and pursuant to, the terms and provisions thereof, and Secured Party shall not have any obligation or liability under any such agreement by reason of, or arising out of, this Agreement or any other document related thereto prior to its acquisition thereof by enforcement of its rights under this Agreement or transfer in lieu of such enforcement, nor shall Secured Party have any obligation to make any inquiry as to the nature or sufficiency of any payment received by it or have any obligation to take any action to collect or enforce any rights under any agreement included in the Collateral, and (iii) the exercise by Secured Party of any of its rights hereunder shall not release Grantor from any of its duties or obligations under the contracts and agreements included in the Collateral.

4. CERTAIN PERFECTION REQUIREMENTS, ETC.

- (a) Grantor shall (i) accept amounts provided by Secured Party, in its capacity as Capital Provider, to Grantor pursuant to the LW Capital Provision Agreement; (ii) receive any funds paid as the result of a Claim Resolution and to promptly pay to Secured Party, in its capacity as Secured Party, any amounts owed to Secured Party in accordance with the LW Capital Provision Agreement; and (iii) irrevocably instruct the Nominated Lawyers, by means of instructions in the form attached hereto as Exhibit 1, to keep the Secured Party informed about developments in the Claim, notify the Secured Party upon becoming aware of the receipt of any Claim Proceeds and notify the Secured Party of any change in their representation of Grantor.
- (b) The Grantor hereby agrees to establish the Capital Provider Funding Account as defined in the LW Capital Provision Agreement, and to accept therein (and maintain therein in accordance with the LW Capital Provision Agreement) all amounts provided by Secured Party, in its capacity as Capital Provider, for use by the Grantor pursuant to the LW Capital Provision Agreement and the Trust Agreement.
- (c) Subject to applicable law, the Nominated Lawyers hereby waive their right to take any action (or assist others in so doing) to: (i) contest or challenge the attachment, perfection, or priority of any security interest of Secured Party in any Collateral, whether through settlement, arbitration, or court resolution; (ii) contest or challenge any action taken by Secured Party to enforce its rights in any recovery in the Claim or to dispose of the same; or (iii) otherwise hinder any exercise of remedies by Secured Party, which remedies may be undertaken by Secured Party in its sole discretion, as authorized by this Agreement or Article 9 of the UCC.

5. REPRESENTATIONS AND WARRANTIES

Grantor hereby makes all the representations and warranties set forth in the LW Capital Provision Agreement as if such representations and warranties were set forth herein, and Grantor further represents and warrants that:

5.1 Grantor Information and Status, Authority, etc.

(a) Schedule 2 (as such schedule may be amended or supplemented from time to time) is true and accurate in all respects; and except as provided thereon, Grantor has not changed its name, identity, jurisdiction of organization or of principal residence, type of organization, corporate or company structure (e.g., by merger, consolidation, change in organizational form or otherwise), or sole place of business or chief executive office, in each case, within the past five (5) years.

5.2 Ownership of Collateral and Absence of Other Liens

- (a) With respect to the Claim, Grantor is the sole legal and beneficial owner of and has good title to (A) such Claim; (B) if applicable, all Intellectual Property and other rights underlying such Claim; and (C) all Collateral related to such Claim; in each case free and clear of any Adverse Claims, except for any lien of the Nominated Lawyers on Claim Proceeds pursuant to an Engagement Agreement with the Nominated Lawyers or pursuant to the Litigation Cost Advance Agreement, the Settlement Agreement, Section 6.1(d) of the Trust Agreement, or the Security Agreement dated as of June 2, 2017 (the "2017 Security Agreement") made by Grantor in favor of Cynthiana LLC and Earlham LLC, each a Delaware limited liability company (collectively the "CE Capital Providers").
- (b) Other than financing statements filed in favor of Secured Party, the DIP Lenders or the CE Capital Providers, no effective financing statement, notice of tax lien or other instrument similar in effect under any applicable law covering all or any part of the Collateral is on file in any filing office or registry where such documents could be filed, including without limitation the United States Patent and Trademark Office, the United States Copyright Office, the various Secretaries of State, and the foreign counterparts of any of the foregoing.

5.3 Status of Security Interest

(a) Except for any lien of the Nominated Lawyers on Claim Proceeds pursuant to an Engagement Agreement with the Nominated Lawyers or pursuant to the Litigation Cost Advance Agreement, Settlement Agreement, Section 6.1(d) of the Trust Agreement or the 2017 Security Agreement, the security interest of Secured Party in the Collateral shall constitute a valid, perfected lien upon the Collateral, third and junior in priority to the liens of the DIP Lenders and the CE Capital Providers in the Collateral and senior in priority to any lien of any other party upon (i) the filing of a financing statement naming Grantor as "debtor" and Secured Party as "secured party" and describing the Collateral in the office of the Secretary of State of the appropriate jurisdiction and/or any foreign counterpart thereof, at Secured Party's election, and (ii) the entry by the Bankruptcy

- Court of the order described in Section 2.1(d)(iii) of the LW Capital Provision Agreement.
- (b) Except as set forth in the Litigation Cost Advance Agreement, Settlement Agreement and Trust Agreement, no authorization, consent, approval or other action by, and no notice to or filing with, any Governmental Authority or any other Person is required for either (i) the pledge or grant by Grantor of the security interest hereby created in favor of Secured Party hereunder or (ii) the exercise by Secured Party of any rights or remedies in respect of any Collateral (whether specifically granted or created hereunder or created or provided for by applicable law), except (A) for the filings contemplated by clause (a) above and (B) as may be required in connection with the disposition of any securities, by laws generally affecting the offering and sale of securities.

6. COVENANTS AND AGREEMENTS

Grantor hereby covenants and agrees that:

6.1 Grantor Information and Status

- (a) Grantor shall not change its name, identity, jurisdiction of organization or of principal residence, type of organization, corporate or company structure (e.g., by merger, consolidation, change in organizational form or otherwise), or sole place of business or chief executive office, unless it shall have (i) notified Secured Party in writing at least thirty (30) Business Days prior to any such change, identifying such change and providing such other information in connection therewith as Secured Party may reasonably request, and (ii) taken all actions reasonably requested by Secured Party to maintain the continuous validity, perfection and priority of Secured Party's security interest in the Collateral granted or intended to be granted and agreed to hereby which, in the case of any merger or other change in corporate structure shall include, without limitation, executing and delivering to Secured Party a supplement hereto (in form and substance satisfactory to Secured Party) upon completion of such merger or other change in corporate structure confirming the grant of the security interest hereunder.
- (b) Grantor shall maintain its existence at all times in good standing (as a Registered Organization if applicable) in its jurisdiction of organization and shall remain qualified to do business in each other jurisdiction in which the nature of its business so requires.

6.2 Ownership of Collateral, Absence of Other Adverse Claims, and Further Duties

(a) Except for the security interest created by this Agreement, any lien of the Nominated Lawyers on Claim Proceeds pursuant to an Engagement Agreement with the Nominated Lawyers or pursuant to the Litigation Cost Advance Agreement, Settlement Agreement, Section 6.1(d) of the Trust Agreement or the 2017 Security Agreement, (i) Grantor shall not create or suffer to exist any Adverse Claim upon or with respect to any of the Collateral, and (ii) Grantor shall defend the Collateral against all Persons and Governmental Authorities at any time claiming any interest therein.

- (b) Upon Grantor's obtaining knowledge thereof, it shall promptly notify Secured Party in writing of any event that could reasonably be expected to have a material adverse effect on the value of the Collateral or any portion thereof, the ability of Grantor or Secured Party to dispose of the Collateral or any portion thereof, or the rights and remedies of Secured Party in relation thereto, including, without limitation, the levy of any legal process against the Collateral or any portion thereof or, if applicable, a material change in the status or enforceability of any Intellectual Property relevant to the Claim.
- (c) Grantor shall not sell, transfer or assign (by operation of law or otherwise) to any Person any portion of the Collateral unless and to the extent permitted by the LW Capital Provision Agreement.

6.3 Status of Security Interest

Grantor shall maintain the security interest of Secured Party hereunder in all Collateral as a valid, perfected lien, third in priority to the liens of the DIP Lenders and the CE Capital Providers in the Collateral and superior in priority to any security interest of any other party.

6.4 Reliance

Grantor agrees and acknowledges that the duties contained in Sections 6.1 through 6.3 above are a material inducement for Secured Party to enter into this Agreement, and that Secured Party may, if such covenants are breached, immediately exercise all of the remedies provided herein or under law. Grantor further agrees that a breach of any of the covenants contained in Sections 6.1 through 6.3 will cause irreparable injury to Secured Party, that Secured Party has no adequate remedy at law in respect of such breach and, as a consequence, that each and every covenant contained in Sections 6.1 through 6.3 shall be specifically enforceable against Grantor, and Grantor hereby waives and agrees not to assert any defenses against an action for specific performance of such covenants except for a defense that no default has occurred by virtue of a violation of Section 6.1, 6.2 or 6.3 above. Nothing in this Section shall in any way limit the rights of Secured Party hereunder.

6.5 Enforcement Costs

Grantor agrees to pay or reimburse Secured Party for all reasonable costs and expenses incurred by Secured Party in a substantially successful enforcement of any of their rights and remedies hereunder or under the LW Capital Provision Agreement, including but not limited to attorneys' fees and all other costs and expenses of any sale of, collection from or other realization upon all or any part of the Collateral (all costs and expenses described in this Section 6.5, collectively, "Enforcement Costs").

7. ACCESS; RIGHT OF INSPECTIONS AND FURTHER ASSURANCES

7.1 Access; Right of Inspection

Secured Party shall at all times have full and free access during normal business hours upon three (3) Business Days' prior written notice and during business hours to those books, correspondence and records of Grantor that relate to the Collateral, and Secured Party and its

representatives may examine the same, take extracts there from and make photocopies thereof, and Grantor agrees to render to Secured Party, at Secured Party's cost and expense such clerical and other assistance as may be reasonably requested with regard thereto. Notwithstanding the foregoing, Grantor is not obliged to provide to Secured Party any information that is subject to attorney-client privilege; <u>provided</u> that if Grantor withholds from Secured Party information otherwise required by this Agreement on the basis of attorney-client privilege or any similar doctrine of confidentiality, it shall provide prompt notice thereof to Grantor.

7.2 Further Assurances

- (a) Grantor agrees that from time to time, at the expense of Grantor, it shall promptly execute and deliver all further instruments and documents and take all further action, that may be necessary or desirable, or that Secured Party may reasonably request, in order to create and/or maintain the validity, perfection or priority of, and protect, any security interest granted or purported to be granted hereby or to enable Secured Party to exercise and enforce its rights and remedies hereunder with respect to any Collateral. Without limiting the generality of the foregoing, Grantor shall:
 - (i) file, or authorize the filing of, such financing or continuation statements, or amendments thereto, and execute and deliver such other agreements, stock certificates and other tangible financial assets, instruments, mortgages, endorsements, stock powers, powers of attorney or notices, or take such other actions as may be necessary or desirable, or as Secured Party may reasonably request, in order to effect, reflect, perfect and preserve the security interests granted or purported to be granted hereby;
 - (ii) at Secured Party's request, appear in and defend any action or proceeding that may affect Grantor's title to, or Secured Party's security interest in, all or any part of the Collateral;
 - (iii) furnish Secured Party with such information regarding the Collateral, including, without limitation, the location thereof, as Secured Party may reasonably request from time to time; and
 - (iv) upon request by Secured Party, join with Secured Party in notifying any Opponent or issuer of a Supporting Obligation of Secured Party's security interest hereunder.
- (b) Grantor hereby authorizes Secured Party to file a Record or Records, including financing or continuation statements, other security agreements and amendments, and supplements to any of the foregoing, in any jurisdictions and with any filing offices as Secured Party may determine, in its sole discretion, are necessary or advisable to perfect or otherwise protect the security interest granted to Secured Party herein; and Grantor hereby ratifies such actions if taken by Secured Party prior to execution and delivery of this Agreement. Such financing statements may describe the Collateral in the same manner as described herein or may contain an indication or description of the Collateral that describes such property in any other manner as Secured Party may determine, in its sole discretion, is

necessary, advisable or prudent to ensure the perfection of the security interest in the Collateral granted to Secured Party herein. Grantor shall furnish to Secured Party from time to time statements and schedules further identifying and describing the Collateral and such other reports in connection with the Collateral as Secured Party may reasonably request, all in reasonable detail.

8. REMEDIES

8.1 Generally

Subject to the rights of the DIP Lenders to exercise their remedies pursuant to the Litigation Cost Advance Agreement and the rights of the CE Capital Providers to exercise their remedies pursuant to the 2017 Security Agreement:

- (a) If (i) Grantor has breached Section 4 hereof, (ii) Grantor has materially breached any provision hereof other than Section 4, or any provision of the LW Capital Provision Agreement, and any such breach remains in existence for five (5) days following notice thereof (a breach as described in the foregoing clause (i) or an uncured breach as described in this clause (ii), a "Material Breach"), (iii) a Remedy Event under the LW Capital Provision Agreement occurs or (iv) an Insolvency Proceeding has been commenced by or against Grantor and has not been dismissed within thirty (30) days, then Secured Party may exercise in respect of the Collateral, in addition to all other rights and remedies provided for herein or otherwise available to it at law or in equity, all the rights and remedies of a secured party on default under the UCC (whether or not the UCC applies to the affected Collateral) to collect, enforce or satisfy any Secured Obligations then owing, whether by acceleration or otherwise, and also may pursue any of the following separately, successively or simultaneously:
 - (A) notify the Account Debtor or other Person obligated on the Collateral to make payment or otherwise render performance to or for the benefit of Secured Party;
 - (B) prior to the disposition of the Collateral, prepare the Collateral for disposition in any manner to the extent Secured Party deems appropriate; and
 - (C) without notice except as specified below or under the UCC, sell, assign, transfer or otherwise dispose of the Collateral or any part thereof in one or more parcels at public or private sale, at any office of Secured Party or elsewhere, for cash, on credit or for future delivery, at such time or times and at such price or prices and upon such other terms as Secured Party may deem commercially reasonable.
- (b) Secured Party may be the purchaser of any or all of the Collateral at any public or private (to the extent the portion of the Collateral being privately sold is of a kind that is customarily sold on a recognized market or the subject of widely distributed standard price quotations) sale in accordance with the UCC, and Secured Party shall be entitled, for the purpose of bidding and making settlement or payment of the purchase price for all

or any portion of the Collateral sold at any such sale made in accordance with the UCC, to use and apply any of the Secured Obligations as a credit on account of the purchase price for any Collateral payable by Secured Party at such sale; *provided, however*, that any bid by Secured Party may not include a "credit bid" in respect of any Secured Obligations unless the cash proceeds of such bid are sufficient to satisfy in full the Senior Obligations and the Second Ranking Obligations (as each such term is defined in the Subordination Agreement). Each purchaser at any such sale shall hold the property sold absolutely free from any claim or right on the part of Grantor, and Grantor hereby waives (to the extent permitted by applicable law) all rights of redemption, stay and/or appraisal which it now has or may at any time in the future have under any rule of law or statute now existing or hereafter enacted. Grantor agrees that, to the extent notice of sale shall be required by law, at least ten (10) days' notice to Grantor of the time and place of any public sale or the time after which any private sale is to be made shall constitute reasonable notification.

- Secured Party shall not be obligated to make any sale of Collateral regardless of notice of (c) sale having been given. Secured Party may adjourn any public or private sale from time to time by announcement at the time and place fixed there for, and such sale may, without further notice, be made at the time and place to which it was so adjourned. Grantor agrees that it would not be commercially unreasonable for Secured Party to dispose of the Collateral or any portion thereof by using internet sites that provide for the auction of assets of the types included in the Collateral or that have the reasonable capability of doing so, or that match buyers and sellers of assets. Grantor hereby waives any claims against Secured Party arising by reason of the fact that the price at which any Collateral may have been sold at such a private sale was less than the price which might have been obtained at a public sale, even if Secured Party accepts the first offer received and does not offer such Collateral to more than one offeree. If the proceeds of any sale or other disposition of the Collateral are insufficient to pay all the Secured Obligations, Grantor shall be liable for the deficiency and the fees of any attorneys employed by Secured Party to collect such deficiency.
- (d) Grantor further agrees that a breach of any of the covenants contained in this Section will cause irreparable injury to Secured Party, that Secured Party has no adequate remedy at law in respect of such breach and, as a consequence, that each and every covenant contained in this Section shall be specifically enforceable against Grantor, and Grantor hereby waives and agrees not to assert any defenses against an action for specific performance of such covenants except for a defense that no default has occurred giving rise to the Secured Obligations becoming due and payable prior to their stated maturities. Nothing in this Section shall in any way limit the rights of Secured Party hereunder.
- (e) Secured Party may sell the Collateral without giving any warranties as to the Collateral. Secured Party may specifically disclaim or modify any warranties of title or the like. This procedure shall not be considered to adversely affect the commercial reasonableness of any sale of the Collateral.
- (f) Secured Party shall have no obligation to marshal any of the Collateral.

8.2 Application of Proceeds

Except as expressly provided elsewhere in this Agreement, all proceeds received by Secured Party in respect of any sale of, any collection from, or other realization upon all or any part of the Collateral shall be applied in full or in part by Secured Party in the following order of priority: first, against secured obligations of the DIP Lenders; second, against secured obligations of the CE Capital Providers; third, against the Secured Obligations; fourth, to the extent of any excess of such proceeds, to the payment to any subordinate secured party of all amounts to which such subordinate secured party is lawfully entitled; and fifth, to the extent of any further excess of such proceeds, to the payment to or upon the order of Grantor or as a court of competent jurisdiction may direct.

8.3 Cash Proceeds

Upon the occurrence of a Material Breach, proceeds of any Collateral received by Grantor consisting of Cash Proceeds shall be held by Grantor in trust for Secured Party, segregated from other funds of Grantor, and shall, forthwith upon receipt by Grantor, be turned over to Secured Party in the exact form received by Grantor (duly indorsed by Grantor to Secured Party, if required) and held by Secured Party. Any Cash Proceeds received by Secured Party (whether from Grantor or otherwise) may, in the sole discretion of Secured Party, (A) be held by Secured Party as collateral security for the Secured Obligations (whether matured or unmatured) and/or (B) then or at any time thereafter may be applied by Secured Party against the Secured Obligations then due and owing.

9. CONTINUING SECURITY INTEREST

This Agreement shall create a continuing security interest in the Collateral and shall remain in full force and effect until the payment in full of all Secured Obligations, be binding upon Grantor, its successors and assigns, and inure, together with the rights and remedies of Secured Party hereunder, to the benefit of Secured Party and its successors, transferees and assigns. Upon the payment in full of all Secured Obligations, this Agreement, and the security interest granted hereby, shall automatically terminate without any further action by any Person and all rights to the Collateral shall revert to Grantor. Upon any such termination Secured Party shall execute and deliver to Grantor or otherwise authorize the filing of such documents as Grantor shall reasonably request to evidence such termination, including UCC termination statements.

10. STANDARD OF CARE; SECURED PARTY MAY PERFORM

The powers conferred on Secured Party hereunder are solely to protect their interest in the Collateral and shall not impose any duty upon Secured Party to exercise any such powers. Except for the exercise of reasonable care in the custody of any Collateral in its possession and the accounting for moneys actually received by it hereunder, Secured Party shall have no duty as to any Collateral or as to the taking of any necessary steps to preserve rights against prior parties or any other rights pertaining to any Collateral. Secured Party shall be deemed to have exercised reasonable care in the custody and preservation of Collateral in its possession if such Collateral is accorded treatment substantially equal to that which Secured Party accords its own property. Secured Party shall not, and each of its directors, officers, employees and agents shall

not, be liable for failure to demand, collect or realize upon all or any part of the Collateral or for any delay in doing so, and shall not be under any obligation to sell or otherwise dispose of any Collateral upon the request of Grantor or otherwise. If Grantor fails to perform any agreement contained herein, Secured Party may itself perform, or cause performance of, such agreement, and the expenses of Secured Party incurred in connection therewith shall be payable by Grantor.

11. NOTICES

11.1 Effectiveness of Notices

Any notice or other communication in respect of this Agreement shall be in writing and may be given in any manner described below to the address or number provided for the recipient on Schedule 1 and shall be deemed effective as indicated:

- (a) if delivered in person or by courier, on the date it is received;
- (b) if sent by certified or registered mail or the equivalent (return receipt requested), on the date it is received;
- (c) if sent by facsimile transmission, on the date the sender receives an acknowledgement from the recipient; or
- (d) if sent by e-mail, on the date the sender receives an acknowledgement from the recipient, unless the date of receipt is not a Business Day or the communication is received after the close of business on a Business Day, in which case such communication shall be deemed given and effective on the first following day that is a Business Day. Notices to Grantor shall be copied to the Nominated Lawyers, and the time of effectiveness of each such notice shall be the effective time of receipt by either Grantor or the Nominated Lawyers, whichever occurs earlier.

11.2 Change of Details

Either party may by notice to the other in accordance with Section 11.1 change the address, facsimile number or e-mail details at which notices or other communications are to be given to it.

12. MISCELLANEOUS

12.1 Remedies Cumulative

The rights, powers, remedies and privileges provided in this Agreement are cumulative; may be exercised singularly, concurrently or successively at the Secured Party's option; and may be exercised or enforced without constituting a bar to the exercise or enforcement of any other such rights, powers, remedies and privileges.

12.2 Covenants Independent

All covenants hereunder shall be given independent effect so that if a particular action or condition is not permitted by any of such covenants, the fact that it would be permitted by an

exception to, or would otherwise be within the limitations of, another covenant shall not avoid the occurrence of a breach if such action is taken or condition exists.

12.3 Successors and Assigns

This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns.

12.4 No Transfer

Grantor shall not transfer or delegate any right or obligation in or under this Agreement without the prior written consent of Secured Party, except that with thirty (30) days' advance written notice to Secured Party, Grantor may make a transfer of all (but not less than all) of its rights and obligations under this Agreement pursuant to a consolidation or amalgamation with, or merger with or into, or transfer of all or substantially all its assets to, another entity (but without prejudice to any other right or remedy of Secured Party hereunder), provided that such transfer would not cause a Material Adverse Effect. Secured Party may assign its rights and obligations under this Agreement, in whole or in part, to another Person without the consent of Grantor. Secured Party may also appoint a servicing entity to administer this Agreement. Any purported transfer that is not in compliance with this Section shall be void.

12.5 Amendments

No amendment, modification or waiver in respect of this Agreement shall be effective unless in writing and executed by each of the parties.

12.6 Entire Agreement

This Agreement and the other Transaction Documents collectively constitute the entire agreement and understanding of the parties with respect to their subject matter and supersede all oral communication and prior writings with respect thereto.

12.7 Counterparts

This Agreement (and each amendment, modification and waiver in respect of it) may be executed and delivered in counterparts (including by facsimile or digital transmission), each of which shall be deemed an original.

12.8 No Waiver

A failure or delay in exercising any right, power or privilege in respect of this Agreement shall not be presumed to operate as a waiver, and a single or partial exercise of any right, power or privilege shall not be presumed to preclude any subsequent or further exercise, of that right, power or privilege or the exercise of any other right, power or privilege.

12.9 Severability

If any term of this Agreement, or the application thereof to either party or any circumstance, is held to be unenforceable, invalid or illegal (in whole or in part) for any reason (in any relevant jurisdiction), the remaining terms, modified by the deletion of the unenforceable, invalid or illegal portion, shall continue in full force and effect, and such unenforceability, invalidity, or illegality shall not otherwise affect that of the remaining terms, so long as this Agreement as so modified continues to express, without material change, the original intentions of the parties as to the subject matter hereof and the deletion of such portion of this Agreement shall not substantially impair the respective expectations or reciprocal obligations of the parties or the practical realization of the benefits that would otherwise be conferred upon the parties. The parties shall endeavour in good faith negotiations to replace any prohibited or unenforceable provision with a valid provision, the economic effect of which comes as close as possible to that of the prohibited or unenforceable provision.

13. APPLICABLE LAW AND JURISDICTION

THIS AGREEMENT AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES HEREUNDER AND ALL CLAIMS AND CONTROVERSIES ARISING OUT OF THE SUBJECT MATTER HEREOF WHETHER SOUNDING IN CONTRACT LAW, TORT LAW OR OTHERWISE SHALL BE GOVERNED BY, AND SHALL BE CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK, WITHOUT REGARD TO CONFLICTS OF LAW PROVISIONS THAT WOULD RESULT IN THE APPLICATION OF ANY OTHER LAW (OTHER THAN ANY MANDATORY PROVISIONS OF THE UCC RELATING TO THE LAW GOVERNING PERFECTION, THE EFFECT OF PERFECTION OR NONPERFECTION AND THE PRIORITY OF THE SECURITY INTEREST).

SUBJECT TO CLAUSE (E) OF THE FOLLOWING SENTENCE, AND SOLELY FOR THE PURPOSES OF ENFORCEMENT OF THIS AGREEMENT AND NO OTHER PURPOSE, ALL JUDICIAL PROCEEDINGS BROUGHT AGAINST ANY PARTY ARISING OUT OF OR RELATING HERETO, OR ANY OF THE OBLIGATIONS, SHALL BE BROUGHT EXCLUSIVELY IN THE UNITED STATES BANKRUPTCY COURT FOR THE SOUTHERN DISTRICT OF NEW YORK. IF THE BANKRUPTCY COURT DECLINES JURISDICTION, THE JUDICIAL PROCEEDING MAY BE BROUGHT IN ANOTHER COURT OF COMPETENT JURISDICTION IN THE STATE OF NEW YORK, CITY OF NEW YORK, BOROUGH OF MANHATTAN. EXECUTING AND DELIVERING THIS AGREEMENT, GRANTOR IRREVOCABLY (A) ACCEPTS GENERALLY AND UNCONDITIONALLY THE EXCLUSIVE JURISDICTION AND VENUE OF SUCH COURTS; (B) WAIVES ANY DEFENSE OF FORUM NON CONVENIENS; (C) AGREES THAT SERVICE OF ALL PROCESS IN ANY SUCH PROCEEDING IN ANY SUCH COURT MAY BE MADE BY REGISTERED OR CERTIFIED MAIL, RETURN RECEIPT REQUESTED, TO THE GRANTOR AT ITS ADDRESS PROVIDED IN ACCORDANCE WITH THIS AGREEMENT; (D) AGREES THAT SERVICE AS PROVIDED IN CLAUSE (C) ABOVE IS SUFFICIENT TO CONFER PERSONAL JURISDICTION OVER THE APPLICABLE PARTY IN ANY SUCH PROCEEDING IN ANY SUCH COURT, AND OTHERWISE CONSTITUTES EFFECTIVE AND BINDING SERVICE IN EVERY RESPECT; AND (E) AGREES THAT SECURED PARTY RETAINS THE RIGHT TO SERVE PROCESS IN ANY OTHER MANNER PERMITTED BY LAW.

EACH OF THE PARTIES HERETO HEREBY AGREES TO WAIVE ITS RESPECTIVE RIGHTS TO A JURY TRIAL OF ANY CLAIM OR CAUSE OF ACTION BASED UPON OR ARISING HEREUNDER OR ANY DEALINGS BETWEEN THEM RELATING TO THE SUBJECT MATTER HEREOF. THE SCOPE OF THIS WAIVER IS INTENDED TO BE ALL-ENCOMPASSING OF ANY AND ALL DISPUTES THAT MAY BE FILED IN ANY COURT AND THAT RELATE TO THE SUBJECT MATTER HEREOF, INCLUDING CONTRACT CLAIMS, TORT CLAIMS, BREACH OF DUTY CLAIMS AND ALL OTHER COMMON LAW AND STATUTORY CLAIMS. EACH PARTY HERETO ACKNOWLEDGES THAT THIS WAIVER IS A MATERIAL INDUCEMENT TO ENTER INTO A BUSINESS RELATIONSHIP, THAT EACH HAS ALREADY RELIED ON THIS WAIVER IN ENTERING INTO THIS AGREEMENT, AND THAT EACH WILL CONTINUE TO RELY ON THIS WAIVER IN ITS RELATED FUTURE DEALINGS. EACH PARTY HERETO FURTHER WARRANTS AND REPRESENTS THAT IT HAS REVIEWED THIS WAIVER WITH COUNSEL AND THAT IT KNOWINGLY AND VOLUNTARILY WAIVES ITS JURY TRIAL RIGHTS **FOLLOWING CONSULTATION** WITH COUNSEL. **THIS** WAIVER IRREVOCABLE, MEANING THAT IT MAY NOT BE MODIFIED EITHER ORALLY OR IN WRITING OTHER THAN BY A MUTUAL WRITTEN WAIVER SPECIFICALLY REFERRING TO THIS SECTION AND EXECUTED BY EACH OF THE PARTIES HERETO, AND THIS WAIVER SHALL APPLY TO ANY SUBSEQUENT AMENDMENTS, RENEWALS, SUPPLEMENTS OR MODIFICATIONS HERETO. IN THE EVENT OF LITIGATION, THIS AGREEMENT MAY BE FILED AS A WRITTEN CONSENT TO A TRIAL BY THE COURT.

14. Intentionally Omitted.

15. SUBORDINATION AGREEMENT

The terms and conditions of this Agreement are expressly subject to the terms and conditions of the Subordination Agreement.

[Remainder of this page intentionally left blank.]

Signature Page to Security Agreement

IN WITNESS WHEREOF, Grantor and Secured Party have caused this Agreement to be duly executed and delivered by their respective officers thereunto duly authorized as of the date first written above.

SECURED PARTY:	NOMINATED LAWYERS:				
	Accepted and Agreed (as to Section 4 only):				
LW Holdco VI LLC	Binder & Schwartz LLP				
By Name: Boaz Weinstein	By:, a Partner				
Title: Authorized Signatory	Print Name				
GRANTOR:	TRUST MONITOR:				
Motors Liquidation Company Avoidance	Acknowledged and approved: Action Trust				
By: Wilmington Trust Company, as Trust Administrator and Trustee	Arthur Gonzalez, as Trust Monitor				
ByName: Title:					

Signature Page to Security Agreement

IN WITNESS WHEREOF, Grantor and Secured Party have caused this Agreement to be duly executed and delivered by their respective officers thereunto duly authorized as of the date first written above.

SECURED PARTY:	NOMINATED LAWYERS:				
e e	Accepted and Agreed (as to Section 4 only):				
LW Holdco VI LLC	Binder & Schwartz LLP				
ByName: Title:	By: A Partner Ne.: 1 5. Binder Print Name				
GRANTOR:	TRUST MONITOR:				
Motors Liquidation Company Avoidance	Acknowledged and approved: Action Trust				
By: Wilmington Trust Company, as Trust Administrator and Trustee By Name: Sch Andrus Title: Viu Husidest	Arthur Gonzalez, as Trust Monitor				

Signature Page to Security Agreement

IN WITNESS WHEREOF, Grantor and Secured Party have caused this Agreement to be duly executed and delivered by their respective officers thereunto duly authorized as of the date first written above.

SECURED PARTY:	NOMINATED LAWYERS:				
LW Holdco VI LLC	Accepted and Agreed (as to Section 4 only): Binder & Schwartz LLP				
ByName:	By:, a Partner				
Title:	Print Name				
96 B					
GRANTOR:	TRUST MONITOR:				
Motors Liquidation Company Avoidance By: Wilmington Trust Company, as Trust Administrator and Trustee By	Acknowledged and approved: Action Trust Arthur Gonzalez, as Trust Monitor				

EXHIBIT 1 IRREVOCABLE INSTRUCTIONS TO COUNSEL

As of January 24, 2019

Binder & Schwartz LLP 366 Madison Avenue 6th Floor New York, NY 10017 Attention: Neil Binder

As you know, the Motors Liquidation Company Avoidance Action Trust ("Us" "We", or "Client") has retained Binder & Schwartz LLP ("You" or "Counsel") to represent Us in claims against the Opponent. We are entering into a capital provision agreement (the "LW Capital Provision Agreement") and a Security Agreement (the "Security Agreement") (together with the LW Capital Provision Agreement, the "Agreements") with LW Holdco VI LLC (the "LW Capital Provider"). Copies of the Agreements are attached to this letter. All capitalized terms not defined in this letter refer to defined terms in each Agreement.

Except as may be otherwise provided in an order of the Bankruptcy Court entered after the date hereof, in connection with the Agreements, We hereby give You the following irrevocable instructions:

- 1. We instruct You to keep the LW Capital Provider reasonably informed about developments in the Claim, and about developments in efforts to monetize the Claim. We further permit and instruct You, pursuant to the LW Capital Provision Agreement, to share information with LW Capital Provider as provided therein.
- 2. We instruct You to promptly inform the LW Capital Provider in writing upon your becoming aware of the receipt by Grantor, or any other party on behalf of Grantor, of any funds that would arguably be part of the Claim Proceeds (as defined in the Security Agreement).
- 3. We instruct You to promptly notify the LW Capital Provider in writing in the event Your representation of Us terminates or changes in a material way, including but not limited to any modification of any Engagement Agreement or any other agreement relating to Your compensation. Such notification shall include all specific details of the change.

The provisions of these Irrevocable Instructions to Counsel are for the benefit of Client, and also for the benefit of the LW Capital Provider and its Affiliates, permitted successors and assigns (collectively, the "Third Parties"). The LW Capital Provider would not have entered into the Agreements without these Irrevocable Instructions and Your acknowledgment hereof, and the Agreements are of significant economic value to Us. Client and Counsel explicitly agree that the Third Parties are express third-party beneficiaries hereunder, and the Third Parties shall be permitted to enforce this letter agreement as if they were a party hereto.

[Remainder of this page intentionally left blank.]

Please note that the directions contained in this Irrevocable Instruction to Counsel are irrevocable from the date hereof except in the event the LW Capital Provider grants Us written permission to make such revocation.

Very truly yours,

RUST

MOTORS LIQUIDATION COMPANY AVOIDANCE ACTION T
By: Wilmington Trust Company, as Trust Administrator and Trustee
By Name: Beth Agellus Title: Vile Hesiolent
Address: Wilmington Trust Company Rodney Square North 1100 North Market Street Wilmington, DE 19890-1615 Attn: Corporate Trust Administration

Telephone: (302) 651-1000

Fax: (302) 636-4145

Email: bandrews@wilmingtontrust.com

Accepted and agreed,

BINDER & SCHWARTZ LLP

By _			
Name:			
Title:	Partner	~	

Acknowledged and approved:

ARTHUR GONZALEZ, as Trust Monitor

Please note that the directions contained in this Irrevocable Instruction to Counsel are irrevocable from the date hereof except in the event the LW Capital Provider grants Us written permission to make such revocation.

Very truly yours,

MOTORS LIQUIDATION COMPANY AVOIDANCE ACTION TRUST

By: Wilmington Trust Company, as Trust Administrator and Trustee

Ву

Name: Title:

Sern Andrews Vice Henrickas

Address:

Wilmington Trust Company

Rodney Square North 1100 North Market Street Wilmington, DE 19890-1615

Attn: Corporate Trust Administration

Telephone: (302) 651-1000

Fax: (302) 636-4145

Email: bandrews@wilmingtontrust.com

Accepted and agreed,

BINDER & SCHWARTZ LLP

Name:

Title: Partner

Acknowledged and approved:

ARTHUR GONZALEZ, as Trust Monitor

Exhibit G

SUBORDINATION AGREEMENT

Among

UNITED STATES DEPARTMENT OF TREASURY AND EXPORT DEVELOPMENT CANADA,

as DIP Lenders

CYNTHIANA LLC AND EARLHAM LLC,

as CE Capital Providers

LW HOLDCO VI LLC,

as LW Capital Provider

and

MOTORS LIQUIDATION COMPANY AVOIDANCE ACTION TRUST,

a Delaware Statutory Trust Dated as of January 24, 2019

SUBORDINATION AGREEMENT

This SUBORDINATION AGREEMENT (this "Agreement") is made as of January 24, 2019, by and among (i) the United States Department of Treasury and Export Development Canada, as lenders under the DIP Credit Agreement (collectively, the "DIP Lenders"); (ii) Cynthiana LLC, a limited liability company formed under the laws of the State of Delaware, and Earlham LLC, a limited liability company formed under the laws of the State of Delaware (collectively, the "CE Capital Providers"); (iii) LW Holdco VI LLC, a limited liability company formed under the laws of the State of Delaware (the "LW Capital Provider"); and (iv) Motors Liquidation Company Avoidance Action Trust, an entity formed under the laws of Delaware (the "Trust"), and together with the DIP Lenders, the CE Capital Providers and the LW Capital Provider, the "Parties"). Capitalized terms not defined herein shall have the meaning ascribed thereto in the Fourth Amended and Restated Motors Liquidation Company Avoidance Action Trust Agreement").

RECITALS

WHEREAS, prior to August 2016, there was a dispute between the Official Committee of Unsecured Creditors of Motors Liquidation Company (the "Committee") and the DIP Lenders as to entitlements to Avoidance Action Proceeds, which dispute was resolved by mutual agreement (the "Litigation Settlement") pursuant to which, among other things, (i) the DIP Lenders provided the Litigation Cost Advance and (ii) the DIP Lenders shall be entitled to be repaid the Litigation Cost Advance out of Distributable Trust Assets; and

WHEREAS, the Trust Administrator and the Trust Monitor determined that the Avoidance Action Trust Administrative Cash and Supplemental Avoidance Action Trust Cash held at the time of the Litigation Settlement by the Trust was insufficient to satisfy projected fees, costs and expenses of the Trust, and, as an integral part of the Litigation Settlement, the DIP Lenders agreed to provide the Litigation Cost Advance to the Trust on the terms set forth in that certain agreement, dated July 14, 2016, executed by the Trust and the DIP Lenders (the "Litigation Cost Advance Agreement"), which included, among other things, that the DIP Lenders would provide the Litigation Cost Advance to fund the prosecution of the Avoidance Action; and

WHEREAS, on August 24, 2016, the Bankruptcy Court entered an order approving the Settlement Agreement and the Litigation Cost Advance Agreement; and

WHEREAS, one of the preconditions to the DIP Lenders' provision of the Litigation Cost Advance was the Trust's agreement, as reflected in Article 6.1(d)(i) of the Second Amended and Restated Trust Agreement, that any additional litigation funding provided to the Trust would be (i) junior and subordinate to the Litigation Cost Advance and any other amounts owed to the DIP Lenders on account of prior funding of the Trust and (ii) subject to a form of subordination acceptable to the DIP Lenders in all respects; and

WHEREAS, subsequent to entering into the Litigation Cost Advance Agreement, the Trust Administrator and the Trust Monitor determined that the Avoidance Action Trust Administrative Cash and Supplemental Avoidance Action Trust Cash held by the Trust remained insufficient to satisfy projected fees, costs and expenses of the Trust; and

WHEREAS, with the consent of the DIP Lenders, the Trust Administrator and Trust Monitor entered into that certain capital provision agreement, dated June 2, 2017 (including any related security agreement, the "CE Capital Provision Agreement"), with the CE Capital Providers, pursuant to which, among other things, the CE Capital Providers agreed to provide up to \$15,000,000 in additional funding to the Trust (the "CE Supplemental Capital") as Other Supplemental Cash; and

WHEREAS, on or prior to the date hereof, the Trust Administrator and the Trust Monitor have determined that the Avoidance Action Trust Administrative Cash and Supplemental Avoidance Action Trust Cash currently held by the Trust remain insufficient to satisfy projected fees, costs and expenses of the Trust; and

WHEREAS, with the consent of the DIP Lenders, the Trust Administrator and Trust Monitor entered into a capital provision agreement, dated January 21, 2019 (including any related security agreement, the "LW Capital Provision Agreement"), with the LW Capital Provider, pursuant to which, among other things, the LW Capital Provider agreed to provide up to \$10,000,000 in additional funding to the Trust (the "LW Supplemental Capital") as Other Supplemental Cash; and

WHEREAS, pursuant to the DIP Lenders' consent rights under Article 6.1(d)(i) of the Third Amended and Restated Trust Agreement, the Trust's and the LW Capital Provider's entry into a form of subordination acceptable to the DIP Lenders is a condition of the DIP Lenders' consent to (i) the Trust's entry into the LW Capital Provision Agreement and (ii) the LW Capital Provider's provision of the LW Supplemental Capital to the Trust.

AGREEMENT

NOW, THEREFORE, in consideration of the premises and other good and valuable consideration, the adequacy and legal sufficiency of which are hereby acknowledged, the Parties hereto agree as follows:

ARTICLE I DEFINITIONS

SECTION 1.1 <u>General Definitions.</u> As used in this Agreement, the terms defined in the Preamble and Recitals hereto shall have the respective meanings specified therein, and the following terms shall have the following meanings:

"<u>Distribution Proceeding</u>" means and includes any distribution of all or substantially all of the assets of the Trust or any liquidation, dissolution, winding up or reorganization of the Trust or its business whether by judicial or court proceedings or otherwise, including, without limitation, any sale, receivership, insolvency, arrangement, reorganization, bankruptcy, assignment for the benefit of creditors, marshalling of assets and liabilities or other proceedings by or against the Trust under any federal or state bankruptcy, reorganization, insolvency or other law or laws relating to the relief of debtors, readjustment of indebtedness, reorganization or composition.

"Governmental Authority" shall mean and include any federal, state, municipal, national, or other government, governmental department, commission, board, bureau, court, agency, or

instrumentality, or political subdivision thereof, or any entity or officer exercising executive, legislative, judicial, regulatory, or administrative functions of or pertaining to any government or any court, in each case whether associated with a state of the United States, the United States, another domestic government, or a foreign entity or government.

"<u>LW Capital Provider's Entitlement</u>" has the meaning ascribed to it in the LW Capital Provision Agreement.

"Paid in Full" or "Payment in Full" shall mean the payment in full in cash of all Senior Obligations, Second Ranking Obligations or LW Capital Provider's Entitlement, as applicable.

"Person" or "person" shall mean and include a natural person, a company, a partnership, a joint venture, a trust, an unincorporated organization or association or a government or any department or agency thereof.

"Second Ranking Obligations" means collectively and includes all present indebtedness, liabilities, and obligations of the Trust to the CE Capital Providers of any kind and nature whatsoever now existing under, or as a result of, the CE Capital Provision Agreement.

"Senior Obligations" means collectively and includes all present and future indebtedness, liabilities, and obligations of the Trust to the DIP Lenders of any kind and nature whatsoever both now existing and hereafter arising under, or as a result of, the Litigation Cost Advance Agreement, the Settlement Agreement and the Trust Agreement.

"Standstill Notice" means a written notice from the LW Capital Provider to the DIP Lenders and the CE Capital Providers identified by its terms as a "Standstill Notice" for purposes of this Agreement and stating that a default under the LW Capital Provision Agreement has occurred and is continuing and that, as a consequence thereof, the LW Capital Provider has declared all of the Subordinated Obligations to be immediately due and payable.

"Standstill Period" means the period of 90 days commencing on the date on which the DIP Lenders and the CE Capital Providers receive the applicable Standstill Notice.

"Subordinated Obligations" means collectively and includes all present and future indebtedness, liabilities, and obligations of the Trust to the LW Capital Provider of any kind and nature whatsoever both now existing and hereafter arising under, or as a result of, the LW Capital Provision Agreement and the Trust Agreement.

SECTION 1.2 Other Definitional Provisions. The words "hereof," "herein" and "hereunder" and words of similar import when used in this Agreement shall refer to this Agreement as a whole and not to any particular provision of this Agreement, and section, subsection, schedule and exhibit references are references to sections or subsections of, or schedules or exhibits to, as the case may be, this Agreement unless otherwise specified. As used herein, the singular number shall include the plural, the plural the singular and the use of the masculine, feminine or neuter gender shall include all genders, as the context may require.

ARTICLE II SUBORDINATION

SECTION 2.1 Subordination of Subordinated Obligations. The Trust and the LW Capital Provider hereby subordinate the payment of all of the Subordinated Obligations to the prior Payment in Full of the Senior Obligations and the Second Ranking Obligations and agree that all of the right, title and interest of the LW Capital Provider in and to the payment of the Subordinated Obligations shall at all times be subject, junior and subordinate to the right, title and interest of the DIP Lenders in and to the Payment in Full of the Senior Obligations and to the right, title and interest of the CE Capital Providers in and to the Payment in Full of the Second Ranking Obligations.

Provider will not take, demand or receive, and the Trust will not make, pay, distribute, give or otherwise permit, directly or indirectly, by set-off, redemption, purchase, distribution or otherwise, any payment or prepayment (whether as principal, interest or otherwise, and whether in cash, securities or property) on all or any part of the Subordinated Obligations, until and unless (a) all of the Senior Obligations have been Paid in Full in accordance with the Litigation Cost Advance Agreement, the Settlement Agreement and the Trust Agreement, including, without limitation, payment of any interest at the applicable contract rates accruing after the institution of any Distribution Proceeding, whether or not allowed as a claim in any such Distribution Proceeding and (b) all of the Second Ranking Obligations have been Paid in Full in accordance with the CE Capital Provision Agreement and the Trust Agreement, including, without limitation, payment of any interest at the applicable contract rates accruing after the institution of any Distribution Proceeding, whether or not allowed as a claim in any such Distribution Proceeding.

In the event that any payment, recovery or distribution of the Subordinated Obligations of any kind or character, whether in cash, property or securities, and whether directly, by exercise of any right of set-off, exercise of any remedy with respect to collateral, or otherwise, shall be received by the LW Capital Provider, such payment, recovery or distribution will be received by the LW Capital Provider solely in trust as agent for the DIP Lenders and the CE Capital Providers, and the LW Capital Provider will immediately turn the same over to the DIP Lenders for application to the Senior Obligations, and to the extent the Senior Obligations have been Paid in Full, to the CE Capital Providers for application to the Second Ranking Obligations, in each case in the form received except for the endorsement(s) of the LW Capital Provider where appropriate, which endorsement(s) the LW Capital Provider agrees to make forthwith. Until so turned over, the LW Capital Provider shall hold such payment in trust for the DIP Lenders and the CE Capital Providers. If the LW Capital Provider fails to so endorse any such payment where necessary or appropriate, the LW Capital Provider hereby irrevocably constitutes and appoints the DIP Lenders (or if the Senior Obligations have been Paid in Full, the CE Capital Providers) as attorneys-in-fact for the LW Capital Provider with full power to make any such endorsement, and with full power of substitution.

SECTION 2.3 <u>Distribution Proceeding</u>. Upon any Distribution Proceeding by or against the Trust, any payment, recovery or distribution of any kind or character, whether in cash, property or securities which, but for the provisions of this Agreement, would be payable or deliverable to the LW Capital Provider upon or in respect of the Subordinated Obligations, shall

be paid or delivered by the LW Capital Provider or any person making such payment, recovery or distribution (whether a trustee in bankruptcy, debtor in possession, a receiver, custodian, liquidating trustee or otherwise) directly to the DIP Lenders for application by the DIP Lenders as a payment or prepayment on account of the Senior Obligations (and to the extent the Senior Obligations have been Paid in Full, to the CE Capital Providers for application by the CE Capital Providers as a payment or prepayment on account of the Second Ranking Obligations), and the LW Capital Provider shall not receive any such payment, recovery or distribution or any benefit therefrom unless and until the Senior Obligations and the Second Ranking Obligations have been In the event that, notwithstanding the foregoing, any payment, recovery or distribution of the assets of the Trust of any kind or character, whether in cash, property or securities, and whether directly, by exercise of any right of set-off or otherwise, shall be received by the LW Capital Provider before all of the Senior Obligations and Second Ranking Obligations have been Paid in Full, such payment, recovery or distribution will be received by the LW Capital Provider solely as agent for the DIP Lenders and the CE Capital Providers, and the LW Capital Provider will immediately turn the same over to the DIP Lenders for application to the Senior Obligations (and to the extent the Senior Obligations have been Paid in Full, to the CE Capital Providers for application to the Second Ranking Obligations) in the form received except for the endorsement(s) of the LW Capital Provider where appropriate, which endorsement(s) the LW Capital Provider agrees to make forthwith. Until so turned over, the LW Capital Provider shall hold such payment, recovery or distribution in trust for the DIP Lenders and the CE Capital Providers. If the LW Capital Provider fails to so endorse any such payment, recovery or distribution where necessary or appropriate, the LW Capital Provider hereby irrevocably constitutes and appoints the DIP Lenders (or if the Senior Obligations have been Paid in Full, the CE Capital Providers) as attorneys-in-fact for the LW Capital Provider with full power to make any such endorsement, and with full power of substitution.

SECTION 2.4 Enforcement.

The LW Capital Provider irrevocably authorizes and empowers the DIP Lenders to demand, sue for, collect and receive any payment or distribution of any kind or character to which the LW Capital Provider may be entitled with respect to any Distribution Proceeding and give acquittance therefor, and to file claims and proofs of claims in any Distribution Proceeding, vote such claims and take such other actions in the DIP Lenders' name, or in the LW Capital Provider's name or otherwise, as the DIP Lenders may deem necessary or advisable for the enforcement of the provisions of this Agreement. Upon any Distribution Proceeding, the LW Capital Provider hereby agrees to properly take such action as may be reasonably requested at any time and from time to time by the DIP Lenders to collect the Subordinated Obligations for the DIP Lenders' account (including, without limitation, to declare all of the Subordinated Obligations to be due and payable), to file appropriate proofs of claims in respect thereof, to file motions for relief from any applicable bankruptcy stays, and to execute and deliver such powers of attorney, assignments or other instruments as the DIP Lenders may request in order to enable the DIP Lenders to enforce any and all claims upon or in respect of the Subordinated Obligations and to collect and receive all payments, recoveries or distributions which may be payable or deliverable at any time upon or in respect of the Subordinated Obligations, or, alternatively, to permit the DIP Lenders to take such actions in the LW Capital Provider's name. Any and all moneys so collected or received by the DIP Lenders shall be retained indefeasibly by the DIP Lenders for application to the Senior Obligations (whether matured or unmatured);

provided, that upon the indefeasible payment to the DIP Lenders of moneys collected or received on account of the Subordinated Obligations, the Senior Obligations, and the Second Ranking Obligations (if any) in an aggregate amount in excess of the matured and unmatured Senior Obligations, the DIP Lenders will (x) first, pay over to the CE Capital Providers the portion of such excess to which the CE Capital Providers are entitled on account of the Second Ranking Obligations, if any, and (y) second, pay over to the LW Capital Provider the portion of such excess to which the LW Capital Provider is entitled on account of the Subordinated Obligations, if any.

(b) If the Senior Obligations have been indefeasibly Paid in Full, the rights of the DIP Lenders pursuant to the foregoing clause (a) shall be deemed to be rights of the CE Capital Providers, *mutatis mutandis*; provided that any and all moneys collected or received by the CE Capital Providers as a result of their exercise of such rights shall be retained indefeasibly by the CE Capital Providers for application to the Second Ranking Obligations (whether matured or unmatured), provided that upon the indefeasible payment to the CE Capital Providers of moneys collected or received on account of the Subordinated Obligations and the Second Ranking Obligations (if any) in an aggregate amount in excess of the matured and unmatured Second Ranking Obligations, the CE Capital Providers will pay over to the LW Capital Provider the portion of such excess to which the LW Capital Provider is entitled on account of the Subordinated Obligations, if any.

Obligations and the Second Ranking Obligations have been Paid in Full, the LW Capital Provider, without the prior written consent of the DIP Lenders and the CE Capital Providers, will not (i) accelerate the maturity of any of the Subordinated Obligations; (ii) exercise any rights or remedies or take any action or proceeding to collect or enforce in any manner whatsoever any of the Subordinated Obligations, other than to file a claim and otherwise exercise its rights in a Distribution Proceeding, consistent with the other terms of this Agreement; (iii) join with any other creditors of the Trust in filing any petition commencing any Distribution Proceeding against or in respect of the Trust; or (iv) exercise any remedy with respect to any collateral securing the Subordinated Obligations; provided, that after the expiration of the Standstill Period, the LW Capital Provider may take any of the foregoing actions if the exercise of remedies with respect to the Senior Obligations and the Second Ranking Obligations is not being diligently pursued by the DIP Lenders and/or the CE Capital Providers at such time.

SECTION 2.6 Amendments to LW Capital Provision Agreement. The Trust and the LW Capital Provider agree that neither the LW Capital Provision Agreement, nor any other instrument or agreement evidencing, creating, or executed and delivered in connection with all or any part of the Subordinated Obligations, without the prior written consent of the DIP Lenders and the CE Capital Providers, will be modified or amended in any way that has a material adverse effect on the Trust's operations or financial condition, or the Senior Obligations or the Second Ranking Obligations, or otherwise inconsistent with the terms of this Agreement.

SECTION 2.7 Amendments to Litigation Cost Advance Agreement and the Settlement Agreement. The Trust and the DIP Lenders agree that, without the prior written consent of the LW Capital Provider, no new agreement will be entered into by the Trust and DIP Lenders providing, nor will any instrument or agreement evidencing, creating, or executed and delivered in connection with all or any part of the Senior Obligations be modified or amended to

provide, for any increase in the amount of the Senior Obligations prior to the Payment in Full of the LW Capital Provider's Entitlement. The Trust and the CE Capital Providers agree that, without the prior written consent of the LW Capital Provider, no new agreement will be entered into by the Trust and the CE Capital Providers providing, nor will any instrument or agreement evidencing, creating, or executed and delivered in connection with all or any party of the Second Ranking Obligations be modified or amended to provide, for any increase in the amount of the Second Ranking Obligations prior to the Payment in Full of the LW Capital Provider's Entitlement.

ARTICLE III GENERAL PROVISIONS

SECTION 3.1 Representations and Warranties of LW Capital Provider. The LW Capital Provider represents and warrants that (i) it is duly organized or formed and validly existing under the laws of the jurisdiction of its organization or formation and, if relevant under such laws, in good standing; (ii) is qualified to do business in each jurisdiction in which the nature of its business so requires except to the extent that the failure to be so qualified would not reasonably be expected to have a material adverse effect on the operations of the LW Capital Provider; (iii) assuming proper execution and delivery by the other parties hereto, this Agreement constitutes the valid and binding obligation of the LW Capital Provider enforceable in accordance with its terms, subject as to enforcement, to applicable bankruptcy, insolvency, reorganization and other laws of general applicability relating to or affecting the LW Capital Provider's rights generally; and (iv) neither the execution and delivery of this Agreement by the LW Capital Provider, the consummation of the transactions herein contemplated, nor the compliance with the terms and provisions hereof by the LW Capital Provider will conflict with or result in the breach of any of the terms, conditions or provisions of the organizational documents of the LW Capital Provider or of any applicable law or regulation or of any agreement or material instrument to which the LW Capital Provider is a party or by which it or its property is bound or constitutes a default thereunder.

SECTION 3.2 <u>Default</u>. The breach or the failure of the DIP Lenders, the CE Capital Providers, the LW Capital Provider or the Trust in the due observance or performance of any of the provisions of this Agreement shall constitute a default under this Agreement.

All rights and interests of the DIP Lenders and the CE Capital Providers hereunder, and all agreements and obligations of the LW Capital Provider or the Trust under this Agreement shall remain in full force and effect and shall in no way be affected, limited, modified or released by the Trust or the LW Capital Provider, and may be enforced by the DIP Lenders until the Senior Obligations shall have been Paid in Full and by the CE Capital Providers until the Second Ranking Obligations shall have been Paid in Full, as applicable, irrespective of (a) any act or failure to act on the part of the LW Capital Provider or the Trust or any noncompliance by the LW Capital Provider or the Trust with the provisions of this Agreement; (b) any attempt, pursuit, enforcement or exhaustion of any rights and remedies the DIP Lenders or the CE Capital Providers may at any time have to collect the Senior Obligations or the Second Ranking Obligations, as applicable, from the Trust or any other maker, guarantor, endorser, surety or other person directly or contingently liable therefor or from any collateral and security for the Senior Obligations or the Second Ranking Obligations; (c) any release or non-perfection of any collateral and security for any or all of the

Senior Obligations or the Second Ranking Obligations; or (d) any other circumstance that might otherwise constitute a defense available to, or discharge of, the Trust in respect of the Senior Obligations (other than Payment in Full of the Senior Obligations) or the Second Ranking Obligations (other than Payment in Full of the Second Ranking Obligations).

- Without notice to, or consent of, the LW Capital Provider, and without in any way affecting, limiting, modifying, discharging or releasing the right of the DIP Lenders or the CE Capital Providers to enforce this Agreement or the liabilities and obligations of the parties hereto, the DIP Lenders and the CE Capital Providers may, to the extent consistent with the Trust Agreement, at any time and from time to time (subject to the relative rights between the DIP Lenders and the CE Capital Providers as set forth in this Agreement) (a) extend, renew, accelerate, modify, amend, restate, change, decrease, substitute, subordinate, exchange, sell, assign, compromise, settle, waive, terminate, release, discharge, grant indulgences with respect to, or otherwise deal with in any manner satisfactory to the DIP Lenders or the CE Capital Providers, as applicable, all or any part of the Senior Obligations or the Second Ranking Obligations, as applicable, any collateral and security therefor, the Trust or any other maker, guarantor, endorser, surety or other person directly or contingently liable or obligated for all or any part of the Senior Obligations or the Second Ranking Obligations, as applicable; (b) fail, omit, lack diligence or delay to perfect, maintain, enforce, assert or exercise any lien, encumbrance, right or remedy conferred on the DIP Lenders or the CE Capital Providers, as applicable; or (c) take or omit to take any other action, whether similar or dissimilar to the foregoing, which may or might in any manner or to any extent constitute a defense available to, or discharge of, the Trust in respect of the Senior Obligations or the Second Ranking Obligations, as applicable, or of the Trust or the LW Capital Provider in respect of this Agreement.
- (iii) The DIP Lenders, the CE Capital Providers and/or the LW Capital Provider may demand specific performance of this Agreement, and the Trust, the DIP Lenders, the CE Capital Providers and the LW Capital Provider acknowledge that there is no adequate remedy at law for a breach of this agreement and expressly waive any defense based on the adequacy of a remedy at law and any other defense that might be asserted to bar any remedy of specific performance requested by the DIP Lenders, the CE Capital Providers and/or the LW Capital Provider in any action or proceeding, including any Distribution Proceeding.
- (iv) The Parties acknowledge that this Agreement, in its entirety, constitutes a "subordination agreement" under section 510(a) of title 11 of the United States Code (the "Bankruptcy Code") and section 9-339 of the Uniform Commercial Code, which subordination agreement will be effective before, during, and after the commencement of a Distribution Proceeding. This Agreement shall remain enforceable and in full force and effect in any and all courts and tribunals or other Governmental Authorities notwithstanding that any bankruptcy court declines to give effect to this Agreement or any provision hereof in any Distribution Proceeding, whether pursuant to section 1129(b) of the Bankruptcy Code or otherwise. All references in this Agreement to the Trust and the LW Capital Provider includes such person as a debtor-in-possession and any receiver or trustee for such person, including in any Distribution Proceeding.

ARTICLE IV MISCELLANEOUS

SECTION 4.1 Course of Dealing; Amendment. No course of dealing between the DIP Lenders, the CE Capital Providers, the LW Capital Provider or the Trust shall be effective to amend, modify or change any provision of this Agreement. The DIP Lenders, the CE Capital Providers, and the LW Capital Provider shall have the right at all times to enforce the provisions of this Agreement in strict accordance with the provisions hereof, notwithstanding any conduct or custom on the part of the DIP Lenders, the CE Capital Providers or the LW Capital Provider in refraining from so doing at any time or times. The failure of the DIP Lenders, the CE Capital Providers or the LW Capital Provider at any time or times to enforce their rights under the provisions hereof, strictly in accordance with the same, shall not be construed as having created a custom in any way or manner contrary to specific provisions of this Agreement or as having in any way or manner modified or waived the same. This Agreement may not be amended, modified, or changed in any respect except by an agreement in writing signed by the DIP Lenders, the CE Capital Providers, the LW Capital Provider and the Trust.

SECTION 4.2 Waiver of Default. The DIP Lenders may, at any time and from time to time, execute and deliver to the Trust, the CE Capital Providers and/or the LW Capital Provider a written instrument waiving, on such terms and conditions as the DIP Lenders may specify in such written instrument, any of the requirements of this Agreement, provided that any such waiver shall be for such period and subject to such conditions as shall be specified in any such instrument. To the extent that the CE Capital Providers have enforcement rights under this Agreement, the CE Capital Providers may, at any time and from time to time, execute and deliver to the DIP Lenders, the Trust and/or the LW Capital Provider a written instrument waiving, on such terms and conditions as the CE Capital Providers may specify in such written instrument, any such rights, provided that any such waiver shall be for such period and subject to such conditions as shall be specified in any such instrument. Each such waiver shall be effective only in the specific instance and for the specific purpose for which given.

SECTION 4.3 Consent to Jurisdiction. Each of the DIP Lenders, the CE Capital Providers, the LW Capital Provider and the Trust (i) consent and submit to the jurisdiction and venue of any state or federal court sitting in the State of New York over any suit, action or proceeding arising out of or relating to this Agreement; (ii) waive, to the fullest extent permitted by law, any objection that the DIP Lenders, the CE Capital Providers, the Trust or the LW Capital Provider may now or hereafter have to the laying of the venue of any such suit, action or proceeding brought in any such court and any claim that any such suit, action or proceeding brought in any such court has been brought in an inconvenient forum; and (iii) consent to the service of process in any such suit, action or proceeding in any such court by such means as is provided for by the jurisdiction in which such suit, action or proceeding is filed.

SECTION 4.4 Waiver of Trial by Jury. The Trust, the LW Capital Provider, the CE Capital Providers and the DIP Lenders hereby voluntarily and intentionally waive any rights they may have to trial by jury in any action or proceeding to which the Trust, the LW Capital Provider, the CE Capital Providers or the DIP Lenders may be parties, arising out of, under or in any way pertaining to this Agreement. This waiver constitutes a waiver of trial by jury of all

claims against all parties to such actions or proceedings, including claims against parties who are not parties to this Agreement.

- SECTION 4.5 Severability. Wherever possible, each provision of this Agreement shall be interpreted in such manner as to be effective and valid under applicable laws, but if any provision contained in this Agreement shall be invalid, illegal or unenforceable in any respect under any applicable law, the validity, legality and enforceability of the remaining provisions contained herein shall remain effective and binding on the parties hereto and shall not be affected or impaired thereby.
- SECTION 4.6 Continuing Agreement. This Agreement shall be a continuing one and shall be binding upon the parties hereto regardless of how long before or after the date hereof any of the Senior Obligations or the Second Ranking Obligations were or are incurred. This Agreement shall continually exist in full force and effect as between all Parties until all of the Senior Obligations have been Paid in Full and no commitments therefor are outstanding, and shall continually exist in full force and effect thereafter as between all Parties (other than the DIP Lenders) until all of the Second Ranking Obligations have been Paid in Full and no commitments therefor are outstanding. This Agreement shall continue to be effective, or be reinstated, as the case may be, as to the DIP Lenders if at any time any payment, or any part thereof, of any of the Senior Obligations is rescinded or must otherwise be restored or returned by the DIP Lenders, and as to the CE Capital Providers if at any time any payment, or any part thereof, of any of the Second Ranking Obligations is rescinded or must otherwise be restored or returned by the CE Capital Providers, in each case upon, under or as a result of any Distribution Proceeding by or against the Trust or otherwise, all as though such payment had not been made.
- SECTION 4.7 <u>Binding Effect</u>. This Agreement shall be binding upon the DIP Lenders, the CE Capital Providers, the Trust and the LW Capital Provider and their respective successors, assigns, heirs, personal representatives, executors, administrators and estate and shall inure to the benefit of the DIP Lenders, the CE Capital Providers and the LW Capital Provider and their successors and assigns; provided that the LW Capital Provider will not sell, assign, transfer or otherwise dispose in any manner whatsoever of all or any part of the Subordinated Obligations unless such assignee agrees to be bound by the terms of this Agreement.
- SECTION 4.8 Applicable Law. This Agreement and the rights and obligations of the parties hereunder shall be construed and interpreted in accordance with the laws of the State of New York (without reference to any conflict of law principles or choice of law doctrine that would have the effect of causing this Agreement to be construed in accordance with or governed by the law of any other jurisdiction).
- **SECTION 4.9** <u>Duplicate Originals and Counterparts</u>. This Agreement may be executed in any number of duplicate originals or counterparts, and each of such duplicate originals or counterparts shall be deemed to be an original and all taken together shall constitute but one and the same instrument.
- **SECTION 4.10** <u>Headings.</u> Article and Section headings in this Agreement are included herein for convenience of reference only, shall not constitute a part of this Agreement for

any other purpose and shall not be deemed to affect the meaning or construction of any of the provisions hereof.

SECTION 4.11 Conflicts. In the event of any conflict between the terms of this Agreement and the terms of the LW Capital Provision Agreement, the terms of this Agreement shall govern and control.

[SIGNATURES APPEAR ON FOLLOWING PAGE]

IN WITNESS WHEREOF, each of the Parties has executed and delivered this Agreement as of the day and year first written above.

LW HOLDCO VI LLC, as
LW Capital Provider
12 07.14
12mm WIT
By:
Name: Boaz Weinstein
Title: Authorized Signatory
ů ,
UNITED STATES DEPARTMENT OF
THE TREASURY, as DIP Lender
GEOFFREY S. BERMAN
United States Attorney for the Southern
District of New York
Counsel for the United States Department of
the Treasury
By:
Name:
Title:
EXPORT DEVELOPMENT CANADA, as
DIP Lender
By:
Name:
Title:
D
By:
Name:
Title:
CYNTHIANA LLC, as CE Capital Provider
, 1
D.
By:
Name:
Title:

IN WITNESS WHEREOF, each of the Parties has executed and delivered this Agreement as of the day and year first written above.

LW Capital Provider By:.__ Name: Title: UNITED STATES DEPARTMENT OF THE TREASURY, as DIP Lender **GEOFFREY S. BERMAN** United States Attorney for the Southern District of New York Counsel for the United States Department of the Treasury DAVID S. JONES Assistant U.S. Attorney, SDNY Name: Title: EXPORT DEVELOPMENT CANADA, as **DIP** Lender By: Name: Title: By: _ Name: Title: CYNTHIANA LLC, as CE Capital Provider Bv: Name: Title:

LW HOLDCO VI LLC, as

IN WITNESS WHEREOF, each of the Parties has executed and delivered this Agreement as of the day and year first written above.

LW Capital Provider By: Name: Title: UNITED STATES DEPARTMENT OF THE TREASURY, as DIP Lender **GEOFFREY S. BERMAN** United States Attorney for the Southern **District of New York** Counsel for the United States Department of the Treasury By:.___ Name: Title: **EXPORT DEVELOPMENT CANADA, as DIP** Lender Bv: Name: Brian Craig Title: Sr. Special Risks Manager By: Name: Essya Lamouchi Title: Sr. Special Risk Manager CYNTHIANA LLC, as CE Capital Provider By: Name:

LW HOLDCO VI LLC, as

Title:

09-50026-mg Doc 14413-7 Filed 02/04/19 Entered 02/04/19 15:31:37 Exhibit G Pg 17 of 19

IN WITNESS WHEREOF, each of the Parties has executed and delivered this Agreement as of the day and year first written above.

LW Capital Provider By:.____ Name: Title: UNITED STATES DEPARTMENT OF THE TREASURY, as DIP Lender **GEOFFREY S. BERMAN United States Attorney for the Southern District of New York** Counsel for the United States Department of the Treasury By:.____ Name: Title: **EXPORT DEVELOPMENT CANADA, as DIP Lender** By: _____ Name: Title: By: Name: Title: CYNTHIANA LLC, as CE Capital Provider

LW HOLDCO VI LLC, as

462344_3

By: __ Name:

Title: Authorized Signatory

EARLHAM	I LLC,	as	CE	Capital	Provider
---------	--------	----	----	---------	-----------------

Name:

Title: Authorized Signatory

Acknowledged and approved:

MOTORS LIQUIDATION COMPANY AVOIDANCE ACTION TRUST,

a Delaware Statutory Trust

Title:

By: Name:

ARTHUR J. GONZALEZ, as

Trust Monitor

09-50026-mg Doc 14413-7 Filed 02/04/19 Entered 02/04/19 15:31:37 Exhibit G Pg 19 of 19

EARLHAM LLC, as CE Capital Provide	EA	RL	HAN	1 LI	C.	as	CE	Car	oital	Provide	er
------------------------------------	----	----	-----	------	----	----	----	-----	-------	---------	----

By: Name: Title:

Acknowledged and approved:

MOTORS LIQUIDATION COMPANY

AVOIDANCE ACTION TRUST, a Delaware Statutory Trust

Trust Monitor

By:

Name:

Title: Vice President